Corridor Study -Red Lodge North Finding of No Significant Impact



Project Number STPP 28-2(25)70

Control Number 4375

Carbon County, Montana

May 2009

U.S. Department of Transportation Federal Highway Administration



FEDERAL HIGHWAY ADMINISTRATION

FINDING OF NO SIGNIFICANT IMPACT

for

Project Number: STPP 28-2(25)70 Project Name: Corridor Study - Red Lodge North **Control Number: 4375**

in

Carbon County, Montana

The Montana Department of Transportation (MDT) and the US Department of Transportation Federal Highway Administration (FHWA) have determined that the Preferred Alternative, as described in the attached Environmental Assessment (EA) dated October 2008, will have no significant impact on the human environment. This Finding of No Significant Impact (FONSI) is based on the October 2008 EA. After independent evaluation of the EA, MDT and FHWA conclude that the EA adequately and accurately discusses needs, environmental issues and environmental impacts of the proposed project and appropriate mitigation measures. The EA provides sufficient evidence and analysis for determining that an Environmental Impact Statement (EIS) is not required. MDT and FHWA take full responsibility for the accuracy, scope, and content of the October 2008 EA.

For purposes of compliance with the Montana Environmental Policy Act (MEPA) (ARM 17.4.609(3)(j) and ARM 18.2.239(3)(j)), this FONSI and conclusion that an EIS is not required should be considered part of the EA.

Montana Department of Transportation

Federal Highway Administration

<u>6 /8 09.</u> Date <u>6 /25 /2009</u>

Table of Contents

FINDING OF NO SIGNIFICANT IMPACT

1.0	Introd	luction	1
2.0	Selec	tion of Preferred Alternative	1
	2.1	Summary of Impacts and Mitigation	4
3.0	Comn	nents and Coordination	10
	3.1	EA Distribution	10
	3.2	News Advertisements and News Releases	10
	3.3	Public Hearings	11
	3.4	Summary of Primary Agency and Public Concerns	11
4.0	Clarif	ications to the EA	14
5.0	Resp	onse to Comments and Questions on the EA	18

TABLES

Table 1	Summary of Preferred Alternative	.2
Table 2	Summary Comparison of Project Alternatives Impacts and Mitigation	.5

APPENDIX A

Public Hearings

APPENDIX B

Agency and Public Comments and Responses

APPENDIX C

Environmental Assessment (October 2008)

Acronym List

BMP	Best Management Practice		
EA	Environmental Assessment		
FHWA	Federal Highway Administration		
LOS	Level of Service		
MDEQ	Montana Department of Environmental Quality		
MDT	Montana Department of Transportation		
NEPA	National Environmental Policy Act		
TWLTL	Two-way Left-Turn Lane		
USC	United States Code		
USFWS	United States Fish and Wildlife Service		

Finding of No Significant Impact

1.0 INTRODUCTION

The purpose of this document is to summarize the final coordination process conducted by MDT (Montana Department of Transportation) and FHWA (Federal Highway Administration) in order to complete the Corridor Study–Red Lodge North EA (Environmental Assessment) and Section 4(f) Evaluation. This document has been developed in accordance with the NEPA (National Environmental Policy Act) of 1969, as amended; Council on Environmental Quality regulations, Parts 1500–1508; the MEPA (Montana Environmental Policy Act); and other applicable laws and regulations. The EA, which is included in Appendix C, describes the potential environmental impacts of reconstruction of US Highway 212 from 8th Street in Red Lodge, Montana to approximately 1,000 feet north of Boyd, Montana.

2.0 SELECTION OF PREFERRED ALTERNATIVE

Following the identification of the purpose and need for the proposed project and the project objectives, numerous improvement scenarios were developed and evaluated. These scenarios were modified and refined based on input from elected officials and the general public as well as data collected pertaining to engineering factors, environmental considerations, and existing and planned development along the corridor. The result of this process was the consideration of two alternatives: a no-build alternative and a build alternative (identified as the Preferred Alternative in the EA). Of the two alternatives analyzed in the EA, the build alternative is the Preferred Alternative. Alternative for the proposed project. *Please refer to Table 1, Summary of Preferred Alternative*.

The Preferred Alternative meets the project purpose and need by:

- Improving the pavement condition along the project corridor by reconstructing the roadway.
- Improving the intersection geometry at key locations along the corridor.
- Incorporating an Access Management Plan for Red Lodge, which was supported by the Red Lodge City Council in March 2007.
- Providing pedestrian and bicycle facilities in Red Lodge.
- Improving level of service by providing passing lanes in rural segments where appropriate and making improvements at the 8th Street/US Highway 212 and MT Highway 78/US Highway 212 intersections.
- Providing wider roadway shoulders in rural segments where appropriate.
- Flattening ditch slopes in rural segments where appropriate.
- Reducing clear zone encroachments in rural segments where appropriate.
- Providing increased snow storage with wider and deeper roadside ditches.
- Clearing thick brush and trees within the clear zone to improve driver visibility of approaching wildlife.
- Providing turning lanes where needed to reduce differential speed conflicts.
- o Improving highway-related storm water drainage in Red Lodge and Roberts.

Table 1 Summary of Preferred Alternative					
Segment	Sub-segment	Typical Section	Intersection(s)	Intersection(s) Improvements	Drainage
Red Lodge	8 th St to MT 78	 Two 12' travel lanes Two 10' parking lanes Two 5' sidewalks 	Oakes Avenue and 8 th Street	 Realign Oakes with 7th Street Convert Oakes to southbound one-way Provide diagonal parking on Oakes Narrow the street width at Carnegie Library Install traffic signal on 8th Street when warranted and justified 	 Curb and gutter Storm water
	MT 78 to Developed Limits of Red Lodge	 Two 12' travel lanes Two 5.5' shoulders One 14' TWLTL One 5' sidewalk on west side of roadway One 10' shared bike/ped path on east side of roadway 	MT 78 and Villard Avenue	 Construct roundabout Close Villard and MT 78 intersection Construct cul-de-sac on Villard south of MT 78 Extend 4th Street 	
	Developed Limits of Red Lodge to Two Mile Bridge Road	 Two 12' travel lanes One 13' raised median Two 5.5' shoulders Two 3.5' median shoulders One 10' shared bike/ped path on east side of roadway 	Two Mile Bridge Road and four other locations	Per access management plan: • Construct roundabouts at full access intersections (3 intersections) • Construct ³ ⁄ ₄ access intersection • Construct ¹ ⁄ ₂ access intersection	Roadside ditches
Red Lodge to Roberts	Two Mile Bridge Road to south end of Roberts	 Two 12' travel lanes Two 8' shoulders Bus turnaround One-mile northbound passing lane 			 Maintain existing drainage patterns and culvert locations May relocate irrigation ditches within right-of- way

Table 1 Summary of Preferred Alternative					
Segment	Sub-segment	Typical Section	Intersection(s)	Intersection(s) Improvements	Drainage
Roberts	South end of Roberts to East Maple Street	 Two 12' travel lanes Two 4' shoulders One 14' TWLTL 	Oak, Cedar, Pine, and East Maple Streets	Construct crosswalk at each intersection	 Construct berms perpendicular to highway to direct storm water Replace existing
	East Maple Street to north end of Roberts	 Two 12' travel lanes Two 4' shoulders One 14' TWLTL 2:1 inslope with guardrail on east 	Cooney Dam Road	 Realign to form a single intersection perpendicular to US Highway 212 	culvert south of Birch Street • Direct storm water within Roberts to open ditches and/or pipes
Roberts to Boyd		 Two 12' travel lanes Two 8' shoulders Bus turnaround One-mile northbound passing lane One-mile southbound passing lane 	Clear Creek Road	 Realign with south access of rest area 	
	North end of Roberts to Boyd		Cooney Dam Road	 Close northern fork Add southbound right-turn lane on US Highway 212 to Cooney Dam Road 	 Maintain existing drainage patterns and culvert locations May relocate irrigation
	Boyd Country Store Boyd Country		Main Street	 Realign to be perpendicular to US Highway 212 Addition of southbound left- turn lane on US Highway 212 	ditches within right-of- way

2.1 SUMMARY OF IMPACTS AND MITIGATION

Based on the analysis provided in the EA, impacts associated with the proposed project were not found to be individually or cumulatively significant. The proposed project would provide numerous benefits, as compared to the No-build Alternative, to the transportation system and highway-related drainage. *Please refer to Table 2, Summary Comparison of Project Alternatives Impacts and Mitigation, for a summary of the Preferred and No-build alternatives impacts and associated mitigation measures.*

		Summary Compariso	Table 2 on of Project Alternatives Impacts and Mitigation	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Land Use		No impact.	Acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way would create direct land use impacts; overall land uses in the area would not be affected.	No mitigation required.
Farmland		No Impact.	Impacts to approximately 275.8 acres (111.6 hectares) of farmland; of which 89.6 acres (36.3 hectares) are prime farmland and 48.9 acres (19.8 hectares) are of statewide importance.	No mitigation required.
Transportation System	Traffic	Inadequate LOS currently at northbound lane between Roberts and Boyd; southbound lane between Red Lodge and Roberts by 2010; and MT Highway 78 and 8 th Street by 2030.	All segments and key intersections expected to operate at acceptable levels through 2030.	No mitigation required.
	Access	Inconsistent with Red Lodge Council Resolution No. 3228.	The ability of the roadway to provide for both access and progression would be improved. Access would be managed between MT Highway 78 and Two Mile Bridge Road. Consistent with Red Lodge Council Resolution No. 3228.	No mitigation required.
	Safety	The number of crashes and existing crash rates are anticipated to increase as traffic continues to increase.	Would include the following safety improvements: intersection realignments, addition of turn lanes where needed, access management in Red Lodge, wider shoulders, rumble strips, flatter ditch slopes, clearing of thick brush and trees within the clear zone, ped/bike facilities in Red Lodge and Roberts, bus turnarounds.	No mitigation required.
	Pedestrian/Bicycle Facilities	Inconsistent with Red Lodge Comprehensive Trails Plan.	Would provide sidewalks and crosswalks, where appropriate, in Red Lodge; a shared bike/ped path between MT Highway 78 and Two Mile Bridge Road; and crosswalks and one-block of sidewalk in Roberts. Consistent with Red Lodge Comprehensive Trails Plan.	No mitigation required.

		Summary Comparise	Table 2 on of Project Alternatives Impacts and Mitigation	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
	Noise	No impact.	By 2030, five residential properties would experience noise levels at or slightly above the noise abatement criteria.	At this time, noise mitigating measures are not considered reasonable and feasible.
Right-of-Way and Relocations		No impact.	Acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way; potential acquisition and/or relocation of up to nine structures (six dwellings, two out-buildings, and one commercial building).	Compliance with Uniform Act. MDT will also attempt to meet individually with affected property owners. Reasonable efforts to avoid and/or minimize impacts will be made.
	Surface Water	No new impacts.	Impacts may result from culvert replacement or extension; ditch realignment; dredge/fill activities in wetlands; the relocation of irrigation ditches outside of the proposed right-of-way; new storm water outfall locations at Rock Creek; and conflict between existing storm drain and new storm drain near the intersection of MT Highway 78 and US Highway 212.	Use of BMPs; compliance with applicable permits, and local, state, and federal regulations.
Water Resources/Quality	Irrigation Facilities	No new impacts.	Would relocate irrigation ditches, as necessary, in consultation with owners to minimize impacts and may impact Mullaney Spring.	Care to avoid altering flow rate to water rights holder of Mullaney Spring. Consultation with affected ditch associations and other landowners/water rights holders to minimize impacts to irrigation facilities.
	Ground Water	No new impacts.	Impacts to ground water resources are not anticipated. The Preferred Alternative may require relocation of domestic wells within the proposed right-of-way.	If domestic wells are displaced, domestic water would be restored to the affected properties.
	Public Water Systems	No new impacts.	No new impacts anticipated.	No mitigation required.
	Waste Water Systems	No new impacts.	Potential relocation of an identified mound septic system in the proposed right-of-way and construction limits. A new storm drain pipe and outfall may need to be constructed or the existing storm drain replaced.	If the mound system were impacted, MDT would relocate the system per County and MDEQ requirements.

		Summary Compariso	Table 2 on of Project Alternatives Impacts and Mitigation	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Water Body Modifications		No new impacts.	New culvert installation at Stanley Creek and minor inlet and outlet ditches may be required; irrigation ditches would be relocated outside of right-of-way.	Structures would be designed to minimize disruption to hydrology and to comply with applicable federal and state regulations.
Wetlands		No new impacts.	Impacts to approximately 40.7 acres (16.5 hectares) of wetlands; of which 24.8 acres (9.7 hectares) are jurisdictional based on preliminary jurisdictional determinations.	Unavoidable impacts would be mitigated according to permit requirements at an approved mitigation site.
Wildlife Habitat and Ecosystems	Vegetation	No new impacts.	Removal of vegetation in select areas for proposed improvements. Clearing of ground cover along the corridor has the potential to open areas to noxious weeds.	Compliance with MDT Standard Specifications.
	Terrestrial and Avian Species	No new impacts.	May result in minor fragmentation, modification, and/or loss of habitat for terrestrial and avian species.	Use of BMPs; implementation of erosion and sediment control plan; compliance with Migratory Bird Treaty Act and MDT's most current depredation permit from the USFWS.
	Aquatic Species	No new impacts.	May result in minor impacts due to water body modifications. No substantive losses of spawning fish species are anticipated.	Use of BMPs; compliance with applicable permits and federal and state regulations.
	Montana Species of Concern	No new impacts.	Impacts are not anticipated; however, the gray wolf would be subject to the same impacts as other terrestrial species.	No mitigation required.
	Animal-Vehicle Collisions	High frequency of animal- vehicle collisions would continue.	May improve driver visibility of approach wildlife by removing thick brush and vegetation from the clear zone.	No mitigation required.

		Summary Comparis	Table 2 on of Project Alternatives Impacts and Mitigation	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Floodplain		No new impacts.	No new impacts anticipated. Coordination with the county floodplain administrator would occur to determine whether minor encroachment of the floodplain would occur and whether a floodplain development permit is required.	No mitigation required.
T Enc	Threatened and dangered Species	No new impacts.	No effect.	No mitigation required.
Cultural Resources		No impact.	SHPO concurrence with finding of <i>No Effect</i> or <i>No Adverse Effect</i> to all historic properties.	No further avoidance/mitigation measures required.
4(f) es	Historic Properties	No impact.	De Minimis Section 4(f) impacts to four historic resources; no additional Section 4(f) use.	
stion 4 opertic	Historic Irrigation Ditches	No impact.	<i>De Minimis</i> Section 4(f) impacts to ten historic irrigation ditches; no additional Section 4(f) use.	No further avoidance/mitigation measures required.
ъŠг	Recreation Areas	No impact.	No impact.	
Section 6(f) Properties		No impact.	No Section 6(f) properties would be converted to a transportation use. Directional and entrance signs that may be removed would be reinstalled following construction. If impacted, the entrance road for the Water Birch fishing access site would be returned to existing or improved condition following construction.	No mitigation required.
Hazardous Materials, Solid Waste, and Underground Storage Tanks		No impact.	No impacts anticipated.	If hazardous materials are discovered, generated, or used they would be stored, handled, and disposed of in accordance with applicable local, State, and Federal laws.
	/isual/Aesthetic Considerations	No impact.	Improved aesthetics in Red Lodge; four roundabouts as desired in Resolution No. 3228; incorporation of elements of the Red Lodge Streetscape Plan.	No mitigation required.

		Summary Comparis	Table 2 on of Project Alternatives Impacts and Mitigation	
Resource		Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Temporary Construction Considerations	Air Quality	No impact.	Temporary increase of fugitive dust and mobile source emissions.	Compliance with standard MDT procedures and applicable permit requirements.
	Transportation System	No impact.	Temporary impacts to local and regional traffic circulation in the project area due to lane closures, delays, temporary travel on unpaved surfaces, and reduced travel speeds.	Development of a transportation management plan according to MDT Standard Specifications.
	Water Resources/Quality	No impact.	Short-term increase in turbidity, potential for erosion, and storm water runoff.	Use of BMPs; compliance with applicable federal and state regulations. Temporary impacts to wetlands would be restored to original contours and re- vegetated at the earliest practicable date following construction.
	Wildlife Habitat and Ecosystems	No impact.	May result in temporary minor disturbances to wildlife communities.	Between Sept. 1 and Apr. 30, vacated swallow or other songbird nests would be physically removed and deterrents would be placed on existing structures. Disturbed areas would be reseeded with desirable seed mix.
	Noise	No impact.	Temporary increase in noise levels within the vicinity of the project.	Compliance with MDT Standard Specifications. As necessary, the contract will include additional requirements for projects located in or near urban areas.
Utilities		No impact.	Some relocation of overhead and underground power lines and underground telephone lines may be required.	Potential impacts would be coordinated with the appropriate utility companies. Rural overhead power lines that are relocated would be raptor proofed per MDT policy.

3.0 COMMENTS AND COORDINATION

In accordance with NEPA and MEPA, MDT has coordinated with agencies, the public, and other interested parties throughout the development of the EA. The following sections summarize the final coordination efforts for the proposed project.

3.1 EA Distribution

MDT distributed the EA to property owners identified as potential relocations and to the following agencies:

- US Army Corps of Engineers Billings Regulatory Office
- US Department of Agriculture Natural Resources Conservation Service
- US Department of Interior US Fish & Wildlife Service
- US Environmental Protection Agency Montana Operations Office
- Montana Department of Environmental Quality
 - Permitting and Compliance Division
 - Water Protection Bureau
 - o Remediation Division
 - o Director's Office
- Montana Fish, Wildlife & Parks
 - o Region 5
 - Parks Division
 - Habitat Protection Bureau
- Montana Department of Natural Resources and Conservation
 - State Water Projects Bureau
 - Water Resources Division Billings Regional Office
- Carbon County Commission
- Red Lodge City Council

The EA was made available to the public for review and comment beginning November 17, 2008, and ending December 19, 2008. The EA was available at the following public viewing locations:

- Carnegie Library 3 W 8th St, Red Lodge
- Red Lodge City Hall 801 N Broadway, Red Lodge
- Roberts Public School 106 E Maple St, Roberts
- Kadrmas, Lee and Jackson 2611 Gabel Rd, Billings
- MDT Billings Area Office 424 Morey St, Billings
- MDT Helena Headquarters Office 2701 Prospect Ave, Helena
- Montana State Library 1515 E 6th Ave, Helena

3.2 News Advertisements and Press Releases

A Notice of Availability of the EA was published in newspapers and local media. Press releases for the Notice of Availability were released on November 13, 2008, to the *Carbon County News* and on November 17, 2008, for the remaining Billings and Red Lodge area broadcast media. Press releases for the public hearings were released on November 24 and December 1, 2008, to Billings and Red Lodge area broadcast media. Media that received the press releases are as follows:

- Billings Gazette
- Billings Outpost
- Billings Times
- Carbon County News
- KTVQ
- KBBB/KBUL/KCTR/KKBR/KMHK
- KBLG/KRKX/KRZN/KYYA
- KBLW FM
- KEMC News and Public Service Announcements
- KGHL FM and AM
- KQBL FM
- KRSQ FM
- KZRV FM
- KHMT TV
- KPBR/KPLN/KWMY FM
- KRZN and KRZN 2
- KULR 8 Billings, TV
- KMZK AM
- KMXE FM
- Road Watch Montana

Newspaper advertisements were published in the *Carbon County News* and *Billings Gazette* to alert the public of the availability of the EA and also the public hearings. The Notice of Availability of the EA appeared in the *Carbon County News* on November 13, 2008, and in the *Billings Gazette* on November 17, 2008. Advertisements to alert the public and other interested parties of the public hearings appeared in the *Carbon County News* on November 20 and December 4, 2008, and in the *Billings Gazette* on November 20 and December 4, 2008, and in the *Billings Gazette* on November 23 and December 7, 2008.

In addition to press releases and newspaper advertisements, postcards were distributed to members of the public and other interested parties who have attended meetings or provided comments throughout the development of the EA. *Please refer to Appendix A, Public Hearings, to view the newspaper advertisements, press releases, and postcards distributed.*

3.3 Public Hearings

Public hearings for the project were held on December 9, 2008, at the Red Lodge Senior Citizens Center and on December 10, 2008, at Roberts Public School. The meetings were held from 5:00 to 9:00 PM with a formal presentation given at 5:30. The meeting in Red Lodge was attended by 23 members of the community and government agencies and the meeting in Roberts was attended by 21 members of the community and government agencies. Seven verbal comments were received in total at the public hearings. At the close of the comment period on December 19, 2008, 40 written comments were received from agencies and members of the community.

3.4 Summary of Primary Public and Agency Concerns

Of the 47 comments received from agencies and the public, four primary concerns were apparent: 1. realignment of the roadway; 2. extension of the sidewalk and/or shared bike/ped path; 3. access to Beartooth Hospital; and 4. raised median lighting requirements. *Please refer*

to Appendix B, Agency and Public Comments and Responses, for agency and public comments received and their corresponding responses.

Realignment of Roadway

Realignment of the roadway has been recommended by the public mainly due to concerns over potential relocations and right-of-way acquisition. Members of the public have primarily suggested a realignment of US Highway 212 to the east, while a number of other citizens have suggested to not realign the roadway to the east.

The projected impacts described in Chapter 3 of the EA were based on preliminary design (approximately 30%) using a standard right-of-way width of 80 feet. In order to minimize impacts to various resources, such as wetlands and the historic Rocky Fork Branch of the Northern Pacific Railroad, and through coordination with various federal, state, and local agencies, the resulting preliminary design of the project maintained the existing centerline of the roadway. Realignment of the roadway was initially explored at the beginning of the Corridor Study–Red Lodge North project in order to avoid and/or minimize impacts to wetlands and again following public comment.

During final design, MDT will make reasonable efforts to avoid and/or minimize impacts to potentially affected property owners. Realignment of the roadway to avoid potential relocations would be determined during final design. MDT will comply with the "Uniform Act", or Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970 (42 U.S.C. 4601 et seq.), and amendments.

Extension of Sidewalk and/or Shared Bike/Ped Path

The City of Red Lodge and its citizens have frequently commented throughout the development of the EA about the desire for the community to be as pedestrian and bicycle accessible as possible. One of the primary issues/concerns raised after the public hearings was the desire for either an extension of the sidewalk, which is proposed on the west side of the roadway within the developed limits of Red Lodge, or the shared bike/ped path, which is proposed on the east side of the roadway between MT Highway 78 and Two Mile Bridge Road. Individuals suggested extending either the sidewalk or shared bike/ped path approximately a half-mile north of Two Mile Bridge Road on the west side of the roadway to connect to a planned trail associated with the Spires Subdivision on the West Bench.

Extending the sidewalk or shared bike/ped path to connect with the proposed Spires subdivision trail system was analyzed. However, because of the proposed footprint of the roadway, potential relocations had been previously identified on both sides of US Highway 212 between Two Mile Bridge Road and the proposed Spires trail. Due to the right-of-way constraints in this area, extending the sidewalk or shared bike/ped path to connect to the Spires subdivision trail system would not be feasible without eliminating the potential to avoid the relocation of at least one of these properties. The trail extension may be explored further during the design phase of the project.

The proposed project would not preclude other entities or individuals from constructing a trail to connect the Spires subdivision trail system to the proposed shared bike/ped path. If this were to occur, the roundabout at the Two Mile Bridge Road intersection would accommodate

pedestrians and bicyclists and allow them to access the proposed shared bike/ped path on the east side of the roadway.

Additionally, the proposed 8-foot shoulders along the rural segments of the roadway, such as north of Two Mile Bridge Road, would be wide enough to accommodate pedestrians and bicyclists.

Access to Beartooth Hospital

During the development of the EA, the Beartooth Hospital was proposed and planned along the segment of the proposed project between the developed limits of Red Lodge and Two Mile Bridge Road, where the access management plan was developed. After the public hearings, a number of comments were received which were concerned that the proposed three-quarters access intersection, which is part of the access management plan, at the main entrance of the Beartooth Hospital would be a safety concern for patients leaving the hospital and also emergency vehicles entering and exiting the hospital. To alleviate these concerns, requests were made for the three-quarters access intersection to be modified to a full access intersection and/or roundabout.

The decision whether to construct a full access intersection or a roundabout at the main entrance of the hospital would be determined through the MDT system impact action process. If a roundabout were to be designed, the hospital site plan would need to accommodate the increased right-of-way needs required for a modern roundabout and to avoid impacts to Rock Creek. Additionally, in order to receive approach approval from MDT, the Beartooth Hospital and their representatives will need to go through the aforementioned process, which was initiated by Beartooth Hospital on January 5, 2009. A guide to the system impact action process can be found online at the following link:

http://mdt.mt.gov/publications/docs/brochures/siap_guide.pdf.

Raised Median Lighting Requirements

A raised median is an element proposed in the Access Management Plan to manage access between the developed limits of Red Lodge and Two Mile Bridge Road. The primary public issue/concern surrounding the raised median is its requirement for light posts and the feeling that it would not provide a unique entrance to the city of Red Lodge.

Initially, a two-way left-turn lane (TWLTL) was considered for the entire stretch between MT Highway 78 and Two Mile Bridge Road to accommodate turning vehicles accessing adjacent properties while improving progression for through traffic. However, the City of Red Lodge opposed the concept of a TWLTL in the developing area due to concerns that it may encourage commercial strip development. To accommodate that concern, MDT and the City developed the Access Management Plan.

During the development of the Access Management Plan, raised and depressed medians were analyzed to manage access between the developed limits of Red Lodge and Two Mile Bridge Road. Through the planning process, it was determined that the raised median was preferred by the City of Red Lodge. The Access Management Plan, which includes the raised median, was supported by Red Lodge City Council Resolutions No. 3223 and 3228, and the City of Red Lodge Growth Policy (2008).

The proposed project would incorporate elements of the Red Lodge Streetscape Plan, as appropriate, and lighting would be designed to minimize light pollution. In addition, there are numerous alternative lighting methods that would allow the median to be lit safely and still allow the City of Red Lodge to maintain its unique entrance into the city. Use of these methods will be determined during design. MDT will work with the City to develop an agreement to address appropriate lighting features to be incorporated into the project.

4.0 CLARIFICATIONS TO THE EA

The purpose of this section is to provide correction to errors and/or omissions, as well as additional information, to the EA released on November 17, 2008. Text deleted is shown as a strikethrough (e.g., strikethrough) and text added is shown underlined (e.g., underline).

Page S-8, Table A, Summary Comparison of Project Alternatives and Impacts

		Summary Com	Table A parison of Project Alternatives and Impacts	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Land Use		No impact.	Acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way would create direct land use impacts; overall land uses in the area would not be affected.	No mitigation required.
Farmland		No Impact.	Impacts to approximately 275.8 acres (111.6 hectares) of farmland; of which 89.6 acres (36.3 hectares) are prime farmland and 48.9 acres (19.8 hectares) are of statewide importance.	No mitigation required.
Transportation System	Traffic	Inadequate LOS currently at northbound lane between Roberts and Boyd; southbound lane between Red Lodge and Roberts by 2010; and MT Highway 78 and 8 th Street by 2030.	All segments and key intersections expected to operate at acceptable levels through 2030.	No mitigation required.
	Access	Inconsistent with Red Lodge Council Resolution No. 3228.	The ability of the roadway to provide for both access and progression would be improved. Access would be managed between MT Highway 78 and Two Mile Bridge Road. Consistent with Red Lodge Council Resolution No. 3228.	No mitigation required.
	Safety	The number of crashes and existing crash rates are anticipated to increase as traffic continues to increase.	Would include the following safety improvements: intersection realignments, addition of turn lanes where needed, access management in Red Lodge, wider shoulders, <u>rumble strips</u> , flatter ditch slopes, clearing of thick brush and trees within the clear zone, ped/bike facilities in Red Lodge and Roberts, bus turnarounds.	No mitigation required.
	Pedestrian/Bicycle Facilities	Inconsistent with Red Lodge Comprehensive Trails Plan.	Would provide sidewalks and crosswalks, where appropriate, in Red Lodge; a shared bike/ped path between MT Highway 78 and Two Mile Bridge Road; and crosswalks and one-block of sidewalk in Roberts. Consistent with Red Lodge Comprehensive Trails Plan.	No mitigation required.

Corridor Study – Red Lodge North Finding of No Significant Impact STPP 28-2(25)70 Control No. 4375 May 2009

		Summary Com	Table A parison of Project Alternatives and Impacts	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Temporary Construction Considerations	Air Quality	No impact.	Temporary increase of fugitive dust and mobile source emissions.	Compliance with standard MDT procedures and applicable permit requirements.
	Transportation System	No impact.	Temporary impacts to local and regional traffic circulation in the project area due to lane closures, delays, temporary travel on unpaved surfaces, and reduced travel speeds.	Development of construction traffic control plan transportation management plan according to MDT Standard Specifications.
	Water Resources/Quality	No impact.	Short-term increase in turbidity, potential for erosion, and storm water runoff.	Use of BMPs; compliance with applicable federal and state regulations. Temporary impacts to wetlands would be restored to original contours and re- vegetated at the earliest practicable date following construction.
	Wildlife Habitat and Ecosystems	No impact.	May result in temporary minor disturbances to wildlife communities.	Between Sept. 1 and Apr. 30, vacated swallow or other songbird nests would be physically removed and deterrents would be placed on existing structures. Disturbed areas would be reseeded with desirable seed mix.
	Noise	No impact.	Temporary increase in noise levels within the vicinity of the project.	Compliance with MDT Standard Specifications. As necessary, the contract will include additional requirements for projects located in or near urban areas.

Page S-12, Table A, Summary Comparison of Project Alternatives and Impacts

Corridor Study – Red Lodge North Finding of No Significant Impact STPP 28-2(25)70 Control No. 4375 May 2009

Page 3-9, Impacts to Safety

Alternative B (Preferred): The Preferred Alternative would provide numerous safety improvements along the project corridor. These include intersection realignments to eliminate skewed and offset intersections, the addition of turn lanes where needed, access management in Red Lodge, wider shoulders, <u>rumble strips</u>, flatter ditch slopes, clearing of thick brush and trees within the clear zone, provision of pedestrian and bicycle facilities in Red Lodge and Roberts, and provision of bus turnarounds for the Roberts school buses. These improvements are anticipated to result in a reduction in the number and severity of crashes as well as crash rates. The proposed roundabout at the intersection of MT Highway 78 adjacent to the Red Lodge Fire Station would reduce traffic congestion impacts on emergency vehicle response time. (Please note that animal-vehicle crashes are discussed further in Section 3.11.5, Animal-Vehicle Collisions.)

Table 3.7Impacts to Irrigation Facilities ¹				
Irrigation Ditch	Location	Description	Impact	
Brewery Ditch	Red Lodge	Serves 4 users and has a flow of 1.99 cfs; irrigates approximately 23.3 acres (9.4 hectares) of hay land.	No impacts anticipated.	
Vincent Ditch	Red Lodge to Roberts	Serves 3 users and has a flow of 2.13 cfs; irrigates approximately 92.7 acres (37.5 hectares) of hay land. Part of the ditch flows inside the west fence line.	Section along west fence line may be moved beyond right-of-way.	

Page 3-19, Table 3.7, Impacts to Irrigation Facilities

¹ Impacts presented in this table are related to the relocation of the irrigation ditches identified along the project corridor. Drainage structures that would be replaced would be designed to address hydrologic conditions and comply with federal and state regulations.

Page 3-45, Mitigation for Temporary Transportation System Impacts

A construction traffic control plan transportation management plan will be developed according to MDT Standard Specifications to include construction phasing devised to maintain two lanes of traffic and uninterrupted side road access along the corridor to the greatest extent practicable. The contractor will coordinate with emergency service providers and schools to solicit input for the construction traffic control plan transportation management plan to provide ongoing information during construction.

Page 6-4, Table 6.1, Agencies Consulted

The ditch associations were removed from the Agencies Consulted table because, while letters were distributed to these associations alerting them of the Environmental Assessment, no formal consultation occurred.

Table 6.1 Agencies Consulted
Municipality
Red Lodge Police
Roberts School Board
Private
BNSF Railway Company
Consolidated Ditch Company
Finn Ditch Company
Montana Land Reliance
Red Lodge Chamber of Commerce
Rock Creek Water Users

5.0 RESPONSE TO COMMENTS AND QUESTIONS ON THE EA

Public hearings for the project were held on December 8 and 9, 2009. A transcript of the verbal comments received at the public hearings is included in Appendix A, Public Hearings. A total of 47 comments were received during the public comment period. These comments and their corresponding responses, including responses to comments made at the public hearings, are included in Appendix B, Agency and Public Comments and Responses. Comment numbers 5 through 10 were received at the public hearings. The remaining comments were received during the public comment 17, 2009, and ended on December 19, 2009.

Appendix A

Public Hearings

- Publicity for EA and Public Hearings

 News ads, press releases, and postcard
- EA Distribution List
- Public Hearing Sign-in Sheets
- Public Hearing Partial Transcripts



Notice of Availability

Corridor Study - Red Lodge North Environmental Assessment - Carbon County

Beginning November 17, 2008, the Environmental Assessment (EA) and Section 4(f) Evaluation will be available for public review and comment. The Preferred Alternative identified in the EA includes elements that best satisfy the need for the project while minimizing impacts. The Preferred Alternative identified in the EA consists of urban and rural reconstruction. Depending upon the specific identified needs along the corridor, reconstruction is expected to include some or all of the following: pavement surfacing, aggregate base, geometric improvements, pedestrian and/or bicycle facilities, lighting, drainage improvements, access management and other miscellaneous items.

Review the EA at:

- Carnegie Library, 3 West 8th Street, Red Lodge
- Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- Roberts Public School, 106 East Maple Street, Roberts
- Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- MDT Billings District Office, 424 Morey Street, Billings
- Montana State Library, 1515 East 6th Avenue, Helena
- MDT Environmental Services Office 2701 Prospect Ave., Helena
- Online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml
- Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Comment Period: November 17, 2008 to December 19, 2008

- Present oral or written comments at the public hearing
 - Written comments to Tom Martin, MDT, PO Box 201001, 2701 Prospect Ave., Helena, MT 59620
 - Online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml

For More Information:

• Stefan Streeter, MDT, (406) 252-4138 • Tom Martin, MDT, (406) 444-0879

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program, or activity of our department. For reasonable accommodations to participate in this hearing, call Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired: TTY (406) 444-7696, (800) 335-7592, or Montana Relay. Alternative accessible formats of information provided on request.

Public Hearings:

Tuesday, December 9, 2008 Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge Wednesday, December 10, 2008 Roberts School Cafeteria, 106 East Maple Street, Roberts

Both Locations: Open House: 6:00 p.m. Presentation: 6:30 p.m. Same presentation will be given on both evenings



Public Hearings

Corridor Study - Red Lodge North Environmental Assessment - Carbon County

The Environmental Assessment (EA) and Section 4(f) Evaluation is now available for public review and comment. The Preferred Alternative identified in the EA includes elements that best satisfy the need for the project while minimizing impacts. The Preferred Alternative identified in the EA consists of urban and rural reconstruction. Depending upon the specific identified needs along the corridor, reconstruction is expected to include some or all of the following: pavement surfacing, aggregate base, geometric improvements, pedestrian and/or bicycle facilities, lighting, drainage improvements, access management and other miscellaneous items.

Review the EA at:

- Carnegie Library, 3 West 8th Street, Red Lodge
- Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- Roberts Public School, 106 East Maple Street, Roberts
- Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- MDT Billings District Office, 424 Morey Street, Billings
- Montana State Library, 1515 East 6th Avenue, Helena
- MDT Environmental Services Office 2701 Prospect Ave., Helena
- Online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml
- Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Comment Period: November 17, 2008 to December 19, 2008

- Present oral or written comments at the public hearing
- Written comments to Tom Martin, MDT, PO Box 201001, 2701 Prospect Ave., Helena, MT 59620
- Online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml

For More Information:

• Stefan Streeter, MDT, (406) 252-4138 • Tom Martin, MDT, (406) 444-0879

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program, or activity of our department. For reasonable accommodations to participate in this hearing, call Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired: TTY (406) 444-7696, (800) 335-7592, or Montana Relay. Alternative accessible formats of information provided on request.

Public Hearings: Tuesday, December 9, 2008 Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge Wednesday, December 10, 2008 Roberts School Cafeteria, 106 East Maple Street, Roberts

Both Locations: Open House: 6:00 p.m. Presentation: 6:30 p.m. Same presentation will be given on both evenings

Becky Rude

From:	Grant, Paul [pgrant@mt.gov]
Sent:	Thursday, November 06, 2008 10:02 AM
To:	Red Lodge - Carbon County News (news@carboncountynews.com)
Cc:	Streeter, Stefan; Neville, Gary; Priebe, Gabe; mike.wamboldt@kljeng.com; 'becky rude@kljeng.com': Bruner, Heidy: Gocksch, Thomas: James, Carl: Grant, Paul: Watt
	Levis, Charity; Martin, Tom
Subject:	MDT announces Notice of Availability: Corridor Study - Red Lodge North Environmental Assessment - beginning November 17, 2008

November 6, 2008

FOR IMMEDIATE RELEASE

For further information, contact:

Mike Wamboldt, Project Manager, Kadrmas, Lee and Jackson,

(406) 245-5499

Stefan Streeter, MDT Billings District Administrator, (406) 252-4138 Tom Martin, MDT Environmental Services, (406) 444-0879 Paul Grant, MDT Public Involvement, (406) 444-9415

MDT announces Notice of Availability: Corridor Study - Red Lodge North Environmental Assessment

Carbon County - Beginning November 17, 2008, an Environmental Assessment (EA) will be available for public review and comment. The EA outlines the need for the proposed project, discusses the Preferred Alternative considered to meet those needs, and examines impacts which may result from the proposed project.

The Montana Department of Transportation (MDT) and the Federal Highway Administration (FHWA) invite all interested parties to review the EA and provide comments at a Public Hearing on either Tuesday, December 9, 2008 at the Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge, or Wednesday, December 10, 2008 at the Roberts School Cafeteria, 106 East Maple Street, Roberts. An open house will begin at 6:00 p.m. with an informational presentation at 6:30 p.m. Project staff will take questions to clarify the understanding of the EA, and then will proceed to a formal Public Hearing. The presentation will summarize the project history, present the Preferred Alternative, and describe the environmental process. Both Public Hearings will follow the same format and agenda. The same presentation will be given on both evenings.

Anyone interested in viewing the EA may view it online at www.mdt.mt.gov/pubinvolve/eis ea.shtml or at one of the following locations:

- * Carnegie Library, 3 West 8th Street, Red Lodge
- * Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- * Roberts Public School, 106 East Maple Street, Roberts
- * Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- * MDT Billings District Office, 424 Morey Street, Billings
- * Montana State Library, 1515 East 6th Avenue, Helena
- * MDT Environmental Services Office 2701 Prospect Ave., Helena
- * Online at <u>www.mdt.mt.gov/pubinvolve/eis ea.shtml</u>
- * Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Community participation is a very important part of the process, and the public is encouraged to attend. Oral or written opinions and comments may be presented at the public hearing.

Alternatively, written comments may also be submitted to Tom Martin, MDT Environmental Services, at 2701 Prospect Avenue, PO Box 201001, Helena, MT 59620-1001, or online at http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

The review period for the EA will conclude on December 19, 2008. All public comments are due by December 19, 2008.

The proposed Corridor Study - Red Lodge North project would be constructed from the city of Red Lodge, MT to approximately 1,000 feet north of the community of Boyd, MT to accomplish the following objectives:

- * Improve pavement condition along the project corridor
- * Improve intersection geometry at key locations
- * Manage access in north Red Lodge
- * Accommodate pedestrian/bicycle users in Red Lodge
- * Improve level of service in rural segments and at two intersections in Red Lodge
- * Provide wider shoulders in rural segments
- * Flatten ditch inslopes in rural segments
- * Reduce encroachments within clear zones where appropriate
- * Increase ditch storage for snow/ice off the roadway in rural segments
- * Reduce frequency of animal-vehicle collisions along project corridor
- * Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program or activity of our department. If you require reasonable accommodations to participate in this hearing, please contact Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired, the TTY number is (406) 444-7696 or 1 (800) 335-7592, or call Montana Relay at 711. Alternative accessible formats of pertinent information will be provided upon request.

-----END-----

Project name: Corridor Study - Red Lodge North Project ID: STPP 28-2(25)70 Control Number: 4375 Carbon County

Becky Rude

From:	Grant, Paul [pgrant@mt.gov]
Sent:	Monday, November 17, 2008 8:50 AM
To:	aaron murphy ktvq; Becky Bohrer (E-mail); Billings Gazette; Billings Outpost; Billings Times; KBBB/KBUL/KCTR/KKBR-FM/KMHK-FM; KBLG; KBLG-KRKX-KRZN-KYYA; KBLW-FM; KEMC - NEWS; KEMC - PSA's; KGHL-AM/KGHL-FM/KQBL-FM/KRSQ-FM/KZRV-FM; KHMT-TV; KHMT-TV; KMHK; KPBR-FM/KPLN-FM/KWMY-FM; KRZN; KRZN 2; KSVI-TV;
	Kulr 8 - Billings (E-mail); KULR/KMZK-AM; KULR-TV; KYYA; Road Watch Montana; KMXE-
	FM; Red Lodge - Carbon County News (E-mail)
Cc:	Priebe, Gabe; Streeter, Stefan; Neville, Gary; 'becky.rude@kljeng.com';
	mike.wamboldt@kljeng.com; Gocksch, Thomas; Martin, Tom; Bruner, Heidy; Conway, Tim; Watt Levis, Charity; Grant, Paul; James, Carl; Woodmansey, Alan; County Commissioners
Subject:	MDT announces Notice of Availability: Corridor Study - Red Lodge North Environmental Assessment

November 17, 2008

FOR IMMEDIATE RELEASE

For further information, contact:

Mike Wamboldt, Project Manager, Kadrmas, Lee and Jackson, (406) 245-5499 Stefan Streeter, MDT Billings District Administrator, (406) 252-4138 Tom Martin, MDT Environmental Services, (406) 444-0879 Paul Grant, MDT Public Involvement, (406) 444-9415

MDT announces Notice of Availability: Corridor Study - Red Lodge North Environmental Assessment

Carbon County - Beginning Monday, November 17, 2008, an Environmental Assessment (EA) will be available for public review and comment. The EA outlines the need for the proposed project, discusses the Preferred Alternative considered to meet those needs, and examines impacts which may result from the proposed project.

The Montana Department of Transportation (MDT) and the Federal Highway Administration (FHWA) invite all interested parties to review the EA and provide comments at a Public Hearing on either Tuesday, December 9, 2008 at the Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge, or Wednesday, December 10, 2008 at the Roberts School Cafeteria, 106 East Maple Street, Roberts. An open house will begin at 6:00 p.m. with an informational presentation at 6:30 p.m. Project staff will take questions to clarify the understanding of the EA, and then will proceed to a formal Public Hearing. The presentation will summarize the project history, present the Preferred Alternative, and describe the environmental process. Both Public Hearings will follow the same format and agenda. The same presentation will be given on both evenings.

Anyone interested in viewing the EA may view it online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml or at one of the following locations:

- * Carnegie Library, 3 West 8th Street, Red Lodge
- * Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- * Roberts Public School, 106 East Maple Street, Roberts
- * Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- * MDT Billings District Office, 424 Morey Street, Billings
- * Montana State Library, 1515 East 6th Avenue, Helena
- * MDT Environmental Services Office 2701 Prospect Ave., Helena

- * Online at www.mdt.mt.gov/pubinvolve/eis ea.shtml
- * Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Community participation is a very important part of the process, and the public is encouraged to attend. Oral or written opinions and comments may be presented at the public hearing. Alternatively, written comments may also be submitted to Tom Martin, MDT Environmental Services, at 2701 Prospect Avenue, PO Box 201001, Helena, MT 59620-1001, or online at http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

The review period for the EA will conclude on December 19, 2008. All public comments are due by December 19, 2008.

The proposed Corridor Study - Red Lodge North project would be constructed from the city of Red Lodge, MT to approximately 1,000 feet north of the community of Boyd, MT to accomplish the following objectives:

- * Improve pavement condition along the project corridor
- * Improve intersection geometry at key locations
- * Manage access in north Red Lodge
- * Accommodate pedestrian/bicycle users in Red Lodge
- * Improve level of service in rural segments and at two intersections in Red Lodge
- * Provide wider shoulders in rural segments
- * Flatten ditch inslopes in rural segments
- * Reduce encroachments within clear zones where appropriate
- * Increase ditch storage for snow/ice off the roadway in rural segments
- * Reduce frequency of animal-vehicle collisions along project corridor
- * Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program or activity of our department. If you require reasonable accommodations to participate in this hearing, please contact Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired, the TTY number is (406) 444-7696 or 1 (800) 335-7592, or call Montana Relay at 711. Alternative accessible formats of pertinent information will be provided upon request.

-----END------

Project name: Corridor Study - Red Lodge North Project ID: STPP 28-2(25)70 Control Number: 4375 Carbon County

Becky Rude

From:	Grant, Paul [pgrant@mt.gov]
Sent:	Monday, November 24, 2008 8:50 AM
То:	KMXE-FM; Red Lodge - Carbon County News (E-mail)
Cc:	Priebe, Gabe; Streeter, Stefan; Neville, Gary; Martin, Tom; Bruner, Heidy; Gocksch, Thomas;
	Watt Levis, Charity; Grant, Paul; James, Carl; mike.wamboldt@kljeng.com;
	'becky.rude@kljeng.com'; County Commissioners
Subject:	MDT announces Notice of Availability and Public Hearing: Corridor Study - Red Lodge North Environmental Assessment

November 24, 2008

FOR IMMEDIATE RELEASE

For further information, contact:

Mike Wamboldt, Project Manager, Kadrmas, Lee and Jackson, (406) 245-5499 Stefan Streeter, MDT Billings District Administrator, (406) 252-4138 Tom Martin, MDT Environmental Services, (406) 444-0879 Paul Grant, MDT Public Involvement, (406) 444-9415

MDT announces Notice of Availability and Public Hearing: Corridor Study - Red Lodge North Environmental Assessment

Carbon County - The Montana Department of Transportation (MDT) has prepared an Environmental Assessment (EA) for the proposed Corridor Study - Red Lodge - North project and will conduct two (2) Public Hearings on that EA.

The Montana Department of Transportation (MDT) and the Federal Highway Administration (FHWA) invite all interested parties to review the EA and provide comments at a Public Hearing on either Tuesday, December 9, 2008 at the Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge, or Wednesday, December 10, 2008 at the Roberts School Cafeteria, 106 East Maple Street, Roberts. An open house will begin at 6:00 p.m. with an informational presentation at 6:30 p.m. Project staff will take questions to clarify the understanding of the EA, then will proceed to a formal Public Hearing. The presentation will summarize the project history, present the Preferred Alternative, and describe the environmental process. Both Public Hearings will follow the same format and agenda. The same presentation will be given on both evenings.

Anyone interested in viewing the EA may view it online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml or at one of the following locations:

- * Carnegie Library, 3 West 8th Street, Red Lodge
- * Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- * Roberts Public School, 106 East Maple Street, Roberts
- * Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- * MDT Billings District Office, 424 Morey Street, Billings
- * Montana State Library, 1515 East 6th Avenue, Helena
- * MDT Environmental Services Office 2701 Prospect Ave., Helena
- * Online at www.mdt.mt.gov/pubinvolve/eis-ea.shtml
- * Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Community participation is a very important part of the process, and the public is encouraged to attend. Oral or written opinions and comments may be presented at the public hearing. Alternatively, written comments may also be submitted to Tom Martin, MDT Environmental Services, at 2701 Prospect Avenue, PO Box 201001, Helena, MT 59620-1001, or online at http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

The review period for the EA will conclude on December 19, 2008. All public comments are due by December 19, 2008.

The proposed Corridor Study - Red Lodge North project would be constructed from the city of Red Lodge, MT to approximately 1,000 feet north of the community of Boyd, MT to accomplish the following objectives:

- * Improve pavement condition along the project corridor
- * Improve intersection geometry at key locations
- * Manage access in north Red Lodge
- * Accommodate pedestrian/bicycle users in Red Lodge
- * Improve level of service in rural segments and at two intersections in Red Lodge
- * Provide wider shoulders in rural segments
- * Flatten ditch inslopes in rural segments
- * Reduce encroachments within clear zones where appropriate
- * Increase ditch storage for snow/ice off the roadway in rural segments
- * Reduce frequency of animal-vehicle collisions along project corridor
- * Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program or activity of our department. If you require reasonable accommodations to participate in this hearing, please contact Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired, the TTY number is (406) 444-7696 or 1 (800) 335-7592, or call Montana Relay at 711. Alternative accessible formats of pertinent information will be provided upon request.

-----END-----END------

Project name: Corridor Study - Red Lodge North Project ID: STPP 28-2(25)70 Control Number: 4375 Carbon County

Becky Rude

From:	Grant, Paul [pgrant@mt.gov]
Sent:	Monday, December 01, 2008 8:51 AM
To:	aaron murphy ktvq; Becky Bohrer (E-mail); Billings Gazette; Billings Outpost; Billings Times; KBBB/KBUL/KCTR/KKBR-FM/KMHK-FM; KBLG; KBLG-KRKX-KRZN-KYYA; KBLW-FM;
	KEMC - NEWS; KEMC - PSA's; KGHL-AM/KGHL-FM/KQBL-FM/KRSQ-FM/KZRV-FM;
	KHMT-TV; KMHK; KPBR-FM/KPLN-FM/KWMY-FM; KRZN; KRZN 2; Kulr 8 - Billings (E-mail);
	KULR/KMZK-AM; KULR-TV; KYYA; Road Watch Montana
Cc:	Priebe, Gabe; Streeter, Stefan; Neville, Gary; Martin, Tom; Gocksch, Thomas; Bruner, Heidy;
	James, Carl; Woodmansey, Alan; Watt Levis, Charity; Grant, Paul;
	mike.wamboldt@kljeng.com; 'becky.rude@kljeng.com'
Subject:	MDT announces Notice of Availability and Public Hearing: Corridor Study - Red Lodge North
	Environmental Assessment

December 1, 2008

FOR IMMEDIATE RELEASE

For further information, contact:

Mike Wamboldt, Project Manager, Kadrmas, Lee and Jackson, (406) 245-5499 Stefan Streeter, MDT Billings District Administrator, (406) 252-4138 Tom Martin, MDT Environmental Services, (406) 444-0879 Paul Grant, MDT Public Involvement, (406) 444-9415

MDT announces Notice of Availability and Public Hearing: Corridor Study - Red Lodge North Environmental Assessment

Carbon County - The Montana Department of Transportation (MDT) has prepared an Environmental Assessment (EA) for the proposed Corridor Study - Red Lodge - North project and will conduct two (2) Public Hearings on that EA.

The Montana Department of Transportation (MDT) and the Federal Highway Administration (FHWA) invite all interested parties to review the EA and provide comments at a Public Hearing on either Tuesday, December 9, 2008 at the Red Lodge Senior Citizens Center, 207 South Villard Avenue, Red Lodge, or Wednesday, December 10, 2008 at the Roberts School Cafeteria, 106 East Maple Street, Roberts. An open house will begin at 6:00 p.m. with an informational presentation at 6:30 p.m. Project staff will take questions to clarify the understanding of the EA, then will proceed to a formal Public Hearing. The presentation will summarize the project history, present the Preferred Alternative, and describe the environmental process. Both Public Hearings will follow the same format and agenda. The same presentation will be given on both evenings.

Anyone interested in viewing the EA may view it online at www.mdt.mt.gov/pubinvolve/eis_ea.shtml or at one of the following locations:

- * Carnegie Library, 3 West 8th Street, Red Lodge
- * Red Lodge City Hall, 801 North Broadway Street, Red Lodge
- * Roberts Public School, 106 East Maple Street, Roberts
- * Kadrmas, Lee and Jackson, Inc., 2611 Gabel Road, Billings
- * MDT Billings District Office, 424 Morey Street, Billings
- * Montana State Library, 1515 East 6th Avenue, Helena
- * MDT Environmental Services Office 2701 Prospect Ave., Helena
- * Online at www.mdt.mt.gov/pubinvolve/eis-ea.shtml
- * Call MDT Environmental Services at (406) 444-7228 for a copy of the EA

Community participation is a very important part of the process, and the public is encouraged to attend. Oral or written opinions and comments may be presented at the public hearing. Alternatively, written comments may also be submitted to Tom Martin, MDT Environmental Services, at 2701 Prospect Avenue, PO Box 201001, Helena, MT 59620-1001, or online at http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

The review period for the EA will conclude on December 19, 2008. All public comments are due by December 19, 2008.

The proposed Corridor Study - Red Lodge North project would be constructed from the city of Red Lodge, MT to approximately 1,000 feet north of the community of Boyd, MT to accomplish the following objectives:

- * Improve pavement condition along the project corridor
- * Improve intersection geometry at key locations
- * Manage access in north Red Lodge
- * Accommodate pedestrian/bicycle users in Red Lodge
- * Improve level of service in rural segments and at two intersections in Red Lodge
- * Provide wider shoulders in rural segments
- * Flatten ditch inslopes in rural segments
- * Reduce encroachments within clear zones where appropriate
- * Increase ditch storage for snow/ice off the roadway in rural segments
- * Reduce frequency of animal-vehicle collisions along project corridor
- * Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts

MDT attempts to provide accommodations for any known disability that may interfere with a person's participation in any service, program or activity of our department. If you require reasonable accommodations to participate in this hearing, please contact Paul Grant at (406) 444-9415 at least two days before the hearing. For the hearing impaired, the TTY number is (406) 444-7696 or 1 (800) 335-7592, or call Montana Relay at 711. Alternative accessible formats of pertinent information will be provided upon request.

-----END------

Project name: Corridor Study - Red Lodge North Project ID: STPP 28-2(25)70 Control Number: 4375 Carbon County



EA Distribution List

Federal Agencies

US Army Corps of Engineers Billings Regulatory Office PO Box 2256 Billings, MT 59103 Attn: Shannon Johnson, Program Manager

U.S. Department of Interior U.S. Fish and Wildlife Service 100 N Park, Suite 320 Helena, MT 59601 Attn: Scott Jackson, Wildlife Biologist

State Agencies

Montana Department of Environmental Quality Water Protection Bureau PO Box 200901 Helena, MT 59620-0901 Attn: Bonnie Lovelace, Bureau Chief

Montana Department of Environmental Quality Director's Office PO Box 200901 Helena, MT 59620-0901 Attn: Thomas Ellerhoff, Administrative Officer

Montana Fish, Wildlife & Parks Region 5 2300 Lake Elmo Drive Billings, MT 59105 Attn: Gary Hammond, Regional Supervisor

Montana Fish, Wildlife & Parks Habitat Protection Bureau 1420 E 6th Ave Helena, MT 59620-0701 Attn: Doug McDonald, SPA Coordinator

Montana Department of Natural Resources and Conservation State Water Projects Bureau PO Box 201601 Helena, MT 59620-1601 Attn: Jim Domino, Environmental Coordinator U.S. Department of Agriculture Natural Resources Conservation Service PO Box 510 Joliet, MT 59041-0229 Attn: Gordon Hill, District Conservationist

U.S. Environmental Protection Agency Montana Operations Office 10 W 15th St, Suite 3200 Helena, MT 59626 Attn: Judy Hanson, Acting Administrator

Montana Department of Environmental Quality Remediation Division PO Box 200901 Helena, MT 59620-0901 Attn: Sandra Olsen, Administrator

Montana Fish, Wildlife & Parks Region 5 2300 Lake Elmo Drive Billings, MT 59105 Attn: Jim Olsen, Fisheries Biologist

Montana Fish, Wildlife & Parks Parks Division 1420 E 6th Ave Helena, MT 59620-0701 Attn: Douglas Habermann, Regional Parks Manager

Montana Department of Natural Resources and Conservation State Water Projects Bureau PO Box 201601 Helena, MT 59620-1601 Attn: Kevin Smith, Bureau Chief

Montana Department of Natural Resources and Conservation Water Resources Division, Billings Regional Office 1371 Rimtop Dr Billings, MT 59105-1978 Attn: Keith Kerbel, Regional Manager Montana Department of Natural Resources and Conservation Water Resources Division PO Box 201601 Helena, MT 59620-1601 Attn: John Tubbs, Administrator

Local Agencies

Carbon County Commissioners PO Box 887 Red Lodge, MT 59068

<u>Private</u>

James Noe PO Box 787 Red Lodge, MT 59068

Vickie Westrick PO Box 1063 Red Lodge, MT 59068

Micky Hawman 7615 US Highway 212 Red Lodge, MT 59068

Roberts Ranch Supply 109 S 1st Street Roberts, MT 59070 Red Lodge City Council PO Box 9 Red Lodge, MT 59068

Fred Javid PO Box 111 Red Lodge, MT 59068

Richard Kording PO Box 61 Red Lodge, MT 59068

Colleen Cook 1000 S Idaho Rd PMB 356 Apache Junction, AZ 85219

Casey Hill 7605 US Highway 212 Red Lodge, MT 59068 ARE CONSIDERED PUBLIC RECORDS.

PUBLIC HEARING CORRIDOR STUDY - RED LODGE NORTH NOTE: SIGN-IN SHEETS DECEMBER 9, 2008 6:00-8:00PM (MST) RED LODGE SENIOR CITIZENS CENTER ENVIRONMENTAL ASSESSMENT RED LODGE, MT CARBON COUNTY, MT PROJECT # STPP-28-2(25)70, CN 4375 BUSINESS/ORGANIZATION ZIP CODE E-MAIL NAME STREET ADDRESS CITY/STATE Please Print RAD LODGE MT Ave South 59068 lindsy jourson Que com lity of Kell Lodge indsy Jourson 59601 Helena, MT MDT - Environmental Burenn Toocksch @ MT. gov 2701 Prospert Au Gocksch Tom MHA HOSPITAL 23 PAUSADES CAMPEROUND DR. RES CONGEM 59068 JOHN TOLERY MT. NET TOHN TOLITZ ip-dbenedette OMSK. 500 59068 eff DiBenedetto 317 Haggin Ave S 10,000 Plannin Bound 59068 marcella@ cbthebrokers.com Hur ZIZ ducell Branker Ked Lodgo apple 71. Hause Box 5-98 59068 etatoya aven, com Rh Eco Dev Corp. 401 RI 14 And 11 ((11 11 11 40 ANDY ANDERSEN CITIZEN 0 59065 barbbecke cablent.net BOX STO Jans Beak 11 400 W 2151 Redhodgi 59068 KEVANSCONTE . 059 St Brantoth Hospital Collect Treama 59068 23 Biorn galeherzberg @ gmail. Cc -itizan 0192 FAND HERZBERG Hespita Segreath HILDERMAN VISh 59068 N' Broadward Lodar Spis (Idladgespose yuhoc. Com Storda Martin 1102 Alle Red lodse Ron 59063 - Jastan Bacher News Deiten contantos . Ce Canbu Cater, Nevs colie 106 5 ADAMS AVE 59069 TOME MAPMURALS. COM R.L. TOM KOHLEY MATY R.L. PALKS BARD 59068 RL aarow predlodge fore.com 801 N Browdowing Lodge Fire Rescue Aaron MeDowell Red 54668 nijschoen. Ve Quwalumni. com RL Lodge C.t. Planning Boud hanning Colt N. Acagona AVP Michael J. Scheenike bab. shannon@kljeng.com MT Billing 2611 GeberRS Bob Shannon Kad may lock Jackson Box 2200 - 59068 dachrist@aol.com Faithful RL ALAA'S LLC MAT -less
NOTE: SIGN-IN SHEETS ARE CONSIDERED PUBLIC RECORDS.

CORRIDOR STUDY - RED LODGE NORTH

ENVIRONMENTAL ASSESSMENT

PROJECT # STPP-28-2(25)70, CN 4375

CARBON COUNTY, MT

BUSINESS/ORGANIZATION NAME STREET ADDRESS CITY/STATE Please Print MT Kadrmas, Lee & Jackson 2611 Gable Rol, Billings Mike Wamboldt Red Lodge Planning/ZUNIKE BOARd 206 N. HARSIN, R-L. MIT PAN GAINEN mt 522 N And (XX 182 21 Silline 2228 ine st nkins 20 N Plat Locks BRTA Thompson oven

PUBLIC HEARING DECEMBER 9, 2008 6:00-8:00PM (MST) RED LODGE SENIOR CITIZENS CENTER RED LODGE, MT E-MAIL ZIP CODE 59102 Mike. Wamboldte Kljeng. com 59068 departer @ insheo.com 59068 59101 59068 Corey thomas a comail, com

NOTE: SIGN-IN SHEETS ARE CONSIDERED PUBLIC RECORDS.

CORRIDOR STUDY - RED LODGE NORTH

ENVIRONMENTAL ASSESSMENT

PROJECT # STPP-28-2(25)70, CN 4375

CARBON COUNTY, MT

NAME	BUSINESS/ORGANIZATION	STREET ADDRESS	CITY/STATE
Please Print Becky Rude Kathy Junes Konny Jethsonne Jane gunbun Rogen zumbun Mary aller	KLS	2611 Carle Rd 182 Shorey Rd 1883 HWY 212 8769 HWY 212 8769 HWY 212 8769 HWY 212 7981 Highway 212	Billings, MT Roberts Roberts Roberts Roberts Roberts

PUBLIC HEARING DECEMBER 10, 2008 6:00-8:00PM (MST) ROBERTS SCHOOL CAFETERIA ROBERTS, MT ZIP CODE E-MAIL 59102 becty.rule (~ etq. com 59070 59070 Kenpe CB The Brokers, Ca. 59070 JZUMBRUNICYA 59070 59070

NOTE: SIGN-IN SHEETS ARE CONSIDERED PUBLIC RECORDS.

CORRIDOR STUDY - RED LODGE NORTH

ENVIRONMENTAL ASSESSMENT

PROJECT # STPP-28-2(25)70, CN 4375 CARBON COUNTY, MT **BUSINESS/ORGANIZATION** NAME CITY/STATE STREET ADDRESS Please Print Poberts Sanford hangoon #5 Comer R. MOST - Hay Cattle Vanch Gable Roa Billings Shannon 2611 mas Loo Jackson 11 1 . adrmas Lee & Jackson Jambold F like Rodladge PU Sor Cuty of iled bodge RL P.D. Br. 1976 NANDA ENALICOTT Dunal - ICec Roberts 8079 US Highway 212 Loc Ke Fav ason Priest City of Red Locke 602 N. Hauser RL Roberts. 8057 45 Hishway Z12 James Binando RURFRETS 7883 11 5 MWY 212 JACKY JEANSONNE Roberts RIM S Karlway 172 Horn Robert 122 Raulway 12h YOWE 28 ROBERTS ROAD A COONEY AP Roberts tighway 212 7.98 Allen 120 MDT - Environment im Gocksch WINA

PUBLIC HEARING DECEMBER 10, 2008 6:00-8:00PM (MST) **ROBERTS SCHOOL CAFETERIA** ROBERTS, MT ZIP CODE E-MAIL 59020 bob.shannon@kljeng.com 59102 15 mike. wambold + Q klipng. com 59068 reshold an and on the Man. com 59068 Kennicottw/ callent, not 59070 real lodge word I@ Smill. com 59068 59070 ibing noto @ Stillwater mining com 59670 lazyKJ @ worldnet.att.ne 58070 Hogan 1050 Comsh. Cen 54070 59070 59070

PARTIAL TRANSCRIPT OF PUBLIC HEARING

CORRIDOR STUDY - RED LODGE NORTH ENVIRONMENTAL ASSESSMENT CN 4375

Red Lodge Senior Citizen Center 207 Villard Avenue Red Lodge, Montana 6 pm – 8 pm December 9, 2008

FORMAL HEARING

Dennis Christ: Is it possible for you to put up the slide showing the section of roadway from Beckwith by Sam's Tap Room? What it appears is that coming from 78 down here (referring to graphic) the roundabout at 78 going north is a two-way turn lane going up this far. Sam's Tap is right about here and there are two more lots in here, and then you hit the Golf Course, and then there is this whole structure on the Golf Course where there is only a need for a right-in/right-out at this point here. It seems to me that it would make a lot more sense to either continue that two-way left-turn lane another few hundred feet to this point and then do whatever you're going to do there. It's a very arbitrary position to make that change down here. My reason is that hopefully within the next two to three years, there will be a very nice large church on this property with people wanting to come in and out from both north and south.

Kelly Evans: I'm the CEO of Beartooth Hospital and Health Center. I too would like to address the access management overview, specifically in the immediacy of the construction of the new facility. We chose this site; this green dot represents the three-quarter entrance that would become the primary access for Beartooth Hospital and Health Center (referring to graphic) with the hospital being built approximately in this area. This three-quarters access would preclude us from allowing an ambulance to leave the facility and immediately make a left-hand turn onto Hwy 212 transporting a patient to Billings. Of the 2,000 approximate patients that we see at Beartooth Hospital and Health Center about 220 per year are of some sort of medical or trauma need that requires immediate transfer to another care area primarily by ground ambulance. By precluding this left-hand turn, you're asking the ambulance to come back into a roundabout and then leave for Billings, or utilize the service road which is being proposed for the backside of the hospital. The purpose of that service road was to receive service trucks and employees and that in fact is where their parking is and where the helipad would also be located. So we have some considerable concerns about the safety of re-directing ambulances through that back part of the hospital. In addition we feel that patients are creatures of habit. By not allowing this left-hand turn back to Billings and creating this roundabout across from Two Mile Bridge Road into a service road, that we may be misrepresenting the primary access to the hospital and thereby confusing the patients at a time they would not need to be confused. Additionally if there is a discussion to bring Hauser out, the intersection that has been proposed to us is actually right now here (referring to graphic). If Hauser is extended this way and this is not allowable as a turn onto the highway, we're going to have an extensive amount of traffic to deal with on what was intended primarily to be a service road. We would like to see a reconsideration or a compromise reached to either allow a left-hand turn or consider a full roundabout there so we are not preventing traffic from leaving the hospital with no direct access to go north. Thank you for the opportunity to comment. We would like to supply some traffic flow information and patient utilization information regarding the amount of patients that seek care at Beartooth Hospital and Health Center whose primary residence is not Red Lodge. Thank you.

Tom Kohley: First I want to thank you guys at MDT and KLJ for taking into consideration the Trail Plan the City developed several years ago. You guys have honored that pretty well. I think it is fair to say that the citizens of Red Lodge do want safe pedestrian and bicycle lanes in and around our community. My comments will center around some new developments and new opportunities within the Trail System that aren't reflected in the current Trail Plan. I want to recommend that you consider an extension of the proposed Trail System right now to Two Mile Bridge Road. Just north of Red Lodge there has been a new residential subdivision called Spires which has now been annexed into the city. It's a piece of land that almost extends from 212 all the way over to Willow Creek Road which is directly north of Hwy 78. In that plan they have planned for a pretty extensive Trial System running the full length from west to east. They've also secured an easement going down into the 212 corridor through the old service road which terminates at the approach at the round barn. This is only less than one half mile distance from the proposed termination of the Bike/Ped Path. If there is any way possible to extend that Bike and Ped Path, preferably on the west side of the highway, another half mile so that we can connect into that easement at the round barn, we would then have almost a near full loop going back out to Hwy 78. I've talked to you about the potential for an additional trail in the right-ofway along Hwy 78; that would give the citizens of Red Lodge a full circular path which we would like to see developed all the way along the north corridor. So if there is any way possible to consider a further extension of that trial, we would appreciate it.

Jeff DiBenedetto: I'm concerned about that section with the raised median and the light posts going down the center of the highway and the seemingly inconsistency with the stated goals of our Growth Policy relative to community entrances. Particularly in terms of maintaining the uniqueness of Red Lode which is one of the stated goals in the Growth Policy and not having a community entrance that is typical of every other town in the Untied States. When I see that drawing of that raised median and a center row of lights going down, what that tells me is that we are going to make this north entrance into Red Lodge just like any other entrance, like any other town in the United States. It will loose its uniqueness. We are going to loose the visual quality that we currently experience as we come into town. I'm speaking primarily of being able to look straight south and have a clear unobstructed view of Mt. Maurice in the backdrop with the Beartooths behind the City. Coming into town now we're going to see a string of light poles and I think it is inconsistent with our current growth policies. So I would like to see a reevaluation of that alternative to better take into account consistency with our growth policy goals and particularly take a close look at the visual impacts and make those consistent with our Growth Policy.

Andy Anderson: It seems to me that all too often we forget all the planning and hours and creativity that go into the development of a project as big as this one, even though it is for a small rural town in Montana. I would just like to say I think that the staff of the Highway Department in Montana and the firm that helped develop the design phase have done a really good job so far. We have a long ways to go. I really appreciate, as a citizen, what's been done to this point.

CLOSING

Does anyone else have any comments? I will close the hearing at this time. Remember that you have until December 19th to get your comments in. We will be around if you have any questions. Thank you for coming tonight; we appreciate your taking the time to do this.

PARTIAL TRANSCRIPT OF PUBLIC HEARING

RED LODGE NORTH ENVIRONMENTAL ASSESSMENT CORRIDOR STUDY CN 4375

Roberts School Cafeteria Roberts, Montana 6 pm – 8 pm December 10, 2008

FORMAL HEARING

Kenny Jeausonne: I appreciate everything you guys have done. I feel, as I've mentioned before, that we are in a state of limbo should my wife and I decide that we wanted to sell our property I feel tied because we cannot do anything with this sort of disclosure. I think everyone along this corridor is going to be affected by it. The people on the west side of that road, it just does not make sense to me, and I'm not the brightest star in the sky, but that east side of the road where the railway bed is. That's all I have to say. Thank you for your time.

Betsy Scanlin: I am the mayor of Red Lodge. I speak sort of as the mayor; as a resident of Red Lodge. I don't speak for the council or any individual members of the council. I presume they have their own feelings about this. Just a comment I think that... (inaudible)...about your proposed alternative, your preferred one, on Oakes Street in Red Lodge. Oakes Street currently, most of you know here, is the second Main Street for Red Lodge. We have many of our big facilities, a super market, grocery market, we have the Health Care Center, we have a fairly small box store, a hardware center, and we have a post office, all on Oakes Street. That has traditionally been a through-way, so when you are talking about putting a stop sign there, you are dramatically changing the traffic flow there, which would cause them to go down Main Street which is more of a retail area. Turn off Main Street and put a stop sign on one of the side streets onto Oakes. So I would ask you to consider keeping Oakes a through-way, eliminating the parking on the east side of Oakes next to the Library, and still allow the other parking on the west side, that does meet the parking objective and continuing the flow on Oakes towards the south. The other part of that is when you stop the traffic on Oakes there's really only space for two cars on 8th Street to go on and off Beartooth Market flow. It's a very congested corner. You are going to put a lot of traffic onto the corner of 8th and Broadway with your alternative—your proposed alternative. So that's just a statement from my perspective. I know that we've hashed over this for a long time and I appreciate all the opportunities we've had to be into this, but before we change the traffic pattern in Red Lodge, I would like you to reconsider that treatment at 8th and Oakes. Thanks.

CLOSING

Does anyone else have any comments? We will close the hearing at this time. This will conclude the hearing. Again we appreciate your coming out tonight and being present for this hearing.

Remember you have until December 19th to get your comments in. We will be around if you have any questions. Thank you for coming.

Appendix B

Agency and Public Comments and Responses

Agency Comments	
Montana Fish, Wildlife & Parks	B-1
Carbon County	
US Fish and Wildlife Service—MT Field Office	B-5
Public Hearing Comments	
Andy Anderson	B-5
Dennis Christ	B-7
Jeff DiBenedetto	B-7
Kellev Evans	B-9
, Kenny Jeansonne	B-9
Tom Kohley	B-11
Betsy Scanlin	B-11
Written/Online Comments	
James M. Allen	B-13
Joseph A. Allen	B-15
Mary L. Allen	B-17
Grant Barnard	B-19
Barb Beck	B-21
R.L. Bertolino	B-25
Elta Bertolino Ayre	B-29
John Clayton	B-33
Jeff DiBenedetto	B-35
Kelley Evans, Mitch Goplen, James G. Shephard	B-41
Curtiss Jay Fleck	B-49
Les Hedquist	B-51
Rand Herzberg	B-53
Sam Hoffmann	B-55
William M. and Deborah Hogan	B-57
Harry Hollman	B-61
Jacqueline and Kenny Jeansonne	B-63
Louise Jenkins	B-67
Lindsy Johnson	B-71
Mary Johnson	B-73
Tom Kohley	B-75
Ray and Louise Locke	B-77
Marcella Manuel	B-79
Ray Masters	B-83
Dennis Meeker	B-83
Joseph Nichols	B-85
Traute Parrie	B-87
Janet Peterson	B-87
Jason S. Priest	B-89
Betsy Scanlin	B-93
Curtis Schwend	B-97
Dave Stauffer	B-101
Corey Thompson	B-103
David Torgerson	B-107
Luke Wright	B-109
Lyle Wright	B-109

COMMENT #1



2300 Lake Elmo Drive Billings, MT 59105

January 7, 2009

Mr. Tom S. Martin, P.E. Bureau Chief, Environmental Services Montana Department of Transportation 2701 Prospect Avenue PO Box 201001 Helena, MT 59620-1001

SUBJECT: Environmental Assessment for Proposed Transportation Project Corridor Study – Red Lodge North

Dear Mr. Taft:

Please consider this a response to your request for information on the fisheries and wildlife resources relative to your proposed project.

- X B ased on a review of the proposed project, the Region 5 Wildlife Staff has no specific comments.
- R eview of the proposed project resulted in concerns about impacts to wildlife as summarized in the comment section below.
- <u>X</u> B ased on a review of the proposed project, the Region 5 Fisheries Staff has no specific comments relating to the potential impacts on fisheries habitat.
- ____R eview of the proposed project resulted in concerns about impacts to fisheries habitat as summarized in the comment section below:

COMMENTS:

The type of fisheries habitat in the proposed project's vicinity is described in the Montana Interagency Stream Database. Montana Fish, Wildlife & Parks reserves the right to address any impacts to any stream or its banks or tributaries by any type or form of construction when detailed plans are furnished with an application as required by the Stream Protection Act and the Natural Streambed and Land Preservation Act.

Sincerely,

Ray Mule

Wildlife Division

Ken Frazer

Fisheries Division

Glenn Phillips

B-1

C:

RESPONSE #1

Thank you for your comment.

Debra Hronek Historic Preservation Officer Carbon County, MT

December 8, 2008

- To: Jason Priest, ward 1 Councilman, Red Lodge, MT Montana Department of Transportation
- RE: Red Lodge North EA & Sec 4(f) Evaluation

I find myself unable to attend the December 9th, 2008 hearing to be held with MDOT in Red Lodge concerning the Highway 212 Corridor project and wish to comment on this project as it affects the community of Red Lodge as well as the Cultural Resources located within Carbon County.

As is noted in the Red Lodge North EA & Sec 4(f) Evaluation, there is a National Register listed Historic Commercial District on Highway 212 Red Lodge. The proposed plan includes a two-way left turn lane on the north entrance to Red Lodge. The Commercial Historic District currently has two-lane with parallel parking on either side. The proposed plan would consist of two-lane roadway, changing to two-lane roadway with a two-way left turn lane, and then back to two-lane roadway as highway 212 enters Red Lodge. This proposed plan shows a lack of continuity in the design. Additionally, the two-way left turn lane may have the effect of speeding up traffic on the outskirts of Red Lodge, rather than slowing the traffic. Two-way left turn lanes are not pedestrian friendly, and Red Lodge is a community that has many pedestrians.

Red Lodge adopted a new Growth Policy in 2008. This growth policy effectively limits access to Highway 212 in ways that are consistent with the network of city streets. I feel that the MDOT plan should be updated to reflect this new growth policy.

The research into the many Cultural Resources that are located with the project area has been well researched and documented. As with any project there is always a chance that something unforeseen will occur. I would request that any new developments or discoveries during the course of this project, concerning historic or prehistoric sites be brought to the attention of the Carbon County Historic Preservation Officer in a timely manner.

Debra Hronek Carbon County Historic Preservation Officer

> P.O. Box 881 224 N. Broadway Red Lodge, MT 59068 Country

PHONE (406) 446-3667 EMAIL preservation@carboncountyhistory.com

2а

The two-way left-turn lane (TWLTL) was prepared in coordination with the public and the Red Lodge City Council. Subsequently, two resolutions (No. 3223 and No. 3228) were passed by the City Council accepting the TWLTL.

MDT needs to plan its immediate roadway improvements to accommodate what is there now. There are approximately 21 existing access points along this part of the corridor. The mix of through traffic attempting to progress along the corridor with traffic turning on and off the corridor at these access points creates a safety concern. These differential speed conflicts and high percentage of following traffic create a situation that is conducive to rear-end collisions. MDT conducted a traffic analysis, which identified that a TWLTL would improve safety conditions along the corridor. The TWLTL is intended to give turning vehicles a place to slow down or wait prior to making turning movements.

The City can petition the Montana Traffic Commission to revise access management in the area. The proposed typical section includes adequate width to accommodate a re-striping from the TWLTL to two parking lanes in the future, if and when existing conditions change such that that layout would better fit the needs of those using the corridor.

2b

Developments wishing to access US Highway 212 would need to submit an approach permit application and development proposal to MDT for review and approval. These approaches would be subject to MDT's approach standards for highways, which includes the standard practice of minimizing the number of driveways accessing a state highway in order to reduce the potential for traffic conflicts. The purpose of the Access Management Plan was to minimize and manage access to US Highway 212.

2с

In order to identify cultural resources along the project corridor and to determine potential impacts, three cultural resource surveys were undertaken, as was a historical inventory within Red Lodge and Roberts. MDT has determined, and the State Historic Preservation Office has concurred, that the proposed project would have No Effect or No Adverse Effect to all of the historic properties identified during the cultural resource and historic inventories. If cultural resources are discovered during construction, construction in the vicinity of the resource would be stopped and the State Historic Preservation Officer and Carbon County Historic Preservation Officer would be contacted.

COMMENT #3

From: Scott_Jackson@fws.gov [mailto:Scott_Jackson@fws.gov] Sent: Thursday, December 18, 2008 11:43 AM To: Martin, Tom Cc: Anne_Vandehey@fws.gov Subject: Red Lodge - North (CN 4375)

Hi Tom,

Just a quick note to let you know that, responding to your November 12, 2008 request, I've taken a look at the EA for the Red Lodge - North Corridor Study. I think it looks fine from my perspective and the FWS has no substantive comments to pass along relative to this document.

Have a great holiday season!

Scott

Scott Jackson, Fish and Wildlife Biologist U.S. Fish and Wildlife Service Montana Field Office 585 Shepard Way Helena, Montana 59601 (406)449-5225, extension 201 (406)449-5339 (fax) scott jackson@fws.gov

COMMENT #4

Andy Anderson

It seems to me that all too often we forget all the planning and hours and creativity that go into the development of a project as big as this one, even though it is for a small rural town in Montana. I would just like to say I think that the staff of the Highway Department in Montana and the firm that helped develop the design phase have done a really good job so far. We have a long ways to go. I really appreciate, as a citizen, what's been done to this point.



Thank you for your comment.

RESPONSE #4

Thank you for your comment.



Dennis Christ

Is it possible for you to put up the slide showing the section of roadway from Beckwith by Sam's Tap
 Room? What it appears is that coming from 78 down here (referring to graphic) the roundabout at 78 going north is a two-way turn lane going up this far. Sam's Tap is right about here and there are two more lots in here, and then you hit the Golf Course, and then there is this whole structure on the Golf Course where there is only a need for a right-in/right-out at this point here. It seems to me that it would make a lot more sense to either continue that two-way left-turn lane another few hundred feet to this point and then do whatever you're going to do there. It's a very arbitrary position to make that change down here.
 My reason is that hopefully within the next two to three years, there will be a very nice large church on this property with people wanting to come in and out from both north and south.

COMMENT #6

Jeff DiBenedetto

I'm concerned about that section with the raised median and the light posts going down the center of the highway and the seemingly inconsistency with the stated goals of our Growth Policy relative to community entrances. Particularly in terms of maintaining the uniqueness of Red Lodge which is one of the stated goals in the Growth Policy and not having a community entrance that is typical of every other town in the United States. When I see that drawing of that raised median and a center row of lights going down, what that tells me is that we are going to make this north entrance into Red Lodge just like any other entrance, like any other town in the United States. It will loose its uniqueness. We are going to loose the visual quality that we currently experience as we come into town. I'm speaking primarily of being able to look straight south and have a clear unobstructed view of Mt. Maurice in the backdrop with the Beartooths behind the city. Coming into town now we're going to see a string of light poles and I think it is inconsistent with our current growth policies. So I would like to see a reevaluation of that alternative to better take into account consistency with our Growth Policy.

RESPONSE #5

5a

A two-way left-turn lane (TWLTL) was initially considered for the entire stretch between MT Highway 78 and Two Mile Bridge Road to accommodate turning vehicles accessing adjacent properties while improving progression for through traffic. However, the City of Red Lodge opposed the concept of a TWLTL in the developing area due to concerns that it may encourage commercial strip development.

To accommodate that concern, MDT and the City developed the Access Management Plan, which is discussed in greater detail in Section 2.3.1.4, Access Management, on page 2-14 of the EA. The plan identified that a TWLTL would be used in the developed area. For the developing area, a TWLTL would not be used. Instead, access would be managed along the developing area through the use of intersection types (full, ¾, and ½ access intersections) and a raised median. Full access intersections, where roundabouts are proposed, are spaced approximately a half-mile apart and would accommodate vehicles needing to make U-turns in order to access a property (See Figure 2-10, Access Management Plan Overview, on page 2-17 of the EA). The plan identified locations for future intersections based upon existing access points, property lines, plats, development potential, and input from adjacent property owners. The Red Lodge City Council passed Resolution No. 3228 on March 28, 2007, which supported this plan. Pursuant to applicable Montana statutes and MDT policy, the plan would be recommended to the Montana Transportation Commission for their adoption.

5b

In order for future developments to receive approval from MDT for a proposed approach to US Highway 212, developers and/or individuals will need to go through the system impact action process. A guide to the system impact action process can be found online at the following link: http://mdt.mt.gov/publications/docs/bro-chures/siap_guide.pdf.

RESPONSE #6

A two-way left-turn lane (TWLTL) was initially considered for the entire stretch between MT Highway 78 and Two Mile Bridge Road to accommodate turning vehicles accessing adjacent properties while improving progression for through traffic. However, the City of Red Lodge opposed the concept of a TWLTL in the developing area due to concerns that it may encourage commercial strip development.

To accommodate that concern, MDT and the City developed the Access Management Plan, which is discussed in greater detail in Section 2.3.1.4, Access Management, on page 2-14 of the EA. In order to manage access between the developed limits of Red Lodge and Two Mile Bridge Road, a raised and depressed median were analyzed during development of the Access Management Plan. Through the planning process, it was determined that the raised median was preferred by the City of Red Lodge. The Access Management Plan, which includes the raised median, was supported by Red Lodge City Council Resolutions No. 3223 and 3228, and the City of Red Lodge Growth Policy (2008).

The proposed project would incorporate elements of the Red Lodge Streetscape Plan, as appropriate, and lighting would be designed to minimize light pollution. In addition, there are numerous alternative lighting methods that would allow the median to be lit safely and still allow the City of Red Lodge to maintain its unique entrance into the city. Use of these methods will be determined during design. MDT will work with the City to develop an agreement to address appropriate lighting features to be incorporated into the project.

COMMENT #7

Kelley Evans

I'm the CEO of Beartooth Hospital and Health Center. I, too, would like to address the access management overview, specifically in the immediacy of the construction of the new facility. We chose this site; this green dot represents the three-quarter entrance that would become the primary access for Beartooth Hospital and Health Center (referring to graphic) with the hospital being built approximately in this area. This threeguarters access would preclude us from allowing an ambulance to leave the facility and immediately make a left-hand turn onto Highway 212 transporting a patient to Billings. Of the 2,000 approximate patients that we see at Beartooth Hospital and Health Center, about 220 per year are of some sort of medical or trauma need that requires immediate transfer to a tertiary care area primarily by ground ambulance. By precluding this lefthand turn, you're asking the ambulance to come back into a roundabout and then leave for Billings, or utilize the service road which is being proposed for the backside of the hospital. The purpose of that service road was to receive service trucks and employees and that in fact is where their parking is and where the helipad would also be located. So we have some considerable concerns about the safety of re-directing ambulances through that back part of the hospital. In addition we feel that patients are creatures of habit. By not allowing this left-hand turn back to Billings and creating this roundabout across from Two Mile Bridge Road into a service road, that we may be misrepresenting the primary access to the hospital and thereby confusing the patients at a time they would not need to be confused. Additionally, if there is a discussion to bring Hauser out, the intersection that has been proposed to us is actually right down here (referring to graphic). If Hauser is extended this way and this is not allowable as a turn onto the highway, we're going to have an extensive amount of traffic to deal with on what was intended primarily to be a service road. We would like to see a reconsideration or a compromise reached to either allow a left-hand turn or consider a full roundabout there so we are not preventing traffic from leaving the hospital with no direct access to go north. Thank you for the opportunity to comment. We would like to supply some traffic flow information and patient utilization information regarding the amount of patients that seek care at Beartooth Hospital and Health Center whose primary residence is not Red Lodge. Thank you.

COMMENT #8

Kenny Jeansonne

I appreciate everything you guys have done. I feel, as I've mentioned before, that we are in a state of limbo should my wife and I decide that we wanted to sell our property I feel tied because we cannot do anything with this sort of disclosure. I think everyone along this corridor is going to be affected by it. The people on the west side of that road, it just does not make sense to me, and I'm not the brightest star in the sky, but that east side of the road where the railway bed is. That's all I have to say. Thank you for your time.



The decision whether to construct a full access intersection or a roundabout at the main entrance of the hospital would be determined through the MDT system impact action process. If a roundabout were to be designed, the hospital site plan would need to accommodate the increased right-of-way needs required for a modern roundabout and to avoid impacts to Rock Creek. Additionally, in order to receive approach approval from MDT, the Beartooth Hospital and their representatives will need to go through the aforementioned process, which was initiated by Beartooth Hospital on January 5, 2009. A guide to the system impact action process can be found online at the following link: http://mdt.mt.gov/publications/docs/brochures/siap_guide.pdf.

RESPONSE #8

The projected impacts described in Chapter 3 of the Environmental Assessment were based on preliminary design (approximately 30%) using a standard right-of-way width of 80 feet (See Section 3.7, Right-of-way and Relocations). In order to minimize impacts to various resources, such as wetlands and the historic Rocky Fork Branch of the Northern Pacific Railroad (See Section 3.14, Wetlands, and Chapter 4 of the EA), and through co-ordination with various federal, state, and local agencies, the resulting preliminary design of the project maintained the existing centerline of the roadway. Realignment of the roadway was initially explored at the beginning of the Corridor Study–Red Lodge North project in order to avoid and/or minimize impacts to wetlands and again following public comment. During final design, MDT will make reasonable efforts to avoid and/or minimize impacts to potentially affected property owners. Realignment of the roadway to avoid potential relocations would be determined during final design. MDT will comply with the "Uniform Act", or Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970 (42 U.S.C. 4601 et seq.), and amendments.

COMMENT #9

Tom Kohley

First I want to thank you guys at MDT and KLJ for taking into consideration the Trail Plan the City developed several years ago. You guys have honored that pretty well. I think it is fair to say that the citizens of Red Lodge do want safe pedestrian and bicycle lanes in and around our community. My comments will center around some new developments and new opportunities within the Trail System that aren't reflected in the current Trail Plan. I want to recommend that you consider an extension of the proposed Trail System right now to Two Mile Bridge Road. Just north of Red Lodge there has been a new residential subdivision called Spires which has now been annexed into the city. It's a piece of land that almost extends from 212 all the way over to Willow Creek Road which is directly north of Highway 78. In that plan they have planned for a pretty extensive Trial System running the full length from west to east. They've also secured an easement going down into the 212 corridor through the old service road which terminates at the approach at the round barn. This is only less than one-half mile distance from the proposed termination of the bike/ped path. If there is any way possible to extend that bike and ped path, preferably on the west side of the highway, another half-mile so that we can connect into that easement at the round barn, we would then have almost a near full loop going back out to Highway 78. I've talked to you about the potential for an additional trail in the right-of-way along Highway 78; that would give the citizens of Red Lodge a full circular path which we would like to see developed all the way along the north corridor. So if there is any way possible to consider a further extension of that trial, we would appreciate it.

COMMENT #10

Betsy Scanlin

I am the mayor of Red Lodge. I speak sort of as the mayor; as a resident of Red Lodge. I don't speak for the council or any individual members of the council. I presume they have their own feelings about this. Just a comment I think that... (inaudible)...about your proposed alternative, your preferred one, on Oakes Street in Red Lodge. Oakes Street currently, most of you know here, is the second Main Street for Red Lodge. We have many of our big facilities, a super market, grocery market, we have the Health Care Center, we have a fairly small box store, a hardware center, and we have a post office, all on Oakes Street. That has traditionally been a through-way, so when you are talking about putting a stop sign there, you are dramatically changing the traffic flow there, which would cause them to go down Main Street which is more of a retail area. Turn off Main Street and put a stop sign on one of the side streets onto Oakes. So I would ask you to consider keeping Oakes a through-way, eliminating the parking on the east side of Oakes next to the Library, and still allow the other parking on the west side, that does meet the parking objective and continuing the flow on Oakes towards the south. The other part of that is when you stop the traffic on Oakes there's really only space for two cars on 8th Street to go on and off Beartooth Market flow. It's a very congested corner. You are going to put a lot of traffic onto the corner of 8th and Broadway with your alternative—your proposed alternative. So that's just a statement from my perspective. I know that we've hashed over this for a long time and I appreciate all the opportunities we've had to be into this, but before we change the traffic pattern in Red Lodge, I would like you to reconsider that treatment at 8th and Oakes. Thanks.

Extending the shared bike/ped path to connect with the proposed Spires subdivision trail system was analyzed. However, because of the proposed footprint of the roadway, potential relocations had been previously identified on both sides of US Highway 212 between Two Mile Bridge Road and the proposed Spires Trail. Due to the right-of-way constraints in this area, extending the shared bike/ped path or sidewalk to connect to the Spires subdivision trail system would not be feasible without eliminating the potential to avoid the relocation of at least one of these properties. The trail extension may be explored further during the design phase of the project.

The proposed project would not preclude other entities or individuals from constructing a trail to connect the Spires subdivision trail system to the proposed shared bike/ped path. If this were to occur, the roundabout at the Two Mile Bridge Road intersection would accommodate pedestrians and bicyclists and allow them to access the proposed shared bike/ped path on the east side of the roadway.

Additionally, the proposed 8-foot shoulders along the rural segments of the roadway, such as north of Two Mile Bridge Road, would be wide enough to accommodate pedestrians and bicyclists.

RESPONSE #10

See Response #41.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

Hote, comments are considered public records.	-	-
PLEASE Name: James M. Allen PRINT Address: 7949 U.S. Highwax 212		_
Raberte art 56170	_	-
illiberis, Min Store		SPA
— My name is James M. Allen, and I have land on both sides of	-	
Highway 212, approximately 3 miles south of Roberts.	-	
It would seem to me like it would be more feasible to extend	- 20	
the reconstructed highway to the East omstead of expanding	- 19 (24)	
to both sides of the existing center line After driving this	- 183	
highway, there are very few obstructions, such as houses,		
cattle guards, gates, pipelines and power lines to the east,		
	-	
opposed to the west, where there are dozens. Thank you for	-	
taking the time to consider input from those of us who will be		
directly affected.	- Maria	
	-	
	1.313	
	State of the	
lease mail your comments by December 19, 2008 to:		
Tom S. Martin	ED	
PO Box 201001 MASTER FIL FDEC of a page		
2701 Prospect Ave		
Helena, MT 59620	TAL	
Comments may also be submitted online at:		
http://www.mdt.mt.gov/pubinvolve/eis_ea.shtml	37.736	



See Response #8.

CORRIDOR STUDY - RED LODGE NORTH COMMENTS ENVIRONMENTAL ASSESSMENT PROJECT # STPP-28-2(25)70, CN 43 TUESDAY, DECEMBER 9, 2008—RED LODGE, MT RED LODGE SENIOR CITIZENS CENTER Please use the space below to tell us your comments regarding the proposed Corridor Study - Red Lodge North Project. Note: Comments are considered public records. oseph en Name: PLEASE G 6 Wax Address: PRINT R Oh 1 7. 07 -THIS SPACE OFFICE USE ONLYboth ave Ind On ray nd ne 0 ma 7 He Please mail your comments by December 19, 2008 to: Montana Department of Transportation MASTE RECEIVED Tom S. Martin CON PO Box 201001 DEC 2 2 2008 2701 Prospect Ave Helena, MT 59620 ENVIRONMENTAL Comments may also be submitted online at: http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

COMMENTS

Corridor Study - Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375

Additional space for comments, if needed. Please print.

Note: Comments are considered public records.

-THIS SPACE m OFFICE USE ONLY-19 P 1

RESPONSE #12

See Response #8.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

ller Mary Name: PLEASE 59071 hynau PRINT Address: D First Me Sa -THIS SPACE OFFICE USE ONLYhave n 13a DI one 1-19 DA 136 n OSEIU the n 0 X15 bom on Please mail your comments by December 19, 2008 to: Montana Department of Transportation RECEIVED Tom S. Martin MASTEP ...LE PO Box 201001 DEC 1 8 2008 COFY 2701 Prospect Ave Helena, MT 59620 **ENVIRONMENTAL** Comments may also be submitted online at: http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

CORRIDOR STUDY - RED LODGE NORTH ENVIRONMENTAL ASSESSMENT PROJECT # STPP-28-2(25)70, CN 4375 Additional space for comments, if needed. Please print. Note: Comments are considered public records.

11010.00111	
1	side of Highway 212, Widening to the
-THIS SPACE	pact would supplie supplie the DOT
OFFICE USE ONLY-	this would shreld shre the DOI
and the second	lots of montel since there are there
13b Continued	
CONTRACTOR OF A DESCRIPTION OF A DESCRIP	homes handaking the highwall, a proceeders
the second second second	nomes pordering the monthly approaches
	Und service lines to be moved.
and the second	
A second second second	

RESPONSE #13

13a

Thank you for your comment.

13b

See Response #8.

COMMENT #14

14a

To:	Tom Martin, MDOT
	2701 Prospect Ave., Helena, MT
From:	Beartooth Recreational Trails Association
	Box 1872, Red Lodge, MT
RE:	Comment on Hwy 212, Red Lodge north project

Thank you for taking our comments and for including our requests in the MDOT plans for the Highway 212 project north of Red Lodge. We would like to stress the following as important for 1) improving <u>safety</u> for all users of the corridor, 2) improving <u>exercise</u> opportunities for locals and visitors, 3) and providing <u>alternative transportation</u> opportunities for locals and visitors:

-the separate bike/pedestrian path on the east side, and the sidewalk on the west side of the highway should be as far from traffic lanes as physically possible; painted lines, physical barriers, ditches, or rumble strips would improve safety by separating motor vehicles from bikers & walkers

- -NO MOTOR VEHICLES signs and physical barriers should be placed at entry and exit of bike/pedestrian paths and sidewalks
 -the sidewalk planned on the west side should continue north as
- ^{14d} a bike/pedestrian path to near the Round Barn, so as to connect with the trail coming downhill from The Spires subdivision
- -any lighting planned for this corridor should be subdued and non-polluting to the skies

Thanks again for taking our comments, Grant Barnard President, BRTA 406-446-3755

RESPONSE #14

14a

Comment noted.

14b <mark>– 14</mark>c

The shared bike/ped path and sidewalks would be a minimum of five feet from the travel lanes within the developed limits of Red Lodge (See Figure 2-3, 8th Street to MT Highway 78, and Figure 2-5, MT Highway 78 to Developed Limits of Red Lodge, on pages 2-8 and 2-10 of the EA). In the developing limits of Red Lodge, the shared bike/ped path would be a minimum of 20 feet from the travel lanes (See Figure 2-6, Developed Limits of Red Lodge to Two Mile Bridge Road, on page 2-11 of the EA). The path and sidewalks would be signed as "No Motor Vehicles", in accordance with design standards. Physical barriers would be required if the path were less than five feet from the travel lanes. Actual distance of the shared bike/ped path and sidewalks from the travel lanes, use of physical barriers, and additional signage for paths and sidewalks will be determined during design.

14d

The character of the roadway and user needs helped determine the roadway section between the developed limits of Red Lodge and Two Mile Bridge Road. The length of a sidewalk in this area would be primarily adjacent to the golf course, which does not generate pedestrians. Because there is a lack of existing and foreseen pedestrian generators, it was determined that pedestrian and bicycle facilities on one side of the roadway would be adequate to meet user needs. Roundabouts are proposed along this segment of the roadway and would accommodate pedestrian and bicycle crossings. Therefore, future developments may construct trail extensions to these crossings to link to the proposed bike/ped path on the east side of the roadway.

Extending the shared bike/ped path to connect with the proposed Spires subdivision trail system was analyzed and determined not practicable at this location. Between Two Mile Bridge Road and the Kent Dairy Round Barn, two potential relocations, one on the east side of the roadway and one of the west side of the roadway, have been identified due to the roadway template of the Preferred Alternative (See Figure 2-15, Two Mile Bridge Road to south End of Roberts, on page 2-21 of the EA). Due to right-of-way constraints in this area, the extension of the shared bike/ped path to connect to the Spires subdivision trail would not allow for the potential avoidance or minimization of impacts to these properties during the design phase of the project. The proposed project would not preclude other individuals or entities from constructing a trail to connect to the Spires subdivision trail.

In addition, Two Mile Bridge Road to the south end of Roberts is a rural segment of the project. In order to meet the needs of pedestrian and bicycle users in the rural segments, 8-foot wide shoulders are proposed. This width would provide 6.5 feet of usable shoulder for pedestrians and bicyclists, allowing for multi-modal access within this area.

14e

The proposed project would incorporate elements of the Red Lodge Streetscape Plan, as appropriate, and lighting would be designed to minimize light pollution. MDT will work with the City to develop an agreement to address appropriate lighting features to be incorporated into the project.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

PLEASE Name: PARB BECK POBOX 870 DI EGOCS	
PRINT Address: 10 DOR OTO, RC 51000	
400 200-0610 000	-THIS SPACE OFFICE USE ONLY-
Preter either no change in speed limits	
150 between RI and two Mile Bridge +/or	
decreasing speed limits,	
Don't put pavement cuts in shoulders	Read and
156 that would be hazandous to bicycles, unless	
shoulders are wide enough to accommodate both	
	the state of the second
15c Design one or more places where ped/bikes	
can cross 212 between Two like + town.	
Consider adding a pullout N edge of	
150 Red lodge up sign's containing historic info	
about Red ladge + a welcome to RL sigh?	
monument etc.	
Places mail your comments by December 19, 2008 to:	
Montana Department of Transportation	
Tom S. Martin PO Box 201001	4.49
2701 Prospect Ave	
Helena, MT 59620 Comments may also be submitted online at:	
http://www.mdt.mt.gov/pubinvolve/eis_ea.shtml	

15a

No modification of posted speed limits are proposed in conjunction with this project. In order to potentially have the speed limits reduced, the city or community would need to request that MDT conduct a speed study.

15b

According to the American Association of State Highway and Transportation Officials design criteria, a minimum 4-foot shoulder is desirable beyond the outside edge of a rumble strip. The rural sections of the US Highway 212, where rumble strips would be located, would have an 8-foot shoulder. Including the width of the rumble strip and the distance beyond the edge of it, there would still be a 6.5-foot usable shoulder.

15c

Crosswalks would be provided, where appropriate, as determined during final design and in coordination with the City of Red Lodge. Additionally, locations of full access intersections, where roundabouts are proposed, between the developed limits of Red Lodge and Two Mile Bridge Road may serve as pedestrian and bicycle crossings of the highway, as the roundabouts would be designed to accommodate pedestrian and bicycle traffic. See Section 3.5.4, Pedestrian/Bicycle Facilities, on page 3-9 of the EA.

15d

Comment noted.

COMMENT #15 CONTINUED

COMMENTS

Corridor Study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375

Additional space for comments, if needed. Please print.

Note: Comments are considered public records.

-THIS SPACE OFFICE USE ONLY-	In concerned about the lighting requirement
	and design, especially on the median
15e	going all the way N from RL to Two mile
and and	Bridge, This area is dank now, id like it to
Sala Sala	stay as dark as possible with any required
	lighting directed downward and poles as
	short as possible.
A star and	
	Please take sertously the need to beduce
15f	Wildlife vehicle collisions. Use every reasonable
a sugar	tactic, signage, speed limits, etc.
150	I love the roundabout at the intersection of
1-9	78 + 212. Great job + also the redesign of 8th + Oaker
The second	(provide time parking anen to)
	Understand the tuture held for a center
<u>15h</u>	Turn lane, but please only design for that
	as a turne possibility and do not construct
	It auring this first project. (Nof KL.)
	Do yubateres was can front to bailairadia
15;	Later Ot AA (Lisian hadred to the Hadred Like
131	Tater UNIV. (design, materials, etc) That includes
	provement, signs, rights, peaponce providenty
<u>15j</u>	Minimize the removal of reastation.
15k	Aninimize impacts to cultural historic sites.
in the second second	

RESPONSE #15 CONTINUED

15e

See Response #6.

15f

As stated in the Environmental Assessment (p. 3-35), MDT recognizes the high frequency of animal-vehicle collisions along the project corridor and will make a good faith effort to incorporate appropriate mitigation measures into the project design. MDT will make decisions based on the best available research and information at the time, as well as the need to balance roadway design criteria, funding constraints, and other factors.

There are no obvious focal zones along the corridor at which mitigation measures are easily identifiable. The topography in the project area is very flat, and the groundwater levels are high; this does not lend itself well to constructing wildlife underpasses, a common mitigation measure. Further, it is anticipated that the travel patterns of deer and other wildlife have and may continue to change in response to the ongoing land use and development changes in the project area.

15g

Comment noted.

15h

See Response #2a.

15i

Comment noted.

15j

Comment noted.

15k

See Response #2c.

COMMENT #16

Corridor study - Red Lodge North Environmental Assessment COMMENTS PROJECT # STPP-28-2(25)70, CN 4 TUESDAY, DECEMBER 9, 2008-RED LODGE. **RED LODGE SENIOR CITIZENS** Please use the space below to tell us your comments regarding the proposed Corridor Study - Red Lodge North Project. Note: Comments are considered public records. Name: PLEASE PRINT Address: -THIS SPACE 16a OFFICE USE ONLY-16b D 01 16c 16d ERFILE 102 COPY Please mail your comments by December 19, 2008 to: IN "ED Montana Department of Transportation RECEIVED Tom S. Martin 1 2 3 008 PO Box 201001 DEC 2 2 2008 2701 Prospect Ave TAL Helena, MT 59620 Comments may also be submitted online at: http://www.mdt.mt.gov/pubinvolve/eis ea.shtml



16a – 16d

Thank you for your comment.
COMMENT #16 CONTINUED

COMMENTS

Corridor Study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375

Additional space for comments, if needed. Please print.

Note: Comments are considered public records.

-THIS SPACE OFFICE USE ONLY-

C 10 () CS 16e 212 allo Une 1 .Sh <

RESPONSE #16 CONTINUED

16e

The purpose of the Environmental Assessment is to analyze and disclose potential impacts related to the proposed project. The Environmental Assessment does not compare the proposed project to other transportation projects for purposes of addressing or analyzing project priority.

COMMENT #17



RECEIVED

DEC 1 7 2008

ENVIRONMENTAL

MASTER FILE

COPY

December 13, 2008

Box 284 • Roberts, Montana 59070 • (406) 445-2453

Tom Martin MDT Environmental Services 2701 Prospect Ave Box 201001 Helena, MT 59620-1001

Dear Sir:

I recently was made aware of the MDT's view regarding widening Highway 212 through the town of Roberts, Montana, regarding curbs, etc. In your statements regarding the project through Roberts it states that curbs, gutters, etc. would not be addressed due to a recent project be the Roberts Community Foundation and CTEP to run a sidewalk a block away from the highway.

I am enclosing a duplicate of a letter I submitted to your department in July of 2000, regarding this CTEP project the Roberts Community Foundation was pursuing. We had originally planned to put a walkway and lighting along the highway through Roberts because of extreme concern about the safety of citizens who walked along there to school, the post office, the store, etc. In the planning stages of this project, we were made aware that the MDT was going to be expanding the highway in the next 4-6 years and would be going over the top of the project we had planned. That meant a waste of the resources and funding we had acquired to do this project. In light of that news, the Foundation, county commissioners and architect we were working with, redirected that CTEP money to put in a sidewalk along the Main Street of Roberts and built it to abut the future sidewalk we had asked the MDT to consider. This was an \$84,000 project, but was in NO WAY meant to deter the MDT from improving the areas along Highway 212.

Because of the urgency of the safety issues for our community members walking along the highway, we went ahead and funded the installation of 7 street lights along that highway area. There had been a young boy hit by a car while crossing the highway because it was so dark, he was not obviously visible to the driver.

The Roberts Community Foundation, again, reiterates to the MDT and architects working on this highway expansion, our desire to have curbs, gutters, a walkway along Highway 212, and to maintain our lighting project along this area as well.

Roberts is an unincorporated community under the governance of our County Commissioners. Any improvements or beautification projects that have occurred in our community these past 12 years have been accomplished with help from the Foundation. Our purpose is to improve the quality and safety of the community for our citizens. We feel that these improvements along Highway 212 through Roberts are necessary, a project we had initiated, but abandoned because of your expansion project. We believe it would be important for the town of Roberts that these improvements be made along the highway through town and ask you to include them in your planning of this project. Not only would this improve the safety of our citizens, it could be designed to help control the flood waters we experience with spring storms, and perhaps people would slow down driving through town because they would have more of an impression of being in a town and not just driving through the country side!

Thank you for your time and consideration in this matter. We look forward to hearing your response to our concerns and request.

Sincerely,

Elta Bertolino Ayre

Vice Chair

Cc: Mike Wombolt

COMMENT #17 ATTACHMENT



July 6, 2000

Dear Gentlemen at the Department of Transportation,

The Roberts Community Foundation was recently the recipient of a grant from the Community Transportation Enhancement Program this past year and had a project in the works for a walkway and lighting along Highway 212 as it traverses through the town of Roberts, Montana. We have now learned that you have a project four widening this highway in 4-6 years, and the area you would expand over would cover the project we had planned. We have, therefore, had to change the direction of our project because of your plans.

We are very concerned about the safety of our citizens traveling along this currently unlit section of the highway which has only barrow pits along the sides. We would like to request that in your planning you would please include curbs, gutters, a sidewalk and lighting. This is definitely a safety issue for this community.

We will redirect our CTEP funding to do safety and beautification work to compliment your project on streets connecting to your project. The architect who is doing the CTEP project planning for us is working on a master plan to this effect.

If you have any questions or concerns regarding this request, please contact the Roberts Community Foundation at 406-445-2453. Thank you for your time and consideration.

Sincerely,

Elta Bertolino Ayre Chairperson

[Page Intentionally Left Blank]

RESPONSE #17

According to the environmental document (a Categorical Exclusion) which was submitted for the CTEP project in March 2001, the project began at the intersection of Oak Street/US Highway 212 and extended one block east, then followed First Street north to the school. The purpose of the project was the following: "To provide a safe walking/biking route from the south end of the community to the school located at the north end. This will proved [*sic*] a much safer route than the current practice of walking/biking along Highway 212." As such, it was understood that getting pedestrians and bicyclists off of US Highway 212 was desired within the community of Roberts and this was used to develop the Preferred Alternative for Roberts (See Figure 3-3, Roberts Pedestrian/Bicycle Facilities, on page 3-13 of the EA).

In addition, highway related stormwater drainage would be addressed via a sloped roadway which would drain to the ditch proposed on the east side of the roadway. For further discussion of drainage within Roberts, see Section 2.3.3.2, Drainage, on page 2-24 of the EA.

COMMENT #18

From: John Clayton [john@johnclaytonbooks.com] Sent: Thursday, December 18, 2008 3:22 PM Cc: mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.com Subject: Comment on Red Lodge North EA

Greetings:

I'd like to comment on the Red Lodge North Corridor study. Thanks for all the work you have put into this project, including thoughtful delineations of issues and alternatives.

I have been involved in land-use planning issues in Red Lodge since 1992, as a concerned citizen and as participant and moderator of several community meetings of the Beartooth Front Community Forum (BFCF).Improving the look of the north entrance to Red Lodge has consistently been a priority of a large segment of the local general public for as long as I've been involved (and apparently many years longer). This construction project represents an important, and fleeting, opportunity to put into place many of the ideas we've been talking about for more than 16 years.

Those ideas, which I wholeheartedly endorse, seek to reinforce the unique and special community values of Red Lodge in its land-use patterns on the north end of town. They include strategies such as:

1. Making everything up to Two Mile Bridge extraordinarily friendly to pedestrians and bicycles.

2. Managing access along that corridor.

3. Employing one or more roundabouts, especially if they represent alternatives to streetlights.

4. Eschewing two-way left-turn lanes, which are symbols of the sort of sprawling development that is the antithesis of Red Lodge.

Over the 16 years that I've been involved in community affairs, Red Lodge has developed an extraordinarily talented and dedicated set of public servants. These include the six members of the Red Lodge City Council, the city planning staff including Forrest and Lindsy, alumni of our city planning staff including David Stauffer and Brent Moore, and the leadership of the Red Lodge Parks Board and the Beartooth Recreational Trails Association. Not only elections, but also annual topical BFCF events, have consistently put these people, these principles, and these policies up in front of a wide range of the Red Lodge populace.So this leadership really does speak for the wider community, and I (for one, if you need at least one saying so in an official comment) wholeheartedly endorse their positions. To the extent that your plans include these perspectives, I commend you. To the extent that your plans fall short of the vision these community leaders have established, I strongly urge you to alter them appropriately.

Thanks very much for the opportunity to comment. Best wishes, John Clayton 123 South Haggin Red Lodge MT 59068

author of "Images of America: Red Lodge" www.johnclaytonbooks.com facebook.com/people/John_Clayton/1438417965 http://www.linkedin.com/in/writerjohnclayton john@johnclaytonbooks.com or 406/446-3843

RESPONSE #18

Thank you for your comment.

From:	Jeff DiBenedetto [jp_dibenedetto@msn.com]	
Sent:	Thursday, December 18, 2008 3:58 PM	
То:	mdtcommentsredlodge@mt.gov	
Cc:	'Forrest Sanderson '; 'Jason Priest '; 'Lindsy	
Johnson '; mike.wam	boldt@kljeng.com; charlotte.brett@kljeng.com;	
becky.rude@kljeng.c	om; 'angie hazelswart '	
Subject:	omment On Red Lodge North EA	
Attachments:	Comments Red Lodge 212 Corridor Study.doc;	
212_endofdeve_pair	ed_view.doc; 212_2milebridge_paired_view.doc	

Attached you will find my comments on the Red Lodge North Corridor Study EA. The comments pertain to the raised median from the end of the developed area to the 2 mile bridge. I have include photos showing current views and approximate views to illustrate my concern about the visual impact of center placed paired street lights.

19a I would also like to state I am in support of extending the trail/bike path beyond 2 mile bridge to link with the Spires trail system, as was brought up at the public meeting in Red Lodge.

Jeff DiBenedetto PO Box 1464 317 Haggin South Ave Red Lodge, Montana 59068

RESPONSE #19

19a

See Response #9.

COMMENT #19 CONTINUED

Jeff DiBenedetto 317 Haggin South Ave. Red Lodge, Montana 59068

December 17, 2008

Tom S. Martin Montana Department of Transportation PO Box 201001 Helena, Montana 59620

I am commenting on the Corridor Study-Red Lodge North Environmental Assessment Red Lodge (Project Number STPP 28-2(25)70, Control Number 4375). Specifically my concerns deal with the proposed raised median and street lighting centered on the median. This proposal seems inconsistent with the Growth Policy goal for Community Entrances.

"Classic strip development and sprawl will not be allowed. Development at each of the four entrances must present Red Lodge as a <u>unique</u> and welcoming community" page 24 0f 73

Regulations, K, page 270f 2 - "The appearance of development in the community entrances should be <u>consistent with Red Lodge's historic image</u>. All development in the entry corridors will be subject to design review"

- 19b My concern isn't with the raised median, but with the placement of paired lighting down the center of the roadway from 2 mile bridge to the Developed Limits of Red Lodge. I have two primary concerns
 - 1. Visual Impact The of a row of street lights obstructing the views south and particularly diminishing the visual impact of Mount Maurice and the Beartooth Mountains dominating the backdrop to Red Lodge (see attached photos). Rather than a clear view down the highway towards the center of Red Lodge with Mount Maurice and the Beartooth Mountains as backdrop, the row of paired street lights will be the dominant visual impact. It is the current views south, along with minimal development, that makes this entrance unique and different from community entrances of other towns. The raised median and associated paired street lights will create an entrance that is not unique, but typical of most big city boulevards, community entrances, and strip development. Development along the north entrance corridor is at a critical stage. Decisions made by the City of Red Lodge about how development proceeds along this corridor will dictate whether the Community Entrance Goal will be achieved, or not. This includes not just paying attention to how private land is developed along the corridor, but also how government entities (City of Red Lodge, Montana Department of Transportation)

adhere to the same standards to achieve those goals. Simply accepting the same cookie cutter design alternatives used elsewhere is not consistent with achieving the goal for community entrances. If the raised median requires paired street lighting down the center of the roadway, than I suggest rethinking this design. Rather than a raised median, I suggest continuation of the turning lane planned for the MT Highway 78 to Developed Limits of Red Lodge.

2. Light Pollution and Carbon Footprint - It appears the paired street lights will double the number of lights as compared to placement of single lights along the shoulders. The paired lights will be use more energy and contribute to more light pollution emanating from the City of Red Lodge. Red Lodge is not doing well in minimizing its carbon footprint. Energy used by streetlights was mentioned as the number one contributing factor in a recent report. Not only is the additional use of energy a concern, but also the additional contribution to night time glare from the City of Red Lodge over the past several years, particularly since installation of the historic lighting downtown. I would suggest exploring lighting solutions that have less impact on the night time sky and are more energy efficient.

I would like to see alternatives explored that will eliminate the need for center lane street lights and reduce the visual impact on views both in and out of town in the north community entrance. It would like to see alternative methods of street lighting evaluated which are more energy efficient and will not contribute to light pollution. Current technology and visualization software should enable the presentation of images displaying the visual impacts of these alternatives.

Jeff DiBenedetto

Cc: Betsy Scanlin, Mayor City of Red Lodge Red Lodge City Council Forrest Sanderson, City Planner Lindsy Johnson, City Planner Red Lodge Planning Board/Zoning Commission

RESPONSE #19 CONTINUED

19b

See Response #6.

19b Continued

COMMENT #19 ATTACHMENT









COMMENT #20



December 19, 2008

Tom S. Martin, PE Environmental Services Bureau Chief Montana Department of Transportation 2701 Prospect Ave. PO Box 201001 Helena, Montana 59620-1001

Re: Environmental Assessment & Section 4(f) Evaluation Corridor Study- Red Lodge North

Mr. Martin:

This letter and attachments represent comments from Beartooth Hospital and Health Center, CTA Architects Engineers and the Billings Clinic. Billings Clinic is providing financial & management support to Beartooth Hospital for this new facility and CTA is our primary design firm. We are signing this letter on behalf of respective organizations and these comments should be considered as the official positions of our organizations.

Attached please find the following items:

- 2 pages listing our specific comments
- Campus plan (hospital & other buildings to be built on our campus) illustrating some traffic flow and campus circulation to support our comments in graphic form
- Diagram of proposed left hand turn overlaying a ³/₄ intersection template provided by MDOT

One of the major reasons we purchased our site was its location near Highway 212 which is the primary circulation for vehicles going north/south in Carbon County. The adjacency of the site to the highway is a major benefit for patient convenience. The ability for incoming and outgoing traffic to access our campus is of paramount importance to our medical facility and will have a direct impact of our quality of care we will be able to deliver.

The access into and out of our site is so important to us that we've requested a traffic study projecting traffic volumes be completed by CTA. That study is well underway and will be submitted to the District Traffic Engineer when it is complete. We expect that to be in the next several weeks. We are submitting some preliminary findings from that traffic study as part of our comments.

600 West 21st Street • P.O. Box 590 • Red Lodge, Montana 59068-0590 • Phone (406) 446-2345 • Fax (406) 446-3182

We serve patients of all ages and physical conditions. Many of these patients are driving to and from our campus under a variety of situations. Many of our senior patients have compromised vision, hearing and motor skills. The simpler the traffic pattern is for these patients as they arrive and depart, the less opportunity for confusion on their part and less risk for accidents.

All of our transfers to hospitals in Billings are by ambulance turning north as they exit our hospital Emergency department adjacent to the main entry. These transfers all have elements of urgency due to the patient's physical condition, and in some cases....their lives are dependent on the time in transit.

In a single word, we believe SAFETY is the key component in requesting the left hand turn capability.

Please review our two graphic attachments and our more specific comments (3 pages) in support of our position.

Sincerely, Kelley Evans Kelley Evans, CEO Mitch Gopler Executive Director - Billings Clinic Beartooth Hospital and Health Center Facility Service Shepard, Presid rchitects Engineers

Beartooth Hospital and Health Center

December 19, 2008

Red Lodge, Montana

The following points are being presented by the Beartooth Hospital and Health Center, CTA Architects Engineers, and Billing Clinic to support an intersection at the main entry to the Hospital that will allow a left turn for vehicles leaving the Hospital so they can proceed immediately north on Highway 212.

- 1. CTA Architects Engineers is currently completing a traffic impact study for the proposed Hospital to determine impacts from the Hospital proposal on the adjacent road network. The traffic impact study will use data from a feasibility study conducted for the Hospital to obtain USDA funding that identifies patient use by zip code to determine the project trip distribution, including the expected northbound egress volumes for the main entry. Preliminary findings indicate that approximately 26% of site traffic would result from trips coming from / going to the north from the site.
- 2. 100% of ground transfers (by ambulance) leaving the Hospital will be to Billings (turning north). There were approximately 200 of these transfers last year. If a left turn is not available, these ambulances will be required to turn south and to the next full intersection or roundabout and this will add a mile of travel distance and related transit time. The other option would be for these ambulances to navigate their way through the campus (through parking lots and related pedestrian traffic) to reach the 2 Mile Road intersection. This would be a more circuitous path and much less direct than using the main entry intersection.

In either case cited above, this unnecessary delay will affect the "golden hour" that medical providers have to provide the appropriate medical intervention and could negatively affect the clinical outcomes related to these patients. A left hand turn capability at the main entry intersection saves precious minutes on ambulance ground transfers.

3. People are creatures of habit. Once patients and visitors, that visit the Hospital campus with some frequency, determine that they cannot go north at the main entry intersection, they will drive through the campus to the 2 Mile Road intersection. This path takes them through parking lots and pedestrian walkways, thus increasing the internal campus traffic. This traffic would greatly increase "in-campus encounters" and be less safe than the left turn capability at the main intersection.

Beartooth Hospital and Health Center

Red Lodge, Montana

- 4. Possible future extension of Hauser. The Hospital agreed to maintain a 60 foot right of way on the "internal" main entry road of our campus to support the City of Red Lodge in providing for a future extension of the City street named Hauser. The volume from this street extension would certainly add to the Hospital campus volume of traffic in a significant way once Hauser is extended north. This added future volume should be considered in the design of the main entry intersection to the Hospital campus. See attached diagram showing Hauser extension.
- 5. Once the Hauser extension is complete and operational, the lack of a left hand turn would certainly cause Red Lodge traffic to take an alternate path through our site to the intersection at 2 mile road. This Hauser traffic would either use the back "service road" or go through our parking lots on the east side of the Hospital. Either route would be very disruptive to the campus and create unsafe conditions for vehicles and pedestrians within the campus due to the additional traffic volume. The Hospital fully supports the extension of Hauser to the main entry intersection as per City future planning. The ability for a left turn lane at the intersection would eliminate the concerns stated earlier in this talking point.
- 6. The Hospital purchased the site in August, 2005 and the specific design of the Hospital campus followed. The planning for the controlled access management plan predated the design of the Hospital campus so those campus needs were not known or considered in the access planning. We submit that a re-evaluation of the highway design is needed and appropriate now that we have new and definitive information regarding the Hospital campus.
- 7. Due to the close proximity of Rock Creek to the highway right-of-way at the proposed Hospital entrance, the feasibility of an east leg to an intersection at this location is questionable. Rather than a 3/4 intersection as proposed in the EA, a tee intersection that would allow northbound left turn movements could be installed at this location to provide the northbound access sought by the Hospital. Please see the attached intersection diagram. (The Hospital would not be opposed to the consideration of a roundabout at this location if northbound access could be provided in that way.)
- 8. All incoming traffic (ambulances, patients driving cars, visitors) will use the main entry for closest access to the Hospital main entry and the emergency entrance. These same vehicles will intuitively return to that same entry point to leave the site. The lack of a left turn lane at this main entry point will certainly cause confusion. Some vehicles will likely attempt a left hand turn to the north despite signage, etc. This is a risk.

COMMENT #20 ATTACHMENT CONTINUED

Beartooth Hospital and Health Center

December 19, 2008

Red Lodge, Montana

- 9. A significant portion of patient visits will come from the north as this is a community hospital that serves the entire county. By census, 37.3% of the population, served by the Hospital, live in areas that will access this site from the north and return that direction to go home. With the new Hospital, the "capture rate" of these patients to the north is likely to increase over current market share, so the numbers of left turning vehicles is expected to continue to rise over time.
- 10. The future development of the campus includes senior living facilities in the south portion of the campus and potential clinic buildings in the north portion of the site. Both future developments will add significant future traffic volume to the campus and certainly increase the volumes at the main entry intersection.
- 11. The southbound traffic will be reducing speed dramatically as they approach the 2 Mile Road intersection. With this reduced speed, the ability of vehicles leaving the campus via the main entry intersection will have increased time gaps to make the left hand turn.



COMMENT #20 ATTACHMENT CONTINUED



North

RESPONSE #20

See Response #7.



From:Curtiss Fleck - Weddings [cjweddings@redlodge.com]Friday, December 19, 2008 7:47 AMSent:Friday, December 19, 2008 7:47 AMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comSubject:Comment On Red Lodge North EA

Dear sirs,

21a Please allow for safe pedestrian and bike routes when you improve the Northern entrance to Red Lodge.

There is an existing trail easement from the Squires sub-division travelling to highway 212 near the new hospital, please extend the bike pedestrian trail at least far enough to link with that trail and farther if possible. If the path is placed as far from traffic as 21c possible snow plows would not cover it and it would get more use.

Thank you,

Curtiss Jay Fleck

21a

Comment noted. Please refer to Section 3.5.4, Pedestrian/Bicycle Facilities, on page 3-9 of the EA.

21b

See Response #9.

21c

See Response #14b-c.

From:Les [lhedquist@gmail.com]Sent:Friday, December 19, 2008 9:34 AMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comComment On Red Lodge North EA

To Whom it May Concern,

I am extremely pleased to learn that round-a-bouts are being planned for Red Lodge. My wife and I lived for a year in Europe where round-a-bouts are common. We became devoted supporters of these safe, efficient and inexpensive traffic control devices. Further reading on their use in the USA showed how effective they are in reducing auto accidents and in providing better pedestrian safety.

An important feature is having an S-turn to negotiate as one approaches the circle that places the incoming car near the center of the circle, rather than tangent to it, which forces incoming cars to slow down or stop, then having a one-way traffic flow to the right around the circle. This eliminates the dangerous scenario of one high-speed incoming car trying to beat the light while another car is crossing the same intersection at 90 degrees to the first car. All cars must slow or stop before entering the circle traffic. A further great benefit was the ability to simply go around the circle again if we were uncertain of which exit to take.

Round-a-bouts require no electricity, have very low maintenance and provide a location for seasonal and event specific decoration to entertain drivers.

Round-a-bouts were very easy to understand and negotiate. We used them in several countries with different languages with no trouble. Around you go until to arrive at your exit. No waiting for long lights, no encouragement to run a red light, everyone keeps moving. Most users in Red Lodge have probably used them already in other places in the USA or Europe.

I applaud and support the planners intent to install round-a-bouts in Red Lodge.

Best Regards, Les Hedquist 406-425-3804 cell POB 1604, Red Lodge, MT 59068 Thank you for your comment.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

DIFASE	Name:	RAND HERZBERG	
PLEASE PRINT	Address:	23 Biorn Lans, Rellolg, M	T 59068
		see comments	-THIS SPACE OFFICE USE ON
	[on back	

CORRIDOR STUDY - RED LODGE NORTH ENVIRONMENTAL ASSESSMENT PROJECT # STPP-28-2(25)70. CN 4375

Additional space for comments, if needed. Please print.

what Note: Comments are considered public records. -THIS SPACE 15 mand 40 OFFICE USE ONLY-8720 he 1759 trance 2 en acces 0 02 rospital SQUAR CQ. new 0 lan Connue Û MM BWM 0 C

the wospotor l vehicles emachia DALMAND will 1) Mane e.M na ·Use ŵ alan and 0 emercen ×1536 Ó MIM 15% 0 An 112 lac man mini ή. a 0 6 lo oul two 224 O DINALOSES Was 10 this SA 10.00 22 2 minutes becomes MA MASI ē. n. other to Le Brun DESC 19 Mar a 00 Nord war 武 NON V 1 ressed KOSpital nas 6 71.57 a in, ACAS 0 Ø NRQ an roll 2 Berle. tace. 101 Q.R. M Q. Brang bu Nospita C megalle South. row 6 M a LANDA upp energy 6 one add nnecessaril will TRA Which Q G. 15 劉 褒 rollindabout

RESPONSE #23

See Response #7.

COMMENT #24

sam hoffmann

 From:
 "sam hoffmann" <sam_rediodgeales@yahoo.com>

 To:
 <mdtcommentsrediodge@mt.gov>

 Cc:
 <mike.wamboldt@kljeng.com>; <charlotte.brett@kljeng.com>; <becky.rude@kljeng.com>

 Sent:
 Friday, December 19, 2008 4:14 PM

 Subject:
 Comment On Red Lodge North EA

To whom it may concern:

I am a developer in the north corridor of Red Lodge and am writing to comment on the proposed highway improvements in the north corridor. My property sits in the zones labeled, "MT Highway 78 to Developed Limits of Red Lodge," and "Developed Limits of Red Lodge to Two-Mile Bridge Road."

24a I like the plan for "MT Highway 78 to Developed Limits of Red Lodge." I like the TWLT, the sidewalk and shared bike path/ ped path.

I do not care for the plan for the "Developed Limits of Red Lodge to Two Mile Bridge Road." The raised median with lighting would be an ugly source of light pollution at our beautiful entrance to Red Lodge. The roundabouts would be confusing for tourists and would not say, "welcome to Red Lodge." Many landowners and developers in this corridor paid hundreds of thousands of dollars for their property. Limiting their access to the highway would reduce their property values. I do not share the City Council and Planning Dpeartment's view that we need to limit commercial development in the north corridor. Red Lodge is hurting for economic development and it ought of be encouraged, not discouraged.

24c I think the plan for "MT Highway 78 to Developed Limits of Red Lodge," ought to be implemented all the way to Two Mile Bridge Road.

Thank you for your consideration. Please feel free to contact me at 406 446 4607 or 406 425 4607 or email me at sam@rediodgeales.net.

Sam Hoffmann President Red Lodge Ales Brewing Company Partner H&H Real Estate, LLC

24b

RESPONSE #24

24a

Thank you for your comment.

24b

See Response #5a and #6.

24с

See Response #5a.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

Name: William NI. Deborah 4 PLEASE Robe 122 907 Address: _ PRINT -THIS SPACE DFFICE USE ONLYconcern with main expansion 21 through 10 heeded 1C taken trom Dac G nnea an 25a be nging 5 SIDA 15 most would Seem mo rea sonabl oroper our THE home highe close to our yard and e too home Please mail your comments by December 19, 2008 to: RECEIVE Montana Department of Transportation Tom S. Martin DEC 2 3 2008 PO Box 201001 2701 Prospect Ave **ENVIRONMENTAL** Helena, MT 59620 Comments may also be submitted online at: http://www.mdt.mt.gov/pubinvolve/eis ea.shtml



RESPONSE #25

25a

See Response #8 and Section 3.5.3, Safety, on page 3.7 of the EA.

COMMENT #25 CONTINUED

The other concern we have is the existing culvert to the south of our property. This culvert is not large enough to handle overflow during any major run off. We have been flooded 256 twice recently i 2005, 2008). We have a floor furnace that has had to be repaired because of the flood. This is not a new problem but an existing one when it floods! The flood waters have also washed out gravel and soil from under our front porch . which we have tried to replace. These issues need to be addressed and evaluated for a betty and Sater highway enviroment not only for the 25c ivers, but most of all for the perty owners in Roberts and along ghway 212

RESPONSE #25 CONTINUED

25b

As stated in the Environmental Assessment, it is beyond the scope of this project to address floodwaters entering Roberts from the drainage to the west (See Section 2.3.3.2, Drainage, on page 2-24 of the Environmental Assessment). Goals of the proposed project would be to convey as much highway-related storm water as practicable toward Rock Creek prior to Roberts, and within Roberts to satisfactorily convey storm water intercepted by the highway towards Rock Creek. For example, berms may be constructed perpendicular to flow east and west of the highway, approximately 1,400 feet (425 meters) south of Birch Street in Roberts. At that location an existing centerline culvert is currently planned to be replaced. The existing downstream drainage channel located on the east side of the highway may be enlarged and re-graded to convey more runoff to Rock Creek.

25с

Thank you for your comment.

COMMENT #26

From: Harry Hollman [harry@greatermontanarealty. com] Sent: Friday, December 19, 2008 1:20 PM To: mdtcommentsredlodge@mt.gov Cc: mike.wamboldt@kljeng.com; charlotte. brett@kljeng.com; becky.rude@kljeng.com **Comment On Red Lodge North EA** Subject:

Dec 19, 2008 12:18 PM

To Whom It May Concern, I have read through the report on the road improvements just north of Red Lodge. I am opposed to 26a more than two roundabouts. I believe one at first street and one at Two Mile Bridge Road will suffice.

I also am opposed to the raised and lighted median between those two roundabouts. I feel a four lane corridor with right and 26b left turn lanes is sufficient. I wou; ld prefer extremely limited lighting. We do not need anymore light pollution in this area. The bare minimum of lighting should be shielded to only illuminate down and NOT up. Thanks You, Harry Hollman

Harold 'Harry' Hollman Broker/ Owner G.R.I. Greater Montana Realty P. O. Box 1071, 215 S.Broadway Red Lodge, MT. 59068 PH: 406-446-9122 Fax: 406-446-9128 Cell: 406-671-9640 harry@greatermontanarealty.com www.greatermontanarealty.com

26a

At the request of the City of Red Lodge, MDT and the City of Red Lodge developed an Access Management Plan for the area between MT Highway 78 and Two Mile Bridge Road. The purpose of the Access Management Plan is to provide a means for MDT and the City to balance the need for vehicular progression along US Highway 212 with the need for access to adjacent properties. Two intersection treatment options were proposed to the public for the full access intersection locations: a conventional full access intersection and a roundabout. At the request of the City of Red Lodge, a roundabout design has been identified as preferred for all of the full access intersection locations. This was to "realize the safety advantages that roundabouts provide, capture the cost advantages of roundabouts over time, and provide a distinctive entrance into the City of Red Lodge" (See City Council Resolution No. 3228). For a full discussion of the Access Management Plan, please see Section 2.3.1.4, Access Management, on page 2-14 of the Environmental Assessment.

26b

See Response #6.
From: Kenny Jeansonne [mailto:ken@cbthebrokers.com] Sent: Thursday, December 18, 2008 2:20 PM To: Martin, Tom Subject: Kenny's Letterhead

Thanks for taking the time to speak with me. I appreciate your consideration.....Kenny



THE BROKERS 5 North Broadway Red Lodge, Montana 59068 406-446-3001 or Fax 406-446-1998 www.cbthebrokers.com



Kenny Jeansonne Sales Associate (406)446-3001 Business (406)818-9551 Cell (406)869-0585 Direct Line <u>ken@cbthebrokers.com</u>

December 17, 2008

Tom S. Martin Montana Department of Transportation P.O. Box 201001 Helena, Montana 59620

Re: Corridor Study-Red Lodge North Environmental Assessment, Roberts

Dear Mr. Martin,

We are writing to express our disapproval of the planed highway expansion between Roberts, and Red Lodge, Montana.

We are property owners in the area which will be adversely affected by this proposed expansion. This expansion will and is currently having a tremendous negative affect on our property all ready. We are located approximately 3 miles south of the town of Roberts at 7883 Highway 212.

Why is it the necessary land needed for the widening of the highway cannot be reclaimed from the East side of highway 212 where the railway bed lies? At the present time there are electrical lines, natural gas lines and telephone lines all located on the west side of highway 212. There are no less than 17 homesteads which will be affected in this claiming of property on the West side of the highway.

On the east side of highway 212 there is no home owner which would be affected. On the east side of the highway the old railway bed is located. None of the property owners have built on this acquired land from the railroad. Why is it the ground necessary for this expansion is being taken from the land owners on the west side of highway 212?

My wife and I have been contacted by an interested party in purchasing our property. Because of the disclosure laws it is imperative we disclose any and all negative features associated with our property. This issue is hanging over our heads as we cannot commit to any sort of a sale as long as there is the possibility of us loosing any of our land to this expansion. Our property will be severely impacted negatively if we are to loose any of the land between our home and the highway.

Currently our mound septic system is located between our home and the highway in the vicinity of where any expansion may occur. The septic system was laid out by the Carbon County engineer and we do not have any other place for it to go without causing a huge odor problem or affecting our water system.

At the public hearing held at the Roberts School Wednesday December 10, 2008 several issues were discussed by local residents. One of he issues discussed was the accommodating of residents in the town of Roberts by placing sidewalks on the east side of the roadway. Also there was discussion of extra wide shoulder on the east side of to accommodate culverts to divert water. If this sort of consideration can be done in the town of Roberts can we please consider our situation as well?

COMMENT #27 CONTINUED

Mr. Martin, my wife and I are getting up in age. Our hearts, souls and life savings have been put into our home and property. We take great pride in our place. Any loss of land would greatly affect the value of our property. It just seems so outrageous to me that with all of the utilities, all the homes which will be adversely affected on the west side of the highway and very minimal impact to homes and property on the east side that this issue is being discussed.

I would very much like to meet with you and show you our concerns. I am quite sure you would agree with us if you would visually see the negative impact this loss of land will create to our property.

Respectfully,

Kenneth Jeansonne

Jacqueline Jeansonne

RESPONSE #27

See Response #8.

COMMENTS

Corridor study - Red Lodge North Environmental Assessment PROJECT # STPP-28-2(25)70, CN 4375 TUESDAY, DECEMBER 9, 2008-RED LODGE, MT RED LODGE SENIOR CITIZENS CENTER

Please use the space below to tell us your comments regarding the proposed Corridor Study - Red Lodge North Project. 8

Note: Comments are considered public records.

Name: PLEASE 0 PRINT Address: 6 THIS SPACE OFFICE USE ONLY-28a na 28b a 28c Che 28d 2 ,12 11 en 21 Please mail your comments by December 19, 2008 to: Montana Department of Transportation Tom S. Martin PO Box 201001 2701 Prospect Ave Helena, MT 59620 Comments may also be submitted online at:

http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

28a

Public involvement conducted for the project is summarized in the Environmental Assessment on page S-15, Executive Summary, and is discussed in greater detail in Chapter 6 of the document. Some of the public involvement included the following:

- March 27, 2002 Kickoff Meeting held to inform elected officials and local, State, Federal, and regional agencies, as well as the public, of the project and to obtain local knowledge of concerns related to the proposed study. This meeting also served as an early notification of the preparation of an EA.
- November 6, 2002 Alternatives Public Workshop held to inform the public of alternatives being considered for the project and to obtain public input.
- May 16, 2006 Public informational meeting held to discuss improvements to highway-related storm water drainage along the project corridor in Roberts.
- November 14, 2006 A property owner meeting was held to discuss current and future access needs from MT Highway 78 to Two Mile Bridge Road.
- January 17, 2007 A second access management meeting was held to present potential alternatives for access management along MT Highway 78 to Two Mile Bridge Road and to solicit public feedback.

These meetings were advertised in the *Carbon County News* and the *Billings Gazette*. In addition, press releases were submitted to local media to alert the public of upcoming meetings. Thirty day comment periods followed all meetings.

28b

Access to US Highway 212 from Villard Avenue would be via 4th Street, which would be extended along the south side of the Visitor Center between US Highway 212 and Villard Avenue (replacing the Visitor Center's south access). Please refer to Figure 2-8, MT Highway 78 / Villard Avenue Intersections, on page 2-15 of the EA.

28с

Comment noted.

28d

See Response #8.

Louise C Jenkins 2228 Pine Street Billings, MT 59101 December 19, 2008 RECEIVE DEC 2 3 2008 ENVIRONMENT

Mr. Tom Martin Montana Department of Transportation PO Box 201001 Helena MT 59620-1001

RE: Property located at the corner of Villard and Hwy. 78

This written comment is made on behalf of Adeline Donnes, P O Box 805, Red Lodge, MT, 59068. Adeline is the owner of the above referenced property and I am her daughter and hold Power Of Attorney for her.

The property is located across from the Red Lodge Fire Station and currently has complete access from both Villard Avenue and Highway 78. There are four contiguous lots and a small house located on one of the lots. The potential value of the property is that it is an excellent commercial location.

If the Corridor Plans for Red Lodge North are implemented and include the proposed round-about, the major issue for this property is the possible total loss of access to the property.

Please address these questions:

*Will parking be available along Highway 78? Currently many fireman park along side the property as do many others including visitors to the property and to the adjacent Visitor Center.

- *Will dirveways be allowed off of Highway 78? There is a currently existing driveway with access off Highway 78. What is the potential for future use of that driveway?
- *If Villard Avenue is closed, will access be available from the proposed cul de sac? Will parking be available in the cul de sac area?
- *Which entity is responsible for the potential closure of Villard Avenue; The Department of Transportation or The City of Red Lodge?

*The closure of Villard Avenue and the potential loss of complete access to the porperty greatly affect its

29d value. What plans does the Department of Transportation have for compensation of loss to the property owner?

I request a written response to the above questions. If you need further information, please contact me at (406) 252-8868.

Thank you for your help.

Sincerely bac Uli

Louise Jenkins

CC: Betsy Scanlin

29a

29a

The proposed improvements related to the project do not extend beyond the US Highway 212 corridor. As such, no improvements to MT Highway 78 are being proposed as part of the project, with the exception of its intersection with US Highway 212 where a roundabout is proposed. The segment of MT Highway 78 in question may be addressed as part of the Red Lodge Northwest project, whose project begins at the MT Highway 78 and US Highway 212 intersection and extends approximately 5.1 miles northwest.

29b

Access would be provided via the cul-de-sac (Please refer to the MT Highway 78 and Villard Avenue discussion on page 2-14 of the EA). The Preferred Alternative is based on preliminary design (approximately 30%); potential parking in the cul-de-sac would be determined during the design phase of the project.

29с

The Montana Department of Transportation and the City of Red Lodge are both responsible for the closure of Villard Avenue. The City would need to abandon its right-of-way along Villard Avenue. Decisions made with regard to the proposed project are ultimately approved by the Montana Department of Transportation and Federal Highway Administration.

29d

As is required by law, property owners who have been identified for potential right-of-way acquisition would receive fair and just compensation, as would be determined during the right-of-way acquisition phase (See the Mitigation for Right-of-Way and Relocation Impacts discussion on page 3-17 of the EA).

From:Lindsy Johnson [lindsyjohnson@vcn.com]Sent:Friday, December 19, 2008 12:37 PMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comSubject:Subject:Comment On Red Lodge North EA

Thank you for working with the community of Red Lodge!

Please remove the TWLTL from the 'MT Highway 78 to Developed Limits of Red Lodge' section of the document. The section from Hwy 78 to the Controlled Access Corridor should be consistent with the typical section from 8th Street to Highway 78 and the Historic Downtown. See the 2008 Red Lodge Growth Policy.

^{30b} Limit access to Highway 212 to ONLY public Rights Of Way i.e. the planned and existing gridded network of public streets and avenues. See the 2008 Red Lodge Growth Policy.

Please add a 5 foot wide sidewalk to the West side of Highway 212 in the Controlled Access Corridor per years of discussion and public comment. This sidewalk has been in many drafts of the proposed design of this section of Highway 212. See the 2008 Red Lodge Growth Policy.

- ^{30d} Please extend a sidewalk north to connect to the Spires ped/ bike trail that comes off the bench approximately 1/2 mile north of Two Mile Bridge Road.
- ^{30e} Please assure that safe pedestrian crossings are designed and built into all access points in the Controlled Access Management Corridor. See the 2008 Red Lodge Growth Policy.

Lindsy Johnson Red Lodge Homeowner 513 N. Cooper Avenue, PO Box 811 Red Lodge, MT 59068

RESPONSE #30

30a

See Response #2a.

30b

See Response #2b.

30с

See Response #14d.

30d

RESPONSE

30e

See Response #15c.

From: ROBERT MARY JOHNSON [mailto:msj1rpj1@msn.com] Sent: Saturday, December 20, 2008 1:17 PM To: MDT Comments - Red Lodge EA Subject: mike.wamboldt@kljeng.com;charlotte.brett@kljeng.com;becky.rude@kljeng.com

To whom it may concern,

Regarding your plans for highway 212 north of Red Lodge, a sidewalk on the west side of 212 connecting to the Spires Subdivision would be excellent for bikers and pedestrians in and around Red Lodge. As it stands, bikers and runners put themselves in a dangerous position on 212. With the new hospital and subdivisions north of the city, pedestrian/bike traffic will most likely increase. It would be a great addition to safe, convenient trails for Red Lodge.

Thank you for your consideration. Mary Johnson Red Lodge, MT

RESPONSE #31

See Response #9.

From:	Tom Kohley [tom@mapmurals.com]	
Sent:	Friday, December 19, 2008 11:06 PM	
To:	mdtcommentsredlodge@mt.gov	
Cc:	mike.wamboldt@kljeng.com; charlotte.	
brett@kljeng.com; b	ecky.rude@kljeng.com	
Subject:	Comment On Red Lodge North EA	

Mike and Others,

First, I would like to thank you for referring to our community's trail plan. I think the ped/bike system you currently have planned is consistent with the plan and the desire of our community interests. I do have a couple specific comments concerning the proposed enhancements along the Highway 212 corridor.

1) I would strongly urge MDT to consider extending the ped/bike path another ½ mile to the round barn on the west side of the highway so that the trail can link to a planned trail segment through the Spires subdivision. This trail segment would extend from Highway 212 to Willow Creek Road (near Highway 78) and would provide an excellent opportunity to complete a connection from both highways. The Spires trail is partially complete, and they have recently secured an non-motorized easement down to Highway 212 according to Russ Squires, the developer of the project.

2) I would suggest that MDT maximize the separation between the ped/bike path and the
 automobile corridor. I have seen too many situations where snow removal from the automobile
 path resulted in the adjacent ped/bike path being buried. Placing the ped/bike path on the
 eastern most extent of the highway ROW will help alleviate this potential problem.

3) The planned round-about at Two Mile Bridge Road should be designed with pedestrians in mind. This will likely be the highway "cross-over" point for the ped/bike lane. Large curb cuts and pavement striping should be incorporated into the design to provide safe and consistent travel from the 10 foot ped/bike trail.

Again, thank you for considering these suggestions. It is important the Red Lodge continue its progress toward a pedestrian/bike friendly community.

Tom Kohley Red Lodge Parks Board Beartooth Recreational Trails Association



32a

See Response #9.

32b

The sidewalk and shared bike/ped path would be a minimum of five feet from the travel lanes, in accordance with design standards. Actual distance of the shared bike/ped path and sidewalks from the travel lanes will be determined during design. Please refer to Figure 2-10, MT Highway 78 to Developed Limits of Red Lodge, on page 2-10 and Figure 2-6, Developed Limits of Red Lodge to Two Mile Bridge Road, on page 2-11 of the EA.

32с

All roundabouts would be designed to accommodate pedestrian and bicycle traffic.

COMMENTS

Corridor study – Red Lodge North Environmental Assessment Project # STPP-28-2(25)70, CN 4375 Tuesday, December 9, 2008—Red Lodge, MT Red Lodge Senior Citizens Center

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

Dec, 2008 Name: RAY + Louise Locke PLEASE 59070 212 PRINT Address: 8079 T Am ncer ting -THIS SPACE OFFICE USE ONLY-2 Righ 0 21 - a-way NAth Follow th erl 5 ho WA thore 11) 5 5 Ka oa road project 11202 10 1004 omos 2 2 e12000 fert Oc. ere Dro 9 ine an a which CERCERNS oter 50 me Serves our home, and aux neighbor. Please mail your comments by December 19, 2008 to: Montana Department of Transportation Tom S. Martin RECEIVED PO Box 201001 MASTER FILE **DEC 1 8 2008** 2701 Prospect Ave COPY Helena, MT 59620 ENVIRONMENTAL Comments may also be submitted online at: http://www.mdt.mt.gov/pubinvolve/eis ea.shtml

CORRIDOR STUDY - RED LODGE NORTH ENVIRONMENTAL ASSESSMENT PROJECT # STPP-28-2(25)70, CN 4375

Additional space for comments, if needed. Please print. Note: Comments are considered public records.

1.	
-THIS SPACE OFFICE USE ONLY-	South of us.
	we have a pond on aur
	property, which is shielded by several
	aspentrees that help keep the noise
	down From the hiway. most of those
	trees would have to be out down,
	and removed by the project.
	my last statement is simply,
	and plain. It would make more
	Sense, and becheaper to build the
	New Hiway project to the East, not
	to the west side at all!
	Thank you For letting us air our
	Concerns,
	Ray & Louise Bocke
	8079 U.S. Hiway 212
	Roberts mt. 59070
	406-445-2321
	12/13/08

RESPONSE #33

See Response #8.

From:	Marcella Manuel [marcella@cbthebrokers.com]	
Sent:	Friday, December 19, 2008 1:07 PM	
То:	mdtcommentsredlodge@mt.gov	
Cc:	mike.wamboldt@kljeng.com; charlotte.	
brett@kljeng.c	om; becky.rude@kljeng.com	
Subject:	Comment On Red Lodge North EA	

To Montana Dept of Transportation, Federal Highway Admin., Kadrmas, Lee and Jackson,

Re: Highway 212 Project

34a

I attended the public meeting at the Red Lodge Senior Center and was very disappointed at the public attendance. I believe the combination of poor advertising, a conflict with the City Council Meeting and the very cold weather caused the very poor attendance. This is a very important project for the community of Red Lodge and Roberts, and it merits an opportunity for further public comment. I do not believe that the public has been very well informed about this project and I urge you to extend the comment period.

34a

Public involvement conducted for the project is summarized in the Environmental Assessment on page S-15, Executive Summary, as follows (public involvement is described in greater detail in Chapter 6 of the Environmental Assessment):

- March 27, 2002 Kickoff Meeting held to inform elected officials and local, State, Federal, and regional agencies, as well as the public, of the project and to obtain local knowledge of concerns related to the proposed study. This meeting also served as an early notification of the preparation of an EA.
- November 6, 2002 Alternatives Public Workshop held to inform the public of alternatives being considered for the project and to obtain public input.
- May 16, 2006 Public informational meeting held to discuss improvements to highway-related storm water drainage along the project corridor in Roberts.
- November 14, 2006 A property owner meeting was held to discuss current and future access needs from MT Highway 78 to Two Mile Bridge Road.
- January 17, 2007 A second access management meeting was held to present potential alternatives for access management along MT Highway 78 to Two Mile Bridge Road and to solicit public feedback.
- January 31, 2007 A meeting was held with the architect working with the City of Red Lodge on the new Bank of Red Lodge. The purpose of the meeting was to discuss when the US Highway 212 project would be constructed, the amount of right-of-way required for the proposed project, and possible layouts for the bank site plan.
- February 16, 2007 A meeting was held with the owners and representatives of the proposed Beartooth Hospital in an effort to coordinate the proposed US Highway 212 project and right-of-way requirements with the future site plan of the Beartooth Hospital.
- December 18, 2007 A public informational meeting was held to discuss improvements to highwayrelated storm water drainage along the project corridor in Roberts.

In addition to the coordination listed above, public hearings were held in Red Lodge on December 9, 2008, and Roberts on December 10, 2008. Both meetings were advertised in the *Carbon County News* and *Billings Gazette* for four consecutive weeks beginning on November 13 in the *Carbon County News*, and November 17 in the *Billings Gazette*. Press releases were also distributed to local media sources in the same intervals as the news ads. In addition, postcards were distributed to individuals who have commented on the project or attended project meetings during the course of the environmental process (See Section 3.0, Comments and Coordination, of the Finding of No Significant Impact). No further public meetings are planned at this time; however, further opportunities for comment will be available to property owners during the design phase of the project.

At least 30 days were provided for public comment on the EA; therefore, the comment period will not be extended.

COMMENT #34 CONTINUED

34b I question whether we really need 4 roundabouts in the less than 2 mile stretch between 2 mile Bridge and the Highway 78 intersection. And can these roundabouts be located to accommodate better access to the new hospital. 34c
I don't see the current need for the second roundabout south of 2 mile bridge. I understand that in the future the land to the west along the golf course could be developed, however, the land on the east of the highway is somewhat limited for development by the creek in this area. I suggest that the installation of that roundabout coincide with the actual development of that site. It has been almost 30 years since the golf course was first developed and there are still many vacant lots to be developed. I believe it will still be a few years before those adjoining lots will be developed and the need for that roundabout will be realized. Please review the need for this roundabout and postpone until a demonstrated need exists.

One of my greatest concerns is for the appearance of the community entrance to Red Lodge. Red Lodge is a unique and beautiful town in an awesome mountain setting. Please don't destroy our community entrance by lining it with Street lights. Residents and visitors all enjoy driving into Red Lodge and looking down the stretch of road at Mount Maurice. The current plan with the raised median calls for street lights from 2 mile bridge into town... while this may serve an access control issue, it does not preserve our unique beauty. We are not "a big city" and prefer to not look like one either. Let us preserve this great view and maintain the things that make Red Lodge unique and special, not just another "city entrance".

34e

I think the vast majority of people would prefer to not have the lights obstructing the mountain view. Additionally, many people will also object to the light "pollution". Red Lodge is a small town and people even comment if the lights are left on at the football field. This will be much more objectionable than lights at a few football games a year. This project has the opportunity to enhance the community entrance if done properly, including burying the overhead power lines that currently exist along the golf course.

I understand the desire for controlled access in this stretch but urge you to look at other alternatives to a lighted raised median. Please look further into the divided road with a center drainage ditch or other alternatives. Please make the visual impacts of this project a high priority.

34f I feel that the balance of the plan in concept is good. I support the proposed Villard and Oakes changes, the Highway 78 roundabout, the Roberts storm water diversion, the passing lanes and changes at Boyd.

Thank you for the opportunity to comment.

Respectfully submitted,

Marcella Manuel

Marcella Manuel Coldwell Banker The Brokers 5 N. Broadway Red Lodge, MT 59068 406 671-3790

RESPONSE #34 CONTINUED

34b

See Response #26a.

34с

See Response #7.

34d

Funding is available for construction of a portion of this corridor. A prioritization plan will be developed in order to meet the highest order needs within the corridor. Once priorities have been established, projects will be programmed accordingly. At this time, the top priorities are the city of Red Lodge and the community of Roberts (See page 1-3 of the EA).

34e

See Response #6.

34f

Thank you for your comment.

16 Dec. 08 Mr. Tom Martin





Dear Mr. Martin,

I am writing in reference to the highway 212 rebuild project near Red lodge. I am especially concerned about overhead lighting in the median strip, and hope it will be held to a minimum. Also any lighting should be of the newer type that shine downward and does not pollute night skies.

Glad there are plans for sidewalks and bike paths, which should extend to the Two Mile Bridge road and as far off the highway as is possible. These
 sidewalks and bike paths should be for non-motor use only.

Sincerely, Ray Masters PO box 573 Red Lodge Mt.

ay Marten

COMMENT #36

From:Dennis Meeker [dmeeker@montana.net]Sent:Friday, December 19, 2008 1:00 PMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comSubject:Comment On Red Lodge North EA

Gentlemen,

36a I am confused about the need for a stop light at 8th Street, yet none at 10th, 11th, 12th and 13th Streets where the real traffic problems exist – especially during busy summer months. I realize that it might not happen for a few years, but the thought of any stop lights in Red Lodge is anathema to our way of life. I also have concerns about all the medians and restricted intersections from the north end of town to Two Mile Bridge Road. It appears that you would have to go one direction first, then make a U-turn a ways down the road to go in the other direction....that doesn't make much sense for the new hospital.

Just my two cents worth, but I've lived here a long time and some of these designs look pretty inconvenient with regards to local traffic. I do think the roundabout at Highways 78 and 212 looks like a good idea and is preferable to a stop light.

Dennis Meeker

RESPONSE #35

35a

See Response #6

35b

See Responses #9 and #14b-c.

35с

Comment noted.

RESPONSE #36

36a

A traffic analysis was not conducted beyond the extents of the project corridor, which begin at 8th Street in Red Lodge and continue generally north-northeast to approximately 1,000 feet north of Boyd. Within the corridor, the 8th Street/US Highway 212 and MT Highway 78/US Highway 212 intersections may warrant traffic signals in the year 2020. To avoid use of a traffic signal, a roundabout is proposed at the MT Highway 78 intersection. A roundabout is not proposed at the 8th Street intersection due to the right-of-way constraints within Red Lodge and the need to avoid impacts to the Carnegie Library, which is listed on the National Register of Historic Places. Additionally, while a traffic signal may be warranted, that does not require a traffic signal to be installed. Installation of traffic signals must be warranted and justified.

36b

See Response #7.

36с

Thank you for your comment.

12-9-08-1 Tom Martin Mt. Dept. Transportation 9.0. Box 201001 2701 Prospect Ave Helena, Mt 59620 Dear Sir " are no letters from the Crow Trib there Indians regularding this project while Appendix C letter recognizes that this corridor was historically occupied by the there is no imput from them Grow Tribe 37a there oral history, occupation, gathering and If we religious use areas don't ask them we don't Know. talks with members of the Crow Tribe they consider the Rocky Fork Branch of the Northern Pacific Railway (24CB1283) as trust land, Accordin treaties with the United to Sta 376 of America (1825, 1851, 1868) any that was abandoned after grante gny treaty reverts to sidnes the trible treaty. Its called a treaty right 2000 Mr. Fredrick had a meeting with the Carbon County

12-9-08ommission ted 57 100 37b wood Continued in via wess ers DV imput 2 **RESPONSE #37**

37a

Section 106 of the National Historic Preservation Act of 1966, as amended, was followed and the process completed for the project (See Section 3.14, Cultural Resources, on page 3-38 of the Environmental Assessment). Upon performing the cultural resources inventories, no tribal issues were identified. Once the reports were completed they were submitted to the Tribe; the Tribe did not provide any comments. Additionally, during preliminary design, reasonable efforts were made to avoid and/or minimize impacts to the railroad bed.

37b

Ownerships of the railroad bed is undetermined at this time and will be addressed during the right-of-way phase of the project. During this phase, the Tribe may have additional opportunities for comment.

From:tparrie@cablemt.netSent:Monday, December 08, 2008 3:08 PMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.com

Thanks for the opportunity to comment on the preliminary design for Highway 212 in Red Lodge.

- ^{38a} I am supportive of the accomodations for pedestrain and bike trafffic, in order to encourage more sustainable commuting options.
- 38b I would like to see the design reflect limited access to 212, in accordance with the gridded network of City streets, per the 2008 Growth Policy. In addition, I am not supportive of the twoway left turn lane. My experience with that concept in Missoula on Reserve St was very
- ^{38c} unfavorable (saw way too many close calls), and I think there are better solutions. I would prefer two 12' travel lanes, two 10' parking lanes, and two 5' sidewalks.

Thanks again, for seeking public comment.

Traute Parrie PO Box 592 Red Lodge, MT 59068 425-2756

COMMENT #39

From:	Janet [ja.peterson.mt@gmail.com]			
Sent:	Friday, December 19, 2008 8:56 AM			
То:	mdtcommentsredlodge@mt.gov			
Cc:	mike.wamboldt@kljeng.com; charlotte.			
brett@kljeng.com; becky.rude@kljeng.com				
Subject:	Comment On Red Lodge North EA			

I am extremely pleased with the plan for roundabouts on the north end of Red Lodge. They are the appropriate solution to current and future traffic concerns. I believe that once people get used to them, their concerns about this new traffic control method will be resolved. This is the right thing to do.

Janet Peterson

RESPONSE #38

38a

Comment noted.

38b

See Response #2b.

38с

See Response #2a.

RESPONSE #39

Thank you for your comment.

From:	Jason S. Priest [jpriest@medipent.com]		
Sent:	Friday, December 19, 2008 6:51 PM		
То:	mdtcommentsredlodge@mt.gov		
Cc:	mike.wamboldt@kljeng.com; charlotte.		
brett@kljeng.c	om; becky.rude@kljeng.com; 'Lindsy Johnson';		
redlodgeplanni	ng@vcn.com; 'Jason Priest'		
Subject:	Comment On Red Lodge North EA		

Hello,

After reading the Carbon County News' coverage of your December meetings I'm moved to submit further comments. As a citizen of Red Lodge, I'm particularly concerned about last minute pleas to modify or eliminate the proposed controlled access scheme that has been consistently supported and implemented after multiple opportunities for property and business owners to review it, after significant community input, and after the many efforts by the City Council to implement the access scheme.

I'd like to state that I'm grateful for Montana Department of Transportation's thorough inquiry, effort, and consideration of our community's needs. I'd also point out that I'm pleased to see new development in the north corridor and that each business, in its own way, is vital to our community.

The proposed controlled access scheme is the result of a well-considered process. A controlled access scheme has been proposed for some time and residents, policy makers, and business and property owners have submitted numerous comments. Furthermore, additional action has been taken to extend the gridded network, particularly Hauser Avenue on the West and Haggin on the East. A summary of the most recent steps follows:

- In October of 2006, the City of Red Lodge adopted Resolution 3223 endorsing controlled access.
- In March of 2007, the City of Red Lodge adopted Resolution 3228 further confirming its preference for controlled access.
- On February 26, 2008 the Red Lodge City Council unanimously approved the Colt Communications Major Subdivision Application subject to access to Highway 212 being limited to the gridded network.
- On March 25, 2008 the Red Lodge City Council unanimously approved the Stampfel Commercial Subdivision Plat with Condition 3 stating "that all the lots within the subdivision shall share a common approach to Highway 212".
- On March 25, 2008 the Red Lodge City Council unanimously approved the application by Red Lodge Ales with Condition 8 stating "that at the time of reconstruction of Highway 212 that the northernmost approach to Highway 212 is limited to right in, right out in keeping with the City Resolutions #3223 and #3228." The applicant has proceeded with the project.
- On March 25, 2008 the Red Lodge City Council voted five to one in favor of approving the application by Beartooth Hospital and Health Center application with Condition 9 stating "that at the time of reconstruction of Highway 212 that the southernmost approach to Highway 212 is limited to ¾ access in keeping with City Resolutions 3223 and 3228". The applicant has proceeded with the project.
- On May 27, 2008 the Red Lodge City Council unanimously approved the application by Mountain Springs Villas subject to Condition 13 which stated "that the portion of 'East Hauser' within the balance of the subdivision is designated as a collector street as provided by Table 4.2 of the Red Lodge Development Code (60 foot right-of-way required). The condition was to facilitate the extension of the gridded network. The applicant has proceeded with the project.
- On August 12, 2008 the City of Red Lodge adopted the 2008 Growth Policy. Several community entrance development strategies were articulated including Strategy B which states: "The Controlled Access Management Plan for the 212 North Corridor shall be implemented consistent with city policy and by coordinating with the Montana Department of Transportation. The city will create clear standards about how the controlled access shall be implemented through city

40a

40a Continued

policy. The city will expand its network of city streets and alleys as development occurs. The expansion will include Hauser and Haggin Avenues to the north.

On August 12, 2008 the City of Red Lodge adopted the 2008 Growth Policy with specific regulator strategies to address the gridded network. Regulation B states "Hauser Avenue shall be extended from its current north terminus to Two Mile Bridge Road. Site development east of U.S. Highway 212 in the Urban Growth Boundary of the 212 North Corridor shall include provisions for Haggin Avenue to be extended from its current northerly terminus to Two Mile Bridge Road, allowing for site considerations to determine the exact location of the public right of way."

An improvement in the overall public health and safety of the community will be the result of these actions. The proposed controlled access scheme presents the best opportunity to enhance the safety of all residents and visitors and not just those businesses in the north corridor. The proposed scheme, plus the extension of the gridded network north, including Hauser Avenue all the way to the full access point at the northern terminus of the controlled access section, increases safety for all motorists and non-motorists for the following important reasons:

- Route options and controlled access will reduce conflicts between through traffic and local traffic.
- Increased route redundancy and controlled access improve safe access to commercial businesses and the Beartooth Hospital since the closure of one route does not prevent access via another.
- Controlled access will slow speeds and route diversity to commercial businesses, and especially the hospital, will increase the chances of customers/patients safely reaching their destination in a timely manner since there are multiple opportunities.

It's true that every business would like the rules modified for any number of reasons specific to their business. The fact is that each business on Highway 212 has chosen to locate on a major state highway. That choice carries risks for their customers and the community at large which all parties in good faith have attempted to mitigate. It's my personal opinion that controlled access and the gridded network represent the best alternatives for reducing conflict for the largest number of people for the most typical uses. To my mind, protecting most people is better than reducing safety for all in order to meet individual user-specific business requirements associated with low probability events. I fear safety improvements for some businesses would come at the expense of all if changes to the proposed scheme were made. I believe the record demonstrates consistent application of that principal.

It is also important to note that each business, after having conditions imposed upon it to conform to the controlled access scheme, has chosen to **proceed** with their development.

Thank you for allowing participation in your process. Since I live on Hauser Avenue and grew up here in Red Lodge, I feel these decisions are important. My comments are respectfully submitted as a member of the community and not as the Ward One City Council representative.

RESPONSE #40

40a

Thank you for your comment.

COMMENT #40 CONTINUED

Regards,

Jason

PS. Here are a few additional comments.

40b Please remove the TWLTL from the 'MT Highway 78 to Developed Limits of Red Lodge' section of the document. The section from Hwy 78 to the Controlled Access Corridor should be consistent with the typical section from 8th Street to Highway 78 and the Historic Downtown. See the 2008 Red Lodge Growth Policy.

40c Please add a 5 foot wide sidewalk to the West side of Highway 212 in the Controlled Access Corridor per years of discussion and public comment. This sidewalk has been in many drafts of the proposed design of this section of Highway 212. See the 2008 Red Lodge Growth Policy.

- 40d Please extend a sidewalk north to connect to the Spires ped/ bike trail that comes off the bench approximately ½ mile north of Two Mile Bridge Road.
- 40e Please assure that safe pedestrian crossings are designed and built into all access points in the Controlled Access Management Corridor. See the 2008 Red Lodge Growth Policy.

Jason S. Priest CEO Medipent, LLC PO Box 743 Red Lodge, MT

jpriest@medipent.com

(406) 446-1787 Phone (206) 350-3122 Fax

RESPONSE #40 CONTINUED

40b

See Response #2a.

40c

See Response #14d.

40d

See Response #9.

40e

See Response #15c.

	From:	Elizabeth Scanlin [bescanlin@msn.com]		
	Sent:	Thursday, December 18, 2008 9:59 AM		
	То:	mdtcommentsredlodge@mt.gov		
	Cc:	mike.wamboldt@kljeng.com; charlotte.		
	brett@kljeng.cor	m; becky.rude@kljeng.com		
	Subject:	Comment On Red Lodge North EA		
TO:	Tom S. Martin Montana Department of Transportation P.O. Box 201001 Helena, MT 59620			
FROM:	Elizabeth Scanlin, Mayor,	Elizabeth Scanlin, Mayor, City of Red Lodge		
DATE:	December 18, 2008			
RE:	Comments to Corridor Stu Environmental Assessment Section of 8 th Street to MT	dy – Red Lodge North t `Highway 78 in Red Lodge, Montana		

Please accept the following written comments to confirm and in addition to those I made at your public hearing on the above matter in Roberts on December 10, 2008:

1. Oakes Ave. from 7th Street south to 13th Street serves as a "second Main Street" for Red Lodge residents and visitors: the major facilities located on this avenue include our only supermarket (at $8^{th} - 9^{th}$ St.), a "box store" hardware store (at $10^{th} - 11^{th}$ St.), our only nursing home ($11^{th} - 12^{th}$ St.), and our only post office ($12^{th} - 13^{th}$ St.). Over the past 15 years, City policy-makers have purposely encouraged, subsidized and facilitated the location of these facilities in the core of our small town to "keep its heart beating." Making access to these facilities more difficult will encourage them to move outside the center of town, creating "sprawl" that we have tried very purposely to avoid.

2. It is my understanding of the current proposed preferred alternative makes Oakes Ave. a one-way route with southerly traffic from 7th St. This pattern would obviously significantly change an established pattern of traffic from and to the major facilities listed in 1. above. Such change in movement may affect the public's use of these facilities and encourage them to move outside the center of town, creating "sprawl" that we have tried very

41b

41a

purposely to avoid.
3. Stop signs on at the north and south sides of the intersection of Oakes and 8th St. also will significantly affect the current steady flow of traffic on Oakes as our "second Main Street," which again may affect the use of the facilities stated in 1. above and

encourage them to move outside the center of town, creating "sprawl" that we have tried very purposely to avoid.

4. By disrupting the flow of traffic on Oakes Ave., traffic to and from Beartooth Market will use 8th Street to turn off of and return to Broadway, already a busy intersection with sight lines limited looking north past Carnegie Library, a significant building with historic designation whose alteration for traffic purposes may be difficult. There is space for only two vehicles waiting on 8th St. to enter Broadway without blocking the intersection of 8th St. and Oakes Ave. Stop signs remaining on the east and west sides of the intersection of Oakes and 8th Street would allow an uninterrupted flow of traffic on Oakes for parties traveling to and from the supermarket, preventing obstruction of that intersection.

5. Changing the flow of traffic on Oakes Ave. at 8th Street would also encourage traffic from the West Bench on Highway 78 to find alternative routes to and from the core of Red Lodge, particularly through the residential historic district of "High Bug" on the northwest corner of the town's valley floor. Both Word and Hauser Avenues would be negatively affected. Word Ave. is a narrow street with only one lane available in many places, making additional traffic hazardous. Hauser Ave. has most of our few "mansions" that housed our early bankers and mine owners, and additional traffic would negatively affect both the character and safety of that street. Most of Red Lodge's substantial development over the past 15 years is on the West Bench, with 800 new lots approved within the past six years alone, with resulting increase in traffic into town that needs to be considered here.

My main concerns are that the proposed traffic changes on Oakes Ave. in the area from 7th to 8th Streets will create a negative impact on the economics of our town by making access to at least four significant businesses and facilities on Oakes more difficult, encouraging them to move outside the center of town and creating "sprawl" that we have tried very purposely to avoid.

RESPONSE #41

41a

Comment noted.

41b

The proposed improvements were developed to address four issues:

1. The Oakes Avenue/US Highway 212 intersection is currently an undefined access, allowing vehicles to enter/ exit US Highway 212 at a sharp angle. The sharp angle restricts the ability of motorists to view other conflicting traffic, which is a safety concern. The proposed improvement would reconstruct the Oakes Avenue access as close to a 90 degree intersection as possible with US Highway 212.

2. The existing sight distance is inadequate for eastbound vehicles on 8th Street that are looking north up US Highway 212 past the Carnegie Library. The proposed improvements would improve the sight distance without negative impacts to the historic Carnegie Library.

COMMENT #41 CONTINUED

- ^{41c} traffic changes will also create a negative impact of traffic at 8th Street as well as in the neighboring historic residential district.
- 41d I therefore request that a traffic study of the area be conducted to address the potential negative impacts of the proposed alternative before this project proceeds.

The

Thank you for this opportunity to comment.

Elizabeth ("Betsy") Scanlin, Mayor

RESPONSE #41 CONTINUED

41b Continued

3. The existing traffic control on 8th Street includes stop signs at Oakes Avenue and US Highway 212, leaving approximately 60 feet of motor vehicle storage space between US Highway 212 and Oakes Avenue. Motorists traveling south on US Highway 212 and then wishing to turn west on 8th Street have the potential to encounter vehicles stopped along 8th Street due to the close proximity of stop signs. There is potential for stopped vehicles on 8th Street to essentially block the US Highway 212 travel lanes, creating a safety issue. The removal, relocation, and/or placement of stop signs within the project corridor, such as the 8th Street and Oakes Avenue intersection, will be determined during final design to ensure safety and traffic operation concerns are addressed.

4. Coordination with the Arts Guild, Carnegie Library and the City of Red Lodge yielded a request from the City to increase parking in the vicinity if possible. Reducing the Oakes Avenue travel lanes to a single one-way lane between 8th Street and US Highway 212 allows for angle parking on both sides of Oakes Avenue in this area, which would increase the number of parking spaces.

41c

Impacts to historic and cultural properties was analyzed in the Environmental Assessment. Coordination with the State Historic Preservation Officer resulted in a finding of No Effect to the Hi Bug Historic District. Please refer to Section 3.14, Cultural Resources, on page 3-38 of the EA.

41d

A traffic analysis was completed on the 8th Street and Oakes Avenue intersection. No significant changes in traffic patterns due to the conversion of Oakes Avenue to a one-way street were identified. Additional traffic analysis will be conducted during final design to ensure safety and traffic operation concerns are addressed.

CONSOLIDATED DITCH CO, INC. 280West Bench Road Roberts, Montana 59070

Mr. Mike Wombolt, Project Engineer Kadrmas, Lee & Jackson, Inc. P.O. Box 80303 Billings, Mt 59108

Received

DEC 1 8 2008

Kadrmas Lee & Jackson

December 15, 2008

Dear Mr. Wombolt;

The Board of Directors on the Consolidated Ditch would like to enter a comment into the record for the ongoing Environmental Assessment (EA) regarding the proposed Highway construction between Red Lodge and Roberts, Montana. As directors of the ditch company, we are concerned that the EA does not adequately address the impact that the new roadway construction will have upon the Consolidated Ditch.

The Consolidated Ditch passes under Highway 212 approximately five miles north of Red Lodge between mile marker 74 and 75. The approximate GPS co-orindates are N45.25982W109.22758 (see attached aerial view). We are concerned that the present culvert facility needs to be lengthened to accommodate the increase in roadway width. Additionally, we believe the culvert, as constructed is undersized to carry the full 28061/2 miners inches (approximately 70 cu ft/sec) plus an adequate margin for storms or sudden increase in flow.

The EA as published on the Dept. of Transportation website indicates, "no impacts anticipated" on page 3-20 and that the Consolidated Ditch Company was consulted on page 6-4. Because we have not been consulted, we would like to bring the following information to your attention:

Presently, when we run a full head of water (2806 ¹/₂ inches), the water rises a foot over the existing structure and runs over the subdivision road and down the highway barrow pit. We continually remove trash from this sight, which also adds to the problem. A larger culvert would accommodate the trash, accept any flood and or wastewater and prevent the spill over.

We would be happy to discuss the matter with you or your representative.

Sincerely,

Curte Schward

Curtis Schwend, President Consolidated Ditch Company, Inc. Board of Directors 406-446-1893

RESPONSE #42

The statement that the Consolidated Ditch Company had been consulted was made incorrectly, having been based on the planned distribution of the Administrative Draft Environmental Assessment to the president of the company. However, it has since been determined that the document was not submitted to the president and, therefore, the Consolidated Ditch Company did not receive a copy of the document. This has been reflected in the Finding of No Significant Impact (FONSI). See Section 4.0, Clarifications to the EA, in the FONSI.

Table 3.7, Impacts to Irrigation Facilities, on page 3-19 of the Environmental Assessment only reflects potential relocations of irrigation facilities. Replacement of drainage structures would be determined during design. Drainage structures would be designed to address hydrologic conditions and comply with federal and state regulations including the Montana Stream Protection Act, Federal Clean Water Act, and Section 404/401, as applicable. Irrigation facilities would be designed in consultation with ditch owners and operators to minimize impacts to farming/ranching operations.
[Page Intentionally Left Blank]

COMMENT #42 ATTACHMENT



COMMENT #43

From:Dave Stauffer [dave@staufferbury.com]Sent:Thursday, December 04, 2008 10:53 PMTo:mdtcommentsredlodge@mt.govCc:Jason Priest; becky.rude@kljeng.com; charlotte.brett@kljeng.com; mike.wamboldt@kljeng.comComment On Red Lodge North EA

RE: Red Lodge North EA 4 December 2008

Dear MDT:

From 1999 to 2008 I served consecutively in the Red Lodge City government as president of the Planning Board, city planner, and City Council member. In those positions, I was continuously involved in development of the plans for this project, and I also attended planning conferences across the U.S. where I learned about many things that do and don't work in context sensitive corridor design.

With that background, I urge you to take the following actions:

 To maintain continuity with downtown Red Lodge, and so that the entrance to our community states definitively that we are someplace unique, access to 212 should be limited to points that are consistent with the gridded network of City streets (per Chapter <u>7</u> of the 2008 Growth Policy).

2. To prevent a City entrance characterized by typical suburban sprawl, the proposed

continuous Two-Way Left Turn Lane should be replaced by two 12 foot travel lanes, two 10 foot parking lanes, two 5 foot sidewalks.

By taking these actions, MDT will be respecting the often expressed opinions of the citizens who live in and near Red Lodge, including two resolutions by the Red Lodge City Council while I served on that body.

Thank you for your attention to these requests. Sincerely, David Stauffer 214 S. Platt Ave.-PO Box 727 Red Lodge, MT 59068-0727 <u>dave@staufferbury.com</u> tel. 406-446-2563

43a

43b

RESPONSE #43

43a

See Response #2b.

43b

See Response #2a.

COMMENTS

Corridor study - Red Lodge North Environmental Assessment

PROJECT # STPP-28-2(25)70. CN 4375 TUESDAY, DECEMBER 9. 2008—RED LODGE, MT RED LODGE SENIOR CITIZENS CENTER

Please use the space below to tell us your comments regarding the proposed Corridor Study – Red Lodge North Project.

Note: Comments are considered public records.

	Name: Corey Thompson	
	PRINT Address: 20 N Platt Reallodge INT 5900	68
	∂	THIS SDACE
	The round about at the intersection	OFFICE USE ONL
	18 Hwy 18 + 212 is a great idea:	
	however 3 may roundabout is overkill.	
14a -	properially if turn of & lanes appind	
-	available in a 2 mile urea	
-	Bike redestrian trait is a unsderte	2
-	Addition, Please consider extending it	
•	approximately to my further anoth	
-	to the round barne to connect with	
146	a trail extending from the West, coming	
	from the spires development. This would	
-	one the citizens of Red Lodge a circular	
-	path that they have planned for plars	
-		
-		
-		
-		
	Please mail your comments by December 19, 2008 to:	
	Montana Department of Transportation Tom S. Martin	12040
	PO Box 201001	
	2701 Prospect Ave Helena, MT 59620	
	Comments may also be submitted online at:	
	http://www.mdt.mt.gov/pubinvolve/eis_ea.shtml	



44a

See Response #26a.

44b

See Response #9.

[Page Intentionally Left Blank]

COMMENT #44 ATTACHMENT



COMMENT #45

From:David Torgerson [sonle@blackfoot.net]Sent:Thursday, December 18, 2008 6:22 PMTo:mdtcommentsredlodge@mt.govCc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comComment On Red Lodge North EA

Dear Sirs:

45a

As I just recently moved to Red Lodge area (60 Two Mile Bridge Rd). I have not had a chance to become familiar with the proposed Red Lodge North Improvement Project. The local paper just came out with the article regarding your Environmental Assessment and mentioned an upcoming hearing but no scheduled date.

Please place me on a cc list for upcoming public meeting dates. Also as a Montana Licensed Civil Engineer P.E.,
please provide information regarding how I can get involved in the project via local committee. Based on my quick
scanning of the EA document this is a project which I would support and feel that it is much needed regarding
differential speeds for the turn onto Two Mile Bridge Road and pedestrian and bicycle safety to get these non-motorized user away from presently dangerous travel lanes along Highway 212.

I look forward to supporting the project and your assistance to define a role that I can provide to see that this proposed project moves into construction as timely as possible.

Best Regards,

David Torgerson, P.E. Civil Engineer 60-Two Mile Bridge Road Red Lodge, MT 406.446.3646 ph 406.431.7137 cell

45a

Public hearings were held on December 9, 2008, in Red Lodge and December 10, 2008, in Roberts. Both meetings were advertised in the *Carbon County News* and *Billings Gazette* for four consecutive weeks beginning on November 13 in the *Carbon County News*, and November 17 in the *Billings Gazette*. Press releases were also distributed to local media sources in the same intervals as the news ads. See Section 3.0, Comments and Coordination, in the Finding of No Significant Impact.

45b

No further public meetings are planned at this time. However, you have been added to the project mailing list and will be notified if further meetings are held.

45c

Thank you for your interest in the project. Please contact the City of Red Lodge and/or Carbon County to determine local involvement opportunities.

45d

Thank you for your comment.

From:Luke Wright [wright8208@yahoo.com]Sent:Friday, December 19, 2008 2:34 PMTo:mdtcommentsredlodge@mt.gov;mdtcommentsredlodge@mt.govmike.wamboldt@kljeng.com; charlotte.Cc:mike.wamboldt@kljeng.com; charlotte.brett@kljeng.com; becky.rude@kljeng.comComment On Red Lodge North EA

MDT,

I oppose the idea of using any personal property on the east side of Highway 212, to expand the highway through Roberts, MT.

Thanks for the opportuinity to comment.

Luke Wright

406-425-1831

12/19/08

COMMENT #47

December 19, 2008

To Montana Dept. of Transportation PO BOX 201001 Helena MT. 59620-1001

Dear Tom Martin:

Please do not allow any personal property encroachments on the EAST side of highway 212 in Roberts MT, when designing the highway.

Thanks:

Lyle Wright Wright Company 406-425-1825

PS: 1 tried to comment on line but the e-mail address is not good!!

RESPONSE #46

See Response #8.

RESPONSE #47

See Response #8.

Appendix C

Environmental Assessment (October 2008)

Corridor Study -Red Lodge North Environmental Assessment &

Section 4(f) Evaluation

Project Number STPP 28-2(25)70

Control Number 4375

Carbon County, Montana

October 2008

U.S. Department of Transportation Federal Highway Administration



Environmental Assessment & Section 4(f) Evaluation Corridor Study – Red Lodge North STPP 28-2 (25) 70 (Control Number 4375) Carbon County, Montana

This document is prepared in conformance with the Montana Environmental Policy Act (MEPA, 75-1-201 MCA) requirements and contains the information required for an Environmental Assessment (EA) under the provisions of Administrative Rules of Montana (ARM) 18.2.237(2) and 18.2.239. This EA is also prepared in conformance with the National Environmental Policy Act (NEPA) requirements for an Environmental Assessment under 23 CFR 771.119, and Section 4(f) of the United States Department of Transportation Act under 23 CFR 774.

Submitted pursuant to 42 USC 4332(2)(c), 49 USC 303, Sections 75-1-201 & 2-3-104, MCA and Executive Orders 11990, 11988, and 12898 by the United States Department of Transportation, Federal Highway Administration and the Montana Department of Transportation Cooperating Agencies: US Army Corps of Engineers City of Red Lodge Carbon County

Submitted by:

Montana Department of Transportation

Reviewed and Approved for Distribution:

US Department of Transportation Federal Highway Administration

The following persons may be contacted for additional information concerning this document:

Tom S. Martin, PE Environmental Services Bureau Chief Montana Department of Transportation 2701 Prospect Ave. PO Box 201001 Helena, Montana 59620-1001 (406) 444-0879 Alan C. Woodmansey, PE Montana Division Operations Engineer Federal Highway Administration US Department of Transportation 585 Shepard Way Helena, Montana 59601 (406) 449-5302

ABSTRACT: The proposed action is to reconstruct approximately 21.2 miles (34.1 kilometers) of US Highway 212 between Red Lodge and Boyd, Montana. The proposed project starts at 8th Street in Red Lodge and ends approximately 1,000 feet (304.8 meters) north of Boyd. The purpose of the proposed project is to improve the roadway deficiencies and to improve the safety and operational characteristics of the roadway.

Comments on this Environmental Assessment & Section 4(f) Evaluation are due by <u>December 19, 2008</u>, and should be sent to Tom S. Martin, Montana Department of Transportation, at the address above. This document may also be viewed on the MDT website at <u>http://www.mdt.mt.gov/pubinvolve/eis ea.shtml</u>. MDT attempts to provide accommodations for any known disability that may interfere with a person participating in any service, program or activity of the Department. Alternative accessible formats of this information will be provided upon request. For further information call (406) 444-6331 or TTY (800) 335-7592, or by calling Montana Relay at 711.

Date:

Date 15 AT

Table of Contents

EXECUTIVE SUMMARY

Ι.	Introduction	S-1
II.	Purpose and Need for the Project	S-1
	i. Project Objectives	S-2
	ii. Supporting Element	S-2
III.	Alternatives	S-2
	i. Alternative A: No-Build	S-3
	ii. Alternative B: Preferred Alternative	S-3
	a. Red Lodge (8 th Street to Two Mile Bridge Road)	S-3
	b. Red Lodge to Roberts	S-5
	c. Roberts	S-5
	d. Roberts to Boyd	S-6
IV.	Affected Environment, Impacts, & Mitigation	S-7
V.	Permits and Authorizations	S-14
VI.	Coordination	S-14
	i. Cooperating Agencies	S-14
	ii. Coordination with Other Agencies, Organizations, and Interested	
	Parties	S-15
	iii. Coordination with the Public	S-15
	iv. Public Hearing	S-16
	v. Conclusion	S-16

CHAPTER 1 PURPOSE OF AND NEED FOR ACTION

1.1	Introduction	1-1
1.2	Purpose and Need for the Project	1-3
	1.2.1 Need to Improve Safety and Operational Characteristics of the	
	Roadway	1-3
	1.2.2 Supporting Element	1-9
1.3	Project Objectives	1-9
1.4	History of the Planning & Scoping Process	1-10
1.5	Relevant Plans	1-10

CHAPTER 2 ALTERNATIVES

2.1	Introd	luction	2-1
2.2	Altern	ative A: No-Build Alternative	
2.3	Altern	ative B: Preferred Alternative	
	2.3.1	Red Lodge (8 th Street to Two Mile Bridge Road)	
		2.3.1.1 Typical Sections	
		2.3.1.2 Drainage	
		2.3.1.3 Intersections	
		2.3.1.4 Access Management	
	2.3.2	Red Lodge to Roberts	
		2.3.2.1 Typical Section	
dor Study	– Red I	odge North	i

		2.3.2.2	Drainage	
	2.3.3	Roberts.	~	
		2.3.3.1	Typical Sections	
		2.3.3.2	Drainage	
		2.3.3.3	Intersection	
		2.3.3.4	Pedestrian Facilities	
	2.3.4	Roberts	to Boyd	
		2.3.4.1	Typical Sections	
		2.3.4.2	Drainage	
		2.3.4.3	Intersections	
2.4	Alterna	atives Elin	ninated from Further Study	
	2.4.1	Red Lod	ge	
	2.4.2	Red Lod	ge to Roberts	
	2.4.3	Roberts.	-	
	2.4.4	Roberts	to Boyd	2-36
	2.4.5	Project C	Corridor	

CHAPTER 3 AFFECTED ENVIRONMENT, IMPACTS, & MITGIATION

3.1	Introduction	3-1
3.2	Environmental Considerations Unaffected by the Project	3-1
3.3	Land Use	3-2
3.4	Farmland	3-4
3.5	Transportation System	3-5
	3.5.1 Traffic	3-5
	3.5.2 Access	3-6
	3.5.3 Safety	3-7
	3.5.4 Pedestrian/Bicycle Facilities	3-9
3.6	Traffic Noise	3-14
3.7	Right-of-Way and Relocations	3-17
3.8	Water Resources/Quality	3-18
	3.8.1 Surface Water	3-18
	3.8.2 Irrigation Facilities	3-19
	3.8.3 Ground Water	3-22
	3.8.4 Public Water Supplies	3-23
	3.8.5 Waste Water Systems	3-24
3.9	Water Body Modifications	3-24
3.10	Wetlands	3-25
3.11	Wildlife Habitat and Ecosystems	3-31
	3.11.1 Vegetation	3-31
	3.11.2 Terrestrial and Avian Species	3-32
	3.11.3 Aquatic Species	3-33
	3.11.4 Montana Species of Concern	3-33
	3.11.5 Animal-Vehicle Collisions	3-34
3.12	Floodplain	3-36
3.13	Threatened or Endangered Species	3-37
3.14	Cultural Resources	3-38
3.15	Section 6(f) Properties	3-40
3.16	Hazardous Materials, Solid Waste, and Underground Storage Tanks.	3-41
or Study	– Red Lodge North	ii

3.17	Visual/Aesthetic Considerations	3-44
3.18	Temporary Construction Considerations	3-44
	3.18.1 Air Quality	3-44
	3.18.2 Transportation System	3-45
	3.18.3 Water Resources/Quality	3-45
	3.18.4 Wildlife Habitat and Ecosystems	3-46
	3.18.5 Noise	3-46
3.19	Utilities	3-47
3.20	Cumulative Considerations	3-47
	3.20.1 Past, Present, and Reasonably Foreseeable Actions	
	3.20.2 Cumulative Impacts	3-49
3.21	Irreversible and Irretrievable Commitment of Resources	3-51
3.22	Short-term Impacts versus Long-term Benefits	
3.23	Permits and Authorizations	
3.24	Environmental Commitments/Mitigation	

CHAPTER 4 SECTION 4(f) EVALUATION

4.1	Introduction	4-1
4.2	Section 4(f) Properties	4-1
	4.2.1 Historic Properties	4-1
	4.2.2 Historic Irrigation Ditches	4-3
	4.2.3 Recreation Areas	

CHAPTER 5 PREPARERS AND REVIEWERS

5.1	Introduction	5-	1
5.2	Preparers and Reviewers	5-	1

CHAPTER 6 COMMENTS AND COORDINATION

6.1	Introduct	tion	6-1
6.2	Coopera	iting Agencies	6-1
	6.2.1 N	leetings with the City of Red Lodge	6-1
	6.2.2 R	Red Lodge Resolutions	6-2
	6.2.3 N	leetings with Carbon County	6-2
6.3	Coordina	ation with Other Agencies, Organizations, and Interested Parties	6-2
6.4	Coordina	ation with the Public	6-4
	6.4.1 K	Kickoff Meeting	6-4
	6.4.2 A	Alternatives Public Workshop	6-4
	6.4.3 S	Storm Water Drainage Meetings in Roberts	6-5
	6.4.4 A	Access Management in North Red Lodge – Meeting #1	6-5
	6.4.5 A	Access Management in North Red Lodge – Meeting #2	6-5
	6.4.6 C	Other Meetings	6-6
6.4.7	Public He	earing	6-6

CHAPTER 7 REFERENCE LIST/SOURCE DOCUMENTS

7.1	Introduction	7-1
7.2	Subconsultant Reports	7-1
7.3	Other Reports	7-1
7.4	Community Plans/Policies	7-2
7.5	Websites	7-2

REFERENCE MATERIALS

Acronym List	vi
Approximate Route Post Key	vii
Conversion Table	viii

TABLES

Table A	Summary Comparison of Project Alternatives and Impacts	S-8 1-8
Table 2.1	Summary of Preferred Alternative	
Table 3.1	Population Statistics	3-3
Table 3.2	Planned Development	3-3
Table 3.3	Average Annual Daily Traffic (AADT)	3-6
Table 3.4	Crash Summary	3-8
Table 3.5	Noise Abatement Criteria, Exterior Noise Levels	3-16
Table 3.6	Existing and Future Noise Levels for Impacted Properties	3-16
Table 3.7	Impacts to Irrigation Facilities	3-19
Table 3.8	Estimated Wetland Impacts	3-27
Table 3.9	Impacts to Historic Properties	3-39
Table 3.10	Leaking Underground Storage Tanks	3-42
Table 3.11	Underground Storage Tanks	3-42
Table 5.1	Preparers and Reviewers	5-1
Table 6.1	Agencies Consulted	6-3

FIGURES

Figure 1-1	Project Location Map	1-2
Figure 1-2	Conceptual Intersection Angles	1-4
Figure 1-3	Conceptual Offset Intersection Overview	1-5
Figure 1-4	Conceptual Ditch Inslope Illustration	1-7
Figure 1-5	Conceptual Clear Zone Illustration	1-8
Figure 2-1	Segment/Intersection Locations	2-3
Figure 2-2	Red Lodge Typical Section Locations	2-7
Figure 2-3	8 th Street to MT Highway 78	2-8
Figure 2-4	TWLTL (Two-Way Left-Turn Lane) Example	2-9
Figure 2-5	MT Highway 78 to Developed Limits of Red Lodge	2-10
Figure 2-6	Developed Limits of Red Lodge to Two Mile Bridge Road	2-11
r Study – Red Lo	odge North	iv

Figure 2-7	Oakes Avenue/8 th Street Intersections	2-13
Figure 2-8	MT Highway 78/Villard Avenue Intersections	2-15
Figure 2-9	Developed and Developing Limits of Red Lodge	2-16
Figure 2-10	Access Management Plan Overview	2-17
Figure 2-11	Conceptual Full Access Intersection	2-18
Figure 2-12	Conceptual Roundabout Intersection	2-18
Figure 2-13	Conceptual ¾ Access Intersection	2-19
Figure 2-14	Conceptual 1/2 Access Intersection	2-19
Figure 2-15	Two Mile Bridge Road to South End of Roberts	2-21
Figure 2-16	Roberts Typical Section Locations	2-22
Figure 2-17	South End of Roberts to East Maple Street	2-23
Figure 2-18	East Maple Street to North End of Roberts	2-24
Figure 2-19	Cooney Dam Road Intersection (near East Maple Street)	2-26
Figure 2-20	Roberts to Boyd Typical Section Locations	2-27
Figure 2-21	North End of Roberts to Boyd	2-29
Figure 2-22	North End of Roberts to Boyd – Boyd Country Store	2-30
Figure 2-23	Clear Creek Road Intersection	2-32
Figure 2-24	Cooney Dam Road and Main Street Intersections	2-33
Figure 3-1	City of Red Lodge Existing and Planned Trails	3-11
Figure 3-2	Preferred Alternative Consistency with City of Red Lodge	
-	Comprehensive Trails Plan	
Figure 3-3	Roberts Pedestrian/Bicycle Facilities	3-13
Figure 3-4	Passing Zones	3-15
-		

APPENDIX A

Cooperating Agencies

APPENDIX B

NRCS Coordination & CPA-106 Forms

APPENDIX C

March 2002 Agency Scoping

APPENDIX D

May 2007 Agency Scoping

APPENDIX E

Cultural Resources

APPENDIX F

Section 4(f) Properties

APPENDIX G

Section 4(f) De Minimis Evaluations

Acronym List

AADT	Average Annual Daily Traffic
AASHTO	American Association of State Highway Transportation Officials
ADA	Americans with Disabilities Act
ARM	Administrative Rules of Montana
AVC	Animal-Vehicle Collision
ВМР	Best Management Practice
CERCLA CEQ CFR CFS CTEP CWA	Comprehensive Environmental Response, Compensation, and Liability Act Council on Environmental Quality Code of Federal Regulations Cubic Feet Per Second Community Transportation Enhancement Program Clean Water Act
EA	Environmental Assessment
EIS	Environmental Impact Statement
EPA	Environmental Protection Agency
FEMA	Federal Emergency Management Agency
FHWA	Federal Highway Administration
KL&J	Kadrmas, Lee & Jackson, Inc.
KPH	Kilometers Per Hour
LOS	Level of Service
LUST	Leaking Underground Storage Tank
MCA	Montana Code Annotated
MDEQ	Montana Department of Environmental Quality
MDT	Montana Department of Transportation
MCA	Montana Code Annotated
MT FWP	Montana Fish, Wildlife and Parks
MLR	Montana Land Reliance
MOA	Memorandum of Agreement
MPDES	Montana Pollutant Discharge Elimination System
MPH	Miles Per Hour
MSAT	Mobile Source Air Toxics
NEPA	National Environmental Policy Act
NPDES	National Pollutant Discharge Elimination System
NRCS	Natural Resources Conservation Service
NRHP	National Register of Historic Places
RCRA	Resource Conservation and Recovery Act
RDM	Road Design Manual

SAFETEA-LU	Safe, Accountable, Flexible, Efficient Transportation Equity Act: A Legacy for Users
SHPO	State Historic Preservation Officer
SPA	Stream Protection Act
STIP	State Transportation Improvement Program
SWLF	Solid Waste Landfill
TWLTL	Two-way Left-Turn Lane
USACE	United States Army Corps of Engineers
USC	United States Code
USFWS	United States Fish and Wildlife Service
UST	Underground Storage Tank

VMT Vehicle Miles Traveled

Approximate Route Post Key				
Location	Route Post			
Project Corridor	69.83 to 91.00			
Red Lodge	69.83 to 71.54			
Oakes Avenue	69.85			
MT Highway 78 / US Highway 212	70.19			
Developed Limits of Red Lodge	70.68			
Two Mile Bridge Road	71.54			
Red Lodge to Roberts	71.54 to 81.84			
Approximate Northbound Passing Zone Location	72.0 to 74.0			
Fox Road (Approximate Bus Turnaround Location)	76.86			
Roberts	81.84 to 82.62			
South End of Roberts	81.84			
Cooney Dam Road (Roberts)	82.45			
North End of Roberts	82.62			
Roberts to Boyd	82.62 to 91.00			
Approximate Northbound Passing Zone Location	82.7 to 83.7			
Clear Creek Road	84.55			
District Boundary (Approximate Bus Turnaround Location)	87.58			
Approximate Southbound Passing Zone Location	90.2 to 88.2			
Cooney Dam Road (Boyd)	90.70			
Boyd Country Store (Main Street)	90.77			

Conversion Table			
Metric English			
1 meter	3.281 feet		
1 kilometer	0.621 miles		
1 hectare	2.471 acres		
English	Metric		
1 foot	0.305 meters		
1 mile	1.609 kilometers		
1 acre	0.405 hectares		

EXECUTIVE SUMMARY CORRIDOR STUDY – RED LODGE NORTH PROJECT NUMBER: STPP 28-2(25)70, CONTROL NUMBER: 4375

I. INTRODUCTION

The MDT (Montana Department of Transportation) plans to reconstruct approximately 21.2 miles (34.1 kilometers) of US Highway 212 in Carbon County. The proposed project begins at 8th Street in Red Lodge and continues generally north-northeast to approximately 1,000 feet (305 meters) north of Boyd.

The existing facility is a rural two-lane roadway with 12-foot (3.6-meter) driving lanes and approximately 2-foot (0.6-meter) shoulders. The proposed project corridor extends through the northern portion of the city of Red Lodge and through the unincorporated communities of Fox, Roberts, and Boyd. See Figure 1-1, Project Location Map, on page 1-2.

US Highway 212 in the project area is functionally classified as a rural minor arterial and is on Montana's Primary Highway System. US Highway 212 serves local, commuter, tourist, and agricultural traffic and is considered an integral part of the regional transportation network. In addition, US Highway 212 connects a number of local roads to the regional transportation network.

MDT estimates \$21.3 million will be available for projects on this corridor from the SAFETEA-LU Section 1934 Transportation Improvement Project #246 to develop and construct US 212 Red Lodge North. At this time, the top priorities for construction are the city of Red Lodge and the community of Roberts. The other segments (between Red Lodge and Roberts as well as Roberts and Boyd) of the project would be constructed as funding becomes available.

II. PURPOSE AND NEED FOR THE PROJECT

The purpose of the proposed project is to improve the safety and operational characteristics of the roadway, as discussed below. Safety and operational concerns along the project corridor include:

- Pavement—Nineteen of the 21 miles (30.6 of the 33.8 kilometers) of the project corridor have exceeded the design life for asphalt pavement.
- Intersections—Numerous intersections along the project corridor have geometric deficiencies, meaning they are either skewed or offset. Two intersections in Red Lodge also have inadequate capacity for anticipated future traffic.
- Access—MDT worked with the City of Red Lodge to develop an access management plan for US Highway 212, between MT Highway 78 and Two Mile Bridge Road.
- Pedestrian/Bicycle Facilities—There are discontinuous sidewalks and no bicyclist accommodations along the project corridor. The City of Red Lodge Comprehensive Trails Plan of May 2006 identified the need for sidewalk and shared bike/ped path facilities along US Highway 212 within Red Lodge.
- LOS (level of service)—Based on increased traffic, the existing two-lane roadway does not provide an adequate LOS along the rural segments. Two intersections in Red Lodge would also provide inadequate LOS by the project design year.

- Shoulders—The existing 2-foot (0.6-meter) roadway shoulders are narrow, and the public has indicated support for wider shoulders to accommodate emergency parking, wide agricultural vehicles, etc.
- *Ditch Slopes*—Many of the roadside ditches along the project corridor have steep slopes. A flatter inslope is considered more desirable for a roadway such as US Highway 212 in that an errant vehicle has a greater chance of recovery.
- Clear Zones—Obstacles are present to various degrees within the clear zone along US Highway 212.
- *Safety*—In the 15-year analysis period (1992-2006), ten fatal crashes were reported within the project corridor, all at different locations. Additionally, this corridor has a higher than statewide average crash history.

i. Project Objectives

The overall project objective is to improve safety and operational characteristics of the roadway by improving roadway deficiencies to meet MDT and AASHTO (American Association of State Highway and Transportation Officials) standards to the greatest extent practicable. Listed below are specific project objectives.

- Improve pavement condition along the project corridor.
- Improve intersection geometry at key locations.
- Manage access in north Red Lodge.
- Accommodate pedestrian/bicycle users in Red Lodge.
- Improve LOS in rural segments and at two intersections in Red Lodge.
- Provide wider shoulders in rural segments.
- Flatten ditch inslopes in rural segments.
- Reduce encroachments within clear zones where appropriate.
- Increase ditch storage for snow/ice off the roadway in rural segments.
- Reduce frequency of animal-vehicle collisions along project corridor.
- Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts.

ii. Supporting Element

Public concerns have been raised throughout the project planning process regarding highwayrelated storm water drainage in Red Lodge and Roberts. Public concerns in Roberts were exacerbated due to flooding occurring in May 2005 and June 2007. As a result, improving highway-related storm water drainage through Red Lodge and Roberts has been identified as a supporting element of this project.

III. ALTERNATIVES

Analysis of the US Highway 212 project corridor resulted in the development of a no-build alternative (Alternative A) and a build alternative (Alternative B), which is the Preferred

Alternative. Because the project corridor is relatively long (approximately 21.2 miles [34.1 kilometers]), and the character and needs of the adjacent communities change along this length, the project corridor will be discussed as four segments: Red Lodge, Red Lodge to Roberts, Roberts, and Roberts to Boyd. See Figure 2-1, Segment/Intersection Locations, on page 2-3.

i. Alternative A: No-Build

Alternative A would leave the existing roadway in place as it exists today, with a continuation of current maintenance practices. This would not meet the project objectives. There would be no construction costs associated with Alternative A, aside from routine maintenance.

ii. Alternative B: Preferred Alternative

Alternative B would meet the project objectives. The Preferred Alternative would:

- Improve the pavement condition along the project corridor by reconstructing the roadway.
- Improve the intersection geometry at key locations along the corridor.
- Incorporate an Access Management Plan for Red Lodge, which was supported by the Red Lodge City Council in March 2007. See Appendix A, Letter #8.
- Provide pedestrian and bicycle facilities in Red Lodge.
- Improve LOS by providing passing lanes in rural segments where appropriate and making improvements at the 8th Street/US Highway 212 and MT Highway 78/US Highway 212 intersections.
- Provide wider roadway shoulders in rural segments where appropriate.
- Flatten ditch slopes in rural segments where appropriate.
- Reduce clear zone encroachments in rural segments where appropriate.
- Provide increased snow storage with wider and deeper roadside ditches in rural segments where appropriate.
- Clear thick brush and trees within the clear zone to improve driver visibility of approaching wildlife.
- Provide turning lanes where needed to reduce differential speed conflicts.
- Improve highway-related storm water drainage in Red Lodge and Roberts.

a. Red Lodge (8th Street to Two Mile Bridge Road)

The project corridor in Red Lodge consists of three distinct sub-segments: 8th Street to MT Highway 78, MT Highway 78 to developed limits of Red Lodge, and developed limits of Red Lodge to Two Mile Bridge Road.

Typical Sections

- Eighth Street to MT Highway 78. See Figure 2-3, 8th Street to MT Highway 78, on page 2-8.
 - 44-foot (13.2-meter) curb-to-curb urban section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 10-foot (3.0-meter) parking lanes
 - Two 5-foot (1.5-meter) sidewalks

- MT Highway 78 to the developed limits of Red Lodge. See Figure 2-5, MT Highway 78 to Developed Limits of Red Lodge, on page 2-10.
 - 49-foot (14.9-meter) curb-to-curb section
 - o Two 12-foot (3.6-meter) travel lanes
 - Two 5.5-foot (1.7-meter) shoulders
 - One 14-foot (4.2-meter) TWLTL (two-way left-turn lane)
 - One 5-foot (1.5-meter) sidewalk on the west side of the roadway
 - One 10-foot (3.0-meter) shared bike/ped path on the east side of the roadway
- Developed limits of Red Lodge to Two Mile Bridge Road. See Figure 2-6, Developed Limits of Red Lodge to Two Mile Bridge Road, on page 2-11.
 - o 55-foot (15.8-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - One 13-foot (3.9-meter) raised median
 - Two 5.5-foot (1.7-meter) shoulders
 - Two 3.5-foot (1.05-meter) median shoulders
 - One 10-foot (3.0-meter) shared bike/ped path on the east side of the roadway

Drainage

The urban-developed portion of Red Lodge within the project limits extends from 8th Street to an area approximately 1,500 feet (460 meters) north of MT Highway 78. Within this area, a curb and gutter section and storm water conveyance system (such as a storm drain, trunk line, and/or open ditch) are proposed to accommodate highway-related storm drainage. North of the developed limits of Red Lodge, the typical section of the roadway is anticipated to change from an urban section to a more rural section, which would likely include roadside ditches. If a suitable location for a storm water conveyance system outfall is not identified before the start of the rural typical section, then the storm water conveyance system may discharge into the roadside ditches and flow north to an outfall location to Rock Creek.

Intersections

The proposed project would realign Oakes Avenue to directly oppose 7th Street and convert it to a southbound one-way street. Diagonal parking would be provided on both sides of the street. Additionally, a bulbed out curb line would be constructed along US Highway 212 at the Carnegie Library to improve sight distance at 8th Street. A traffic signal would be installed when warranted and justified; it is anticipated that signal warrants may be met by 2020. See Figure 2-7, Oakes Avenue/8th Street Intersections, on page 2-13.

The proposed project would also replace the MT Highway 78 and US Highway 212 intersection with a single lane roundabout. In addition, the intersection of Villard Avenue and MT Highway 78 would be closed. A cul-de-sac would be constructed at Villard Avenue just south of MT Highway 78. Fourth Street would be extended along the south side of the Visitor Center between US Highway 212 and Villard Avenue, replacing the Visitor Center's south access. *See Figure 2-8, MT Highway 78/Villard Avenue Intersections, on page 2-15.*

Access Management

At the request of the City of Red Lodge, MDT and the City of Red Lodge developed an Access Management Plan for the area between MT Highway 78 and Two Mile Bridge Road. The purpose of the Access Management Plan is to provide a means for MDT and the City to balance

the need for vehicular progression along US Highway 212 with the need for access to adjacent properties.

A shared TWLTL is proposed for the developed area immediately north of MT Highway 78 to accommodate turning vehicles accessing adjacent properties while improving progression for through traffic. Future access types and locations have been identified for the developing area south of Two Mile Bridge Road. See Figure 2-9, Developed and Developing Limits of Red Lodge, on page 2-16. The City of Red Lodge supported the Access Management Plan with Resolution 3228. Pursuant to applicable Montana statutes and MDT policy, the plan would be recommended to the Montana Transportation Commission for their adoption. See Figure 2-10, Access Management Plan Overview, on page 2-17 and Appendix A, Letter #8.

b. Red Lodge to Roberts

The Preferred Alternative includes one typical section and drainage improvements for US Highway 212 between Red Lodge and Roberts.

Typical Section

- Two Mile Bridge Road to the south end of Roberts. See Figure 2-15, Two Mile Bridge to the south end of Roberts, on page 2-21.
 - 40-foot (12.0-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 8-foot (2.4-meter) shoulders
 - A bus turnaround approximately one-mile (1.6 kilometers) north of Fox Road at the present state maintenance site
 - A one-mile (1.6-kilometer) northbound passing lane north of Two Mile Bridge Road

Drainage

Some culverts between Red Lodge and Roberts carry water generally from the west side of the roadway to the east side of the roadway. The existing drainage patterns generally parallel the roadway northward. Drainage patterns and culvert locations are expected to remain the same with implementation of the proposed project. Numerous irrigation ditches are currently located within the right-of-way limits of the proposed project. The proposed project may involve relocating those irrigation ditches outside of the proposed right-of-way, in accordance with MDT's general practice.

c. Roberts

The Preferred Alternative within Roberts includes one primary typical section, drainage improvements, intersection improvements, and safety improvements for pedestrians.

Typical Section

- South end of Roberts to East Maple Street. See Figure 2-17, South End of Roberts to East Maple Street, on page 2-23.
 - 46-foot (13.8-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 4-foot (1.2-meter) shoulders

• One 14-foot (4.2 meter) TWLTL

A modification of the typical section was developed for the sub-segment between East Maple Street and the north end of Roberts to avoid impacts to Roberts School and to meet the request of Roberts School for a guardrail along school property.

- East Maple Street to the north end of Roberts. See Figure 2-18, East Maple Street to North End of Roberts, on page 2-24.
 - 46-foot (13.8-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 4-foot (1.3-meter) shoulders
 - o One 14-foot (4.2-meter) TWLTL
 - A 2:1 inslope with guardrail on the east side of the roadway

Drainage

A large drainage area lies west of the roadway. Within this area, flood irrigation operations contribute additional water to the drainage. The drainage parallels the roadway from south of the Fox Road and drains northward along the west edge of Roberts. Because the portion of Roberts west of the roadway sits in one of the swales of the drainage, the potential exists to flood homes in Roberts, as occurred in 2005 and 2007. It is beyond the scope of this project to address floodwaters entering Roberts from the drainage to the west. Goals of the proposed project would be to convey as much highway-related storm water as practicable toward Rock Creek prior to Roberts, and within Roberts to satisfactorily convey storm water intercepted by the highway towards Rock Creek.

Within the community of Roberts, it is anticipated that runoff from the roadway would be directed to open ditches and/or pipes. The ditches and/or pipes would carry the runoff to roadside ditches located north of Roberts.

Intersection

The Preferred Alternative would realign Cooney Dam Road to form a single intersection perpendicular with US Highway 212 approximately 430 feet (130 meters) north of East Maple Street. See Figure 2-19, Cooney Dam Road Intersection (near East Maple Street), on page 2-26.

Pedestrian Facilities

The Preferred Alternative would include one new block of sidewalk along Pine Street from US Highway 212 east to connect to the existing sidewalk on First Street. Crosswalks would also be provided at the intersections of Oak, Cedar, Pine, and East Maple Streets along US Highway 212. School advance warning and school crosswalk warning signs, including flashing beacons, would be installed in accordance with current design guidelines.

d. Roberts to Boyd

For the area from Roberts to Boyd, the Preferred Alternative includes one primary typical section, drainage improvements, and intersection improvements at three locations.

Typical Section

- North end of Roberts to Boyd. See Figure 2-21, North End of Roberts to Boyd, on page 2-29.
 - 40-foot (12.0-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 8-foot (2.4-meter) shoulders
 - A bus turnaround approximately 3 miles (4.8 kilometers) south of Boyd at the school district boundary
 - A one-mile (1.6-kilometer) northbound passing lane immediately north of Roberts
 - A one-mile (1.6-kilometer) southbound passing lane south of Boyd

The exception of this typical section would be at the Boyd Country Store where measures were taken to avoid the store and improve safety conditions.

- Boyd Country Store. See Figure 2-22, North End of Roberts to Boyd Boyd Country Store, on page 2-30.
 - o 52-foot (15.6-meter) rural section
 - Two 12-foot (3.6-meter) travel lanes
 - Two 8-foot (2.4-meter) shoulders
 - A 12-foot (3.6 meter) southbound left-turn lane
 - A reverse curb to delineate the Boyd Country Store parking lot from the roadway and to manage access to US Highway 212.

Drainage

Existing drainage patterns and culvert locations are expected to remain the same in the design of the proposed project. Several irrigation ditches that are currently located within the right-ofway limits for the proposed project are expected to be relocated outside of the new right-of-way in accordance with MDT's general practice.

Intersections

The Preferred Alternative would realign Clear Creek Road with the south access of the rest area. Improvements to the rest area sidewalks and ramps may be included as part of the proposed project. See Figure 2-23, Clear Creek Road Intersection, on page 2-32.

The Preferred Alternative would also close the northern fork of Cooney Dam Road and add a southbound right-turn lane on US Highway 212 to Cooney Dam Road. See Figure 2-24, Cooney Dam Road and Main Street Intersections, on page 2-33.

Additionally, the Preferred Alternative would realign Main Street to be perpendicular with US Highway 212. A southbound left-turn lane would be added on US Highway 212 to provide protection for slowed or stopped traffic accessing Main Street. A reversed curb line would be constructed between the Boyd Country Store parking lot and US Highway 212 to delineate access. Access to the Boyd Country Store would be provided off of Main Street. *See Figure 2-24, Cooney Dam Road and Main Street Intersections, on page 2-33.*

IV. AFFECTED ENVIRONMENT, IMPACTS, & MITIGATION

The following table provides a summary of environmental impacts and mitigation measures associated with each alternative. See Table A, Summary Comparison of Project Alternatives and Impacts.

Table A Summary Comparison of Project Alternatives and Impacts				
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
	Land Use	No impact.	Acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way would create direct land use impacts; overall land uses in the area would not be affected.	No mitigation required.
	Farmland	No Impact.	Impacts to approximately 275.8 acres (111.6 hectares) of farmland; of which 89.6 acres (36.3 hectares) are prime farmland and 48.9 acres (19.8 hectares) are of statewide importance.	No mitigation required.
	Traffic	Inadequate LOS currently at northbound lane between Roberts and Boyd; southbound lane between Red Lodge and Roberts by 2010; and MT Highway 78 and 8 th Street by 2030.	All segments and key intersections expected to operate at acceptable levels through 2030.	No mitigation required.
Transportation System	Access	Inconsistent with Red Lodge Council Resolution No. 3228.	The ability of the roadway to provide for both access and progression would be improved. Access would be managed between MT Highway 78 and Two Mile Bridge Road. Consistent with Red Lodge Council Resolution No. 3228.	No mitigation required.
	Safety	The number of crashes and existing crash rates are anticipated to increase as traffic continues to increase.	Would include the following safety improvements: intersection realignments, addition of turn lanes where needed, access management in Red Lodge, wider shoulders, flatter ditch slopes, clearing of thick brush and trees within the clear zone, ped/bike facilities in Red Lodge and Roberts, bus turnarounds.	No mitigation required.
	Pedestrian/Bicycle Facilities	Inconsistent with Red Lodge Comprehensive Trails Plan.	Would provide sidewalks and crosswalks, where appropriate, in Red Lodge; a shared bike/ped path between MT Highway 78 and Two Mile Bridge Road; and crosswalks and one-block of sidewalk in Roberts. Consistent with Red Lodge Comprehensive Trails Plan.	No mitigation required.
				Continued

		Summary Com	Table A parison of Project Alternatives and Impacts	
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
	Noise	No impact.	By 2030, five residential properties would experience noise levels at or slightly above the noise abatement criteria.	At this time, noise mitigating measures are not considered reasonable and feasible.
R	ight-of-Way and Relocations	No impact.	Acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way; potential acquisition and/or relocation of up to nine structures (six dwellings, two out-buildings, and one commercial building).	Compliance with Uniform Act. MDT will also attempt to meet individually with affected property owners. Reasonable efforts to avoid and/or minimize impacts will be made.
	Surface Water	No new impacts.	Impacts may result from culvert replacement or extension; ditch realignment; dredge/fill activities in wetlands; the relocation of irrigation ditches outside of the proposed right-of-way; new storm water outfall locations at Rock Creek; and conflict between existing storm drain and new storm drain near the intersection of MT Highway 78 and US Highway 212.	Use of BMPs; compliance with applicable permits, and local, state, and federal regulations.
Water Resources/Quality	Irrigation Facilities	No new impacts.	Would relocate irrigation ditches, as necessary, in consultation with owners to minimize impacts and may impact Mullaney Spring.	Care to avoid altering flow rate to water rights holder of Mullaney Spring. Consultation with affected ditch associations and other landowners/water rights holders to minimize impacts to irrigation facilities.
	Ground Water	No new impacts.	Impacts to ground water resources are not anticipated. The Preferred Alternative may require relocation of domestic wells within the proposed right-of-way.	If domestic wells are displaced, domestic water would be restored to the affected properties.
	Public Water Systems	No new impacts.	No new impacts anticipated.	No mitigation required.
	Waste Water Systems	No new impacts.	Potential relocation of an identified mound septic system in the proposed right-of-way and construction limits. A new storm drain pipe and outfall may need to be constructed or the existing storm drain replaced.	If the mound system were impacted, MDT would relocate the system per County and MDEQ requirements.

Continued...

	Table A Summary Comparison of Project Alternatives and Impacts				
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation	
Water Body Modifications		No new impacts.	New culvert installation at Stanley Creek and minor inlet and outlet ditches may be required; irrigation ditches would be relocated outside of right-of-way.	Structures would be designed to minimize disruption to hydrology and to comply with applicable federal and state regulations.	
Wetlands		No new impacts.	Impacts to approximately 40.7 acres (16.5 hectares) of wetlands; of which 24.8 acres (9.7 hectares) are jurisdictional based on preliminary jurisdictional determinations.	Unavoidable impacts would be mitigated according to permit requirements at an approved mitigation site.	
	Vegetation	No new impacts.	Removal of vegetation in select areas for proposed improvements. Clearing of ground cover along the corridor has the potential to open areas to noxious weeds.	Compliance with MDT Standard Specifications.	
nd Ecosystems	Terrestrial and Avian Species	No new impacts.	May result in minor fragmentation, modification, and/or loss of habitat for terrestrial and avian species.	Use of BMPs; implementation of erosion and sediment control plan; compliance with Migratory Bird Treaty Act and MDT's most current depredation permit from the USFWS.	
Habitat a	Aquatic Species	No new impacts.	May result in minor impacts due to water body modifications. No substantive losses of spawning fish species are anticipated.	Use of BMPs; compliance with applicable permits and federal and state regulations.	
Wildlife	Montana Species of Concern	No new impacts.	Impacts are not anticipated; however, the gray wolf would be subject to the same impacts as other terrestrial species.	No mitigation required.	
	Animal-Vehicle Collisions	High frequency of animal- vehicle collisions would continue.	May improve driver visibility of approach wildlife by removing thick brush and vegetation from the clear zone.	No mitigation required.	

	Table A Summary Comparison of Project Alternatives and Impacts				
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation	
	Floodplain	No new impacts.	No new impacts anticipated. Coordination with the county floodplain administrator would occur to determine whether minor encroachment of the floodplain would occur and whether a floodplain development permit is required.	No mitigation required.	
T Enc	Threatened and dangered Species	No new impacts.	No effect.	No mitigation required.	
Cu	ultural Resources	No impact.	SHPO concurrence with finding of <i>No Effect</i> or <i>No Adverse Effect</i> to all historic properties.	No further avoidance/mitigation measures required.	
4(f) es	Historic Properties	No impact.	<i>De Minimis</i> Section 4(f) impacts to four historic resources; no additional Section 4(f) use.	No further avoidance/mitigation	
ction 4 operti	Historic Irrigation Ditches	No impact.	<i>De Minimis</i> Section 4(f) impacts to ten historic irrigation ditches; no additional Section 4(f) use.	Minimis Evaluations.	
Se Pr	Recreation Areas	No impact.	No impact.		
Sec	tion 6(f) Properties	No impact.	No Section 6(f) properties would be converted to a transportation use. Directional and entrance signs that may be removed would be reinstalled following construction. If impacted, the entrance road for the Water Birch fishing access site would be returned to existing or improved condition following construction.	No mitigation required.	
Hazardous Materials, Solid Waste, and Underground Storage Tanks		No impact.	No impacts anticipated.	If hazardous materials are discovered, generated, or used they would be stored, handled, and disposed of in accordance with applicable local, State, and Federal laws.	
V.	/isual/Aesthetic Considerations	No impact.	Improved aesthetics in Red Lodge; four roundabouts as desired in Resolution No. 3228; incorporation of elements of the Red Lodge Streetscape Plan.	No mitigation required.	

	Table A Summary Comparison of Project Alternatives and Impacts				
	Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation	
	Air Quality	No impact.	Temporary increase of fugitive dust and mobile source emissions.	Compliance with standard MDT procedures and applicable permit requirements.	
IS	Transportation System	No impact.	Temporary impacts to local and regional traffic circulation in the project area due to lane closures, delays, temporary travel on unpaved surfaces, and reduced travel speeds.	Development of construction traffic control plan according to MDT Standard Specifications.	
iction Consideration	Water Resources/Quality	No impact.	Short-term increase in turbidity, potential for erosion, and storm water runoff.	Use of BMPs; compliance with applicable federal and state regulations. Temporary impacts to wetlands would be restored to original contours and re- vegetated at the earliest practicable date following construction.	
Temporary Constru	Wildlife Habitat and Ecosystems	No impact.	May result in temporary minor disturbances to wildlife communities.	Between Sept. 1 and Apr. 30, vacated swallow or other songbird nests would be physically removed and deterrents would be placed on existing structures. Disturbed areas would be reseeded with desirable seed mix.	
	Noise	No impact.	Temporary increase in noise levels within the vicinity of the project.	Compliance with MDT Standard Specifications. As necessary, the contract will include additional requirements for projects located in or near urban areas.	
	Continued				

Table A Summary Comparison of Project Alternatives and Impacts			
Resource	Alternative A: No-Build	Alternative B: Preferred	Proposed Mitigation
Utilities	No impact.	Some relocation of overhead and underground power lines and underground telephone lines may be required.	Potential impacts would be coordinated with the appropriate utility companies. Rural overhead power lines that are relocated would be raptor proofed per MDT policy.
V. PERMITS AND AUTHORIZATIONS

The following permits and authorizations are likely to be required prior to construction activities:

- CWA (Clean Water Act) Section 402/MPDES (Montana Pollutant Discharge Elimination System) authorization from MDEQ (Montana Department of Environmental Quality) Permitting and Compliance Division. The MPDES permit requires a storm water pollution prevention plan that includes a temporary erosion and sediment control plan. The erosion and sediment control plan identifies BMPs, as well as site-specific measures to minimize erosion and prevent eroded sediment from leaving the work zone.
- CWA Section 404 permit from the USACE (United States Army Corps of Engineers) for any activities that may result in the discharge or placement of dredged or fill materials in waters of the US, including wetlands.
- Montana Stream Protection Act (SPA 124) from the MT FWP (Montana Fish, Wildlife & Parks)-Fisheries Division. The Montana SPA 124 is required for projects that may affect the bed or banks of any stream in Montana.
- Short-Term Water Quality Standard for Turbidity related to construction activity (318 Authorization) from the MDEQ-Water Quality Bureau for any activities that may cause unavoidable violations of state surface water quality standards for turbidity, total dissolved solids, or temperature.
- Floodplain Development Permit from the County Floodplain Administrator.

VI. COORDINATION

The following discussion briefly details coordination efforts made with cooperating agencies, other interested parties, and the public. Coordination efforts are detailed further in Chapter 5 of this document.

i. Cooperating Agencies

The City of Red Lodge and Carbon County are both cooperating agencies for this project. MDT has coordinated with Red Lodge and Carbon County throughout the development of the project.

Meetings with the City of Red Lodge

- March 28, 2003 Alignment and Grade review consisting of an overall project review.
- February 25, 2005 Preliminary design concepts discussion; City Administrator express interest in roundabout concept at MT Highway 78 intersection.
- June 28, 2006 Updated the City Council on alternatives under consideration in Red Lodge.
- August 10, 2006 Meeting with Red Lodge City Planner and Red Lodge Public Works Director to discuss Oakes Avenue, access management, and land use development.
- February 27, 2007 Presentation of the Access Management Plan to the City Council requesting approval of said plan.
- June 27, 2008 Meeting with the City of Red Lodge to discuss the City's comments on the Administrative Draft EA.

Red Lodge Resolutions

 October 10, 2006 – City of Red Lodge passed Resolution No. 3223 which identified support for the preferred alternative within Red Lodge, with the exception of a TWLTL between the developed limits of Red Lodge and Two Mile Bridge Road. See Appendix A, Cooperating Agencies.

 March 27, 2007 – City of Red Lodge passed Resolution No. 3228 which identified support for the preferred alternative within Red Lodge. See Appendix A, Cooperating Agencies.

Meetings with Carbon County

- March 1, 2007 Informational meeting held to talk about proposed improvements in Boyd.
- March 29, 2007 Follow-up to the March 1 meeting is held to further discuss the Dakota Avenue intersection in Boyd.

ii. Coordination with Other Agencies, Organizations, and Interested Parties

A scoping package was distributed to federal, state, and local agencies and other interested parties in March 2002. Due to modifications in the proposed project and the passage of time, agencies and interested parties were solicited again in May 2007. A total of 25 agency comments were received regarding the proposed project. These comments provided valuable information used to identify potential environmental impacts.

iii. Coordination with the Public

- March 27, 2002 Kickoff Meeting held to inform elected officials and local, State, Federal, and regional agencies, as well as the public, of the project and to obtain local knowledge of concerns related to the proposed study. This meeting also served as an early notification of the preparation of an EA.
- November 6, 2002 Alternatives Public Workshop held to inform the public of alternatives being considered for the project and to obtain public input.
- May 16, 2006 Public informational meeting held to discuss improvements to highwayrelated storm water drainage along the project corridor in Roberts.
- November 14, 2006 A property owner meeting was held to discuss current and future access needs from MT Highway 78 to Two Mile Bridge Road.
- January 17, 2007 A second access management meeting was held to present potential alternatives for access management along MT Highway 78 to Two Mile Bridge Road and to solicit public feedback.
- January 31, 2007 A meeting was held with the architect working with the City of Red Lodge on the new Bank of Red Lodge. The purpose of the meeting was to discuss when the US Highway 212 project would be constructed, the amount of right-of-way required for the proposed project, and possible layouts for the bank site plan.
- February 16, 2007 A meeting was held with the owners and representatives of the proposed Beartooth Hospital in an effort to coordinate the proposed US Highway 212 project and right-of-way requirements with the future site plan of the Beartooth Hospital.
- December 18, 2007 A public informational meeting was held to discuss improvements to highway-related storm water drainage along the project corridor in Roberts.

iv. Public Hearing

Two Public Hearings are planned for this project, one in Red Lodge and the other in Roberts. A Notice of Availability of the EA and Public Hearing dates will be advertised following the approval of this document.

v. Conclusion

There are no areas of controversy, substantive issues raised, or issues remaining to be resolved that resulted from coordination with cooperating agencies, other interested parties, and the public.

Chapter 1 Purpose of and Need for Action

1.1 INTRODUCTION

The MDT (Montana Department of Transportation) plans to reconstruct approximately 21.2 miles (34.1 kilometers) of US Highway 212 in Carbon County. The proposed project begins at 8th Street in Red Lodge and continues generally north-northeast to approximately 1,000 feet (305 meters) north of Boyd.

The south logical terminus in Red Lodge was originally the intersection of MT Highway 78 (3rd Street). However, transitions of intersection and highway improvements would place the end of construction one to two blocks beyond the intersection of MT Highway 78, and so the southern terminus was extended to 5th Street. Following the initial public input meeting, the City of Red Lodge requested MDT extend the project terminus to 8th Street, where there is an abrupt change in the roadside design and character, from urban fringe development without sidewalks and continuous curb and gutter, to downtown commercial development with full sidewalks and curb and gutter. Ending the highway improvements three blocks north of 8th Street (5th Street) would leave a short stretch of US Highway 212 without pedestrian and drainage improvements; therefore the logical terminus was extended to 8th Street. It is anticipated that improvements at the 8th Street intersection.

The north logical terminus at Boyd was selected to tie into and match an existing section of US Highway 212, which has widened shoulders, and avoid leaving a narrow section of roadway for a future project.

The existing facility is a rural two-lane roadway with 12-foot (3.6-meter) driving lanes and approximately 2-foot (0.6-meter) shoulders. The proposed project corridor extends through a portion of the city of Red Lodge and through the three unincorporated communities of Fox, Roberts, and Boyd. The urban portion of the proposed project is through Red Lodge, with variable roadway widths and no turning lanes. Within the corridor, there are no signalized intersections and traffic control typically consists of two-way stop control on intersecting roads. *See Figure 1-1, Project Location Map.*

US Highway 212, in the project area, is functionally classified as a rural minor arterial and is on Montana's Primary Highway System. It serves local, commuter, tourist and agricultural traffic and is considered an integral part of the regional transportation network. The corridor provides transportation for a variety of users, from wide agricultural vehicles to bicyclists and pedestrians who frequent the area between 8th Street and Two Mile Bridge Road in Red Lodge. The highway is a major travel route used by residents commuting between Red Lodge and Laurel or Billings. It also serves tourist traffic as a through route to Yellowstone National Park, the Beartooth Highway, Red Lodge Mountain Ski Area, and other recreational areas. In addition, US Highway 212 connects a number of local roads to the regional transportation network.



Figure 1-1, Project Location Map

The proposed reconstruction of US Highway 212 is being administered by the MDT under a project designated as Corridor Study-Red Lodge North (Project Number STPP 28-2(25)70, Control Number 4375). MDT estimates \$21.3 million will be available for projects on this corridor from the SAFETEA-LU Section 1934 Transportation Improvement Project #246 to Develop and Construct US 212 Red Lodge North. At this time, the top priorities are the city of Red Lodge and the community of Roberts. The other segments (between Red Lodge and Roberts as well as Roberts and Boyd) of the project would be constructed as funding becomes available.

1.2 PURPOSE AND NEED FOR THE PROJECT

The purpose of the proposed project is to improve the safety and operational characteristics of the roadway.

The proposed project is needed because there are numerous roadway deficiencies along the corridor which create safety and operational concerns. These include aging pavement; intersection deficiencies; a need for access management, particularly in Red Lodge; a lack of pedestrian and bicycle facilities in Red Lodge; a need for improved LOS (level of service) in rural segments and at two intersections in Red Lodge; and narrow roadway shoulders and steep ditch slopes in rural segments. This corridor also has a higher than statewide average crash history.

Additionally, the City of Red Lodge is experiencing continued population growth and development pressure. This project stems in part from a request by the City to work cooperatively to plan for and accommodate future development in Red Lodge along the US Highway 212 project corridor.

Furthermore, the community of Roberts has voiced concerns of storm water drainage problems along US Highway 212. These concerns surfaced following the May 2005 flooding in Carbon County and have been perpetuated as a result of additional flooding in June 2007. The proposed project would help address these concerns by improving highway-related storm water drainage near Roberts.

1.2.1 Need to Improve Safety and Operational Characteristics of the Roadway

US Highway 212 within the project corridor was originally constructed in 1921. Since that time numerous improvements have been completed, the most recent being the Red Lodge North project, administered under Project Number STPP 28-2(22)70, which was completed in June 2002 and included pavement preservation construction and replacement of the Rock Creek Bridge north of Roberts. The roadway has numerous deficiencies, which are described below.

Pavement

The pavement on US Highway 212 is nearing the end of its serviceability, and a stronger pavement section is required to serve the next 20 years. The pavement was overlaid in 1984–85. A typical design life for asphalt pavement is approximately 20 years. The Red Lodge North project completed in 2002 included a mill and overlay from the north end of this project south for 2 miles (3.2 kilometers), replacement of the Rock Creek Bridge north of Roberts as well as new bridge approach sections for 1/8-mile (0.2 kilometers) north and south of the bridge, and a seal coat on the remaining 19 miles (30.6 kilometers) of the project corridor. Nineteen of the 21 miles (30.6 of the 33.8 kilometers) have exceeded the design life for asphalt pavement.



Figure 1-2, Conceptual Intersection Angles

Intersections

Numerous intersections along the project corridor have geometric deficiencies, meaning that they are either skewed or offset.

A skewed intersection is one that has an angle of less than 60 degrees. Generally, intersecting roads should meet at right angles (90 degrees) wherever practical and should not intersect at an angle less than 60 degrees. A skew of no less than 75 degrees is preferable to accommodate older drivers. Skewed intersections need extensive turning roadway areas and tend to limit motorist visibility of other vehicles, particularly for drivers of trucks. Skewed intersections also increase the exposure time for vehicles crossing the main traffic flow. Figure 1-2, Conceptual See Intersection Angles.

An offset intersection is one where two opposing streets are not directly lined up with each other. Closely spaced offset intersections are undesirable. Offset intersections in close proximity to each other create additional motor vehicle conflict points for a motorist to consider while executing a crossing or merge maneuver; this may create greater potential for collisions. Offset intersections can also increase the potential for collisions for two opposing vehicles that simultaneously attempt a left turn. See Figure 1-3, Conceptual Offset Intersection Overview.

In addition, two intersections within Red Lodge do not have adequate capacity to accommodate anticipated future traffic volumes.



Figure 1-3, Conceptual Offset Intersection Overview

Access

US Highway 212 accommodates both through traffic and local traffic; therefore, it is important to provide for progression as well as access to adjacent properties. Overall operation of rural minor arterials (such as the project corridor) is not typically governed by capacity or intersection operations; such roadways usually provide free flow under all conditions. However, the conditions along the project corridor, particularly in northern Red Lodge, are continually changing due to population growth and development. The City of Red Lodge identified the need for an access management plan for US Highway 212, between MT Highway 78 and Two Mile Bridge Road, to provide a balance between the need for progression of through traffic and the need to access existing and planned developments adjacent to the corridor.

Pedestrian/Bicycle Facilities

There are discontinuous sidewalks and no bicyclist accommodations along the project corridor. The City of Red Lodge prepared a trails plan in 2006¹, which outlined an improved system of trails for non-motorized transportation in and around the City. The plan also identified the need for sidewalk and shared bike/ped path facilities along US Highway 212 within Red Lodge.

Level of Service

A traffic operations analysis was conducted for the project corridor to determine the LOS for the project segments and key intersections. Based on increased traffic, the analysis identified that the existing two-lane roadway does not provide an adequate LOS along the rural segments. As mentioned previously, two intersections in Red Lodge would also provide inadequate LOS by the project design year. Adequate LOS for rural segments would be LOS B, while urban segments would be LOS C.

Shoulders

US Highway 212 is a rural two-lane roadway that has 12-foot (3.6-meter) driving lanes and approximately 2-foot (0.6-meter) shoulders, with a total roadway width of 28 feet (8.4 meters); immediately north of the project limits the shoulders widen to 8 feet (2.4 meters). Comments received throughout the public input process indicated support for wider shoulders. Wider shoulders are more desirable because they provide recovery room for errant vehicles, space for emergency parking outside the travel lanes, and more comfortably accommodate large agricultural equipment.



What is LOS (Level of Service)?

¹ Beck, Barb, Tom Kohley, and Allie Wood, *The City of Red Lodge Comprehensive Trails Plan* (May 2006).

Ditch Slopes

Many of the roadside ditches along the project corridor have steep slopes. A 3:1 or steeper inslope (the ditch slope closest to the roadway) is considered a "nontraversable" slope in that an errant vehicle could roll over if the driver attempted to turn back onto the roadway. A flatter inslope, such as a 6:1 inslope, is considered more desirable for a roadway such as US Highway 212 in that an errant vehicle has a greater chance of recovery. Where practicable, 6:1 slopes would be used. See Figure 1-4, Conceptual Ditch Inslope Illustration.

What does 3:1 mean?

Ditch slopes are measured using a ratio of horizontal distance compared to vertical distance. For example, a 3:1 ratio indicates that there are three feet of horizontal distance for every one-foot of vertical distance. As the first number gets larger (the horizontal distance increases), the slope becomes flatter and more recoverable.



Figure 1-4, Conceptual Ditch Inslope Illustration

Clear Zones

The clear zone is an area adjacent to the roadway that is kept free of obstacles to prevent impact by an errant vehicle. Examples of obstacles or encroachments that compromise the clear zone are large culverts, trees, non-approved mailboxes and signs, utility poles, steep side-slopes on approaches, irrigation facilities and streams. All of these encroachments are present to varying degrees on US Highway 212. The width of the clear zone is measured from the outside edge of the driving lane and varies with traffic volume, design speed and the slope off of the roadway. Therefore, clear zone width requirements vary along the project corridor. See Figure 1-5, Conceptual Clear Zone Illustration.

	CLEAR ZONE	TRAVEL LANE	TRAVEL LANE	CLEAR ZONE
	VARIABLE			VARIABLE
VARIABLE	UNOBSTRUCTED CLEAR ZON	E	OBST	RUCTED CLEAR ZONE

Figure 1-5, Conceptual Clear Zone Illustration

Crash Rate

A crash analysis for this project was performed for the dates January 1, 1992 through December 31, 2001 and for the dates January 1, 1992 through December 31, 2006². See Table 1.1, Collision Summary (1992–2006).

Table 1.1 Crash Summary (1992–2006)					
Crash Type	US 212 Study Area	Statewide Rural Primary Average			
Clear Conditions	65%	55%			
Nighttime (dark conditions)	42%	31%			
Wildlife Related	37%	14%			

There were 10 fatal crashes during the 15-year analysis period (1992–2006), all at different locations. Seven crash cluster locations were identified along the project corridor. The crash analysis indicated contributing factors to traffic crashes along the corridor, as described below.

- Snow/ice build-up on roadway many of the roadside ditches along the project corridor are narrow and shallow, and they annually become filled with snow. This contributes to the formation of compacted snow and ice on the highway. Inclement weather, including icy or slushy road conditions, was identified as a contributing factor in nearly 10 percent of the total recorded crashes.
- *High frequency of animal-vehicle collisions* Animal-vehicle collisions within the project corridor are close to three times the statewide average and comprise over 54

² Information and analysis are as reported in an April 15, 2002 memorandum, May 24, 2002 memorandum, and July 27, 2007 engineering study evaluation.

percent of the nighttime collisions. The combination of thick brush and trees within the clear zone and narrow roadside ditches with steep inslopes obscures driver visibility of approaching animals and contributes to animal-vehicle collisions.

• *Differential speed conflicts*³ - There are a large number of turning vehicles in the segments through Red Lodge and Roberts, which presents differential speed conflicts with through traffic and slower turning traffic.

1.2.2 Supporting Element

Public concerns have been raised throughout the project planning process regarding highwayrelated storm water drainage in Red Lodge and Roberts. Public concerns in Roberts were exacerbated due to flooding occurring in May 2005 and June 2007. As a result, improving highway-related storm water drainage through Red Lodge and Roberts has been identified as a supporting element of this project.

1.3 PROJECT OBJECTIVES

The overall project objective is to improve safety and operational characteristics of the roadway by improving roadway deficiencies to meet MDT and AASHTO (American Association of State Highway and Transportation Officials) standards to greatest extent practicable. Listed below are specific project objectives.

Improve the Safety and Operational Characteristics of the Roadway

- Improve pavement condition along project corridor.
- Improve intersection geometry at key locations.
- Improve access in north Red Lodge.
- Accommodate pedestrian/bicycle users in Red Lodge.
- Improve LOS in rural segments and at two intersections in Red Lodge.
- Provide wider shoulders in rural segments.
- Flatten ditch inslopes in rural segments.
- Reduce encroachments within clear zones where appropriate.
- Increase ditch storage for snow/ice off the roadway in rural segments.
- Reduce frequency of animal-vehicle collisions along project corridor.
- Reduce differential speed conflicts with turning vehicles in Red Lodge and Roberts.

Supporting Element

• Improve highway-related drainage through Red Lodge and Roberts

³ A "differential speed conflict" is when a vehicle is slowing down, or moving slower, than a vehicle behind them. If the faster vehicle doesn't notice that the car in front is moving slower or slowing down, there could be a rear end accident.

1.4 HISTORY OF THE PLANNING & SCOPING PROCESS

The proposed project is the result of a cooperative planning effort by the FHWA (Federal Highway Administration), MDT, the City of Red Lodge, and Carbon County. Coordination with the City of Red Lodge and the Carbon County Commissioners has been ongoing. Numerous meetings have been held with the City of Red Lodge to discuss the proposed project, including typical section, pedestrian and bicycle facilities, parking, intersection concerns, and access management. In addition, two meetings were held with the Carbon County Commissioners to discuss improvements through the community of Boyd⁴.

In addition to cooperating agencies, an effort was made to solicit views from agencies and other interested parties to ensure that social, economic, and environmental impacts were considered in the development of the EA. Scoping packages were sent in March 2002 and May 2007 to numerous local, State, and Federal agencies to determine the consistency of the project with current and proposed plans, programs, and policies. These coordination efforts have occurred throughout the development of the EA via consultation with agencies having jurisdiction over potentially impacted resources.

Further, throughout the development of this project, efforts were also made to reach out to the public and incorporate public input into plans to improve the highway. Five public meetings were held to discuss items such as the purpose and need of the proposed project; proposed improvement alternatives; storm drainage issues in the community of Roberts; and access management within Red Lodge. In addition, coordination meetings were held with representatives of the proposed Bank of Red Lodge and the proposed Beartooth Hospital in Red Lodge, both of which are planned adjacent to the project corridor.

1.5 RELEVANT PLANS

Both the City of Red Lodge and Carbon County have developed Growth Policies, which are intended to guide development, growth patterns, and land use decisions in the area. The Red Lodge Growth Policy was adopted in May 2001. The Carbon County Growth Policy was adopted in September 2003.

Using the Red Lodge Growth Policy as guidance, the City of Red Lodge adopted the City of Red Lodge Comprehensive Trails Plan in May 2006. This Plan was adopted to satisfy one of the goals of the Growth Policy, which was to develop a trail system linking parks, residential areas, and open space areas. Recommendations for pedestrian and bicycle accommodations along US Highway 212 in Red Lodge have been incorporated into the proposed project where practicable and appropriate.

⁴ Additional information about the coordination process can be found in Chapter 6 of this document.

Chapter 2 Alternatives

2.1 INTRODUCTION

This chapter provides information on the development and evaluation of project alternatives. Following the identification of the purpose and need for the proposed project and the project objectives, numerous improvement scenarios were developed and evaluated. These scenarios were modified and refined based on input from elected officials and the general public as well as data collected pertaining to engineering factors, environmental considerations, and existing and planned development along the corridor. The result of this process is presented in this chapter. Two alternatives are under consideration for this project: a no-build alternative and a preferred alternative.

2.2 ALTERNATIVE A: NO-BUILD

Alternative A is the no-build alternative. A no-build alternative is included in this type of analysis to provide a baseline condition against which other alternatives are evaluated. The no-build alternative is used to describe the existing conditions and anticipate what would happen if no improvements were made. It is important for the public and the decision-makers to understand whether the specific problems associated with the roadway are likely to improve on their own, remain stagnant, or worsen without specific actions to correct such problems.

Alternative A would leave the existing roadway in place as it exists today, with a continuation of current maintenance practices. This would not meet the project objectives. There would be no construction costs associated with Alternative A, aside from routine maintenance.

2.3 ALTERNATIVE B: PREFERRED ALTERNATIVE

Alternative B, hereafter referred to as the Preferred Alternative, would meet the project objectives outlined in the purpose and need chapter. The Preferred Alternative would:

- Improve the pavement condition along the project corridor by reconstructing the roadway.
- Improve the intersection geometry at key locations along the corridor.
- Incorporate an Access Management Plan for Red Lodge, which was supported by the Red Lodge City Council in March 2007. See Appendix A, Letter #8.
- Provide pedestrian and bicycle facilities in Red Lodge.
- Improve LOS by providing passing lanes in rural segments where appropriate and making improvements at the 8th Street/US Highway 212 and MT Highway 78/US Highway 212 intersections.
- o Provide wider roadway shoulders in rural segments where appropriate.
- Flatten ditch slopes in rural segments where appropriate.
- Reduce clear zone encroachments in rural segments where appropriate.
- Provide increased snow storage with wider and deeper roadside ditches.
- Clear thick brush and trees within the clear zone to improve driver visibility of approaching wildlife.
- Provide turning lanes where needed to reduce differential speed conflicts.
- o Improve highway-related storm water drainage in Red Lodge and Roberts.

Because the project corridor is relatively long (approximately 21.2 miles [34.1 kilometers]), and the character and needs of the adjacent communities change along this length, the project corridor will be discussed as four segments: Red Lodge, Red Lodge to Roberts, Roberts, and Roberts to Boyd. See Figure 2-1, Segment/Intersection Locations. The Preferred Alternative would have an estimated construction cost of approximately \$44.0 million; of this, approximately \$40.0 million would be for construction of the project.

The following sections describe in more detail the proposed improvements associated with the Preferred Alternative for each of the four segments. In addition to the improvements proposed to meet the specific project objectives, other improvements may be included to bring the roadway to current standards and/or meet the needs of the adjacent communities. See Table 2.1, Summary of Preferred Alternative. It should be noted that the proposed improvements presented as the Preferred Alternative and associated impact estimates are based on the conceptual design (approximately 30 percent) that is available at this early stage of the design process. Some minor adjustments may be necessary as the design process continues to evolve.



Figure 2-1, Segment/Intersection Locations

Table 2.1 Summary of Preferred Alternative						
Segment	Sub-segment	Typical Section	Intersection(s)	Intersection(s) Improvements	Drainage	
Red Lodge	8 th St to MT 78	 Two 12' travel lanes Two 10' parking lanes Two 5' sidewalks 	Oakes Avenue and 8 th Street	 Realign Oakes with 7th Street Convert Oakes to southbound one-way Provide diagonal parking on Oakes Narrow the street width at Carnegie Library Install traffic signal on 8th Street when warranted and justified 	 Curb and gutter Storm water conveyance system 	
	MT 78 to Developed Limits of Red Lodge	 Two 12' travel lanes Two 5.5' shoulders One 14' TWLTL One 5' sidewalk on west side of roadway One 10' shared bike/ped path on east side of roadway 	MT 78 and Villard Avenue	 Construct roundabout Close Villard and MT 78 intersection Construct cul-de-sac on Villard south of MT 78 Extend 4th Street 		
	Developed Limits of Red Lodge to Two Mile Bridge Road	 Two 12' travel lanes One 13' raised median Two 5.5' shoulders Two 3.5' median shoulders One 10' shared bike/ped path on east side of roadway 	Two Mile Bridge Road and four other locations	Per access management plan: • Construct roundabouts at full access intersections (3 intersections) • Construct ³ / ₄ access intersection • Construct ¹ / ₂ access intersection	Roadside ditches	
Red Lodge to Roberts	Two Mile Bridge Road to south end of Roberts	 Two 12' travel lanes Two 8' shoulders Bus turnaround One-mile northbound passing lane 			 Maintain existing drainage patterns and culvert locations May relocate irrigation ditches within right-of- way 	

Continued...

Table 2.1 Summary of Preferred Alternative						
Segment	Sub-segment	Typical Section	Intersection(s)	Intersection(s) Improvements	Drainage	
Roberts	South end of Roberts to East Maple Street	 Two 12' travel lanes Two 4' shoulders One 14' TWLTL 	Oak, Cedar, Pine, and East Maple Streets	Construct crosswalk at each intersection	 Construct berms perpendicular to highway to direct storm water Replace existing culvert south of Birch Street Direct storm water within Roberts to open ditches and/or pipes 	
	East Maple Street to north end of Roberts	 Two 12' travel lanes Two 4' shoulders One 14' TWLTL 2:1 inslope with guardrail on east 	Cooney Dam Road	 Realign to form a single intersection perpendicular to US Highway 212 		
Roberts to Boyd	North end of Roberts to Boyd North end of Roberts to Boyd North end of Roberts to Boyd North end of Roberts to Boyd Nore-mile northbound passing lane One-mile southbound passing lane	• Two 12' travel lanes	Clear Creek Road	 Realign with south access of rest area 		
		Cooney Dam Road	 Close northern fork Add southbound right-turn lane on US Highway 212 to Cooney Dam Road 	 Maintain existing drainage patterns and culvert locations May relocate irrigation 		
	Boyd Country Store	 Two 12' travel lanes Two 8' shoulders One southbound 12' left-turn lane Reverse curb along store parking lot 	Main Street	 Realign to be perpendicular to US Highway 212 Addition of southbound left- turn lane on US Highway 212 	ditches within right-of- way	

2.3.1 RED LODGE (8TH STREET TO TWO MILE BRIDGE ROAD)

The Preferred Alternative includes the following improvements for Red Lodge:

- Three distinct typical sections to accommodate user needs while minimizing impacts to adjacent properties
 - 8th Street to MT Highway 78 (See page 2-8)
 - MT Highway 78 to developed limits of Red Lodge (See page 2-9)
 - Developed limits of Red Lodge to Two Mile Bridge Road (See page 2-10)
- Drainage improvements (See page 2-12)
- o Intersection improvements at the following locations:
 - Oakes Avenue and 8th Street (See page 2-12)
 - MT Highway 78 and Villard Avenue (See page 2-14)
- Access management plan (See page 2-14)
 - Roundabouts at Two Mile Bridge and two other locations
 - One three-quarter access intersection
 - One half-access intersection
- Crosswalks would be provided, where appropriate, as determined during final design and in coordination with the City of Red Lodge.

The City of Red Lodge supported the Preferred Alternative as described below for the Red Lodge project segment in Resolution No. 3223, dated October 10, 2006, and Resolution No. 3228, dated March 27, 2007. See Appendix A, Cooperating Agencies.

2.3.1.1 Typical Sections

The project corridor in Red Lodge consists of three distinct sub-segments. South of MT Highway 78, the corridor is within a mixed use, downtown area with urban residential and commercial use. North of MT Highway 78, the corridor is adjacent to a developed area with mixed use that is predominantly commercial and suburban residential in nature and an undeveloped/developing area. The need for items such as parking lanes, pedestrian and bicycle facilities, turning lanes, and drainage improvements varies along each sub-segment, as does the proximity of adjacent development, which to some extent determines the available space for roadway improvements. Therefore, a different typical section has been proposed for each sub-segment within Red Lodge. See Figure 2-2, Red Lodge Typical Section Locations.

Additionally, Red Lodge has prepared a streetscape plan for US Highway 212 south of 8th Street. MDT will work with the City of Red Lodge to develop an agreement to address appropriate lighting and landscaping features to be incorporated into this proposed project. The Red Lodge Streetscape Plan, available funding, and maintenance responsibilities will be taken into consideration when developing the agreement.



Figure 2-2, Red Lodge Typical Section Locations

8th Street to MT Highway 78

The proposed improvement from 8th Street to MT Highway 78, in Red Lodge, includes a 44-foot (13.2-meter) curb-to-curb urban typical section consisting of:

- Two 12-foot (3.6-meter) travel lanes
- Two 10-foot (3.0-meter) parking lanes
- Two 5-foot (1.5-meter) sidewalks

See Figure 2-3, 8th Street to MT Highway 78.



Figure 2-3, 8th Street to MT Highway 78

MT Highway 78 to Developed Limits of Red Lodge

The proposed improvement from MT Highway 78 to the developed limits of Red Lodge includes a 49-foot (14.9-meter) curb-to-curb urban typical section with:

- Two 12-foot (3.6-meter) travel lanes
- Two 5.5-foot (1.7-meter) shoulders
- One 14-foot (4.2-meter) TWLTL (two-way left-turn lane)¹. See Figure 2-4.
- One 5-foot (1.5-meter) sidewalk on the west side of the roadway
- One 10-foot (3.0-meter) shared bike/ped path on the east side of the roadway

See Figure 2-5, MT Highway 78 to Developed Limits of Red Lodge.

What is a TWLTL ("Twittle")?

A TWLTL is a turn lane in the middle of a road that is used for left turning vehicles from either direction. The benefit of a TWLTL is that a motorist wishing to turn left can do so while waiting in the turn lane, out of the way of vehicles approaching from behind. Rear end accidents can be reduced by using TWLTLs in areas with frequent driveways or intersections along the road.



Figure 2-4, TWLTL (Two-Way Left-Turn Lane) Example

¹ MDT's standard width for a TWLTL is 14 feet (4.2 meters). The TWLTL at this location may be reduced to a 12-foot (3.6-meter) TWLTL in order to reduce right-of-way impacts. This would be determined during the design phase of the proposed project.



Figure 2-5, MT Highway 78 to Developed Limits of Red Lodge

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

Developed Limits of Red Lodge to Two Mile Bridge Road

The proposed improvement from the developed limits of Red Lodge to Two Mile Bridge Road includes a 55-foot (15.8-meter) rural typical section with:

- Two 12-foot (3.6-meter) travel lanes
- One 13-foot (3.9-meter) raised median
- Two 5.5-foot (1.7-meter) shoulders
- Two 3.5 foot (1.05-meter) median shoulders
- One 10-foot (3.0-meter) shared bike/ped path on the east side of the roadway

See Figure 2-6, Developed Limits of Red Lodge to Two Mile Bridge Road.



Figure 2-6, Developed Limits of Red Lodge to Two Mile Bridge Road

2.3.1.2 Drainage

The urban-developed portion of Red Lodge within the project limits extends from 8th Street to an area approximately 1,500 feet (460 meters) north of MT Highway 78. Within this area, a curb and gutter section and storm water conveyance system (such as a storm drain, trunk line, and/or open ditch) are proposed to accommodate highway-related storm drainage. Additionally, a new storm drain pipe and outfall may need to be constructed or the existing storm drain pipe (at the intersection of MT Highway 78 and US Highway 212) may need to be replaced, which would be determined during design. Construction of a new storm drain pipe may result in replacement of the existing storm drain and may also result in minor modifications to the existing water or wastewater piping systems within Red Lodge. North of the developed limits of Red Lodge, the typical section for the roadway is anticipated to change from an urban section to a more rural section, which would likely include roadside ditches. If a suitable location for a storm water conveyance system outfall is not identified before the start of the rural typical section, then the storm water conveyance system may discharge into the roadside ditches and flow north to an outfall location to Rock Creek. As appropriate, design of the roadside ditches may include permanent erosion and sediment control measures to manage runoff water quality.

Generally, conveying storm water long distances in roadside ditches is an undesirable situation. However, if site-specific issues make this method of storm water conveyance necessary, two potential Rock Creek outfall locations have been identified on the east side of the highway. The first potential location is approximately 1,150 feet (350 meters) south of Two Mile Bridge Road, where some runoff currently reaches Rock Creek. Use of that outfall location may include installation of a new pipe under the railroad grade and some bank erosion protection along the edge of the creek. The second potential location is along the south side of Two Mile Bridge Road. Use of that outfall location would likely involve conveying runoff via a new vegetated swale, constructed along the south side of Two Mile Bridge Road, to Rock Creek. At that location, bank erosion protection along the edge of Rock Creek may be required to protect the bridge abutment. As the design process continues to evolve, other outfall location sites may be determined to be appropriate.

2.3.1.3 Intersections

There are two locations in Red Lodge with proposed intersection improvements. The first location includes the intersections of Oakes Avenue and 8th Street. The second location includes the intersections of MT Highway 78 and Villard Avenue.

Oakes Avenue and 8th Street

Oakes Avenue intersects US Highway 212 at approximately a 70-degree skew through uncontrolled access in front of the former Pony Express Convenience Store. Nearly all eastbound 8th Street vehicles that turn north onto US Highway 212 are using Oakes Avenue instead of the intersection of 8th Street and US Highway 212. The adjacent Carnegie Library building restricts sight distance from the west approach of 8th Street. Additionally, the Carnegie Library building includes a stairway entrance on US Highway 212, which restricts available room to add pedestrian facilities along US Highway 212. On-street parking in front of the stairway entrance at 8th Street.

The proposed project would realign Oakes Avenue with 7th Street and convert it to a southbound one-way street. Diagonal parking would be provided on both sides of the street. Additionally, the total street width would be narrowed along US Highway 212 at the Carnegie Library to improve sight distance at 8th Street and provide a sidewalk along the east side of the

library. A traffic signal would be installed at 8th Street when warranted and justified; it is anticipated that the signal warrant may be met by year 2020. See Figure 2-7, Oakes Avenue/8th Street Intersections.



Figure 2-7, Oakes Avenue/8th Street Intersections

MT Highway 78 and Villard Avenue

The intersection of MT Highway 78 and US Highway 212 does not have adequate capacity to accommodate the anticipated future traffic volumes (please see Section 3.5.1 for more information).

The intersection of Villard Avenue and US Highway 212 is skewed at about 20 degrees. The area of Villard Avenue in front of the Fire Station consists of open pavement with undefined access. Also, the close proximity of the intersections of Villard Avenue and MT Highway 78 creates additional vehicular conflict points.

The proposed project would replace the MT Highway 78 and US Highway 212 intersection with a single lane roundabout. In addition, the intersection of Villard Avenue and MT Highway 78 would be closed. A cul-de-sac would be constructed at Villard Avenue just south of MT Highway 78. Fourth Street would be extended along the south side of the Visitor Center between US Highway 212 and Villard Avenue, replacing the Visitor Center's south access. *See Figure 2-8, MT Highway 78 / Villard Avenue Intersections.*

2.3.1.4 Access Management

At the request of the City of Red Lodge, MDT and the City of Red Lodge developed an Access Management Plan for the area between MT Highway 78 and Two Mile Bridge Road. This includes the developed area from MT Highway 78 north approximately 2,500 feet (760 meters) as well as the developing area from that point north to Two Mile Bridge Road. See Figure 2-9, Developed and Developing Limits of Red Lodge. The purpose of the Access Management Plan is to provide a means for MDT and the City to balance the need for vehicular progression along US Highway 212 with the need for access to adjacent properties.

In the developed area, there are approximately 21 existing access points on US Highway 212 to adjacent properties. The roadway currently consists of two 12-foot travel lanes and two 2-foot shoulders. With the absence of turning lanes, there is potential for differential speed conflicts between vehicles slowing to turn into driveways of adjacent properties and vehicles attempting to use the highway as a through route. This creates a safety concern with an increased potential for rear-end collisions and was cited in MDT's collision analysis as a contributing factor to traffic collisions.

A TWLTL was initially considered for the entire stretch between MT Highway 78 and Two Mile Bridge Road to accommodate turning vehicles accessing adjacent properties while improving progression for through traffic. However, the City of Red Lodge opposed the concept of a TWLTL in the developing area due to concerns that it may encourage commercial strip development.

To accommodate that concern, MDT and the City developed the aforementioned Access Management Plan. The plan identified that a TWLTL would be used in the developed area. For the developing area, a TWLTL would not be used. Instead, access would be managed along the developing area through the use of intersection types (full, ³/₄, and ¹/₂ access intersections) and a raised median. The plan identified locations for future intersections based upon existing access points, property lines, plats, development potential, and input from adjacent property owners. The Red Lodge City Council passed Resolution No. 3228 on March 28, 2007, which supported this plan. Pursuant to applicable Montana statutes and MDT policy, the plan would be recommended to the Montana Transportation Commission for their adoption. *See Appendix A, Letter #8, and Figure 2-10, Access Management Plan Overview.*



Figure 2-8, MT Highway 78 / Villard Avenue Intersections



Figure 2-9, Developed and Developing Limits of Red Lodge



Figure 2-10, Access Management Plan Overview

Two intersection treatment options were proposed to the public for the full access intersection locations: a conventional full access intersection and a roundabout. See Figure 2-11, Conceptual Full Access Intersection and Figure 2-12, Roundabout Intersection. At the request of the City of Red Lodge, a roundabout design has been identified as preferred for all of the full access intersection locations. See Appendix A, Letter #8.



Figure 2-11, Conceptual Full Access Intersection

What is a Roundabout?

Roundabouts should not be confused with traffic circles. Traffic circles have been used for many years in European countries and in parts of the United States in the early half of the 1900s. Roundabouts are a more modern intersection that coincidently also use a circular island in the center of the intersection. However, roundabouts are designed for modern vehicles, including fire trucks, buses, various sized emergency vehicles, truck and trailer combinations, and snow plows. Roundabouts require entering vehicles to yield to those already in the intersection, while the older traffic circles were susceptible to gridlock by allowing entering vehicles to have the right-of-way over vehicles within the intersection.



Figure 2-12, Conceptual Roundabout Intersection



The ³/₄ access intersection locations would allow right and left turns onto side streets. However, it would restrict access by eliminating left turns onto major streets and cross-street movements. This intersection option would require concrete islands to direct side street vehicles and would also require a median on US Highway 212 at the intersection. See Figure 2-13, Conceptual ³/₄ Access Intersection.

Figure 2-13, Conceptual ³/₄ Access Intersection

The ½ access intersection locations would only allow right-in/right-out turn movements, thereby eliminating left turns and cross-street movements. See Figure 2-14, Conceptual ½ Access Intersection.

Figure 2-14, Conceptual 1/2 Access Intersection

2.3.2 RED LODGE TO ROBERTS

The Preferred Alternative includes the following improvements for US Highway 212 between Red Lodge and Roberts:

- One typical section
- Drainage improvements

2.3.2.1 Typical Section

Two Mile Bridge Road to south end of Roberts

The proposed improvement from Two Mile Bridge Road to the south end of Roberts includes a 40-foot (12.0-meter) rural section with:

- Two 12-foot (3.6-meter) travel lanes
- Two 8-foot (2.4-meter) shoulders²
- A bus turnaround approximately one-mile (1.6 kilometers) north of Fox Road at the present state maintenance site
- A one-mile (1.6-kilometer) northbound passing lane

See Figure 2-15, Two Mile Bridge Road to South End of Roberts.

2.3.2.2 Drainage

Some culverts between Red Lodge and Roberts carry water generally from the west side of the roadway to the east side of the roadway. The existing drainage patterns generally parallel the roadway northward. Drainage patterns and culvert locations are expected to remain the same with implementation of the proposed project.

Numerous irrigation ditches are currently located within the right-of-way limits of the proposed project. The proposed project may involve relocating those irrigation ditches outside of the new right-of-way, in accordance with MDT's general practice.

² This would provide adequate width for future overlays.



Figure 2-15, Two Mile Bridge Road to South End of Roberts

2.3.3 ROBERTS

The Preferred Alternative includes the following improvements for Roberts:

- One primary typical section, with a modification by Roberts School to avoid impacts (See page 2-23)
- Drainage improvements (See page 2-24)
- Intersection improvements at Cooney Dam Road (See page 2-25)
- Safety improvements for pedestrians (See page 2-25)

2.3.3.1 Typical Sections

Within Roberts, there is one typical section. A slight modification of the typical section was required adjacent to Roberts School, between East Maple Street and the north end of Roberts. See Figure 2-16, Roberts Typical Section Locations.

South end of Roberts to East Maple Street

The proposed improvement from the south end of Roberts to East Maple Street includes a 46-foot (13.8meter) rural section with:

- Two 12-foot (3.6meter) travel lanes
- Two 4-foot (1.2meter) shoulders
- One 14-foot (4.2-meter) TWLTL³

See Figure 2-17, South End of Roberts to East Maple Street.



Figure 2-16, Roberts Typical Section Locations

³ MDT's standard width for a TWLTL is 14 feet (4.2 meters). The TWLTL at this location may be reduced to a 12-foot (3.6-meter) TWLTL in order to reduce right-of-way impacts. This would be determined during the design phase of the proposed project.



Figure 2-17, South End of Roberts to East Maple Street

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

East Maple Street to north end of Roberts

A modification of the typical section was developed for the sub-segment between East Maple Street and the north end of Roberts to avoid impacts to Roberts School and to meet the request of Roberts School for a guardrail along school property.

The proposed improvement from East Maple Street to the north end of Roberts includes a 46-foot (13.8-meter) rural section with:

- Two 12-foot (3.6-meter) travel lanes
- Two 4-foot (1.2-meter) shoulders
- One 14-foot (4.2-meter) TWLTL⁴
- 2:1 inslope with guardrail on east side

See Figure 2-18, East Maple Street to North End of Roberts.

⁴ MDT's standard width for a TWLTL is 14 feet (4.2 meters). The TWLTL at this location may be reduced to a 12-foot (3.6-meter) TWLTL in order to reduce right-of-way impacts. This would be determined during the design phase of the proposed project.


Figure 2-18, East Maple Street to North End of Roberts

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

2.3.3.2 Drainage

A large drainage area lies west of the roadway. Within this area, flood irrigation operations contribute additional water to the drainage. The drainage parallels the roadway from south of the Fox Road and drains northward along the west edge of Roberts. Because the portion of Roberts west of the roadway sits in one of the swales of the drainage, the potential exists to flood homes in Roberts, as occurred in 2005 and 2007. It is beyond the scope of this project to address floodwaters entering Roberts from the drainage to the west. Goals of the proposed project would be to convey as much highway-related storm water as practicable toward Rock Creek prior to Roberts, and within Roberts to satisfactorily convey storm water intercepted by the highway towards Rock Creek. For example, berms may be constructed perpendicular to flow east and west of the highway, approximately 1,400 feet (425 meters) south of Birch Street in Roberts. At that location an existing centerline culvert is currently planned to be replaced. The existing downstream drainage channel located on the east side of the highway may be enlarged and re-graded to convey more runoff to Rock Creek.

Within the community of Roberts, it is anticipated that runoff from the roadway would be directed to open ditches and/or pipes. The ditches and/or pipes would carry the runoff to roadside ditches located north of Roberts.

2.3.3.3 Intersection

There is one location in Roberts with proposed intersection improvements: the intersection of Cooney Dam Road and US Highway 212, near East Maple Street. Cooney Dam Road forms a 27-degree skewed intersection with US Highway 212 and is offset from East Maple Street. This is the busiest intersection in Roberts, with a convenience store (Y-Stop) located on the west side of US Highway 212 directly across from the school. The convenience store has uncontrolled access on its frontage with US Highway 212 and along Cooney Dam Road.

The Preferred Alternative would realign Cooney Dam Road to form a single intersection perpendicular with US Highway 212 approximately 430 feet (130 meters) north of East Maple Street. Access to the Y-Stop would be provided as determined through coordination with the business owner during project design. See Figure 2-19, Cooney Dam Road Intersection (near East Maple Street).

2.3.3.4 Pedestrian Facilities

The community of Roberts completed a sidewalk project in 2005, which placed sidewalk along First Street, one block east of and parallel to US Highway 212. As part of that project, sidewalks were also placed on East Maple Street, Cedar Street, and Oak Street, to provide connections from US Highway 212 to the new sidewalk on First Street. The goal of that project was to provide a facility for pedestrians, including those walking to and from school, on a local road rather than on the highway. There are existing crosswalks at Oak Street and north of East Maple Street. School crossing signs, including advance warning signs, and post-mounted flashing beacons are also associated with these crossing locations.

The Preferred Alternative would include one block of sidewalk along Pine Street from US Highway 212 east to connect to the sidewalk on First Street. Crosswalks would also be provided at the intersections of Oak, Cedar, Pine, and East Maple Streets along US Highway 212. School advance warning and school crosswalk warning signs, including flashing beacons, would be installed in accordance with current design guidelines.

2.3.4 ROBERTS TO BOYD

The Preferred Alternative includes the following improvements for the area from Roberts to Boyd:

- o One primary typical section, with a modification at Boyd Country Store
- Drainage improvements (See page 2-28)
- Intersection improvements at three locations (See page 2-31)
 - o Clear Creek Road
 - o Cooney Dam Road
 - o Main Street

2.3.4.1 Typical Sections

From Roberts to Boyd, one typical section is proposed for the majority of the segment with a modification in the vicinity of the Boyd Country Store to improve safety. See Figure 2-20, Roberts to Boyd Typical Section Locations.



Figure 2-19, Cooney Dam Road Intersection (near East Maple Street)

Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.



Figure 2-20, Roberts to Boyd Typical Section Locations

Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

North end of Roberts to Boyd

The proposed improvement from the north end of Roberts to Boyd includes a 40-foot (12.0-meter) rural typical section with:

- Two 12-foot (3.6-meter) travel lanes
- Two 8-foot (2.4-meter) shoulders⁵
- A bus turnaround approximately 3 miles (4.8 kilometers) south of Boyd at the school district boundary
- A one-mile (1.6-kilometer) northbound passing lane north of Roberts
- A one-mile (1.6-kilometer) southbound passing lane south of Boyd

See Figure 2-21, North End of Roberts to Boyd.

The exception to this typical section would be at the Boyd Country Store where the road would become a 52-foot (15.6-meter) rural typical section with:

- Two 12-foot (3.6-meter) travel lanes
- Two 8-foot (2.4-meter) shoulders
- A southbound 12-foot (3.6-meter) left-turn lane
- A reverse curb to delineate the Boyd Country Store parking lot from the roadway and to manage access to US Highway 212

See Figure 2-22, North End of Roberts to Boyd – Boyd Country Store.

2.3.4.2 Drainage

Existing drainage patterns and culvert locations are expected to remain the same in the design of the proposed project. Several irrigation ditches that are currently located within the right-ofway limits for the proposed project are expected to be relocated outside of the new right-of-way in accordance with MDT's general practice.

⁵ This would provide adequate width for future overlays.



Figure 2-21, North End of Roberts to Boyd

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.



Figure 2-22, North End of Roberts to Boyd – Boyd Country Store

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

2.3.4.3 Intersections

There are three locations between the north end of Roberts and Boyd with proposed intersection improvements: (1) Clear Creek Road, (2) Cooney Dam Road, and (3) Main Street.

Clear Creek Road

Clear Creek Road is located across from the rest area between the two rest area access approaches, forming an offset intersection with both approaches. The Preferred Alternative would realign Clear Creek Road with the south access of the rest area. In addition, improvements to the rest area sidewalks and ramps may be included as part of the proposed project. See Figure 2-23, Clear Creek Road Intersection.

Cooney Dam Road

The Cooney Dam Road intersection is located on the northwest side of Boyd and has two forks, both of which intersect US Highway 212. The north fork intersects US Highway 212 at a skew, while the south fork intersects US Highway 212 at a right angle. On Cooney Dam Road, westbound traffic from the north and south forks must merge, creating a conflict point. The proposed project would close the northern fork of Cooney Dam Road and add a southbound right-turn lane on US Highway 212 to Cooney Dam Road. See Figure 2-24, Cooney Dam Road and Main Street Intersections.

Main Street

Main Street is located on the east side of US Highway 212, north of the Boyd Country Store. Main Street intersects US Highway 212 at a skewed angle. The Boyd Country Store has uncontrolled access along its frontage with US Highway 212 and Main Street. Additionally, there are no deceleration lanes on US Highway 212 to provide protection for turning vehicles accessing Main Street or the Boyd Country Store; this creates a differential speed conflict. The Preferred Alternative would realign Main Street to be perpendicular with US Highway 212. In addition, a southbound left-turn lane would be added on US Highway 212 to provide protection for slowed or stopped highway traffic accessing Main Street. A reversed curb line would be constructed between the Boyd Country Store parking lot and US Highway 212 to delineate access. Access to the Boyd Country Store would be provided off of Main Street. See Figure 2-24, Cooney Dam Road and Main Street Intersections.



Figure 2-23, Clear Creek Road Intersection

Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.



Figure 2-24, Cooney Dam Road and Main Street Intersections

Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

2.4 ALTERNATIVES ELIMINATED FROM FURTHER STUDY

Throughout the EA process the options under consideration for the project corridor have continuously evolved based on input from the Red Lodge City Council and Carbon County Commission, public comments, existing and planned development, and data collected pertaining to engineering factors and environmental considerations.

In addition to the Preferred Alternative discussed in the previous sections, numerous other ideas were evaluated to varying levels of detail and discarded from further analysis for a number of possible reasons, such as a failure to meet the purpose and need, anticipated undesirable and unavoidable impacts, or public comments. Following is a brief summary of options that were considered and then eliminated from further study in this EA.

2.4.1 Red Lodge

In Red Lodge, options that were considered and then eliminated from further study include:

- Options for bicycle lanes, sidewalks, and shared bike/ped paths. The City identified a need to accommodate pedestrians and bicyclists along US Highway 212. In the early stages of project planning, the Red Lodge City Council requested that a shared bike/ped path on the east side of the roadway from 8th Street to Two Mile Bridge Road and a sidewalk on the west side of the roadway from 8th Street to the city limits be incorporated into this project and the Red Lodge Park Board requested a shared bike/ped path from 8th Street to the city limits. These preferences were used to develop options for various locations (in relation to the roadway) and widths (based on MDT guidelines and available space) of pedestrian and bicycle facilities. Options mainly consisted of combinations of 5-foot sidewalks and 10-foot shared bike/ped paths between 8th Street and Two Mile Bridge Road. These options were eventually eliminated based on impacts to the surrounding area and City preferences, which were solidified in 2006 when Red Lodge approved a Comprehensive Trails Plan; this is reflected in the Preferred Alternative.
- **Options for parking lanes.** The City identified a need for additional downtown parking in Red Lodge, primarily between 8th Street and MT Highway 78. A 9.5-foot (2.9-meter) and 12-foot (3.6-meter) parking lane width (based on MDT guidelines and available space) and location (based on need for parking) were evaluated in cooperation with the City prior to identification of the preferred typical sections. These options were eventually eliminated based on impacts to the surrounding area and City preferences.
- A TWLTL between the developed limits of Red Lodge and Two Mile Bridge Road. Based on anticipated growth in northern Red Lodge, particularly the area from approximately 2,500 feet (760 meters) north of MT Highway 78 to Two Mile Bridge Road, preliminary project recommendations included the use of a shared TWLTL between MT Highway 78 and Two Mile Bridge Road. The use of a TWLTL would have reduced the potential for differential speed conflicts from turning vehicles while providing access to the developed and developing areas adjacent to the corridor. However, the City of Red Lodge opposed the concept of a TWLTL along the developing northern segment. Through coordination with the City, adjacent property owners, and the general public, an access management plan was developed for this area. This plan identified the use of a TWLTL from MT Highway 78 north for approximately 2,500 feet (760 meters), to the northern limits of the developed portion of Red Lodge. For the developing portions of the corridor (from that point north to Two Mile Bridge Road), a raised median and predetermined access locations/types were agreed upon to manage access.

- **Options for Oakes Avenue Intersection.** The Oakes Avenue intersection with US Highway 212 is skewed, presenting traffic safety concerns, particularly for vehicles traveling north on Oakes Avenue and wanting to turn onto US Highway 212. Preliminary options for Oakes Avenue included closing Oakes Avenue between its junction with US Highway 212 and 8th Street; closing the Oakes Avenue/US Highway 212 intersection and turning Oakes Avenue into a dead end; and realigning the Oakes Avenue/US Highway 212 intersection to be perpendicular and leaving it open for two-way traffic. Through coordination with the City, it was determined that the conversion of Oakes Avenue to a one-way, southbound roadway with diagonal parking on both sides would be more advantageous for the City than the other options. The Preferred Alternative provides additional downtown parking, which is needed in the City, while improving safety conditions. This concept also reflects coordination with the proposed Bank of Red Lodge development.
- **Options for MT Highway 78 and Villard Avenue intersections.** Several options were explored for the MT Highway 78 and Villard Avenue intersections prior to identification of the Preferred Alternative. These included use of a standard, signalized intersection at MT Highway 78 and options for closure of Villard Avenue north or south of MT Highway 78. However, the City requested that a roundabout be considered for the MT Highway 78 intersection instead of a traffic signal. MDT investigated the roundabout option, and for a number of reasons previously described, the roundabout became the preferred option for this location. The traffic signal was eliminated from further study. Likewise, it was determined through coordination with the City that maintaining closure of Villard Avenue to a dead end, would meet the needs of the adjacent property owners while improving safety conditions.
- **Turn lanes at Two Mile Bridge Road.** Northbound and southbound turn lanes were requested by the Mayor and deemed to be warranted by MDT standards. Throughout development of the Access Management Plan it was determined to use a roundabout at this location instead of a conventional intersection design.

2.4.2 Red Lodge to Roberts

Between Red Lodge and Roberts, one option was considered and eliminated from further study:

• **Narrower shoulder widths.** These included 4-foot shoulders, and a combination of 4-foot paved shoulders with 4-foot gravel shoulders. The wider shoulders (now preferred) would provide additional safety, an improved level of operation, and were also desired by the public.

2.4.3 Roberts

In Roberts, options considered and then eliminated from further study include:

- **Parking lanes and sidewalks.** Parking lanes were eliminated due to the higher speed roadway and no identified need for on-street parking. Sidewalks were not needed because a previous project placed sidewalks one block east of and parallel to 212 to provide pedestrian accommodations off of the highway.
- **Typical Section presented at a public meeting.** At a public meeting in 2003, a typical section consisting of two 12-foot travel lanes, two 7.5-foot shoulders, two 7-foot sidewalks, and curb and gutter on both sides of the roadway was presented to the public as an option. However, following the collection and analysis of traffic data, it was determined that a TWLTL between the south end of Roberts and East Maple Street was

warranted and that sidewalks were not needed, for the reasons described above. Also, there was available right-of-way to accommodate storm water drainage using a cross-sloped roadway and roadside ditch rather than curb and gutter.

• **Options for Cooney Dam Road and East Maple Street intersections.** Cooney Dam Road in Roberts is a skewed intersection and also forms an offset intersection with East Maple Street. Options evaluated for these intersections included realignment of East Maple Street to directly oppose Cooney Dam Road, realignment of Cooney Dam Road to a right angle, and realignment Cooney Dam Road to directly oppose East Maple Street. The realignment of East Maple Street was ruled out to avoid impacts to Roberts School, which is a Section 4(f) and 6(f) property as discussed further in Chapters 3 and 4. The other options for realigning Cooney Dam Road were ruled out because they would not provide substantial reductions in traffic conflicts.

2.4.4 Roberts to Boyd

From Roberts to Boyd, options considered and eliminated from further study include:

- **Narrower shoulder widths.** These included 4-foot shoulders, and a combination of 4-foot paved shoulders with 4-foot gravel shoulders. The wider shoulders would provide additional safety, an improved level of operation, and were also desired by the public.
- **Options for Clear Creek Road/Rest Area intersections.** Options for these intersections included realigning both rest area accesses to 90-degree angles, combining both rest area accesses into one access, the addition of acceleration/ deceleration lanes at the rest area, and not realigning Clear Creek Road. These options were ruled out for the following reasons: The existing rest area accesses are approximately at an 85-degree skew, which is acceptable under MDT design standards; combining the rest area accesses would not allow for adequate internal traffic circulation; the addition of acceleration/deceleration lanes was not warranted; and maintaining an offset intersection at this location would not improve traffic, safety, and operation.
- **Two options for Dakota Avenue intersection.** The closure of Dakota Avenue or its conversion to a right-in/right-out access only were both considered as potential safety improvements and then ruled out from further analysis. The Carbon County Commissioners and community of Boyd do not want Dakota Avenue closed, so that it may still accommodate school buses and emergency vehicles. The right-in/right-out access option would require a median, which would be a safety concern due to the close proximity to a high speed highway.
- **Options for Cooney Dam Road intersection.** Early options for the Cooney Dam Road intersection were to realign the north fork of Cooney Dam Road to the north to directly oppose Main Street or to maintain both approaches. These options were eliminated to minimize wetland impacts and improve safety conditions.
- **Delineate the parking lot of the Boyd Country Store with guardrail.** The guardrail option was eliminated because it was not desirable for maintenance purposes and may pose safety concerns due to the traffic speed limit.

2.4.5 Project Corridor

For the project corridor as a whole, the following options were considered and eliminated from further study:

• Additional travel lanes. Based on public input, a four-lane divided roadway for the entire project corridor was considered. A traffic operations analysis was conducted to evaluate this option. Based on the existing and projected urban and rural traffic volumes,

the existing two-lane roadway would provide adequate capacity through the project design year.

- **Realign the project corridor.** Due to roadway geometry and existing wetlands, three portions of the project were considered for alignment shifts to minimize wetland impacts. Upon investigation it was determined that realigning in either direction would not minimize wetland impacts substantially and would incur additional impacts such as relocations.
- **Speed limit changes.** Due in part to request from the public and the City of Red Lodge, potential speed limit changes were evaluated. The project corridor is located on level terrain, with the maximum grade of the roadway not exceeding 3 percent. The MDT design speed criterion for this type of roadway (rural minor arterial on level terrain) is 60 mph (105 kph). Design speed affects various design criteria, such as sight distance, length of deceleration/turn lanes, and width of clear zones. A lower design speed would reduce clear zone width, possibly reducing tree and brush clearing, and would reduce project costs with shorter centerline culverts across the roadway. A higher design speed would have the opposite effect. Further, there is a public perception that a lower design speed would improve safety conditions along the corridor. MDT conducted a speed study through the project area in 2003, yielding a recommendation to maintain the existing speed limits. Motorists have a tendency to drive at a speed that they feel comfortable with based on conscious and subconscious information they receive through their eyes, ears, and the road. MDT studies have indicated that simply lowering a speed limit is not likely to lower driving speeds. However, as development continues along the corridor, particularly in north Red Lodge, another speed study may yield the recommendation to reduce speed limits along parts of the project corridor.
- **Transit Management.** A transit management alternative was not carried forward due to the rural nature of the area and low traffic volume.

Chapter 3 Affected Environment, Impacts, & Mitigation

3.1 INTRODUCTION

This chapter describes the existing conditions and potential environmental impacts along the project corridor. The inventory and evaluation of the existing conditions, or affected environment, provides the necessary baseline from which to determine the impacts of the proposed project alternatives. This chapter uses this baseline to identify the positive and negative environmental impacts of the preferred and no-build alternatives presented in Chapter 2. Potential direct, indirect, and cumulative environmental impacts are presented, as well as avoidance, minimization, and mitigation measures that may be implemented.

3.2 ENVIRONMENTAL CONSIDERATIONS UNAFFECTED BY THE PROJECT

The following environmental considerations were reviewed and were found to be unaffected by the proposed project.

Air Quality — The purpose of this project is to improve the safety and operational characteristics of the roadway by making improvements to pavement, intersections, access, pedestrian/bicycle facilities, LOS, shoulders, ditch slopes, clear zones, and safety, as discussed in Chapter 2. This project would not result in any meaningful changes in traffic volumes, vehicle mix, location of the existing facility, or any other factor that would cause an increase in emissions impacts relative to the no-build alternative. As such, FHWA has determined that this project would generate minimal air quality impacts for Clean Air Act criteria pollutants and has not been linked with any special MSAT (Mobile Source Air Toxics) concerns. Consequently, this effort is exempt from analysis for MSATs.

Moreover, EPA (Environmental Protection Agency) regulations for vehicle engines and fuels will cause overall MSATs to decline significantly over the next 20 years. Even after accounting for a 64 percent increase in VMT (Vehicle Miles Traveled), FHWA predicts MSATs will decline in the range of 57 percent to 87 percent, from 2000 to 2020, based on regulations now in effect, even with a projected 64 percent increase in VMT. This will both reduce the background level of MSATs as well as the possibility of even minor MSAT emissions from this project.

Wild and Scenic Rivers — There are no Wild and Scenic Rivers located in the study area.

Coastal Barriers/Coastal Zone Impacts — The proposed project is not located in a coastal barrier or coastal zone area.

Energy — The proposed project would require the consumption of energy and resources that would not be used if US Highway 212 was not reconstructed. This is necessary in order to maintain a safe and efficient transportation corridor in the area. The benefits of the project to the traveling public would compensate for the energy lost during construction by improving the efficiency of travel along US Highway 212. Additionally, a minor amount of energy may be saved at roundabout intersections versus standard signalized intersections, as they do not require vehicles to idle but allow a continual flow of traffic.

Environmental Justice — US Census block group data (year 2000) were evaluated in regards to low-income populations in the project corridor. A "block group" reflects a sampling of households rather than all households. The block groups along the project corridor range from 4

to 18 percent low-income populations and are found throughout the project corridor. This compares to a 10.5 percent average for the state of Montana.

Census block data (also year 2000) were evaluated in regards to minority populations in the project area. The census "block" reflects data collected from all households. Montana's population consists of 9.4 percent minority populations. Along the project corridor, there are three blocks that contain greater than 5 percent minority populations. They are located in Red Lodge (22 percent minority), between Roberts and Boyd (9 percent minority), and in Boyd (14 percent minority). The block in Red Lodge has a total population of nine with the minority population consisting of two American Indians/Alaska Natives. The block between Roberts and Boyd has a total population of 33 with the minority population consisting of one American Indian/Alaska Native and two individuals that are American Indian/Alaska Native and White. The block in Boyd consists of a total population of seven with the minority population consisting of one American Indian/Alaska Native.

Impacts to residences along the corridor would be mainly a result of right-of-way acquisition to meet standard MDT right-of-way widths, as discussed in Section 3.7. These impacts would be uniform along the project corridor and none of the identified low-income and/or minority populations are expected to bear the brunt of these impacts. Therefore, the proposed project is not expected to result in disproportional adverse effects on minority and/or low-income populations.

Economic Considerations — The proposed project is not expected to significantly affect tax revenues and public expenditures, employment opportunities, accessibility, and retail sales on the regional and/or local economy; the economic vitality of existing highway-related businesses; or established business districts. The location of development may be marginally affected by changes in speed and access. If access control is effective, there may be a somewhat greater tendency of businesses to locate near established settlements and access points. Improvements in time and safety of travel through the corridor should improve consumer access to wider markets, which may be expected to encourage greater competition for consumer and producer spending, with price and selection benefits for the area population. Given the relatively small scale of this project, however, these effects are expected to be marginal.

3.3 LAND USE

The proposed project is located in Carbon County, Montana. The project corridor consists of urban and rural roadway. The land uses surrounding the project corridor are primarily mandominated systems that have been altered for residential, commercial, agricultural, transportation, and utility purposes.

The urban area of the project corridor is within Red Lodge. As previously discussed, the project corridor in Red Lodge consists of three distinct sub-segments. South of MT Highway 78, the corridor is within a mixed use, urbanized, downtown area with residential and commercial use. The second sub-segment, north of MT Highway 78, is adjacent to a developed area with mixed use that is predominantly commercial in nature and the third sub-segment is adjacent to an undeveloped/developing area.

The remainder of the project corridor is rural in nature, and adjacent land is primarily used for agriculture. The corridor also traverses the three unincorporated communities of Fox, Roberts, and Boyd; in these areas, residential and commercial properties abut the corridor. Also in the rural segments, there are three conservation easements owned by the Montana Land Reliance.

US Census data shows that the population in Carbon County grew faster than the national and state averages from 1990-2000. Since 2000, Red Lodge has experienced substantial population increases relative to the comparative samples. *See Table 3.1, Population Statistics.*

	Table 3.1 Population Statistics	
Location	Population Growth 1990-2000	Population Growth 2000-2006
Red Lodge	11.2%	12.8%
Carbon County	18.2%	3.7%
Montana	12.9%	4.7%
United States	13.2%	6.4%

The reasons for population increases in this area are varied and complex, and likely include factors such as economic conditions, the nearby availability of amenities such as hiking, skiing, etc., and the aesthetics of the area. As the population in Carbon County, particularly in Red Lodge, increases, land use changes result. Information received from Red Lodge and Carbon County Planning staff indicates a number of new developments planned adjacent to or near to the project corridor, including five subdivisions, a hospital, and a bank in Red Lodge; one subdivision in Roberts; and two subdivisions between Roberts and Boyd. *See Table 3.2, Planned Development.*

Table 3.2				
Planned Development				
Location	Name	Description		
Red Lodge	City Lights Subdivision	9-lot subdivision in Country Club Estates on Lazy "M" Street		
Red Lodge	Diamond C. Links Subdivision	138-lot subdivision on West Bench Road		
Red Lodge	Woodlands Subdivision	76-unit subdivision near Rock Creek		
Red Lodge	Luoma Annexation – Spires Subdivision	305-400 unit subdivision on 119 acres, on West Bench near MT Highway 307		
Red Lodge	Remington Ranch Subdivision	170-unit subdivision along Remington Ranch Road		
Red Lodge	Beartooth Hospital	New hospital, nursing home, assisted living facility, and medical offices southwest of Two Mile Bridge Road intersection		
Red Lodge	Bank of Red Lodge	New bank northwest of 8 th Street intersection		
Roberts	Merritt Village South Subdivision	22-unit subdivision in Roberts		
Roberts to Boyd	Sapphire Springs Subdivision	47-lot subdivision on the west side of US Highway 212		
Roberts to Boyd	Dot Calm Ranches Subdivision	159-unit subdivision southeast of Clear Creek Road		

Both the City of Red Lodge and Carbon County have developed guidance intended to control growth and development patterns. Red Lodge adopted the Red Lodge Growth Policy in May 2001. The Red Lodge Growth Policy attempts to balance real estate market forces and the interests of the public by requiring that proposed developments on large vacant parcels within the City and areas around Red Lodge be reviewed and evaluated using a permit system, rather than through traditional zoning. In September 2003, Carbon County adopted the Carbon County

Growth Policy, which is applicable to the rural segments within the project corridor, including the unincorporated communities of Fox, Roberts, and Boyd. In May 2006, using the Red Lodge Growth Policy as guidance, Red Lodge adopted the City of Red Lodge Comprehensive Trails Plan. This Plan was adopted to satisfy one of the goals of the Red Lodge Growth Policy, which was to develop a trail system linking parks, residential areas, and open spaces.

Impacts to Land Use

Alternative A (No-Build): If no action were taken, the proposed project would not directly impact land use in the area. Population growth and development in the area would still be expected to occur even if the highway were not improved. However, Alternative A would not provide the necessary roadway improvements to accommodate these land use changes and would not be consistent with the Red Lodge or Carbon County Growth Policies or the Red Lodge Comprehensive Trails Plan.

Alternative B (Preferred): The Preferred Alternative would be consistent with the Red Lodge and Carbon County Growth Policies, as well as the Red Lodge Comprehensive Trails Plan. It would accommodate the land use changes near the project corridor, but would not drive these changes. Some new right-of-way would be needed along the corridor for roadway improvements and to achieve the standard MDT right-of-way width for the facility type; lands in various uses, including agricultural, commercial, residential, and conservation easements, would be converted from their existing use to part of the transportation corridor. However, the highway has been in place for many years, and the Preferred Alternative would improve the roadway on its existing alignment. The Preferred Alternative is not expected to induce additional traffic since it would not provide additional capacity (aside from passing lanes at three locations in the rural segments) nor is it expected to change growth patterns in the area.

During the EA scoping process, the Red Lodge City Planner expressed verbal concerns about potential indirect land use and development impacts that the project may have on the undeveloped area in north Red Lodge. A preliminary project proposal was to use a TWLTL for the length of the project corridor in Red Lodge. The purpose of the TWLTL would have been to provide for access to all developed and developing properties adjacent to the roadway while reducing differential speed conflicts between turning vehicles and through-traveling vehicles. However, the Red Lodge City Planner expressed concern that the use of a TWLTL may encourage commercial strip development in northern Red Lodge, which was not the type of land use development the City wished to see. MDT worked cooperatively with Red Lodge to develop an Access Management Plan for this area, which has been incorporated into the Preferred Alternative (as discussed in Chapter 2). The Preferred Alternative has been modified such that a TWLTL is no longer proposed for the undeveloped area in north Red Lodge, and indirect impacts to land use and development are not anticipated.

Mitigation for Land Use Impacts

No adverse impacts to land use are anticipated as a result of the Preferred Alternative; therefore, no mitigation is required.

3.4 FARMLAND

The Farmland Protection Policy Act of 1981 (7 U.S.C. 4201 et seq.) provides protection to prime and unique farmlands. Prime farmlands are those that have the best combination of physical and chemical characteristics for producing food, feed, forage, fiber, and oilseed crops and is also available for these uses (not developed land or water). It has the soil quality, growing season, and moisture oversupply needed to economically produce sustained high yields of crops when treated and managed, including water management (irrigation), according to acceptable farming methods. Unique farmland is land that is used for production of specific high value food, feed, and fiber crops. Section 658.5 of the Farmland Protection Policy Act provides criteria for federal agencies to identify and take into account the adverse effects of federal programs on the protection of farmland. Federal agencies are to consider alternative actions, as appropriate, that could lessen adverse effects; and to assure that such federal programs, to the extent practicable, are compatible with State, unit of local government, and private programs and policies to protect farmland.

Agricultural lands surround the rural segments of the project corridor. Portions of these segments are located within prime and statewide important farmland.

Impacts to Farmland

Alternative A (No-Build): If no action were taken, there would be no impacts to prime, unique, or statewide important farmland in the project area.

Alternative B (Preferred): The Preferred Alternative would impact approximately 275.8 acres (111.6 hectares) of farmland. Of this, approximately 89.6 acres (36.3 hectares) are considered prime farmland¹ and 48.9 acres (19.8 hectares) are considered to be of statewide importance. Farmland Conversion Impact Rating form CPA-106 has been completed in cooperation with the NRCS (Natural Resources Conservation Service). The CPA-106 form computation resulted in a total point value of 133 out of a possible score of 260. Section 658.4(c)(2) of the Farmland Protection Policy Act states that sites receiving a total score of less than 160 on the Form need not be given further consideration for protection. *See Appendix B, NRCS Coordination & CPA-106 Forms.*

Mitigation for Farmland Impacts

As the CPA-106 score was less than 160, no mitigation is required for impacts to farmland.

3.5 TRANSPORTATION SYSTEM

The following discussion addresses components of the transportation system within the project corridor: traffic, access, safety, and pedestrian and bicycle facilities.

3.5.1 Traffic

MDT provided traffic volume estimates for project corridor. The estimates were for the current year (2007), the approximate year that the proposed improvement would be open to traffic (2010), and the project design year (2030). Due to development along the corridor, traffic volumes are expected to increase approximately 158 percent in the rural segments and 220 percent in Red Lodge by 2030. See Table 3.3, Average Annual Daily Traffic (AADT).

¹ The estimated impacts to prime farmland includes impacts to land designated "prime-if-irrigated".

Table 3.3 Average Annual Daily Traffic (AADT)					
Location		Percent			
Location	2007	2010	2030	Increase	
8 th Street to MT Highway 78	5,440	6,030	12,010	220	
MT Highway 78 to Two Mile Bridge Road	2,980	3,300	6,570	220	
Two Mile Bridge Road to Boyd	2,840	3,010	4,480	158	

The LOS was analyzed for the project segments and key intersections². The analysis was completed for the no-build alternative and the Preferred Alternative for years 2007, 2010, and 2030. The desired LOS is B for rural areas and C for urban areas; this represents stable operations.

Currently, all project segments and key intersections are functioning at an acceptable LOS, with the exception of the northbound traffic lane in the rural segment between Roberts and Boyd.

Impacts to Traffic

Alternative A (No-Build): Under the no-build alternative, the intersections of 8th Street and MT Highway 78 would not function with an acceptable LOS by the project design year. The 8th Street intersection would be expected to operate at a LOS E, and the MT Highway 78 intersection would be expected to operate at a LOS F by 2030. Traffic flow in the northbound lane between Roberts and Boyd, which is currently functioning at a LOS C, would not be improved. The southbound lane in the rural segment between Red Lodge and Roberts is anticipated to deteriorate to LOS C by 2010 if no improvements are made.

Alternative B (Preferred): Under the Preferred Alternative, all of the segments and key intersections would be expected to operate at acceptable LOS (B or greater in rural segments and C or greater in urban segments) through 2030.

Mitigation for Traffic Impacts

The Preferred Alternative would be consistent with MDT design guidelines for a minimum of LOS B for rural segments and LOS C for urban segments; therefore no mitigation of traffic impacts would be required.

3.5.2 Access

It is important to balance the competing needs of access (to adjacent properties) and progression (of through traffic) along a roadway corridor. The existing corridor provides one travel lane in each direction and narrow roadway shoulders, with access to adjacent properties provided directly from US Highway 212. In other words, vehicles wishing to turn and those wishing to go straight are sharing the same narrow roadway. In areas along the corridor with urban development (numerous properties requiring access in close proximity to each other), particularly Red Lodge and Roberts, this creates the potential for disorderly traffic flow and unsafe conditions. The project area is expected to continue to experience population growth and

² More detailed information can be found in the Traffic Report Technical Memorandum dated October 16, 2007, which is available for review from MDT.

associated land development and traffic increases, which will result in a higher demand for both access and progression. The City of Red Lodge has already noted that there is inadequate public parking to support the downtown arts and crafts community, including the Arts Guild and Carnegie Library.

Safe and efficient access is particularly important for emergency response vehicles at the Red Lodge Fire Station and proposed Beartooth Hospital, both located adjacent to the project corridor in Red Lodge.

At the request of the City of Red Lodge, MDT and the City developed an Access Management Plan for the project corridor from MT Highway 78 to Two Mile Bridge Road.

Impacts to Access

Alternative A (No-Build): The no-build alternative would not be consistent with the Access Management Plan supported by the Red Lodge City Council in Resolution No. 3228. If no action were taken, the conflicting needs of drivers wishing to access adjacent properties and those wishing to progress through the corridor would not be addressed. In the short term, this would perpetuate the ingress/egress conditions that are experienced today in areas of Red Lodge and Roberts. In the longer term, conflicts between turning vehicles and through-traveling vehicles may be expected to develop in other areas adjacent to the corridor that become more urbanized (such as northern Red Lodge). Access to the Red Lodge Fire Station and proposed Beartooth Hospital could be compromised as traffic congestion increases.

Alternative B (Preferred): The Preferred Alternative would improve the ability of the roadway to provide for both access and progression. In Red Lodge, the Preferred Alternative would be consistent with the Access Management Plan, developed at the request of the City of Red Lodge, which would manage access between MT Highway 78 and Two Mile Bridge Road. Reasonable access would be maintained for adjacent properties throughout the project area while minimizing traffic operational conflicts. Additional downtown parking would be added along Oakes Avenue to serve the downtown Arts Guild and Carnegie Library. Access to the Red Lodge Fire Station would be enhanced with a roundabout at MT Highway 78, which would minimize long vehicle queues in front of the Fire Station driveways. Access to the new Beartooth Hospital would also be accommodated, allowing for efficient emergency vehicle access approaching the hospital from both the north and south directions on US Highway 212. In Roberts, the addition of a TWLTL would improve traffic flow and safety conditions for those wishing to access adjacent properties or progress along the corridor. Proposed modifications at key intersections, including Cooney Dam Road in Roberts and Boyd, would also result in improved access for adjacent properties.

Mitigation for Access Impacts

The Preferred Alternative would maintain reasonable access for adjacent property owners while managing access and vehicular progression; therefore, no mitigation is required.

3.5.3 Safety

MDT is concerned about traffic safety along the project corridor. Between January 1, 1992 and December 31, 2006 a total of 441 crashes were recorded; there were 132 injuries and 10 fatalities. *See Table 3.4, Crash Summary.* The project corridor as a whole has a higher than average crash history; with an all-vehicle crash rate of 1.93 crashes/million vehicle miles,

compared to the statewide average of 1.40 crashes/million vehicle miles. There are numerous intersections with geometric deficiencies, including a lack of turn lanes, undefined access within the functional area of the intersection, and highly skewed intersections. The public has also voiced safety concerns with the narrow shoulder widths. The roadway ditches are shallow, narrow, and steep; resulting in a lack of storage volume for rain or snow following precipitation events and increasing safety concerns for vehicles that drive off the road. Thick brush and trees within the clear zone obscure visibility of approaching wildlife as well as create safety concerns for errant vehicles.

Table 3.4 Crash Summary				
Type of Crash	Number of Crashes			
Involved "wild" animals	153			
Rear-end	29			
Sideswipe	18			
Involved left turns	8			
Involved right angles	22			
Head on	5			
Other	55			
Not coded	151			

In Red Lodge, City officials have raised concerns about safety and accessibility at the Fire Station. The Beartooth Hospital is also proposed to be constructed along the project corridor. Additional public and City comments have been received concerning a lack of pedestrian and bicycle facilities in Red Lodge.

In Roberts, many social activities center on Roberts School. The School District and general public have raised safety concerns stemming from the skewed and offset intersection of Cooney Dam Road and East Maple Street and uncontrolled access at the Y-Stop. The School District has requested installation of guardrail adjacent to the school property to provide improved protection of the children using the school playground. The School District has also requested bus turnarounds at two locations to facilitate bus turning movements and improve traffic safety conditions. Two years ago, the community of Roberts installed new sidewalks east of and parallel to the project corridor, to provide a safe location for pedestrians traveling to and from school off the highway.

Between Roberts and Boyd, the intersection of the rest area accesses and Clear Creek Road form an offset intersection. Additionally, on Cooney Dam Road in Boyd, westbound traffic from the north and south forks must merge, creating a conflict point.

Impacts to Safety

Alternative A (No-Build): The no-build alternative would leave the existing road in its present configuration. With no improvements, the number of crashes and existing crash rates are anticipated to increase as traffic continues to increase. Public concerns about safety conditions, particularly in the urban areas of Red Lodge and Roberts, would not be addressed. Increasing congestion at the intersection of MT Highway 78 adjacent to the Red Lodge Fire Station may impact emergency vehicle response time.

Alternative B (Preferred): The Preferred Alternative would provide numerous safety improvements along the project corridor. These include intersection realignments to eliminate skewed and offset intersections, the addition of turn lanes where needed, access management in Red Lodge, wider shoulders, flatter ditch slopes, clearing of thick brush and trees within the clear zone, provision of pedestrian and bicycle facilities in Red Lodge and Roberts, and provision of bus turnarounds for the Roberts school buses. These improvements are anticipated to result in a reduction in the number and severity of crashes as well as crash rates. The proposed roundabout at the intersection of MT Highway 78 adjacent to the Red Lodge Fire Station would reduce traffic congestion impacts on emergency vehicle response time. (Please note that animal-vehicle crashes are discussed further in Section 3.11.5, Animal-Vehicle Collisions.)

Mitigation for Safety Impacts

The Preferred Alternative would be consistent with MDT design guidelines for roadway safety features, therefore no mitigation of safety features are required.

3.5.4 Pedestrian/Bicycle Facilities

The Bicycle Transportation and Pedestrian Walkways provisions of Section 217 of Title 23 U.S.C. was amended by the 2005 SAFETEA-LU (Safe, Accountable, Flexible, Efficient Transportation Equity Act: A Legacy for Users) Section 1954. This states that bicycle transportation facilities and pedestrian walkways shall be considered, where appropriate, in conjunction with all new construction and reconstruction of transportation facilities, except where bicycle and pedestrian use are not permitted; and that transportation plans and projects shall provide due consideration for safety and contiguous routes for bicyclists and pedestrians.

Currently, there are no pedestrian and bicyclist facilities along the project corridor. The existing roadway shoulders are approximately 2 feet (0.6 meters) wide, which is below the AASHTO recommended minimum usable shoulder width (i.e., clear of rumble strips) of 4 feet (1.2 meters) for accommodating bicyclists or pedestrians along the road.

US Highway 212 is used by pedestrians and bicyclists in Red Lodge between 8th Street and Two Mile Bridge Road. The City of Red Lodge developed a trail plan in May 2006, which identified the need for sidewalk and shared bike/ped path facilities along US Highway 212 within Red Lodge. In addition, numerous public comments were received requesting pedestrian and bicycle facilities in Red Lodge. *See Figure 3-1, City of Red Lodge Existing and Planned Trails.*

In 2005, Roberts used CTEP (Community Transportation Enhancement Program) funding to construct sidewalks one-block east and parallel to US Highway 212 on First Street. The purpose of the project was to improve safety and accessibility for children walking to school by keeping them safely off of US Highway 212.

Impacts to Pedestrian/Bicycle Facilities

Alternative A (No-Build): The no-build alternative would leave the existing road in its present configuration. No additional provisions for accommodating pedestrians and bicycles would be made. The majority of the roadway would essentially remain inaccessible to pedestrians and bicycles within the communities of Red Lodge and Roberts. The rural portions of the roadway would continue to have inadequate roadway shoulders for accommodating rural pedestrians and bicyclists along the roadway.

Alternative B (Preferred): The Preferred Alternative would be consistent with the local pedestrian/bicycle plan in Red Lodge by providing a shared bike/ped path from MT Highway 78 to Two Mile Bridge Road. The Preferred Alternative would also provide ADA accessible pedestrian facilities within the corporate limits of Red Lodge. See Figure 3-2, Preferred Alternative Consistency with City of Red Lodge Comprehensive Trails Plan. Additional ADA accessible pedestrian facilities and improved school crossings would be provided in Roberts. See Figure 3-3, Roberts Pedestrian/Bicycle Facilities. The rural areas would include an eightfoot wide roadway shoulder to accommodate pedestrians and bicyclists along the roadway.

Mitigation for Pedestrian/Bicycle Facility Impacts

The Preferred Alternative would provide for planned and unmet pedestrian and bicycle accommodations; therefore, no mitigation is required.



Figure 3-1, City of Red Lodge Existing and Planned Trails

*Note: The trails shown in this figure are based on the City of Red Lodge Comprehensive Trails Plan (2006).



Figure 3-2, Preferred Alternative Consistency with City of Red Lodge Comprehensive Trails Plan

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.



Figure 3-3, Roberts Pedestrian/Bicycle Facilities

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

3.6 TRAFFIC NOISE

A traffic noise analysis was conducted to determine the existing traffic noise levels and the projected future traffic noise levels at proposed passing lane locations in the rural segments of the project corridor: Red Lodge to Roberts and Roberts to Boyd. The noise analysis did not include the urban segments of the project corridor (Red Lodge and Roberts), since no additional passing lanes were proposed within those segments.

Two-mile long passing zones (Passing Zones 1 and 3) were analyzed between Red Lodge and Roberts and between Roberts and Boyd to allow for flexibility during the design phase. However, the final length of each passing lane would be approximately one-mile. Additionally, Passing Zone 2 was analyzed between Roberts and Boyd for the northbound passing lane. Due to geometric constraints, a two-mile passing zone could not be analyzed in this location; instead a one-mile passing zone was evaluated. These passing zones are located at the following areas:

- Red Lodge to Roberts—Passing Zone 1 was identified as a two-mile segment north of Two Mile Bridge Road on the east side of the roadway
- Roberts to Boyd—Passing Zone 2 was identified as a one-mile segment north of Roberts on the east side of the roadway
- Roberts to Boyd—Passing Zone 3 was identified as a two-mile segment south of Boyd on the west side of the roadway

See Figure 3-4, Passing Zones.

The analysis was conducted in accordance with 23 CFR Part 772, Procedures for Abatement of Highway Traffic Noise and Construction Noise, and MDT's Traffic Noise Analysis and Abatement: Policy and Procedure Manual (2001).

Traffic noise impacts can occur under two separate conditions: (1) when noise levels are unacceptably high (absolute level); or (2) when noise levels would substantially increase by the project design year. MDT thresholds for absolute noise levels vary with land use, *as shown in Table 3.5, Noise Abatement Criteria, Exterior Noise Levels.* The MDT threshold for a substantial noise increase is 13 dB.



Figure 3-4, Passing Zones

*Note: This is a conceptual figure based on the preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

Table 3.5 Noise Abatement Criteria, Exterior Noise Levels				
Land Use Category	Description	Absolute Level Threshold (dB)	Found in Project Area	
A	Lands on which serenity and quiet are of extraordinary significance. These lands serve an important public need and the preservation of these qualities is essential if the area is to continue to serve its intended purpose.	56	No	
В	Picnic areas, recreation areas, playgrounds, active sports areas, parks, residences, motels, hotels, schools, churches, libraries, and hospitals	66	Yes	
С	Developed lands, properties, or activities not includes in Categories A or B above	71	Yes	
D	Undeveloped lands	_	Yes	

Currently, the majority of the land adjacent to the passing zones is undeveloped; however, there is some residential and commercial development.

Traffic Noise Impacts

Alternative A (No-Build): Currently, one property is at MDT's noise threshold, meaning it is already experiencing noise impacts. Traffic is expected to increase over time due to increased development in the area. By 2030, traffic noise would increase by approximately 2 dB. This has the potential to result in noise levels approximately at, or slightly above, the MDT's noise threshold (65 dB) at five residential properties, including the property currently experiencing noise impacts. See Table 3.6, Existing and Future Noise Levels for Impacted Properties.

Table 3.6 Existing and Future Noise Levels for Impacted Properties					
Location	Residence	Location	Existing Noise Levels (dB)	2030 Noise Levels (dB)	
	1	109°14' 20.833" N / 45° 13' 24.669" W ³	64	66	
Red Lodge to Roberts	2	7242 Highway 212 Roberts, MT	64	66	
	3	7367 Highway 212 Roberts, MT	64	66	
	4	#4 Broken Spoke Roberts, MT	66	68	
Roberts to Boyd	5	109° 4' 57.41" N / 45° 26' 46.807" W	64	66	

Alternative B (Preferred): Under the Preferred Alternative, the five residential properties, as identified in Table 3.6, would potentially experience 2030 noise levels similar to the no-build alternative.

³ Latitude and longitude was provided for properties that did not have a designated physical address.

Mitigation for Traffic Noise Impacts

Under both alternatives, five residential properties would experience noise levels at or slightly above the absolute threshold by 2030. Potential mitigation measures could include items such as speed limit reductions, horizontal or vertical alignment shifts, construction of noise barriers, or property acquisition for buffer zones. However, noise mitigating measures such as these are not reasonable and feasible at this time. MDT will provide Carbon County a Traffic Noise Report that includes recommended setbacks for future development along the corridor to assist the County in their efforts of minimizing potential noise impacts.

3.7 RIGHT-OF-WAY AND RELOCATIONS

The Preferred Alternative may result in relocations and property acquisition. However, the "Uniform Act", or Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970 (42 U.S.C. 4601 et seq.), and amendments, provides protections and assistance for people affected by such federally funded projects.

The location and width of existing right-of-way and easements varies greatly throughout the project corridor. Along the corridor there are approximately 148.7 acres (60.2 hectares) of MDT right-of-way. Additionally, there are 45.6 acres (18.5 hectares) of easements along the corridor that are being used by MDT for roadway purposes.

Right-of-Way and Relocation Impacts

Alternative A (No-Build): If no action were taken, right-of-way would not need to be acquired, and there would be no property acquisitions and/or relocations.

Alternative B (Preferred): The Preferred Alternative would require the acquisition of right-of-way in fee to achieve the standard right-of-way width of 80 feet (25 meters). In addition, the existing MDT easements would be converted to fee right-of-way. Alternative B would require the acquisition of approximately 317.2 acres (128.4 hectares) of right-of-way, including the conversion of existing easements currently being used for roadway purposes.

Additionally, the Preferred Alternative may require the acquisition and/or relocation of structures on up to nine properties, including six dwellings, two outbuildings, and one commercial building. This analysis is based on preliminary (approximately 30 percent) design that is available at this early stage of the design process. As the design process continues and as additional avoidance, minimization, and mitigation strategies are evaluated, potential impacts may change slightly.

Mitigation for Right-of-Way and Relocation Impacts

MDT will comply with the "Uniform Act", or Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970 (42 U.S.C. 4601 et seq.), and amendments. MDT will also attempt to meet individually with affected property owners to discuss potential impacts. MDT will make reasonable efforts to avoid and/or minimize impacts to potentially affected property owners.

3.8 WATER RESOURCES/QUALITY

The Federal Water Pollution Control Act, as amended by the Clean Water Act of 1977 (33 U.S.C. 1251 et seq.), provides the authority to establish water quality standards, control discharge into surface and subsurface waters, develop waste treatment management plans and practices, and issue permits for discharges (Section 402) and for dredged or fill material (Section 404).

3.8.1 Surface Water

There are several permanent or semi-permanent water bodies along the project corridor. They are as follows:

- Rock Creek: Rock Creek parallels US Highway 212 over the entire project length, at distances ranging from 0 (at the crossing) to 1.5 miles (0.8 kilometers). South of the crossing, Rock Creek is located east of the highway. North of the crossing, the creek is on the west side of the highway. Rock Creek is listed on Montana's 2006 list of "Impaired and Threatened Water Bodies in Need of Water Quality Restoration." Probable causes of impairment include low flow alteration, while probable sources of impairment include flow alteration from water diversions and irrigated crop production.
- Stanley Creek: US Highway 212 crosses Stanley Creek along the proposed project corridor. Downstream of the crossing, Stanley Creek becomes the Carbonado Ditch. Carbonado Ditch terminates several miles below this crossing, prior to reaching Rock Creek. Stanley Creek is not listed on Montana's 2006 list of "Impaired and Threatened Water Bodies in Need of Water Quality Restoration."
- Irrigation Ditches: Numerous irrigation ditches cross the highway through culverts. These ditches are sourced by Rock Creek; however, most terminate before returning to Rock Creek or other streams.

In addition, there are approximately 26 intermittent or ephemeral surface water crossings along the project corridor.

Impacts to Surface Water

Alternative A (No-Build): If no action were taken, there would be no new impacts to surface water.

Alternative B (Preferred): Impacts to water quality may result from culvert replacement or extension; ditch realignment; dredge/fill activities in wetlands; the relocation of irrigation ditches outside of the proposed right-of-way, as discussed further in Section 3.8.2; and new storm water outfall locations at Rock Creek.

Mitigation for Surface Water Impacts

Impacts to surface water would be minimized with the implementation of BMPs (Best Management Practices) during construction. Construction activities would be in compliance with applicable permits and regulations, specifically Section 402 and 404 of the CWA (Clean Water Act) and the Montana Stream Protection Act. Mitigation measures for impacts to irrigation

facilities and wetlands are discussed further in Section 3.8.2, Irrigation Facilities, and Section 3.10, Wetlands.

3.8.2 Irrigation Facilities

An Irrigation Report was prepared for the Red Lodge North corridor (Kadrmas, Lee & Jackson, Inc., 2003). The following information was summarized from that report.

Twenty-seven irrigation ditches cross US Highway 212 throughout the corridor. Nineteen of the ditches have decreed irrigation rights, while the other eight have appropriated rights, which are not controlled by the water commissioner. Consolidated Ditch, Finn Ditch, and Highline Ditch are incorporated ditches. (Please note that the historic nature of the irrigation ditches are discussed further in Chapter 4)

Rock Creek is the source for all of the irrigation ditches within the project. Two drainages are also used as irrigation sources, Stanley Creek and an unnamed drainage opposite of the rest area near Clear Creek Road. According to United States Geological Survey streamflow data, the average annual peak flow rate of Rock Creek four miles south of Red Lodge is 1,285 cfs (cubic feet per second), while the average annual peak flow rate of Rock Creek at Joliet is 1,343 cfs. During the peak of irrigation season, water right holders may draw as much as 450 cfs from Rock Creek.

Impacts to Irrigation Facilities

Alternative A (No-Build): If no action were taken, irrigation facilities would not be impacted.

Alternative B (Preferred): The Preferred Alternative would relocate irrigation ditches as necessary in consultation with owners to minimize impacts. As appropriate, removal of ditches would be done during construction of the new roadway and would include removal of concrete head gates, pipes, and structures. New facilities would be located outside the proposed right-of-way. Additionally, the water rights holder at Mullaney Spring has expressed concern over potential impacts to the spring, which irrigates his tree farm. Preliminary assessments show that Mullaney Spring may be impacted; however, this will be further evaluated during project design. *See Table 3.7, Impacts to Irrigation Facilities.*

Table 3.7 Impacts to Irrigation Facilities			
Irrigation Ditch	Location	Description	Impact
Brewery Ditch	Red Lodge	Serves 4 users and has a flow of 1.99 cfs; irrigates approximately 23.3 acres (9.4 hectares) of hay land.	No impacts anticipated.
Vincent Ditch	Red Lodge to Roberts	Serves 3 users and has a flow of 2.13 cfs; irrigates approximately 92.7 acres (37.5 hectares) of hay land. Part of the ditch flows inside the west fence line.	Section along west fence line may be moved beyond right-of-way.

Table 3.7 Impacts to Irrigation Facilities				
Irrigation Ditch	Location	Description	Impact	
Taylor Ditch	Red Lodge to Roberts	The Rock Creek Water Users' Association regulates this ditch. Irrigation water is purchased from the Rock Creek Drainage. Water is released from the Cooney Reservoir to match what was taken from the upper Rock Creek. Part of the ditch flows inside the east fence line.	Section along east fence line may be moved beyond right-of-way.	
Joki Ditch	Red Lodge to Roberts	Serves 14 users and has a flow of 8.44 cfs; irrigates approximately 397.3 acres (160.9 hectares) of hay land.	No impacts anticipated.	
Consolidated Ditch Company	Red Lodge to Roberts	Serves 27 users and the initial capacity of the main canal is 76.79 cfs. In 1965 it served approximately 2,171 acres (879.3 hectares) of hay land with the potential to irrigate 59 additional acres (23.9 hectares).	No impacts anticipated.	
Wallis Ditch	Red Lodge to Roberts	Serves 19 users and has a flow of 3.70 cfs; irrigates approximately 83.2 acres (33.7 hectares) of hay land. Part of the ditch flows just outside the west fence line.	Section flowing just beyond west fence line may be moved beyond right-of-way.	
Kivikangas Ditch	Red Lodge to Roberts	Services 8 users and has a flow of 16.75 cfs; irrigations approximately 661.1 acres (267.7 hectares) of hay land.	No impacts anticipated.	
Curry Ditch	Red Lodge to Roberts	Services 2 users and has a flow of 0.35 cfs; irrigates approximately 22.2 acres (9.0 hectares) of hay land.	No impacts anticipated.	
Price Ditch	Red Lodge to Roberts	Services 4 users and has a flow of 8.38 cfs; irrigates approximately 352.5 acres (142.8 hectares) of hay land. Part of the ditch flows near each fence line.	Sections of the ditch flowing near the fence lines may be moved beyond right-of-way.	
Hunter-Russet Ditch	Red Lodge to Roberts	Services 7 users and has a flow of 18.99 cfs; irrigates approximately 748.9 acres (303.3 hectares) of hay land. Part of the ditch flows just outside the west fence line.	Section of ditch just outside west fence line may be moved beyond right-of-way.	
Price Ditch (Lower Branch)	Red Lodge to Roberts	Lower branch serves one user and has a flow of 1.19 cfs; irrigates approximately 31.49 acres (12.8 hectares) of hay land.	No impacts anticipated.	
Finn Ditch Company	Red Lodge to Roberts	Serves 15 users and capacity of the main ditch is 83.53 cfs with a length of 9 miles (14.5 kilometers); irrigates approximately 2,682.3 acres (1,086.3 hectares) of hay land.	No impacts anticipated.	
McKenzie-Allen Ditch	Red Lodge to Roberts	Serves 5 users and has a flow of 16.58 cfs; irrigates approximately 777.1 acres (314.7 hectares) of hay land.	No impacts anticipated.	

Continued...

Table 3.7 Impacts to Irrigation Facilities				
Irrigation Ditch	Location	Description	Impact	
Mullaney Spring	Red Lodge to Roberts	Patrick Mullaney has water rights to a spring that surfaces in the highway ditch. This spring serves one user and has a flow of 0.14 cfs. It serves 3.7 acres (1.5 hectares) of a tree farm. An underdrain system is currently in place. This water flows through a gravel pit that is 147.6 feet (45 meters) long by 49.2 feet (15 meters) wide by 3.9 feet (1.2 meters) deep.	No impacts anticipated.	
Hill-Hopkins Ditch	Red Lodge to Roberts	Serves 2 users and has a flow of 7.25 cfs; irrigates approximately 290 acres (117.5 hectares) of hay land.	No impacts anticipated.	
Barry-Pitts Ditch	Red Lodge to Roberts	Serves 4 users and has a flow of 6.86 cfs; irrigates approximately 310 acres (125.6 hectares) of hay land.	No impacts anticipated.	
Hill Brothers Ditch	Red Lodge to Roberts	Serves 6 users and has a flow of 11.26 cfs; irrigates approximately 440 acres (178.2 hectares) of hay land.	No impacts anticipated.	
Rule-Thompson (Glantz-Schanck) Ditch Upper Branch	Roberts	Serves 2 users and has a flow of 6.38 cfs; irrigates approximately 600 acres (243 hectares) of hay land.	No impacts anticipated.	
Rule-Thompson (Glantz-Schanck) Ditch Lower Branch	Roberts	Serves 5 users and has a flow of 12.04 cfs; irrigates approximately 885 acres (358.4 hectares) of hay land.	No impacts anticipated.	
Duncan-Aiken Ditch	Roberts to Boyd	Serves 7 users and has a flow of 8.36 cfs; irrigates approximately 492 acres (199.3 hectares) of hay land.	No impacts anticipated.	
Hunt Ditch (Upper and Lower Branches)	Roberts to Boyd	Serves one user and has a flow of 7.00 cfs; irrigates approximately 324 acres (131.2 hectares) of hay land. Water flows in the right- of-way east of both highway culverts.	Channel changes may be required at both locations.	
Highline Ditch	Roberts to Boyd	Serves 14 users and has a flow of 32.51 cfs; irrigates approximately 1,219.4 acres (493.9 hectares) of hay land. Part of the ditch flows inside the east right-of-way.	Section may be moved beyond right-of-way.	
Rooney Ditch	Roberts to Boyd	Serves 5 users and has a flow of 10.83 cfs; irrigates approximately 558 acres (226 hectares) of hay land. Part of the ditch flows inside the east right-of-way.	Section may be moved beyond right-of-way.	
Drake Ditch	Roberts to Boyd	Serves 4 users and has a flow of 11.51 cfs; irrigates approximately 386.5 acres (156.5 hectares) of hay land. Part of the ditch flows inside the east right-of-way and parts flow just inside of the west fence line.	Section may be moved beyond right-of-way.	

Table 3.7 Impacts to Irrigation Facilities				
Irrigation Ditch	Location	Description	Impact	
Stanley Creek	Roberts to Boyd	This crossing is a natural drainage basin that people use for irrigation. This ditch serves ten users and has a flow of 14.92 cfs. It irrigates 472.2 acres of hay land. Part of the drainage channel flows inside the east right-of-way.	Section may be moved beyond right-of-way.	
Ward Ditch	Roberts to Boyd	Serves 4 users and has a flow of 4.40 cfs; irrigates approximately 93.0 acres (37.7 hectares) of hay land. There are two parallel approaches west of the crossing that may be combined into one approach.	If approaches are combined, changes to the channel may be necessary.	
Beerwart Ditch	Roberts to Boyd	Serves 3 users and has a flow of 6.98 cfs; irrigates approximately 307.9 acres (124.7 hectares) of hay land. Part of the ditch flows inside the west right-of-way, paralleling the roadway.	Section may be moved beyond right-of-way.	
#48 Carbonado Ditch	Roberts to Boyd	Serves 48 users and has a flow of 46.58 cfs; irrigates approximately 2,139.1 acres (866.3 hectares) of hay land.	No impacts anticipated.	
Carbonado – Hoyle Ditch	Roberts to Boyd	Serves 9 users and has a flow of 26.28 cfs; irrigates approximately 929.3 acres (376.4 hectares) of hay land. Parallels the roadway but does not cross it. Part of the ditch flows inside the west right-of-way.	May be moved beyond west right-of-way, if feasible.	

Mitigation for Irrigation Facility Impacts

Construction in the vicinity of Mullaney Spring will require care to avoid altering the flow rate to the water rights holder. Consultation with affected ditch associations and other landowners/ water rights holders will take place to minimize impacts to irrigation facilities. BMPs will be implemented as needed.

3.8.3 Ground Water

There are two aquifers within the project corridor; one spans from Red Lodge to approximately 2 miles (3.2 kilometers) south of Boyd and the other from approximately 2 miles (3.2 kilometers) south of Boyd for the remainder of the corridor. These aquifers are part of the Northern Great Plains Aquifer system. Additionally, a preliminary assessment has indicated that there are approximately 67 domestic wells within 100 feet of the project corridor. These wells range from a maximum depth of approximately 220 feet (67.1 meters) to a minimum depth of 11 feet (3.4 meters), averaging a depth of approximately 38.6 feet (11.8 meters).

Impacts to Ground Water

Alternative A (No-Build): If no action were taken, there would be no new impacts to ground water.
Alternative B (Preferred): Impacts to ground water resources are not anticipated as a result of the Preferred Alternative. However, the Preferred Alternative may require the relocation of domestic wells within the proposed right-of-way.

Mitigation for Ground Water

If domestic wells are displaced by the proposed project, domestic water would be restored to the affected properties. The manner in which this activity would be accomplished would be determined on a case-by-case basis.

3.8.4 Public Water Supplies

According to the MDEQ Source Water Assessment website, there are 34 PWS (public water supplies) in Carbon County, of which six are near the project corridor. These include the City of Red Lodge (MT0000314), Red Lodge KOA (MT0002030), Round Barn Restaurant (MT0002506), Crystal Springs Ice and Water (MT0003429), community of Roberts (MT0000317), and MDT's Rest Area north of Roberts (MT0001970). Following is a description of each system.

- The wells and surface water intake for the City of Red Lodge are located south of Red Lodge and are not adjacent to this project.
- The wells for the Red Lodge KOA are located east of the highway with the closest being over 1,000 feet from the highway. Additionally, the highway is not within the 100-foot radius control zone of either well for the Red Lodge KOA.
- The well for the Round Barn Restaurant is located approximately 200 feet southwest of the restaurant building. MDEQ's Source Water Assessment indicates that this water supply is not currently active.
- Limited information is available regarding the Crystal Springs Ice and Water other than the source of the water is a spring.
- The wells for the community of Roberts and the well for the Roberts Rest Area are each located approximately 150 feet from the roadway centerline. However, the 100-foot control zone for these wells does not include the roadway, but would include a portion of the highway right-of-way.

MDEQ has completed Source Water Assessment Plans for the Red Lodge, Round Barn Restaurant, Red Lodge KOA, and Rest Area systems. The majority of the plans indicated no, or a low, potential hazard posed by US Highway 212. However, the Red Lodge KOA plan indicated a moderate potential hazard from the highway. The plan for the community of Roberts has been prepared but not finalized. The draft plan indicates a low to moderate potential hazard posed by the highway, due to the potential for a spill. No plan has been developed for the Crystal Springs Ice and Water System.

Impacts to Public Water Supplies

Alternative A (No-Build): If no action were taken, there would be no new impacts to PWS.

Alternative B (Preferred): The Preferred Alternative would not impact the existing PWS and would not alter any of the assessments.

Mitigation for Public Water Supply Impacts

No impacts to PWS are anticipated as a result of the Preferred Alternative; therefore, mitigation is not required.

3.8.5 Waste Water Systems

One mound septic system has been identified within the project corridor. Additionally, a storm drain pipe has been identified along MT Highway 78. The City of Red Lodge has connected to this pipe and extended it further east to a discharge point.

Impacts to Waste Water Systems

Alternative A (No-Build): If no action were taken, there would be no new impacts to waste water systems.

Alternative B (Preferred): The existing mound system has been determined to be within the proposed right-of-way as well as the proposed construction limits associated with the Preferred Alternative. The County Sanitarian has indicated that the current mound system location is the only location that has been approved by the MDEQ; however, it would be possible to obtain MDEQ approval to relocate a portion of the system further to the west. As a result, it is anticipated that the proposed project may require the relocation of a portion of the mound system, but not a relocation of the residence.

Additionally, a new storm drain pipe and outfall may need to be constructed or the existing storm drain pipe (at the intersection of MT Highway 78 and US Highway 212) may need to be replaced, which would be determined during design. Construction of a new storm drain pipe may result in replacement of the existing storm drain and may also result in minor modifications to the existing water or waste water piping systems within Red Lodge.

Mitigation for Waste Water Systems Impacts

If the mound system were impacted, MDT would relocate the system per County and MDEQ requirements. Mitigation for the potential impacts to the existing storm drain pipe would be determined during design.

3.9 WATER BODY MODIFICATIONS

There are presently a total of 55 water crossings within the project limits. Just south of the Roberts rest area the roadway crosses Rock Creek at the Rock Creek Bridge, which was constructed in 2001. Approximately two miles south of Boyd, the roadway crosses Stanley Creek with a 10-foot by 5-foot concrete box culvert. Additionally, there are 26 other locations where the roadway crosses surface water drainages. There are also 27 locations where the roadway crosses irrigation ditches. The existing culverts generally extend from the toe of the roadway slope on one side to the toe of the roadway slope on the other side.

Water Body Modification Impacts

Alternative A (No-Build): If no action were taken, there would be no new water body modifications.

Alternative B (Preferred): No modifications or impacts to the Rock Creek Bridge or Rock Creek are anticipated at the bridge crossing. The Preferred Alternative consists of a new culvert installation at the Stanley Creek crossing. Minor inlet and outlet ditches may be required for installation of the new culvert at this location. In addition, where existing irrigation ditches are within the proposed right-of-way, MDT's standard practice is to move the irrigation ditch outside of the right-of-way.

The Preferred Alternative would consist of a wider roadway and new culverts. Minor inlet and outlet ditches may be required for installation of new culverts. The new culverts would extend the entire width of the new right-of-way and would generally be longer than the existing culverts. The longer culverts would connect to the existing ditches in their current locations and generally would not result in the need to realign ditches.

Mitigation for Water Body Modification Impacts

Drainage structures would be designed to address hydrologic conditions and comply with federal and state regulations including the Montana Stream Protection Act, Federal Clean Water Act, and Section 404/401, as applicable. Irrigation facilities would be designed in consultation with ditch owners and operators to minimize impacts to farming/ranching operations.

3.10 WETLANDS

Wetlands are defined both in the 1977 Executive Order 11990, Protection of Wetlands, and in Section 404 of the Clean Water Act of 1986, as those areas that are inundated by surface or groundwater with a frequency to support and under normal circumstances do or would support a prevalence of vegetative or aquatic life that requires saturated or seasonally saturated soil conditions for growth and reproduction. Three parameters that define a wetland, as outlined in the Federal Manual for Delineating Jurisdictional Wetlands (US Army Corps of Engineers, 1987), are hydric soils, hydrophytic vegetation, and hydrology. Wetlands generally include swamps, marshes, bogs, and similar areas. Wetlands are important natural resources that often serve many functions, such as providing habitat for wildlife, storing floodwaters, recharging groundwater, and improving water quality through purification.

The Biological Resources Report⁴, which is on file at MDT, provides a detailed account of wetland resources and impacts within the study area. The information is summarized below.

Wetland delineations were conducted in 2004 and 2007, resulting in a total of 98 wetland sites identified along the project corridor. Wetlands along the proposed project are generally comprised of emergent communities, including roadside ditches, sub-irrigated pastures, fringes of irrigation ditches or canals, and streams (Rock Creek and Stanley Creek). Hydrology at a large percentage of the wetlands appears to be related to irrigation ditch flows or groundwater influenced by the presence of irrigation ditches adjacent to the wetland. Wetland soils generally consist of silty or sandy clay loams.

⁴ Land & Water Consulting, Inc., *Red Lodge-North Final Biological Resources Report* (Apr. 2004, Supplement 2007)

The majority of wetlands were rated as Category III and IV sites using the 1999 MDT Montana Wetland Assessment Method. Three sites (64, 68, and 68a) were rated as Category II sites and received high ratings in the General Wildlife Habitat variable. All existing sites were considered highly disturbed in the immediate project area due to road and right-of-way maintenance activities, proximity to roads and residential sources of disturbance, and physical encroachment resulting from access roads, bridges, and culverts. All of the wetland sites scored high in the Nutrient, Toxicant. Removal Sediment/Shoreline Stabilization variables as a result of the borrow ditch location and/or irrigation ditch components of the wetlands.

How are wetlands categorized?

MDT assesses functions and values of a wetland based on a point scale. The points for each function are added up and the sum determines which category of wetland it is. The larger the total point value, the higher value and quality a wetland is. Category I is the highest overall ranking a wetland can receive, followed by Category II, Category III, and Category IV.

On May 6, 2003, representatives from MDT, USACE (US Army Corps of Engineers), KL&J (Kadrmas, Lee & Jackson, Inc.), and PBS&J/Land and Water Consulting, Inc., conducted a field review and USACE provided preliminary jurisdictional determinations of wetlands along the project corridor. Preliminary jurisdictional determinations were not conducted for the 2007 wetland delineations. Coordination with USACE will continue throughout design of the proposed project.

Impacts to Wetlands

Alternative A (No-Build): If no action were taken, no new impacts to wetlands would occur.

Alternative B (Preferred): The Preferred Alternative would result in an estimated 40.7 acres (16.5 hectares) of wetland impacts. Of this, approximately 24.8 acres (9.7 hectares) are considered jurisdictional wetlands based on preliminary determinations⁵. See Table 3.8, *Estimated Wetland Impacts.*

There are no feasible alternatives that would completely avoid wetland impacts. Many of the impacts would occur at sites immediately adjacent to the road, often on both sides, and would be largely unavoidable even if the roadway centerline were shifted in either direction. However, minimization efforts have been included in preliminary project design; the Preferred Alternative was modified at the Cooney Dam Road intersection in Boyd to avoid impacts to a meandering stream channel.

⁵ In light of the *Rapanos vs. United States* decision in 2006 and the subsequent EPA and USACE joint guidance regarding the outcome of *Rapanos* in 2007, the initial jurisdictional determinations may change during the permitting stage of the project.

Table 3.8 Estimated Wetland Impacts					
Site	MDT Wetland Rating Category	Likely Jurisdictional ⁶	Estimated Impacts (Acres [Hectares])	Source of Wetland Hydrology	
1	IV		0.3 (0.1)	Irrigation and storm water	
2	III	\checkmark	0.4 (0.2)	Ground water	
3	IV	✓	0.4 (0.2)	Spring from the hillside on west side of roadway	
4	III	✓	< 0.1 (< 0.04)	Site 3 ditch leakage	
5	111	✓	< 0.1 (< 0.04)	Open water pond within the golf course	
6	III	✓	< 0.1 (< 0.04)	Ground water	
6a	III	✓	0.1 (0.04)	Ground water	
7	III	✓	0.3 (0.1)	Ground water	
8	111	✓	0.1 (0.04)	Ground water; open water pond on left side of road.	
9	Ш	\checkmark	0.1 (0.04)	Ground water, hillside spring and open water pond.	
10	III	\checkmark	0.5 (0.2)	Ground water and irrigation flows.	
10a		\checkmark	< 0.1 (< 0.04)	Ground water and irrigation flows.	
11	IV	\checkmark	< 0.1 (< 0.04)	Ground water	
12	IV	\checkmark	< 0.1 (< 0.04)	Ground water	
12a	IV	\checkmark	< 0.1 (< 0.04)	Irrigation ditch and hillside springs	
13			< 0.1 (< 0.04)	Ground water and irrigation source.	
14	Ш	\checkmark	0.9 (0.4)	Irrigation ditch and influenced by Site 62 hydrology.	
14a	IV		< 0.1 (< 0.04)	Irrigation ditch	
15			< 0.1 (< 0.04)	Likely storm and irrigation water	
15a	III		0.1 (0.04)	Likely storm and irrigation water	
16			< 0.1 (< 0.04)	Taylor Ditch	
17			0.4 (0.2)	Ground water	
17a	IV		< 0.1 (< 0.04)	Irrigation ditch/Rock Creek	
18	111		0.1 (0.04)	Leakage from irrigation ditch to east	
19	III		0.1 (0.04)	Irrigation ditch/Rock Creek	

Continued...

⁶ Sites 1-64 were delineated in 2004 and USACE gave preliminary jurisdictional determinations for these sites. Due to modifications of the Preferred Alternative, a wetland delineation was conducted for additional areas in 2007. Sites 65- 72a reflect the additional areas delineated. These wetlands were not included in the 2003 preliminary jurisdictional determination; therefore, PBS&J determined likely jurisdictional status. Final jurisdictional and non-jurisdictional status will be determined by USACE.

Table 3.8 Estimated Wetland Impacts					
Site	MDT Wetland Rating Category	Likely Jurisdictional ⁶	Estimated Impacts (Acres [Hectares])	Source of Wetland Hydrology	
19a	111		0.2 (0.08)	Irrigation ditches/Rock Creek	
20	III		< 0.1 (< 0.04)	Irrigation ditch/Rock Creek	
20a	IV		0.2 (0.08)	Irrigation ditch/Rock Creek	
21	III		< 0.1 (< 0.04)	Finn Ditch/Rock Creek	
21a	III		< 0.1 (< 0.04)	Finn Ditch/Rock Creek	
22	III		0.1 (0.04)	Ground water and potential surface water	
22a	III		< 0.1 (< 0.04)	Irrigation ditch/Rock Creek	
23	III	~	0.1 (0.04)	Ground water via a pipe from unknown source	
24	111		0.2 (0.08)	Irrigation ditch/Ground water	
25	III	\checkmark	1.4 (0.6)	Irrigation ditches/Ground water/Rock Creek	
25a	111	\checkmark	< 0.1 (< 0.04)	Irrigation ditches/Ground water/Rock Creek	
26	III	\checkmark	0.3 (0.1)	Ground water/Irrigation ditch	
26a	111	\checkmark	4.7 (1.9)	Ground water/Irrigation ditches	
27	IV	~	0.2 (0.08)	Irrigation ditch/Ground water/Rock Creek	
27a	IV	~	0.6 (0.2)	Irrigation ditch/Ground water/Rock Creek	
28	IV	~	0.3 (0.1)	Irrigation ditch/Ground water/Rock Creek	
28a	IV	~	< 0.1 (< 0.04)	Irrigation ditch/Ground water/Rock Creek	
29	IV		< 0.1 (< 0.04)	Irrigation ditch/Rock Creek	
30		✓	0.5 (0.1)	Ground water	
30a	III	✓	0.3 (0.1)	Rock Creek	
31	111		0.1 (0.04)	Ground water/Irrigation ditch	
32			2.2 (0.9)	Ground water/Irrigation ditch	
33	III		0.1 (0.04)	Ground water/Highline ditch	
34	III		0.6 (0.2)	Ground water/Highline ditch	
35	III		< 0.1 (< 0.04)	Ground water	
36			1.2 (0.5)	Ground water/Irrigation ditch	
37		✓	2.8 (1.1)	Ground water and ditch seeps from Rooney Ditch	
38		✓	0.1 (0.04)	Stanley Creek	
				Continued	

Table 3.8 Estimated Wetland Impacts					
Site	MDT Wetland Rating Category	Likely Jurisdictional ⁶	Estimated Impacts (Acres [Hectares])	Source of Wetland Hydrology	
38a		\checkmark	2.6 (1.1)	Irrigation ditches	
39	IV		< 0.1 (< 0.04)	Ward Ditch/Rock Creek	
39a	IV		< 0.1 (< 0.04)	Ward Ditch/Rock Creek	
40	III		0.1 (0.04)	Ward Ditch	
41	III		0.4 (0.2)	Irrigation Ditch	
41a	III	✓	0.4 (0.2)	Source unknown	
42	IV	✓	0.2 (0.08)	Source unknown	
43	III	✓	0.1 (0.04)	Ground water	
44	III		0.9 (0.4)	Ground water	
45	III	✓	0.8 (0.3)	Hoyle Ditch/Rock Creek	
46	III	✓	0.1 (0.04)	Lateral ditch/Stanley Creek	
47	IV	✓	0.1 (0.04)	Stanley Ditch/Carbonado Ditch	
47a	IV	\checkmark	1.3 (0.5)	Stanley Ditch/Carbonado Ditch	
48	III		0.4 (0.2)	Irrigation laterals	
49	III		5.0 (2.0)	Ground water/Irrigation ditch	
50	III	✓	0.6 (0.2)	Rock Creek ground water	
50a	III	✓	0.2 (0.08)	Rock Creek ground water	
51	III	✓	0.5 (0.2)	Subirrigation	
52	III		0.4 (0.2)	Irrigation ditch/Rock Creek	
53	III	✓	0.2 (0.08)	Ground water	
54	III	✓	0.5 (0.2)	Ground water	
55	III	✓	1.2 (0.5)	Ground water/Irrigation ditches	
56	III		0	Storm water	
57	III		0	Storm water	
58	III		0.1 (0.04)	Ground water	
59	III		< 0.1 (< 0.04)	Ground water	
60	IV		0.1 (0.04)	Irrigation ditch	
61	111		< 0.1 (< 0.04)	Kivikangas Ditch/Rock Creek	
61a	111		< 0.1 (< 0.04)	Kivikangas Ditch/Rock Creek	
62	III	✓	1.1 (0.4)	Ground water/Irrigation ditches	
63	III		< 0.1 (< 0.04)	Irrigation ditch	
64	II	\checkmark	2.2 (0.9)	Ground water/Hillside spring	

Continued...

Table 3.8 Estimated Wetland Impacts					
Site	MDT Wetland Rating Category	Likely Jurisdictional ⁶	Estimated Impacts (Acres [Hectares])	Source of Wetland Hydrology	
65	111	Unknown	0.1 (0.04)	Site 66, an irrigation ditch, may supply groundwater for the site.	
66	IV	~	0	Likely Rock Creek. Another undelineated wetland is located upslope of ditch.	
67	III	Unknown	< 0.1 (< 0.04)	Wetlands occur as a result of a man-made drain ditch.	
67a	111	Unknown	0.1 (0.04)	Wetlands occur as a result of a man-made drain ditch.	
67b	III	Unknown	< 0.1 (< 0.04)	Groundwater likely supports these wetlands.	
68	II	✓	0	Rock Creek overflow.	
68a	II	✓	< 0.1 (< 0.04)	Rock Creek overflow.	
69		✓	< 0.1 (< 0.04)	Slope wetland upslope of Site 70	
70	III	Unknown	0	Irrigation water	
71	IV	Unknown	0	Irrigation water	
71a	IV	Unknown	0	Irrigation water	
72	III	Unknown	0	Irrigation water	
72a	III	Unknown	0	Irrigation water	
TOTAL		40.7 (16.5) ⁷			

Mitigation for Wetland Impacts

Compensation for unavoidable impacts to wetlands would involve the evaluation of both on-site and off-site mitigation opportunities in an effort to develop replacement wetlands to offset impacts to these natural resources within the project corridor. On-site wetland mitigation opportunities would be evaluated in areas adjacent to the new roadway. The purpose of on-site mitigation would be to reduce permanent loss of wetland functions and values, maintain hydrologic connectivity with other existing wetlands, restore drained and degraded wetlands, and replace wildlife habitat in the region associated with impacted wetlands by the road project.

Off-site mitigation would be pursued after all on-site mitigation opportunities have been evaluated and it has been determined that additional wetland mitigation is needed to offset the project impacts. For the purpose of this project, off-site mitigation would occur at existing USACE-approved MDT Wetland Mitigation Reserves that have been established within the Watershed # 13—Upper Yellowstone River Basin. Potential off-site mitigation options currently include the MDT Stillwater, Wagner Pit, and/or DH Ranch Mitigation Reserves. At the time of

⁷ Due to the amount of individual wetland impacts that totaled less than 1/10 of an acre, the total reflects the summation of wetland impacts computed to the thousandth of an acre and then rounded to the tenth of an acre.

construction, other mitigation sites may be available. MDT will coordinate with the appropriate agencies to determine where off-site mitigation, if necessary, will be carried out.

3.11 WILDLIFE HABITAT AND ECOSYSTEMS

Pursuant to the Fish and Wildlife Coordination Act (16 U.S.C. 662), if the proposed improvements would affect water resources, then consultation with the USFWS (US Fish and Wildlife Service) and with the state agency having administrative responsibilities over wildlife resources must be initiated. This consultation is to determine the possible wildlife resources, the means and measures that should be adopted to prevent the loss of, or damage to, those resources, and to provide concurrently for the development and improvement of such resources. The Act also provides for the protection of any publicly owned wildlife or waterfowl refuge of national, state or local significance as addressed under Section 4(f) discussed in Chapter 4, as well as threatened and endangered species discussed in Section 3.13.

A Biological Resources Report was prepared for this project⁸. Methods used to obtain information regarding the wildlife and habitat conditions within the project area included coordination with the USFWS, MNHP (Montana Natural Heritage Program), and MT FWP (Montana Fish, Wildlife & Parks). Pertinent literature and information, including the Montana Rivers Information System, were reviewed, and a field survey was conducted on July 30–August 1, 2002, which provided detailed information pertaining to the ecology of the project corridor. Following is a summary of the information provided in the Biological Resources Report.

3.11.1 Vegetation

Vegetation along the project corridor predominantly consists of both upland grass and wetland species such as orchardgrass, meadow foxtail, redtop, fox-tail barley, reed canary grass, reed manna grass, Kentucky bluegrass, smooth brome, timothy, western wheatgrass, bluejoint reedgrass, yellow sweetclover, alfalfa, Canada thistle, spotted knapweed, houndstongue, bladderwort campion, showy milkweed, spreading dogbane, and western salsify. In addition to these grassland species, low shrubs such as western snowberry, big sage, rose, red-osier dogwood, sandbar willow, Pacific willow, Bebb's willow, choke cherry, river birch, aspen, and narrow-leaf cottonwood can be found scattered along the project corridor.

How are weeds categorized?

The Montana Department of Agriculture separates noxious weeds into three categories. Category I species are those that are generally widespread and established throughout the State and counties; Category II species have recently been introduced to the State or are rapidly spreading from their current site; and Category III species have not been detected in the State or are found in small, localized areas. Ten noxious weeds were identified within the project corridor: spotted knapweed, Russian knapweed, field bindweed, oxeye daisy, Canada thistle, houndstongue, leafy spurge, dalmation toadflax, sulfur cinquefoil, and tall buttercup. All of these exotic species are listed as Category I species with the exception of tall buttercup, which is listed as a Category II species. Canada thistle and houndstongue were commonly observed within the project corridor. No Category III species were identified within the project corridor.

Corridor Study – Red Lodge North Environmental Assessment STPP 28-2 (25) 70 Control No. 4375 October 2008

⁸ Land & Water Consulting, Inc., *Red Lodge-North Final Biological Resources Report* (Apr. 2004)

Impacts to Vegetation

Alternative A (No-Build): If no action were taken, there would be no new impacts to vegetation.

Alternative B (Preferred): The Preferred Alternative would remove thick brush and trees from the clear zone to improve driver visibility of approaching wildlife, as discussed further in Section 3.11.5. Additionally, vegetation would be removed in select areas along the project corridor for ditch slope flattening, shoulder widening, and pedestrian/bicycle facilities construction. The clearing of ground cover along the corridor has the potential to open areas to noxious weeds.

Mitigation for Vegetation Impacts

Clearing of vegetation would be done in accordance with MDT Standard Specifications. Noxious weeds would be controlled as discussed further in Section 3.18.4.

3.11.2 Terrestrial and Avian Species

A variety of wildlife species are found throughout the project corridor. Wildlife species in this area are typical of those that occur in grasslands, cultivated lands, and riparian areas of central Montana. Common mammals that within or near the project area include mule deer, white-tailed deer, moose, porcupine, raccoon, striped skunk, badger, coyote, thirteen-lined ground squirrel, white-tail jackrabbit, red fox, deer mouse, and meadow vole. Common amphibious and reptilian species within or near the project area include tiger salamander, western toad, woodhouse's toad, western chorus frog, northern leopard frog, spotted frog, short-horned lizard, painted turtle, rubber boa, racer, gopher snake, western rattlesnake, common garter snake, and western terrestrial garter snake. Birds observed during the Biological Resources Report survey included raven, European starling, black-billed magpie, brown-headed cowbird, and an osprey.

Impacts to Terrestrial and Avian Species

Alternative A (No-Build): If no action were taken, there would be no new impacts to terrestrial or avian species.

Alternative B (Preferred): The Preferred Alternative may result in minor fragmentation, modification, and/or loss of habitat for terrestrial and avian species. This may occur due to water body modifications, increased impervious surfaces resulting in greater runoff, and clearing of thick brush and trees within the clear zone. However, the Preferred Alternative would remain on the existing roadway alignment and the project corridor mainly consists of areas previously disturbed by human activities, such as residential and agricultural activities. Also, the clearing of vegetation in the clear zone may reduce wildlife mortality caused by animal-vehicle collisions, as discussed in Section 3.11.5. Therefore, substantial impacts to terrestrial and avian species are not anticipated.

Mitigation for Terrestrial and Avian Species Impacts

Substantial adverse impacts to terrestrial and avian species are not anticipated. However, BMPs would be implemented. Wetlands would be mitigated, as discussed in Section 3.10, disturbed areas would be reseeded and an erosion and sediment control plan would be implemented. Activities would be in compliance with the Migratory Bird Treaty Act and MDT's most current depredation permit from the USFWS, as discussed further in Section 3.18.4.

3.11.3 Aquatic Species

Fish and aquatic species can be found in Rock Creek, Stanley Creek, and several irrigation ditches, all of which are crossed by the project corridor. Fish species that may occur within the project area include brook trout, brown trout, longnose dace, mottled sculpin, mountain sucker, mountain whitefish, rainbow trout, and white sucker.

Impacts to Aquatic Species

Alternative A (No-Build): If no action were taken, there would be no new impacts to aquatic species.

Alternative B (Preferred): The Preferred Alternative may result in impacts to aquatic species via culvert replacement, highway fill placement, the increase of impervious surfaces which may cause increased runoff, and increased water temperature due to potential clearing and grubbing in riparian areas. However, no substantive losses of spawning habitat for fish species are anticipated and the increase in water temperature is expected to be negligible as most clearing and grubbing would not occur in riparian areas.

Mitigation for Aquatic Species Impacts

Impacts to aquatic species habitat would be minimized with the implementation of BMPs during construction. Unavoidable impacts would be mitigated as required by the applicable permits and regulations, specifically Section 402 and 404 of the CWA and the Montana Stream Protection Act.

3.11.4 Montana Species of Concern

Information regarding Montana Species of Concern was obtained through coordination with the MNHP (Montana Natural Heritage Program). Five species of concern have been identified within or near the project corridor: Beautiful Fleabane, Gray Wolf, Greater Sage-grouse, Milksnake, and Preble's shrew.

- Beautiful Fleabane (*Erigeron formosissimus*). The beautiful fleabane has a G5/S1 ranking, meaning globally the species is common and widespread in most of its range while within Montana it is considered at high risk because of extremely limited and/or rapidly declining numbers, range, and/or habitat, making it highly vulnerable to extinction or extirpation. It is found in meadows and forest openings in the montane and subalpine zones. It was last observed in the project area in 1919.
- Gray Wolf (Canis lupus). The gray wolf has a G4/S3 ranking, meaning globally the species is uncommon but not rare and is usually widespread; however, within Montana they are considered potentially at risk because of limited and/or declining numbers, and/or habitat. Historically gray wolves have been found in a variety of biomes such as boreal forest, temperate deciduous forest, and temperate grassland. The gray wolf was last observed in the project area in 2006.
- Greater Sage-grouse (Centrocercus urophasianus). The greater sage-grouse has a G4/S3 ranking and are typically found in sagebrush habitat. There have been no known observations of greater sage-grouse within the project area.
- Milksnake (Lampropeltis triangulum). The milk snake has a G5/S2 ranking, meaning globally it is common and often widespread while in the State it is at high risk for

extinction due to declining numbers, range, and/or habitat. Milk snakes have been most often reported in open sagebrush habitat and ponderosa pine savannah with sandy soils. There have been no known observations of milk snakes within the project area.

• **Preble's Shrew (Sorex preblei).** The preble's shrew has a G4/S3 ranking and have been observed in sageland-grassland habitats. The preble's shrew has not been observed in the project area since 1968.

Impacts to Montana Species of Concern

Alternative A (No-Build): If no action were taken, there would be no impacts to Montana Species of Concern.

Alternative B (Preferred): Many of the species of concern have either not been observed or have not been observed in recent years within the project corridor, with the exception of the gray wolf. Therefore, impacts to species of concern are not anticipated. Adverse impacts to the gray wolf are not anticipated, as discussed further in Section 3.13. However, the gray wolf would be subject to the same impacts as other terrestrial species, as discussed in Section 3.11.2.

Mitigation for Montana Species of Concern Impacts

No adverse impacts to species of concern are anticipated; therefore, no mitigation is required.

3.11.5 Animal-Vehicle Collisions

AVC (animal-vehicle collisions) are a concern along the project corridor, as evidenced by the crash history data, public comments, and comments from the MT FWP. According to MDT crash history data, there were 441 reported crashes along the project corridor during the 15-year period between January 1992 and December 2006. AVC accounted for approximately 36.9 percent of these accidents, which is nearly three times the statewide average of 13.7 percent. MDT maintenance records identified that during the 9-year period of November 1997 through December 2006, MDT maintenance staff removed 1,103 animal carcasses from the roadway along the project corridor. This indicates that many AVC along the corridor have not been reported. Previous findings support this trend; a paper published in the Wildlife Society Bulletin suggests that AVC estimates should be increased by 16-50 percent when based on accident reports⁹.

According to MDT maintenance records, approximately 98.5 percent of the AVC were with deer (94 percent white-tailed deer, 4 percent mule deer, and 0.5 percent unknown deer). The remaining AVC occurred approximately equally with elk, moose, other wild animals such as raccoons, and domestic animals such as horses and dogs.

The data were reviewed to determine whether there were any focal zones along the project corridor. A focal zone is a stretch of roadway where wildlife movement is notably concentrated and which offers distinct opportunities for implementing effective mitigation measures to improve

⁹ Romin, L.A. and J.A. Bissonnette. 1996. Deer-vehicle collisions: Status of state monitoring activities and mitigation efforts. Wildlife Society Bulletin 24 (2):276-283

highway permeability for wildlife and to reduce AVC¹⁰. No focal zones were identified; the AVC were distributed along the entire length of the project corridor.

The maintenance data were also analyzed to determine whether there was a seasonal pattern to the AVC. This analysis identified that 67 percent of these collisions occurred during the months of October through January.

Complex interactions of many factors influence the frequency of AVC on a roadway. A primary factor is wildlife population density. Some other factors include traits inherent to individual wildlife species such as mobility, food preferences, behavior, reproductive patterns and movement. Other factors may be related to wildlife habitat, such as the location of resources like water, food, cover, breeding areas, or migration routes. Additional factors apply to the road itself, including road design (width, alignment, grade, clear zone width, number of lanes), vehicle speed and traffic volume. Also affecting the frequency of AVC are factors related to driver characteristics and behaviors including vehicle type, attentiveness, and reaction time.

Impacts to Animal-Vehicle Collisions

Alternative A (No-Build): If no action were taken, the high frequency of AVC would continue and would not be addressed.

Alternative B (Preferred): The Preferred Alternative would involve removing thick brush and trees within the clear zone. This vegetation removal may discourage animals that like cover from approaching the road, improve driver visibility of approaching wildlife, and increase driver response time.

Mitigation for Animal-Vehicle Collision Impacts

MDT recognizes the high frequency of AVC along the project corridor and will make a good faith effort to incorporate appropriate mitigation measures into the project design. MDT will make decisions based on the best available research and information at the time, as well as the need to balance roadway design criteria, funding constraints, and other factors.

There are no obvious focal zones along the corridor at which mitigation measures are easily identifiable. The topography in the project area is very flat, and the groundwater levels are high; this does not lend itself well to constructing wildlife underpasses, a common mitigation measure. Further, it is anticipated that the travel patterns of deer and other wildlife have and may continue to change in response to the ongoing land use and development changes in the project area.

In addition to vegetation removal from the clear zone, MDT will evaluate the practicality of the following mitigation measures during project design¹¹:

 Wildlife-friendly fencing is a fence design that enables wildlife to leap over or crawl under the fence without injury. The most common wildlife friendly fencing design is a three to four wire fence no more than 40 inches high. The bottom wire should be smooth and at

¹⁰ Southern Rockies Ecosystem Project. 2006. Linkage Assessment Methodology, Linking Colorado's Landscapes Phase II Report. Southern Rockies Ecosystem Project. Denver, CO.

¹¹ Descriptions of the mitigation measures were taken from the Southern Rockies Ecosystem Project. 2006. Linkage Assessment Methodology, Linking Colorado's Landscapes Phase II Report. Southern Rockies Ecosystem Project. Denver, CO.

least 16 inches from the ground to allow wildlife to pass under safely. There should be at least 12 inches between the top two wires to prevent deer, elk, and moose from getting caught in the fence when they jump due to the way they kick their hind legs backwards. Although wildlife friendly fencing does not stop wildlife from crossing the roadway, it does allow them to cross quickly in and out of the right-of-way—limiting the time spent in the AVC zone. The installation of wildlife-friendly fencing would be negotiated with adjacent landowners during right-of-way negotiations.

- Signage may be used to alert drivers to the presence of wildlife along the right-of-way, particularly during the months of October through January, when most of the AVC occur along the project corridor. Research has shown that static, permanent signage is ineffective; however, variable message signs are more successful at encouraging drivers to reduce their speeds.
- A wildlife detection system, such as laser detectors, motion sensors, or heat detection systems, may be used. Such systems are activated when large animals interrupt sensors set up along the right-of-way, thereby alerting drivers via a message board or flashing lights to the presence of animals entering the right-of-way. Wildlife detection systems are effective for large animals such as deer (which account for 98.5 percent of the AVC along the project corridor). Wildlife detection systems can be cost effective for relatively straight and flat stretches of highway, such as the project corridor, where the sensors can be placed further apart.

3.12 FLOODPLAIN

Floodplains constitute lands situated along rivers and their tributaries that are subject to periodic flooding with a one-percent chance of being flooded in any given year, on the average interval of 100 years or less. Pursuant to Executive Order 11988, Floodplain Management, issued in 1977, the following criteria apply to this project: potential effects on floodplains must be evaluated; alternatives that avoid adverse effects and incompatible development in floodplains must be evaluated; and if it is found that the only practicable alternative requires siting in a floodplain, it is necessary to design or modify the project in order to minimize potential harm to or within the floodplain.

The project corridor follows Rock Creek Valley for the extent of the project between two benches, east and west, which rise 100 feet (30 meters) above the stream valley floor. The southern end of the project corridor is located just west of Rock Creek. The project corridor remains west of Rock Creek until about 1.5 miles (2.4 kilometers) past Roberts, then crosses Rock Creek and remains east of Rock Creek for the remainder of the corridor. One portion of the project corridor, near Clear Creek Road, is located within the FEMA (Federal Emergency Management Agency)-identified 100-year base flood elevation (4413NGVD29) for Rock Creek. Since the development of the 100-year base flood elevation map, portions of the Rocky Fork Branch of the Northern Pacific Railroad have been removed; this may have altered the base flood elevation.

Floodplain Impacts

Alternative A (No-Build): If no action were taken, there would be no new impacts to floodplains.

Alternative B (Preferred): The Preferred Alternative may include minor approach work at the new Rock Creek Bridge, which would require contact and coordination with the county floodplain administrator. Coordination with the county floodplain administrator would occur to determine whether minor encroachment of the floodplain would occur and whether a floodplain

development permit is required. However, the build alternative is not anticipated to increase the 100-year base flood elevation over existing conditions, pursuant to MCA (Montana Code Annotated) Title 76, Land Resources and Use; Chapter 5, Floodplain and Floodway Management. The Preferred Alternative may also include one or more new storm water outfall locations to Rock Creek.

Mitigation for Floodplain Impacts

The Preferred Alternative is not anticipated to increase the 100-year base flood elevation over existing conditions. However, coordination with the county floodplain administrator would occur to determine potential encroachment of the floodplain, and permit and mitigation requirements.

3.13 THREATENED OR ENDANGERED SPECIES

Section 7 of the Endangered Species Act of 1973, as amended (16 U.S.C. 1536), requires each federal agency to ensure that any action funded or carried out by such agency is not likely to jeopardize the continued existence of any federally-listed endangered or threatened species or species proposed to be listed, or likely to result in the destruction or adverse modification of habitat of such species which is determined to be critical by the Secretary of the Interior. An endangered species is in danger of extinction throughout all or a significant portion of its range. A threatened species is one that is likely to become endangered in the near future.

Consultation with USFWS and a review of the lists of listed, proposed, or candidate threatened and endangered species in Montana were used to identify any such species that may be located in the project area. Range/habitat descriptions found in technical literature were also reviewed to determine which listed or proposed species may be located in the project corridor. Following is a summary of the findings¹².

Two species were considered with respect to the project: the bald eagle (*Haliaeetus leucocephalus*) and the Yellowstone nonessential experimental population of gray wolves (*Canis lupus*)¹³.

No known, confirmed, or suspected bald eagle nests occur within several miles of the proposed project corridor. Although the bald eagle was identified as a federally-listed threatened species at the time of project coordination with the USFWS, the USFWS de-listed the bald eagle with an effective date of August 8, 2007¹⁴. However, the bald eagle is still afforded protection under the Bald and Golden Eagle Protection Act of 1940, 16 USC 668-668d as amended, and the Migratory Bird Treaty Act of 1918, 16 USC 703-712 as amended.

The gray wolf was reclassified from endangered to threatened on March 18, 2003, and the USFWS concurrently published an Advance Notice of Proposed Rulemaking to notify the public that the USFWS will soon begin work to propose delisting certain populations, including those in the state of Montana. Newly formed wolf pack activity has been reported a few miles

¹² Land & Water Consulting, Inc., Red Lodge-North Final Biological Resources Report (Apr. 2004)

¹³ The black-footed ferret *(Mustela nigripes)*, an endangered species; and the black-tailed prairie dog *(Cynomys ludovicianus)*, which was, and is no longer, a candidate to be listed as threatened, were also evaluated. Neither of these species are known to inhabit the project corridor, and it was determined that the proposed project would have no effect to these species. ¹⁴ 72 FR 37345

south/southeast of Red Lodge. However, this wolf population is considered a non-essential experimental population, and no active wolf dens are known to occur in the project corridor.

Impacts to Threatened or Endangered Species

Alternative A (No-Build): If no action were taken, there would be no new impacts to threatened or endangered species.

Alternative B (Preferred): The USFWS has determined that the proposed build alternative is not likely to affect individual bald eagles or jeopardize the continued existence of non-essential experimental gray wolves. See Appendix C, March 2002 Agency Scoping, Letter #5 and Appendix D, May 2007 Agency Scoping, Letter #3.

Mitigation for Threatened or Endangered Species Impacts

Adverse impacts to threatened and endangered species are not anticipated; therefore, no mitigation is required.

3.14 CULTURAL RESOURCES

Section 106 of the National Historic Preservation Act of 1966 (16 U.S.C. 470), as amended, requires that federally funded projects be evaluated for the effects¹⁵ on historic and cultural properties included in, or eligible for listing on, the NRHP (National Register of Historic Places). The Archaeological and Historic Preservation Act of 1974 (16 U.S.C. 461 et seq., and 23 U.S.C. 305) provides for the survey, recovery, and preservation of significant scientific, prehistoric, archaeological, or paleontological data when such data may be destroyed or irreparably lost due to a federally licensed or federally funded project.

In order to identify cultural resources along the project corridor and to determine potential impacts, three cultural resource surveys were undertaken^{16,17, 18}, as was a historical inventory¹⁹ within Red Lodge and Roberts. These surveys identified numerous historic properties, including:

- Two historic districts in Red Lodge
- Eight properties in Red Lodge that are listed on or eligible for the NRHP, or contributing to the historic districts
- Three properties eligible for the NRHP between Red Lodge and Roberts
- Four properties eligible for the NRHP in Roberts
- One property eligible for the NRHP in Boyd
- One property listed on the NRHP, which spans the project corridor (Rocky Fork Branch of the Northern Pacific Railroad)
- Eleven historic irrigation ditches along the project corridor

¹⁵ Effect means alteration to the characteristics of a historic property qualifying it for inclusion in or eligibility for the National Register (36 CFR § 800.16).

¹⁶ Historical Research Associates, Inc., *Report of a Cultural Resource Inventory of the Red Lodge-North Project Corridor, Along Highway 212 in Carbon County, Montana* (Feb. 1997)

¹⁷ Metcalf Archaeological Consultants, Inc., *Red Lodge North, Highway 212 Improvements: A Cultural Resource Inventory from Roberts to Boyd, Carbon County, Montana* (Oct. 2002)

¹⁸ Earthworks, Inc., *Highway 212 North: A Cultural Resource Inventory, Carbon County, Montana* (June 2007).

¹⁹ Joan Brownell, *Historic Inventory: Red Lodge North Project, Carbon County, Montana;* Project STPP 28-2 (25) 70 (Feb. 2003)

Impacts to Cultural Resources

Alternative A (No-Build): If no action were taken, there would be no impacts to historic and cultural properties.

Alternative B (Preferred): MDT has determined, and SHPO (State Historic Preservation Office) has concurred, that the proposed project would have *No Effect* or *No Adverse Effect* to all of the historic properties. See Table 3.9, Impacts to Historic Properties, and Appendix E, Cultural Resources.

Mitigation for Cultural Resources Impacts

The Preferred Alternative would have *No Effect* or *No Adverse Effect* to cultural resources; therefore, no further mitigation/avoidance measures are required.

Table 3.9 Impacts to Historic Properties					
Site #	General Description	Location	Status	SHPO Finding	
24CB145	Red Lodge Commercial Historic District	Red Lodge	Listed	No Effect ¹	
24CB1030	Hi Bug Historic District	Red Lodge	Listed	No Effect ¹	
24CB145	Carnegie Library	Red Lodge	Listed	No Adverse Effect ⁵	
24CB1819	One and a half story front gable cottage	Red Lodge	Not Eligible but Contributing	No Effect ¹	
24CB1820	One and a half story front gable cottage	Red Lodge	Not Eligible but Contributing	No Effect ¹	
24CB1821	One and a half story front gable cottage	Red Lodge	Not Eligible but Contributing	No Effect ¹	
24CB1822	Two story Queen Anne residence (Finley House)	Red Lodge	Eligible and Contributing	No Effect ¹	
24CB1827	One story wood frame bungalow (Richardson Bungalow)	Red Lodge	Eligible	No Effect ¹	
24CB1830	Self contained elevator (MT Dakota Grain Company Elevator)	Red Lodge	Eligible	No Effect ¹	
24CB1833	Dairy Delite Drive-In	Red Lodge	Eligible	No Effect ¹	
24CB1320	Kent Dairy Round Barn	Red Lodge to Roberts	Eligible	No Effect ⁴	
24CB1336	Carbon County Dairy/Maryott Ranch	Red Lodge to Roberts	Eligible	No Effect ¹	
24CB1339	Maryott agricultural complex	Red Lodge to Roberts	Eligible	No Effect ⁴	
24CB1705	One story Craftsman residence (O'Shea House)	Roberts	Eligible	No Adverse Effect ¹	

Continued...

Table 3.9 Impacts to Historic Properties					
Site #	General Description	Location	Status	SHPO Finding	
24CB1712	One story American Four- Square/Craftsman residence (Silakka House)	Roberts	Eligible	No Effect ¹	
24CB1717	St. Thomas Catholic Church	Roberts	Eligible	No Effect ⁴	
24CB1720	One story vernacular-style residence (Monahan House)	Roberts	Eligible	No Effect ⁴	
24CB1831	Boyd Country Store (Boyd Mercantile)	Roberts to Boyd	Eligible	No Effect ¹	
24CB1283	Rocky Fork Branch of the Northern Pacific Railroad	Red Lodge to Boyd	Listed	No Adverse Effect ²	
24CB1722	Brandt Ditch	Roberts	Eligible	No Effect ³	
24CB1723	Rule-Thompson Ditch	Roberts	Eligible	No Adverse Effect ³	
24CB1724	Duncan-Aiken Ditch	Roberts to Boyd	Eligible	No Effect ³	
24CB1725	Bernhardt Ditch	Roberts to Boyd	Eligible	No Effect ³	
24CB1726	Hunts Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
24CB1727	Highline Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
24CB1728	Rooney Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
24CB1729	Drakes Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
24CB1730	Ward Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
24CB1731	Carbonado Ditch	Roberts to Boyd	Eligible	No Effect ³	
24CB1761	Hoyle Ditch	Roberts to Boyd	Eligible	No Adverse Effect ³	
¹ April 22, 2003 Montana SHPO concurrence ² September 25, 2003 Montana SHPO concurrence ³ June 10, 2005 Montana SHPO concurrence					

⁴ July 12, 2005 Montana SHPO concurrence

⁵ September 17, 2007 Montana SHPO concurrence

3.15 SECTION 6(f) PROPERTIES

Section 6(f)(3) of the Land and Water Conservation Fund Act of 1965, as amended, specifies that no property acquired or developed with assistance from Section 6(f) Funds shall, without the approval of the Secretary of the Interior, be converted to other than public outdoor recreation uses. The Secretary shall approve such conversion only when: 1) the Secretary finds it to be in accord with the [current] comprehensive statewide outdoor recreation plan and 2) the recreation

Corridor Study – Red Lodge North Environmental Assessment STPP 28-2 (25) 70 Control No. 4375 October 2008 properties are replaced with other public outdoor recreation properties of at least equal fair market value and/or reasonable equivalent usefulness and location.

There are four Section 6(f) properties within the project corridor. Three are fishing access sites: Horse Thief Station, Bull Springs, and Water Birch. The fourth is Roberts Public School.

Impacts to Section 6(f) Properties

Alternative A (No-Build): If no action were taken, there would be no impacts to Section 6(f) properties.

Alternative B (Preferred): The Preferred Alternative would not result in the conversion of the three fishing access sites or Roberts Public School to a transportation facility. In order to avoid the Roberts Public School playground, the typical section along the school was altered to include a guardrail and 2:1 ditch slopes.

In addition, any directional and entrance signs that may be removed would be reinstalled following construction. If impacts occur to the entrance road for the Water Birch fishing access site, it would be returned to existing or improved condition following construction. Therefore, Alternative B would not convert Section 6(f) properties into a transportation facility.

Mitigation for Section 6(f) Property Impacts

All Section 6(f) properties would be avoided; therefore, no mitigation is required.

3.16 HAZARDOUS MATERIALS, SOLID WASTE, AND UNDERGROUND STORAGE TANKS

The RCRA (Resource Conservation and Recovery Act) (33 U.S.C 1241 et seq.) and CERCLA (Comprehensive Environmental Response, Compensation, and Liability Act) (42 U.S.C. 9601 et seq.) regulate hazardous material, hazardous waste, or environmentally contaminated sites.

A hazardous waste survey was conducted to identify known and potential hazardous waste/materials sites and USTs (Underground Storage Tanks) within the project corridor²⁰. Records were reviewed according to Standard Practice for Environmental Site Assessments: Phase I Environmental Site Assessment Process (ASTM, 2000)²¹.

A single site has been officially considered and reviewed for Federal Superfund Status within one-mile (1.6 kilometers) of the proposed project site: Island at Rock Creek. The site is located at the eastern edge of Red Lodge between 8th and 16th Streets, on the east side of Rock Creek.

There is a single SWLF (Solid Waste Landfill) within 0.5-miles (0.8-kilometers) of the proposed project site. This SWLF is a closed Red Lodge City facility that is located at 900 Bonner Avenue, approximately 1,500 feet (450 meters) east of US Highway 212. This facility was closed in 1983 and reclaimed afterward (Tomisich, 2001). It is currently the location of the Beartooth Nature

²⁰ Hyalite Environmental, LLP., *Amended Initial Site Assessment Report, Red Lodge North Reconstruction* (Aug. 2002)

²¹ Additional records research on the Montana Natural Resource Information System was conducted in 2007. This resulted in the identification of one additional site, the Ski Station in Red Lodge, which is both a UST and LUST.

Center. This site is considered by the MDEQ to be a good example of reclamation and re-use of community lands.

Fifteen LUST (Leaking Underground Storage Tanks) sites have been identified adjacent to US Highway 212 or within 0.5-miles (0.8-kilometers) of the beginning of the proposed project in Red Lodge. See Table 3.10, Leaking Underground Storage Tanks.

Fourteen UST sites were identified adjacent to the proposed project. See Table 3.11, Underground Storage Tanks.

Table 3.10 Leaking Underground Storage Tanks					
Site Name	Site Location	Confirmed Release Date	Active Site		
RED LODGE					
Anderson Conoco	910 W. Villard	22-Jul-88	No		
Beartooth Electric Co-Op Inc	Box 1119 (1306 N. Broadway)	20-Jul-93	No		
Carbon County Abstract Title Co	105 N. Broadway	24-Oct-89	No		
Carter's Bulk Plant	Hwy 212	01-Aug-91	No		
Ray Judd Ford	116 N. Broadway	07-Jul-99	Yes		
Red Lodge Travel Center	403 S. Broadway	30-Jul-98	No		
T & D Pump	Hwy 212	09-Dec-93	Yes		
Cowger, Nick	Rte 1 Box 4165	12-Jun-91	No		
Ransdell Union 76 Bulk Plant	Address Unknown	29-Jul-93	No		
The Ski Station	510 N. Broadway Ave	09-Sept-06	Yes		
ROBERTS					
Former Roberts Exxon	#1 Railroad Ave (Hwy 212)	22-Apr-97	Yes		
Laurel Coop Assoc	Box 11 (Hwy 212)	30-Oct-90	No		
Y-Stop	Box 85 (Hwy 212)	27-May-92	No		
BOYD					
Boyd Country Store	Box 236 (Hwy 212)	11-Jun-93	No		
Old West Trading Post	Box 128 (Hwy 212)	17-Dec-92	Yes		

Table 3.11 Underground Storage Tanks						
Facility ID	Site Name	Site Location	Active Tanks	Non-Active Tanks		
RED LODG	RED LODGE					
05-08860	Beartooth Electric Co-Op Inc	Box 1119 (1306 N. Broadway)				
05-03138	King Oil Co	N. Edge of Town		2		
05-06961	Pony Express	401 N. Broadway	3			
05-01404	Red Lodge KOA	4 Mi. N. on Hwy 212		2		
05-13924	Rock Creek North	902 N. Broadway	4			
05-04228	T & D Pump	Hwy 212	3			
05-04737	Old Town Square	410 N. Broadway		1		
05-12391	Biorn, Terry	Waples Tracts		1		
60-15056	The Ski Station	510 N Broadway Ave		3		
				Continued		

Table 3.11 Underground Storage Tanks					
Facility ID	Site Name	Site Location	Active Tanks	Non-Active Tanks	
ROBERTS					
05-06599	Y-Stop C Store	Hwy 212 & Cooney Rd S.	4	4	
05-05374	Former Roberts Exxon	#1 Railroad Ave (Hwy 212)		1	
05-04499	Town and Country Supply	Box 11 (Hwy 212)		2	
05-08035	Wright, Dale	Rt 1 Box 2004		1	
BOYD					
05-05710	Boyd Country Store	Box 236 (Hwy 212)		2	

Impacts to Hazardous Materials, Solid Waste, and Underground Storage Tanks

Alternative A (No-Build): The no-build alternative would not impact hazardous materials, solid waste, or underground storage tank sites.

Alternative B (Preferred): It is unlikely that the proposed project would impact the Island at Rock Creek site or that impacts from the Island at Rock Creek site would affect the proposed project. Mine tailings, which may have impacted ground water at this site, characterize the site. It is also unlikely that the reclaimed SWLF would have impacts on the proposed project, or vice versa.

The LUST identified at the former Exxon station in Roberts may indirectly impact the proposed project. The identified plume appears to be down gradient of the roadway; however, groundwater is extremely shallow under the roadbed and excavation should be monitored. It is the responsibility of the Contractor to have any groundwater encountered sampled and analyzed for organic contaminants to determine safety, handling, and disposal measures. Additionally, a LUST, known as the Ski Station, was identified adjacent to the project corridor in Red Lodge. As a result of storm water improvements in Red Lodge, approximately 90 cubic yards (70 cubic meters) of material is anticipated to be removed along US Highway 212 near the Ski Station. However, coordination with MDEQ revealed that it is unlikely contaminated soil would be encountered during project construction. *See Appendix D, May 2007 Agency Scoping, Letter #4.*

Excavation activities could potentially expose or otherwise affect subsurface hazardous materials. Property owners and subsurface utility locators will be consulted during the design phase of the proposed project concerning the exact locations of the identified USTs to verify potential encounters with hazardous materials. MDT expects that the Contractor will ensure that hazardous materials discovered, generated or used during implementation of the project would be stored, handled, and disposed in accordance with applicable local, State, and Federal laws.

Mitigation for Hazardous Materials, Solid Waste, and Underground Storage Tank Impacts

If contaminated soils or hazardous materials are encountered during construction, excavation and handling will be done in accordance with local, State, and Federal laws. Impacts to hazardous materials, solid waste, and underground storage tanks are not anticipated; therefore, no mitigation is required.

3.17 VISUAL/AESTHETIC CONSIDERATIONS

The City of Red Lodge is preparing a Streetscape Plan for US Highway 212 between 15th Street and 8th Street, which would tie in to the southern terminus of the proposed project. MDT met with the City during the Streetscape planning process to discuss compatibility with both projects/plans. Design concepts for the Streetscape Plan include bulb-outs at intersections, decorative concrete, lighting, landscaping treatments, street furniture, and other concepts.

Visual/Aesthetic Impacts

Alternative A (No-Build): If no action were taken, there would not be visual/aesthetic impacts.

Alternative B (Preferred): The Preferred Alternative would improve aesthetics within Red Lodge. The four roundabouts between MT Highway 78 and Two Mile Bridge Road, proposed as part of the access management plan, would create a distinctive entrance into Red Lodge from the north, as supported in Red Lodge City Council Resolution No. 3228. Further, the proposed project would incorporate elements of the Red Lodge Streetscape Plan, as appropriate, which would improve aesthetics within the City. The proposed typical section from 8th Street to MT Highway 78 is expected to be compatible with the proposed typical section used by Red Lodge from 15th Street to 8th Street. The Preferred Alternative also includes a bulbed out curb line at 8th Street, which is consistent with the Streetscape Plan. MDT will work with the City to develop an agreement to address appropriate lighting and landscaping features to be incorporated into the project. The Red Lodge Streetscape Plan, available funding, and maintenance will be taken into consideration when developing the agreement.

Mitigation for Visual/Aesthetic Impacts

Adverse impacts to visual/aesthetics are not anticipated; therefore, no mitigation is required.

3.18 TEMPORARY CONSTRUCTION CONSIDERATIONS

During construction of the proposed project, specific impacts would occur directly as a result of construction activities. These include, but are not limited to, construction equipment noise, dust from delivery of materials through the local roadways, creation of borrow pits, and disposal of soil. The temporary construction impacts would not occur under the no-build alternative. An analysis of environmental factors affected by construction activities is described below.

3.18.1 Air Quality

Construction activities could have a short-term impact on air quality, primarily during site preparation. Dust is generated during earth moving activities and handling of cement, asphalt, or aggregate. Wind erosion of exposed areas and material stockpiles also generates particulate matter. The amount of dust generated would vary, depending on the construction activity and local weather conditions.

Additionally, construction activities may result in a temporary increase of carbon monoxide. These increases would be a result of slowed traffic due to detours, as discussed in 3.18.2, and slow-moving vehicles performing construction activities.

Mitigation for Temporary Air Quality Impacts

Where excess dust is anticipated to be a problem, effective dust control measures would be implemented in accordance with standard MDT procedures and applicable permit requirements. Dust control would be the responsibility of the contractor.

Federal and State laws regulate emissions from construction equipment. MDT would expect any burning of cleared materials to be conducted in accordance with applicable state and local laws, regulations, and ordinances. Projected traffic volumes are well below typical thresholds for vehicle emission air quality modeling.

3.18.2 Transportation System

Construction delays would likely create temporary impacts to local and regional traffic circulation in the project area due to lane closures, delays, temporary travel on unpaved surfaces, and reduced travel speeds. Traffic diversions and construction equipment and activities close to the travel lanes would also affect speeds and traffic operation within the construction zone. Disruptions to access and parking for business and residences located within the construction zone would occur.

Mitigation for Temporary Transportation System Impacts

A construction traffic control plan will be developed according to MDT Standard Specifications to include construction phasing devised to maintain two lanes of traffic and uninterrupted side road access along the corridor to the greatest extent practicable. The contractor will coordinate with emergency service providers and schools to solicit input for the construction traffic control plan to provide ongoing information during construction.

3.18.3 Water Resources/Quality

Rock Creek, Stanley Creek, and numerous irrigation ditches would be crossed by the proposed project, as described in Section 3.8.1.

Temporary impacts to water quality may result from construction. These may include an increased potential for erosion, reduced slope stability, and increased turbidity caused by disturbing waterway bottoms and re-suspending existing sediments in the water column. During storm events, an influx of fuel and other pollutants from unpaved surfaces could also occur. Increases in turbidity, suspended sediment, and other pollutants can reduce stream productivity and slow biogeochemical and natural purification processes.

Mitigation for Temporary Water Resources/Quality Impacts

The potential for temporary increases in turbidity and other water quality impacts resulting from construction activities would be reduced by the implementation of standard BMPs during construction, compliance with project-specific conditions to be specified in the permits and certifications required for the project (Montana Stream Protection Act; Federal Clean Water Act Section 404/401; MCA Title 75, Environmental Protection, Chapter 5, Water Quality), and compliance with MDT standard water pollution control specification (Section 208). In addition, temporary impacts to wetlands within the right-of-way and construction easement areas would be restored to original contours and re-vegetated at the earliest practicable date following construction.

3.18.4 Wildlife Habitat and Ecosystems

Construction of the project may result in minor indirect disturbance to wildlife communities. The survival of displaced species that relied exclusively within the limits of construction would depend on the carrying capacity of adjacent undeveloped habitat. Minimal impacts to nesting raptors or waterfowl may occur, though little nesting is expected in areas that would be directly impacted by construction. Additionally, construction activities may result in the invasion and/or spread of noxious weeds due to clearing and grubbing activities.

Increases in turbidity, suspended sediment, and other pollutants can reduce feeding opportunities for fish and result in fish avoidance of important habitat. Increased turbidity/suspended sediment may also block light transmission and slow biogeochemical and natural purification processes. Deposited sediments can also reduce habitat volume by filling pools and inter-gravel spaces that are critical to eggs and young fish. Finally, any construction activities in Rock Creek, Stanley Creek, or other water bodies could potentially disrupt fish spawning activity; however, substantive losses of spawning habitat for any fish species are not anticipated.

Mitigation for Temporary Wildlife Habitat and Ecosystems Impacts

In compliance with the Migratory Bird Treaty Act and MDT's most current depredation permit from the USFWS, vacated swallow or other songbird nests that may be directly impacted by construction would be physically removed by MDT staff or the contractor between the dates of September 1 and April 30. Deterrents such as screening or netting and/or Bird-X Repellent would be placed on existing structures (bridges or box culverts) as well as new structures under construction to be impacted by the project to discourage re-nesting until construction activities can be completed.

Disturbed areas would be reseeded and conducted in accordance with MDT Standard Specifications. Noxious weeds would be controlled by MDT, the County Weed Board, and the contractor. Furthermore, an erosion and sediment control would be prepared in accordance with Section 402 and MPDES (Montana Pollutant Discharge Elimination System) regulations.

3.18.5 Noise

Construction would result in temporary increases in noise levels within the vicinity of the project.

Mitigation for Temporary Noise Impacts

MDT Standard Specifications require that contractors comply with applicable laws and regulations to minimize noise impacts. As necessary, the contract will include additional requirements for projects located in or near urban areas.

3.19 UTILITIES

During construction, certain impacts to utilities may occur such as utility relocation or service interruption due to relocation. Overhead and underground power lines and underground telephone lines are located parallel to and on both sides of the existing alignment. The preliminary alignment has been adjusted as much as practical to avoid and minimize impacts to utilities. However, some relocation of these utilities will be required.

Mitigation for Utility Impacts

Utility relocations will be coordinated with appropriate line owner(s) and completed prior to project construction. Notification of service interruption due to relocation will be the responsibility of the appropriate utility line owner. Such disruptions are normally minor in nature, and are limited to the customers connected to the affected lines. In addition, rural overhead power lines that are relocated would be raptor proofed per MDT policy.

3.20 CUMULATIVE CONSIDERATIONS

Cumulative impacts result from the incremental consequences of an action "when added to other past, present, and reasonably foreseeable actions regardless of what agency or person undertakes such other actions" (40 CFR 1508.7). Effects of an action may be insignificant when evaluated in an individual context, but these effects can add to other disturbances and cumulatively may lead to a measurable environmental change. By evaluating the impacts of the proposed action with the effects of other actions, the relative contribution of the proposed action to a projected cumulative impact can be estimated.

The following discussion addresses the potential for cumulative impacts from the US Highway 212 project to numerous other projects and/or actions that have recently occurred, are presently occurring, or may be expected to occur in the reasonably foreseeable future along or near the project corridor. These include projects or actions undertaken by the City of Red Lodge, Carbon County, MDT, and private developers.

3.20.1 Past, Present, and Reasonably Foreseeable Actions

The following discussion focuses on past, present, and reasonably foreseeable future actions undertaken by or in the City of Red Lodge, by or in Carbon County, and by MDT near the project corridor.

The City of Red Lodge and Carbon County have experienced recent population growth and development pressure and expect this trend to continue over the next 15 years. The City and County are actively managing this growth with an emphasis on maintaining or improving the community character. This is evident in the Red Lodge Growth Policy (May 2001), the Carbon County Growth Policy (September 2003), and the City of Red Lodge Comprehensive Trails Plan (May 2006).

Red Lodge has several local roadway projects under consideration. These include the extension of MT Highway 78 from its intersection with US Highway 212 to the east to accommodate commercial development, construction of an alley access to accommodate the Red Lodge Ales Brewing Company south of 6th Street, and streetscape improvements along US Highway 212 from 8th Street to 15th Street.

Additionally, the Red Lodge City Planner provided MDT with information about planned developments along or near the project corridor. As described previously, these include the Beartooth Hospital, Bank of Red Lodge, and five housing subdivisions with a total of nearly 800 new housing units. The Carbon County Planning Director also provided information about planned developments between Red Lodge and Boyd. These included three housing subdivisions with a total over 220 new housing units.

Carbon County completed a sidewalk project in Roberts in 2005 that provided pedestrian facilities on First Street. Additionally, Carbon County Planning staff identified that there is a 159-unit and 40-unit subdivision planned north of Roberts.

MDT also has numerous roadway improvement projects planned near the project corridor. These include three projects on MT Highway 78 located northwest of Red Lodge, one project on MT Highway 72 located southeast of Red Lodge, and one project located on US Highway 212 located north of the project corridor, as summarized in the following sections.

- <u>Red Lodge Safety Improvement Project.</u> MDT has proposed a safety improvement project to address the crash trend at an accident cluster 5 miles (8.1 kilometers) northwest of Red Lodge on MT Highway 78. The accident rate at this cluster is 7.01 with a severity rate of 21.03. The cause of these accidents has been cars leaving the road on a sharp horizontal curve. MDT has proposed reconstructing the curve with a larger radius and flatter slopes. Additionally, the project is designed to meet geometric criteria such as providing stopping sight distance and route segment plan width. The anticipated letting for this project is 2008.
- <u>MT Highway 78 Corridor Study.</u> MDT has initiated a corridor planning process along the MT Highway 78 corridor, beginning about 5 miles (8.1 kilometers) northwest of Red Lodge and ending at the north end of Roscoe. This segment of MT Highway 78 does not meet current design and safety standards. The purpose of the project is to comprehensively address future transportation needs, prioritize future transportation projects, and foster cooperative state and local transportation planning efforts. The anticipated letting date is after 2011.
- <u>Red Lodge Northwest Project</u>. MDT plans to reconstruct a portion of MT Highway 78 in Carbon County to address the higher than average accident rate for this portion of the highway. The proposed work is to reconstruct approximately 5.1 miles (8.1 kilometers) of the existing roadway starting at the MT Highway 78 and US Highway 212 intersection and continuing northwest of Red Lodge. The purpose of the project is to increase the safety of the route and to provide a long-term quality highway. The anticipated letting date is after 2011.
- <u>Belfry North Project.</u> MDT prepared an EA and Nationwide Programmatic Section 4(f) Evaluation, and FHWA issued a Finding of No Significant Impact on June 17, 2005 for the 11.1-mile (17.9-kilometer) section of MT Highway 72 between Montana Secondary 308 in the town of Belfry and US Highway 310 south of the town of Bridger in Carbon County, Montana. The primary purpose of this project is to reconstruct MT Highway 72 to improve safety along the project corridor. The existing project corridor does not meet current MDT standards for a rural arterial highway. The anticipated letting date is after 2011.

<u>US Highway 212 Rockvale to Laurel Project.</u> An EIS (Environmental Impact Statement) is being prepared for the reconstruction of US Highway 212 from Rockvale to Laurel. The purpose of the project is to reconstruct approximately 11.2 miles (18.0 kilometers) of US Highway 212 with added capacity to reduce congestion and the incidence and severity of accidents. The project is needed because this segment of US Highway 212 has several deficiencies and capacity limitations. The planning process for this project is anticipated to take about 7 years. A Draft EIS was approved by MDT and FHWA on October 31, 2007.

3.20.2 Cumulative Impacts

The proposed US Highway 212 project was developed in cooperation with Red Lodge and Carbon County. The proposed project is consistent with the Red Lodge and Carbon County Growth Policies and the City of Red Lodge Comprehensive Trails Plan. The roundabout proposed at MT Highway 78 would accommodate an extension of MT Highway 78 to the east. The proposed improvements at Oakes Avenue and 8th Street would accommodate the proposed Bank of Red Lodge and would not preclude the provision of alley access for the Red Lodge Ales Brewing Company. As appropriate, MDT will incorporate elements of the streetscape design that Red Lodge is proposing for US Highway 212 south of the project corridor into project design for consistency with aesthetic design. The proposed roundabout at the Two Mile Bridge Road intersection would accommodate the proposed Beartooth Hospital. Additional capacity along US Highway 212 is not anticipated to be needed to accommodate the proposed subdivisions in Red Lodge.

The proposed project would also complement the 2005 Roberts sidewalk project by including one block of new sidewalk to tie into the existing sidewalk, as well as crosswalks on US Highway 212. The proposed US Highway 212 project would realign the Clear Creek Road access to eliminate the offset intersection with the rest area approaches. This would improve safety conditions at this intersection.

The proposed project would not directly impact the other MDT projects. Cumulatively, these projects would all improve the safety and operational characteristics of the local and regional transportation system.

Following is a discussion about potential cumulative environmental impacts associated with the proposed projects and the other actions discussed previously.

Land Use — The proposed US Highway 212 project would accommodate the land use changes near the project corridor, but it is not driving these changes. Although some new right-of-way would be needed for the corridor, the project would improve the roadway on its existing alignment. The project is not intended to induce additional traffic since there is not additional capacity nor is it expected to change growth patterns. The additional traffic anticipated as a result of the past, present, or reasonably foreseeable developments would be accommodated and managed by the proposed project, rather than induced. Therefore, the contribution of the proposed project to land use changes associated with other actions is not expected to be significant.

Farmland — As Red Lodge and Carbon County continue to experience increased population and development pressure, farmland will continue to be converted to other uses, primarily residential and commercial. The proposed project would require some additional right-of-way, impacting farmland; however, these impacts would occur adjacent to the existing roadway corridor. In addition, through coordination with NRCS, the impacts to farmland as a result of the US Highway 212 project have been determined to be not significant. Moreover, the contribution of the project to the loss of farmland associated with other actions is not expected to be significant.

Right-of-way and Relocations — The proposed project would require additional right-of-way to accommodate the proposed improvements and provide the MDT standard right-of-way width for the facility type. The proposed project may also result in the acquisition and/or relocation of up to nine structures, including six residences, two outbuildings, and a commercial building. Other MDT projects in the region may also require right-of-way and/or require relocations. However, most roadway projects involve improvements on roadways that have been in place for many years. MDT attempts to minimize impacts to adjacent properties and does provide compensation for unavoidable impacts. Therefore, cumulative right-of-way and relocation impacts are not expected to be significant.

Traffic — Development and roadway projects are actions that can lead to increased traffic and/or changes in traffic patterns. Traffic generation from these past, present, and reasonably foreseeable projects was included for the projected traffic volumes for the proposed project.

Pedestrian and Bicycle Considerations — The proposed project would provide wider shoulders in the rural segments; usable shoulders, sidewalks, a shared bike/ped path, and crosswalks, as appropriate, in Red Lodge; and a sidewalk and crosswalks in Roberts. The proposed project, other improvements proposed in the *Red Lodge Comprehensive Trails Plan*, and the Roberts sidewalk project completed in 2005 cumulatively improve the non-motorized transportation system along and near the project corridor.

Water Quality — Water quality impacts resulting from the proposed project are expected to be minor and will be minimized through the use of BMPs during construction. Road construction and development activities can be contributing factors to increased areas with hard surfaces, such as pavements and buildings. The increased hard surface areas reduce the amount of precipitation that can infiltrate into the ground and increase the amount of precipitation that runs off. The proposed project is expected to contribute to the increase in runoff by creating a wider roadway section. Additional urbanization along the corridor is expected to occur due to ongoing and foreseeable development, particularly in north Red Lodge, and this urbanization is also expected to contribute to the increase in runoff.

The design of the project would accommodate the increased runoff by directing it to natural drainage areas. The project would also reduce the flooding potential for Roberts by incorporating measures that would reduce the amount of runoff that reaches the community from the south. The cumulative impacts of this project would be minor compared to the runoff from the existing roadway and compared to the runoff from the total drainage basin. Therefore, the cumulative effect to water quality would not be significant, when added to the impacts from other development in the area.

Wetlands — Cumulative impacts have occurred and are occurring in Carbon County due to land conversion. Growth in the project area has substantially increased, as discussed in Section 3.3, Land Use, and a number of new developments are planned adjacent or near the project corridor. The proposed project is not anticipated to contribute substantially to the cumulative loss of wetlands in Carbon County due to MDT and FHWA's commitment to avoidance and

minimization of wetland impacts during design, and the development of compensatory wetland mitigation where impacts are unavoidable, in accordance with Executive Order 11990 and Section 404 of the Clean Water Act.

Wildlife — Construction noise, habitat loss or fragmentation, and AVC attribute to impacts to wildlife. However, the project is located on an existing roadway and it is likely that suitable habitat exists outside the project corridor. Additionally, a goal of the proposed project is to minimize the potential for AVC in the project area. Therefore, the cumulative effect to habitat would not be significant, when added to the impacts from other development in the area.

Visual/Aesthetics — The proposed project, combined with the Red Lodge Streetscape Plan, would provide a cumulative aesthetic benefit to Red Lodge.

Other environmental considerations, such as floodplains, threatened and endangered species, cultural resources, Section 4(f) and 6(f) properties, and hazardous materials, are not anticipated to be directly or indirectly impacted by the proposed project, as discussed previously; therefore, they are not anticipated to be cumulatively impacted by the proposed project when coupled with one or more of the aforementioned projects.

3.21 IRREVERSIBLE AND IRRETRIEVABLE COMMITMENT OF RESOURCES

As with any construction project, certain irreversible and irretrievable commitments of natural resources, manpower, materials and fiscal resources are required. Fossil fuels, labor, and highway construction materials such as cement, aggregate, and bituminous material would be expended to complete the project. Additionally, labor and natural resources would be used in the fabrication and preparation of construction materials. These materials are generally not retrievable. However, they are not in short supply, and their use would not have an adverse effect on the availability of these resources. Construction would also require a one-time expenditure of State and Federal funds, which are not retrievable. However, the anticipated beneficial effects would balance the irretrievable commitment of resources caused by construction of the Preferred Alternative.

3.22 SHORT-TERM IMPACTS VERSUS LONG-TERM BENEFITS

The local, short-term impacts and use of resources inherent with the Preferred Alternative would be consistent with the maintenance and long-term functionality of US Highway 212 from Red Lodge to north of Boyd. Short-term impacts would include temporary delays, increased dust, noise, wildlife disruption, and water quality impacts associated with construction. Long-term benefits of the build alternative are related to having a safe, reliable transportation corridor. Such benefits include improving the safety and operational characteristics of the roadway by:

- Improving the pavement condition along the project corridor by reconstructing the roadway with new asphalt pavement.
- Improving the intersection geometry at key locations along the corridor.
- Incorporating an Access Management Plan for Red Lodge, which was supported by the Red Lodge City Council in March 2007. See Appendix A, Letter #8.
- Providing pedestrian and bicycle facilities in Red Lodge.
- Providing wider roadway shoulders in rural segments where appropriate.
- Flattening ditch slopes in rural segments where appropriate.
- Reducing clear zone encroachments in rural segments where appropriate.
- Providing increased snow storage with wider and deeper roadside ditches.

- Clearing thick brush and trees within the clear zone to increase motorist visibility of approaching wildlife.
- Providing turning lanes where needed to reduce differential speed conflicts.
- o Improving highway-related storm water drainage in Red Lodge and Roberts.

3.23 PERMITS AND AUTHORIZATIONS

The following permits and authorizations are likely to be required prior to construction:

- CWA Section 402/MPDES authorization from MDEQ Permitting and Compliance Division. The MPDES permit requires a storm water pollution prevention plan that includes a temporary erosion and sediment control plan. The erosion and sediment control plan identifies BMPs, as well as site-specific measures to minimize erosion and prevent eroded sediment from leaving the work zone.
- CWA Section 404 permit from the USACE for any activities that may result in the discharge or placement of dredged or fill materials in waters of the US, including wetlands.
- Montana Stream Protection Act (SPA 124) from the MFWP-Fisheries Division. The Montana SPA 124 is required for projects that may affect the bed or banks of any stream in Montana.
- Short-Term Water Quality Standard for Turbidity related to construction activity (318 Authorization) from the MDEQ-Water Quality Bureau for any activities that may cause unavoidable violations of state surface water quality standards for turbidity, total dissolved solids, or temperature.
- Floodplain Development Permit from the County Floodplain Administrator.

3.24 ENVIRONMENTAL COMMITMENTS/MITIGATION

The following commitments have been made by MDT:

- MDT will attempt to meet individually with the affected property owners to discuss potential property acquisition and/or relocation impacts. MDT will make reasonable efforts to avoid and/or minimize impacts to potentially affected property owners.
- MDT will consult with affected irrigation ditch associations and other landowners/water rights holders to minimize impacts to irrigation facilities.
- If the mound septic system were impacted, MDT would relocate the system per County and MDEQ requirements.
- MDT recognizes the high frequency of AVC along the project corridor and will make a good faith effort to incorporate appropriate mitigation measures into the project design.

See Table A, Summary Comparison of Project Alternatives and Impacts, on page S-8.

Chapter 4 Section 4(f) Evaluation

4.1 INTRODUCTION

The purpose of this chapter is to evaluate the potential impacts of the Preferred Alternative, as discussed in Chapter 2, to Section 4(f) properties per 49 U.S.C. § 303. The Preferred Alternative was carried forward in this EA to meet the need of the proposed project, as discussed further in Chapter 1, by improving the safety and operational characteristics of the roadway.

4.2 SECTION 4(f) PROPERTIES

Section 4(f) of the Department of Transportation Act of 1966, as codified in the 49 U.S.C. § 303, specifies that the Secretary shall not approve any program or project that requires the use of publicly owned land from a public park, recreation area, wildlife or waterfowl refuge of national, state, or local significance, or land of an historic site of national, state, or local significance, as determined by the officials having jurisdiction thereof, unless (1) there is no feasible or prudent alternative to the use of such land, and (2) such program or project includes all possible planning to minimize harm resulting from the use.

A Section 4(f) impact, or "use," refers to a permanent, temporary, or constructive use as defined in the FHWA/FTA regulations at 23 CFR 774 Parts 15 and 17. These uses are defined as follows:

- Permanent Use Land from a 4(f) property is permanently incorporated into a transportation facility.
- *Temporary Use* There is an adverse temporary occupancy of the 4(f) property.
- Constructive Use The proximity impacts of a project on the 4(f) property are so severe that the activities, features or attributes that qualify the property or resource for protection under Section 4(f) are substantially impaired/diminished.

Section 4(f) properties along the project corridor consist of historic properties, historic irrigation ditches, and recreation areas.

4.2.1 Historic Properties

There are 19 historic properties along the project corridor, excluding historic irrigation ditches, as discussed in Chapter 3, Section 3.14.

On August 10, 2005, Section 6009(a) of SAFETEA-LU (Safe, Accountable, Flexible, Efficient Transportation Equity Act: A Legacy for Users) amended existing Section 4(f) legislation to simplify and streamline the process for projects having only *de minimis* impacts on Section 4(f) properties. Under the new provisions, once the US DOT determines that a transportation use of Section 4(f) property results in a de minimis impact, analysis of avoidance alternatives is not required and the Section 4(f) evaluation process is complete. An impact to a historic site may be determined to be a de minimis impact if the following criteria are met:

1. The process required by Section 106 of the National Historic Preservation Act results in the determination of "no adverse effect" or "no historic properties

affected" with the concurrence of the SHPO and/or THPO, and ACHP if participating in the Section 106 consultation;

- 2. The SHPO and/or THPO, and ACHP if participating in the Section 106 consultation, is informed of FHWA's or FTA's intent to make a de minimis impact finding based on their written concurrence in the Section 106 determination; and
- 3. FHWA or FTA has considered the views of any consulting parties participating in the Section 106 consultation.

Impacts to Historic Properties

Alternative A (No-Build): If no action were taken there would be no impacts to historic properties. Therefore, there would be no impacts to Section 4(f) properties.

Alternative B (Preferred): The Preferred Alternative would result in a de minimis Section 4(f) use of four historic properties: Carnegie Library, Kent Dairy Round Barn, Boyd Country Store, and the Rocky Fork Branch of Northern Pacific Railroad.

The development of the Preferred Alternative included efforts to avoid and minimize impacts to Section 4(f) properties. The Preferred Alternative would avoid impacts to the Carnegie Library structure and minimize/mitigate impacts to the Carnegie Library property, as described below:

- The Carnegie Library building would be avoided with the use of a bulbed out curbline at the 8th Street/US Highway 212 intersection.
- A sidewalk would be added on the east and west sides of US Highway 212 from 8th Street to the north; this would improve pedestrian access to the Library.
- Additional public parking spaces would be added on the west side of the Library along Oakes Avenue, which would improve vehicular access to the Library.
- The existing Mountain Ash tree on the southeast corner of the Library property would be avoided; however, the sign may need to be relocated.
- MDT will work with Carnegie Library representatives during project design to determine an appropriate treatment for the triangular area on the north end of the property (such as planting with grass seed).

MT SHPO concurred with MDT's determination of *No Adverse Effect* to the Carnegie Library site.

The Preferred Alternative would avoid impacts to the Kent Dairy Round Barn structure and minimize impacts to the property. Right-of-way impacts would be minimized with the use of a buried storm water pipe rather than an open ditch at this location. MT SHPO concurred with MDT's determination of *No Effect* to this site.

The Preferred Alternative would avoid impacts to the Boyd Country Store and minimize right-ofway impacts to this property. At this location, highway-related storm water drainage would be diverted to the west side of the roadway, eliminating the need for a full ditch on the east side of the roadway. In addition, a reversed curb would be used to delineate the Boyd Country Store parking lot from the roadway. This would improve the safety and functionality of the parking lot. A temporary construction easement would be required at this location to facilitate construction. Additionally, existing easements currently being used for transportation purposes would be incorporated into permanent right-of-way. However, the temporary construction easement would be extinguished following construction and would not be incorporated into permanent right-ofway. MT SHPO concurred with MDT's determination of *No Effect* to this site.

Additionally, the Preferred Alternative would impact the Rocky Fork Branch of the Northern Pacific Railroad in approximately eight locations. Impacts at four of these locations would be due to relocation of irrigation ditches outside of the proposed right-of-way per standard MDT procedures. At two locations, the impact would be a result of the construction of the highway ditch. One location would be impacted by the addition of a northbound passing lane. Lastly, one location would be impacted by installation of a new culvert under the railroad bed to improve storm water drainage for the community of Roberts. The site consists of approximately 22 acres, of which approximately 3.2 acres would be disturbed by the Preferred Alternative.

This site has already been significantly impacted. The bridges, rails, ties, ballast, and associated features have been long removed from the line. Also, segments of the line have been converted into local access roads, residential developments have encroached on the line, and lack of maintenance has led to deterioration of the line's integrity. The minor impacts from the Preferred Alternative would not substantially alter the railroad line's historical integrity, as it has already been significantly impacted. SHPO determined that the Preferred Alternative would have *No Adverse Effect* to the railroad.

The remaining 15 historic properties in the project corridor would be avoided by the Preferred Alternative; therefore, they would not have a Section 4(f) use.

Mitigation for Historic Property Impacts

The Carnegie Library, Kent Dairy Round Barn, Boyd Country Store, and Rocky Fork Branch of the Northern Pacific Railroad would all have de minimis Section 4(f) impacts and would not require further mitigation measures. See Appendix G, Section 4(f) De Minimis Evaluations. Additionally, to avoid impacts at the Richardson Bungalow and MT Dakota Grain Elevator, the width of the border strip between the road and sidewalk was reduced and no further mitigation measures would be required. The remaining 13 historic properties would have no Section 4(f) use; therefore, no mitigation is required

4.2.2 Historic Irrigation Ditches

There are 11 historic irrigation ditches along the project corridor: Bernhardt, Brandt, Carbonado, Drakes, Duncan-Aiken, Highline, Hoyle, Hunts, Rooney, Rule-Thompson, and Ward Ditches.

Impacts to Historic Irrigation Ditches

Alternative A (No-Build): If no action were taken, there would be no impacts to historic irrigation ditches.

Alternative B (Preferred): The Preferred Alternative would require the relocation of ten historic irrigation ditches outside the proposed right-of-way: Brandt, Carbonado, Drakes, Duncan-Aiken, Highline, Hoyle, Hunts, Rooney, Rule-Thompson, and Ward. The Bernhardt Ditch is not located within the proposed right-of-way and would not require relocation; therefore, there would be no Section 4(f) use to the Bernhardt Ditch.

Mitigation for Historic Irrigation Ditch Impacts

The 10 impacted historic irrigation ditches would all have de minimis Section 4(f) impacts and would not require further mitigation measures. See Appendix G, Section 4(f) De Minimis Evaluations.

4.2.3 Recreation Areas

There are six recreation areas along the project corridor. These include four fishing access sites, a fishing trail, and a school playground. The four fishing access sites (Horse Thief Station, Beaver Lodge, Bull Springs, and Water Birch) are owned and operated by the MT FWP. All of the fishing access sites are located outside the proposed right-of-way; however, access to these sites is from US Highway 212.

Additionally, Roberts Public School is located along US Highway 212 on the north end of Roberts. The school playground is open to the public and serves either organized or recreational purposes (walk-on activity).

The Rock Creek Fishing Trail is located in the Roberts to Boyd project segment. The trail is approximately 300 feet (91.4 meters) in length, from the rest area parking lot to Rock Creek. The Magic City Fly Fishers built and maintains the trail in cooperation with MT FWP and MDT.

Impacts to Recreation Areas

Alternative A (No-Build): If no action were taken, there would be no Section 4(f) use of the recreation areas along the project corridor.

Alternative B (Preferred): The Preferred Alternative would have no Section 4(f) use of the Beaver Lodge, Bull Springs, Horse Thief, Water Birch fishing access sites and the Rock Creek Fishing Trail. In order to avoid the Roberts Public School playground, the typical section along the school was altered to include a guardrail and 2:1 ditch slopes. Therefore, there would be no Section 4(f) use of the playground.

Mitigation for Recreation Area Impacts

There would be no Section 4(f) use of the fishing access sites, fishing trail, or school playground; therefore, no mitigation is required.

Chapter 5 Preparers and Reviewers

5.1 INTRODUCTION

The names and qualifications of the principal persons contributing information to the EA are identified in this chapter. In accordance with Part 1502.6 of the CEQ (Council on Environmental Quality) regulations for implementing the NEPA (National Environmental Policy Act), an interdisciplinary team of planners, environmental scientists, and engineers completed this detailed study.

5.2 PREPARERS AND REVIEWERS

This EA was prepared by KL&J under a contractual agreement with MDT. Listed below are those individuals with primary responsibility for preparation of this EA. See Table 5.1, Preparers and Reviewers.

Table 5.1 Preparers and Reviewers					
Team Member	Affiliation	Role			
PREPARERS					
Mike Wamboldt, PE, Project Manager	KL&J	Project Development, Senior Review			
Charlotte Brett, Environmental Planner	KL&J	Environmental Assessment Development			
Craig Kubas, PE	KL&J	Preliminary Engineering, Roadway Geometrics			
Chad Petersen, PE	KL&J	Preliminary Engineering, Roadway Geometrics			
Mark Peterson, PE	KL&J	Hydraulics/Hydrologic Analysis			
Becky Rude, Environmental Planner	KL&J	Impact Analysis			
Robert Shannon, PE, Senior Engineer	KL&J	Alternatives Analysis, Traffic Analysis			
Skip Skattum, GIS Analyst	KL&J	Impact Analysis, Noise Analysis, Exhibit Creation			
Donovan Slag, PE	KL&J	Preliminary Engineering, Traffic Analysis			
Lauri Travis, Ph.D.	Metcalf Archaeological Consultants, Inc.	Cultural Resource Survey			
Joan Brownell	Independent Contractor	Historic Inventory			
John Morrison	Earthworks	Cultural Resource Survey			
Carol K. Lee-Roark, Ph.D., PG	Hyalite Environmental, LLP	Hazardous Materials Survey			
Jeff Berglund, Senior Wetland Scientist	PBS&J / Land and Water Consulting, Inc.	Biotic & Wetland Surveys			

Continued...

Table 5.1 Preparers and Reviewers					
Team Member	Affiliation	Role			
	REVIEWERS				
Alan C. Woodmansey, PE	FHWA	Operations Engineer, Lead Agency			
Ted Burch	FHWA	Project Development Team Lead			
Tom S. Martin, PE	MDT	Bureau Chief, Senior Review			
Gabe Priebe, PE	MDT	Consultant Design Project Manager			
Heidy Bruner, PE	MDT	EA Review			
Tom Gocksch, PE	MDT	Environmental Project Development Engineer, EA Review			
Bill Semmens	MDT	Wetlands and Wildlife Impacts, EA Review			
Cora Helm	MDT	Noise and hazardous materials, solid waste, and underground storage tank impacts, EA Review			
Jon Axline	MDT	Cultural Resource/Section 4(f) Impacts, EA Review			
Stefan Streeter, PE	MDT	Billings District Administrator			
Gary Neville	MDT	Billings District Engineering Services Supervisor			
Chapter 6 Comments and Coordination

6.1 INTRODUCTION

This chapter provides information about the coordination efforts with agencies and interested parties, which have been ongoing through the development of this EA.

6.2 COOPERATING AGENCIES

The City of Red Lodge and Carbon County are both cooperating agencies for this project. MDT has coordinated with Red Lodge and Carbon County throughout the development of the project. Both cooperating agencies have sent written correspondence identifying needs along the project corridor, which MDT has addressed with the Preferred Alternative. See Appendix A, Cooperating Agencies.

In addition to the written correspondence, MDT has held numerous meetings with the cooperating agencies, as summarized in the following section.

6.2.1 Meetings with the City of Red Lodge

MDT held an Alignment and Grade review meeting with the City of Red Lodge on March 28, 2003 at the Carbon County Courthouse in Red Lodge. The meeting consisted of an overall project review followed by discussion of proposed alternatives and options and a review of the alignment and grade plan set. Nineteen people attended the meeting.

On February 25, 2005, MDT met with the City of Red Lodge to discuss preliminary design concepts for the project corridor within Red Lodge. At that meeting, the City Administrator expressed interest in the concept of a roundabout at the Highway 78 intersection.

MDT met with Red Lodge again on June 28, 2006 to update the City Council on alternatives being considered within Red Lodge. At this meeting, previous alternatives and options that had been evaluated were summarized, along with the current preferred alternative. The City Council had various questions and comments about the preferred alternative; following the meeting, the Red Lodge City Planner indicated that the City wished to see other options for managing access in the developing area in north Red Lodge as well as for Oakes Avenue.

On August 10, 2006, MDT met with the Red Lodge City Planner and the Red Lodge Public Works Director. The primary purpose of this meeting was to discuss Oakes Avenue, access management in north Red Lodge, and potential impacts to land use development in north Red Lodge.

MDT met with the City of Red Lodge, CTA Architects Engineers, and other interested parties on January 18, 2007 to discuss the ongoing Red Lodge Streetscape Plan and its compatibility with this proposed project.

Following two public meetings to discuss access management in north Red Lodge (discussed further below), MDT met with the Red Lodge City Council on February 27, 2007. MDT presented the Access Management Plan to the Council and requested approval of the plan.

On June 27, 2008, MDT met with the City of Red Lodge to discuss the City's comments on the Administrative Draft EA.

6.2.2 Red Lodge Resolutions

The City of Red Lodge passed Resolution No. 3223 on October 10, 2006. This resolution identified support for the preferred alternative within Red Lodge, with the exception of the method of access management proposed for the developing area in north Red Lodge (at that time, a TWLTL was proposed).

Following the development of the Access Management Plan, the City of Red Lodge passed Resolution No. 3228 on March 27, 2007. This resolution identified support for the preferred alternative within Red Lodge.

6.2.3 Meetings with Carbon County

On March 1, 2007, MDT held an informational meeting with the Carbon County Commission and interested members of the public to talk about the proposed improvements in Boyd.

MDT held a follow-up meeting with the Carbon County Commission on March 29, 2007. The purpose of this meeting was to further discuss the Dakota Avenue intersection in Boyd.

6.3 COORDINATION WITH OTHER AGENCIES, ORGANIZATIONS, AND INTERESTED PARTIES

To ensure that social, economic, and environmental impacts are considered in the development of the EA, and pursuant to Section 102 (2) (D) (IV) of NEPA, MDT has also coordinated with other agencies, organizations, and interested parties throughout the development of the EA.

In March 2002, MDT mailed a scoping package to numerous local, State, and Federal agencies to determine the consistency of the project with current and proposed plans, programs, and policies. Their letters provided valuable insight into the evaluation of potential environmental impacts. Their instructions were referenced and incorporated where appropriate within the environmental impact categories in the previous chapter. See Appendix C, March 2002 Agency Scoping.

Due to modifications in the proposed project and the passage of time, agencies and interested parties were solicited again in May 2007. This time, agencies were asked to provide updates or further information regarding the proposed project. See Appendix D, May 2007 Agency Scoping.

The following agencies were consulted concerning potential impacts from the proposed project. *See Table 6.1, Agencies Consulted.*

Table 6.1 Agencies Consulted
Federal
U.S. Army Corps of Engineers
Helena Regulatory Office
Omaha District Planning Branch
U.S. Department of Agriculture
NRCS (Natural Resource Conservation Service)
U.S. Department of the Interior
Bureau of Indian Affairs/Environmental Services Fish & Wildlife Convices/Foolegies/Foolegies/
Fish & Wildlife Service/Ecological Service
• Montana Operations Office
Noticinal Environmental Policy Act Unit
State
Montana Department of Environmental Quality
Director's Office
 Permtting and Compliance Division
Remediation Division
Water Protection Bureau
Montana Department of Fish, Wildlife & Parks
Fisheries Division
Parks Division Region 5 Headquarters
Montana Department of Natural Resource Conservation
Southern Land Office
Special Use Management Bureau
State Water Projects Bureau
Water Resources Division
Montana State Library
Natural Heritage Program
Regional
Rock Creek Irrigation Commission
County
Carbon County Clerk
Carbon County Commissioners
Carbon County School Superintendent
Carbon County Sheriff
Carbon County Treasurer
Municipality
Red Lodge City Clerk
Red Lodge City Council
Red Lodge Director of Public Works
Red Lodge Fire Department
Red Lodge Mayor
Red Lodge Planning and Zoning
Red Lodge Park Board

Table 6.1 Agencies Consulted
Municipality
Red Lodge Police
Roberts School Board
Private
BNSF Railway Company
Consolidated Ditch Company
Finn Ditch Company
Montana Land Reliance
Red Lodge Chamber of Commerce
Rock Creek Water Users

6.4 COORDINATION WITH THE PUBLIC

MDT attempts to provide accommodations for any known disability that may interfere with a person participating in any service, program, or activity of the Department. Alternative accessible formats of public meeting materials were made available if requested and reasonable accommodations were provided if the request was made within 48 hours of a meeting.

6.4.1 Kickoff Meeting

MDT held a public Kickoff Meeting at 6:30 p.m. on March 27, 2002 at the Red Lodge Senior Center in Red Lodge. Notice of the meeting was given to the City, County, and State officials by letter. The meeting was also advertised once per week for two consecutive weeks prior to the meeting date in the Billings Gazette and the Carbon County News. Additional news releases were circulated to local television and radio stations. The intent of this meeting was to inform elected officials and local, State, Federal, and regional agencies, as well as the public, of the project and to obtain local knowledge of concerns related to the proposed study. This meeting also served as an early notification of the preparation of an EA.

Forty-three members of the community and government agencies attended the meeting. Fifteen written comments were received concerning the project; the majority of which included requests for pedestrian and bicycle facilities in Red Lodge and a wider roadway. In general, public comments identified the desire for improved safety and traffic operations along the project corridor.

6.4.2 Alternatives Public Workshop

MDT held an Alternatives Public Workshop at the Roberts School Cafeteria in Roberts on November 6, 2002 at 6:00 p.m. This meeting was held to inform the public of alternatives being considered for the project and to obtain public input. Notice of the meeting was given, as before, to the City, County and State officials by letter. The meeting was again advertised once per week for two consecutive weeks prior to the meeting date in the Billings Gazette and the Carbon County News. News releases were also circulated to local television and radio stations.

Twenty-five people attended the Alternatives Public Workshop. Twelve written comments were received; the majority of which again included requests for a wider roadway.

6.4.3 Storm Water Drainage Meetings in Roberts

During the project planning process, MDT received public comments from property owners residing in the community of Roberts regarding concerns for highway-related storm water drainage. Comments of this nature were received throughout the development of the EA, primarily in response to flooding in Roberts in May 2005 and June 2007. MDT held two public information meetings regarding this issue. The first was held at the Roberts Public School gymnasium on May 16, 2006 at 7:00 p.m. The second meeting was held at the United Methodist Church in Roberts on December 18, 2007 at 6:30 p.m. The meetings were held to discuss the proposed improvements to highway-related storm water drainage along the project corridor in Roberts. Notices were posted in the Carbon County News prior to the meetings. Twenty-seven people attended the first meeting, and three written comments were received following the meeting. Approximately 25 people attended the second meeting, and no written comments were received.

6.4.4 Access Management in North Red Lodge – Meeting #1

MDT held a property owner meeting at the Red Lodge Senior Center on November 14, 2006 at 6:30 p.m. This meeting was held with potentially affected property owners to discuss current and future access needs from MT Highway 78 to Two Mile Bridge Road. Notice of the meeting was given to the City, County, and State officials and potentially affected property owners by letter. In addition, phone calls were placed to property owners notifying them of the meeting. The meeting was advertised prior to the meeting date in local newspapers.

Ten members of the public and government agencies attended the meeting. One written comment was received. The comment suggested half access between the developed limits of Red Lodge and Two Mile Bridge Road along with considerations for aesthetics, speed limit, and maintenance.

6.4.5 Access Management in North Red Lodge – Meeting #2

MDT held a second meeting regarding access management in north Red Lodge at the Red Lodge Senior Center on January 17, 2007 at 6:00 p.m. The purpose of the meeting was to present potential alternatives for access management along MT Highway 78 to Two Mile Bridge Road and to solicit public feedback. As before, a letter of notification was given to the City, County, and State officials and potentially affected property owners. The meeting was advertised for two consecutive weeks prior to the meeting date local newspapers.

Twenty-three members of the public and government agencies attended the meeting. Five written comments were received. Comments included suggestions for roundabouts, median options, access management, and speed limits.

6.4.6 Other Meetings

On January 31, 2007, MDT held a meeting with the architect working with the City of Red Lodge on the remodeling and layout for the new Bank of Red Lodge adjacent to the Oakes Avenue intersection. Others in attendance were the Red Lodge City Planner and other interested parties in the new development. The purpose of the meeting was to discuss when the US Highway 212 project would be constructed, the amount of right-of-way required for the proposed project, and possible layouts for the bank site plan. No written comments were received, but the preferred alternative reflects the discussion at this meeting.

On February 16, 2007, MDT held a meeting at the Billings District office with the owners and representatives of the proposed Beartooth Hospital, located north of the golf course in Red Lodge. The meeting was an effort to coordinate the proposed US Highway 212 project and right-of-way requirements with the future site plan of the Beartooth Hospital. No written comments were received, but the preferred alternative reflects the discussion at this meeting.

A meeting was also held with the Red Lodge Fire Department to discuss access for fire trucks between the Fire Station and the proposed roundabout at the MT Highway 78 and US Highway 212 intersection.

6.4.7 Public Hearing

Two Public Hearings are planned for this project, one in Red Lodge and the other in Roberts. A Notice of Availability of the EA and Public Hearing dates will be advertised in local newspapers.

Chapter 7 Reference List/Source Documents

7.1 INTRODUCTION

This chapter provides information on subconsultant, community plans/policies, websites and other source documents relied upon in the development of this EA.

7.2 SUBCONSULTANT REPORTS

The following subconsultant reports were prepared for this EA:

- Hyalite Environmental, LLP. Initial Site Assessment Report, Red Lodge North Reconstruction (Dec. 2001)
- Land & Water Consulting, Inc. Red Lodge-North Final Biological Resources Report (April 2004)
- PBS&J / Land & Water Consulting, Inc. Red Lodge-North Final Biological Resources Supplement (Anticipated Oct. 2007)
- Historical Research Associates, Inc. Report of a Cultural Resource Inventory of the Red Lodge-North Project Corridor, Along Highway 212 in Carbon County, Montana (Feb. 1997)
- Metcalf Archaeological Consultants, Inc. Red Lodge North, Highway 212 Improvements: A Cultural Resource Inventory from Roberts to Boyd, Carbon County, Montana (Oct. 2002)
- Joan Brownell. Historic Inventory: Red Lodge North Project, Carbon County, Montana; Project STPP 28-2 (25) 70 (Feb. 2003)
- Earthworks, Inc. *Highway 212 Red Lodge North: A Cultural Resource Inventory, Carbon County, Montana* (June 2007)

7.3 OTHER REPORTS

The following reports were relied upon in the development of this EA:

- o Kadrmas, Lee & Jackson. Alignment and Grade Traffic Plans (Nov. 2002)
- o Kadrmas, Lee & Jackson. Location Hydraulic Study Report (Nov. 2002)
- Kadrmas, Lee & Jackson. *Preliminary Traffic* (Nov. 2002)
- o Kadrmas, Lee & Jackson. Preliminary Hydraulic Report Revised (Feb. 2003)
- Kadrmas, Lee & Jackson. US Highway 212 Traffic Study, STTP 28-2(25)70 Control #4375 (March 2004)
- Montana Department of Transportation. US 212 Red Lodge to Laurel: Speed Limit Investigation (Dec. 2003)
- Montana Department of Transportation. Corridor Study Red Lodge North Collision Analysis: January 1, 1992 to December 31, 2001 (2001)
- Romin, L.A. and J.A. Bissonnette. *Deer-vehicle collisions: Status of state monitoring activities and mitigation efforts.* Wildlife Society Bulletin 24 (2):276-283 (1996)
- Southern Rockies Ecosystem Project. Linkage Assessment Methodology, Linking Colorado's Landscapes Phase II Report. Southern Rockies Ecosystem Project. Denver, CO (2006)

7.4 COMMUNITY PLANS/POLICIES

The following community plans/policies were relied upon in the development of this EA:

- City of Red Lodge Comprehensive Trails Plan (May 2006)
- o Carbon County Growth Policy (Sept. 2003)
- Red Lodge Growth Policy (May 2001)
- Red Lodge City Council. *Resolution 3223* (Oct. 2006)
- Red Lodge City Council. *Resolution 3228* (March 2007)

7.5 WEBSITES

To obtain this EA online and comment on the EA, please visit the following address: <u>http://www.mdt.mt.gov/pubinvolve/eis_ea.shtml</u>.

The following websites were also used to compile this EA:

- <u>http://www.co.carbon.mt.us/</u> (Carbon County)
- <u>http://www.cityofredlodge.com/</u> (City of Red Lodge)
- <u>http://redlodgestreetscape.com/</u> (Red Lodge Streetscape Project)
- o http://www.carnivoresafepassage.org/ (Carnivore Safe Passage)
- <u>http://www.epa.gov/enviro/</u> (EPA Envirofacts Data Warehouse)
- o http://www.deq.state.mt.us/ (MDEQ)
- o <u>http://nhp.nris.mt.gov/</u> (Montana Natural Heritage Program)
- <u>http://maps2.nris.mt.gov/mapper/</u> (Montana Natural Resource Information System Digital Atlas of Montana)
- o http://www.census.gov/ (US Census Bureau)
- <u>http://fwp.mt.gov/default.html</u> (MT FWP)
- o <u>http://www.mdt.mt.gov/</u> (MDT)
- <u>http://www.fws.gov/montanafieldoffice/Endangered_Species/Listed_Species.html</u> (USFWS Listed Species of Montana)
- o http://www.wildlifecrossings.info/beta2.htm (Wildlife Crossings Toolkit)

Appendix A

Cooperating Agencies



BOARD of COMMISSIONERS

COUNTY OF CARBON . STATE OF MONTANA

Post Office Box 887 Red Lodge, MT 59068 Phone: (406) 446-1595 Fax: (406) 446-2640

LOWING

115311177

Received

MAR 2 0 2002

Kadrmas Lee & Jackson

Mr. Mike Wambolt, Project Engineer Kadrmas, Lee and Jackson P.O. Box 80303 Billings, Montana 59108

Subject: Project STPP 28-2 (25) 70 Control No. 4375 US Highway 212 from Red Lodge to Roberts

Dear Mike,

14

March 19, 2002

We appreciate your early involvement of Carbon County in this project. We will be interested in approach design onto U. S. 212 from any Carbon County road including the streets of Roberts. We do not know of any other Carbon County property that will be affected, and do not anticipate any problems. We look forward to the completion of this project.

Carbon County is working with the Roberts Community Foundation on a side walk project utilizing CTEP funds. A part of the project involves drainage of water that would normally follow First Street and directing it toward U.S. 212. We are working with MDOT on this issue. The engineer working for Carbon County on this project is Bruce McKee, of McKee Engineering of Red Lodge, Montana. Carbon County will need to place larger culverts at our approaches to U.S. 212 to make this work.

Carbon County would also like to request Cooperating Agency status for this project in accordance with U.S. Department of Transportation, Federal Highway Administrations (FWHA) regulations (23 CFR 771.111 (d).

Thank you for your attention.

John E. Prinkki Carbon County Commissioner

ce: Bruce Barrett, Administrator - MDT Billings Dist.

Kadrmas Lee &	MEM
Jackson	Date:
Engineers Surveyors Planners	То:

Date:	June 28, 2007
То:	Charlotte Brett
Сору То:	File
From:	Mike Wamboldt
Re:	Red Lodge North; Project STPP 28-2(25)70, UPN 4375; Dakota Avenue, Boyd

The purpose of this memo is to document a phone conversation between myself and Dave Davidson, Carbon County Commissioner on June 28, 2007.

I spoke with Dave regarding what the commissioners had decided they would like to see happen at Dakota Avenue in Boyd. Dave confirmed that the County had spoken with their constituents and decided:

- 1. They would prefer to keep the intersection open.
- 2. Their main concern is with the school bus traffic that circulates through Boyd and with being able to have fire access at that location.
- 3. The county said they would be OK with a right-in/right-out access, or at minimum a turnaround within that approach.

I told Dave that we had looked closer at a right-in/right-out access at this location and determined that it couldn't be accomplished well without constructing a raised median. The raised median would be unsafe due to the speed of traffic and the lack of curb and gutter at this location. We agreed to leave the approach as it exists today.

 701
 845
 4980

 1010
 4th
 Avenue
 SW

 PO
 Box
 937

 Valley
 City, ND
 58072-0937

 Fax
 701
 845
 0252

 www.kljeng.com
 Kadtmas, Lee & Jackson, Inc.
 A

 A
 KLJ
 Solutions
 Company

Elstate/mdt/2201102/Correspondence/Davidson_Memo_Dakota Dr.doc



MAR February 28, 2005

Bruce Barrett, District 5 Administrator Montana Department of Transportation P.O. Box 20437 Billings, MT 59104-0437



Dear Mr. Barrett:

Re:

U.S. Highway 212 Corridor Study STPP 28-2(25) 70

Thank you for taking the time to visit with us this past week about the abovereferenced project. We appreciate your candor and flexibility in dealing with the community's changing needs. As you saw from the meeting, the same development pressures that drove the City to request assistance from Montana Department of Transportation are still with us.

In 1999 the City of Red Lodge ("City") began experiencing development and expansion issues on the north end adjacent to U.S. Highway 212. The development requires increased services, and has the potential to create serious conflicts with future improvements along this main access route into the City from Billings.

Subsequently, The Red Lodge City/County Planning Office asked the MDT to provide a design for U.S. Highway 212 to accommodate these changing needs. The area of concern is on the north end of Red Lodge from approximately 5th Street to the Two-Mile Bridge road.

The City specifically asked that the design include future roadway elevations, curb and gutter, sidewalks, storm drain, right-of-way widths, and approach locations. It is the City's intent to use the completed design as a guide for growth policies, community development, and permitting requirements. The City may require a developer to construct some of the features as a permit requirement. With the assistance of MDT, the City expects to accommodate new development, while preserving the character and aesthetics all along the corridor.

On behalf of the City, I want to express our appreciation for MDT's assistance in this planning effort, and anticipate continued mutual cooperation on this project.

City of Red Lodge By: Richard C. Gessling, D.D.S. Mayor, City of Red Lodge file

cc.



MAY 0.9 2005

May 5, 2005 SOUND A DEPI OF TRANSPORTATION SOUND A DEPI OF TRANSPORTATION SOUND SOUND AND A DEPI OF TRANSPORTATION Bruce Barrett, District 5 Administrator Montana Department of Transportation P.O. Box 20437 Billings, MT 59104-0437

Re: <u>Montana State Highway 78</u> STPP 78-1(8)0; Control #4890

Dear Mr. Barrett:

After seeing a notice of preliminary plans being available for the above-captioned project. I contacted your staff and received a Memorandum dated October 25, 2004 ("the Memo"). Please let me use this opportunity to express my appreciation for the many acts of cooperation you and your staff have shown me in the brief time I have been with the City of Red Lodge.

The Memo provides me with a background and status report on this project 1 appreciate having. Development within the project area is moving along very quickly and changes occurring just since the date of the Memo may want to be considered in updating the analysis for this project. The City is budgeting to do a comprehensive recreational trails plan this coming fiscal year, and the notation in the Memo for separated trails in the project area will be taken into consideration as we develop that trails plan.

As you know, I have asked MDT to consider a roundabout at the intersections of Highways 78 and 212. In looking back at the Capital Improvements Plan ("CIP") done by the City in 1997, I found that concerns over that intersection were being expressed even then, and traffic patterns, traffic loads and the need for better traffic flow patterns has only become more obvious with time. Since this intersection is the eastern-most point for the project, I reiterate my request for consideration of a roundabout at this intersection.

Although I alluded to the possibility of moving the intersection either north or south previously, events are transpiring that might influence your considerations for locating this intersection. Red Lodge School District #1 is considering a new high school east on 1st Street from Highway 212, and the owner of the trailer court west of that intersection would be interested in re-developing that property. Serious consideration needs to be given to re-locating the intersection of 78 and 212 to 1st Street rather than its present location on 3st Street.

inc mgnway passes through a draw in the bench, which contains both a pond and a house. This area gets very little sun in the winter, and often has ice flows develop on the highway. The house is now on the market, and MDT should seriously consider purchasing the property to allow better alignment of the highway, which will also remove a serious traffic hazard posed by the lack of parking for that house.

The former "gravel pit" property at the top of that draw is also for sale and development plans are in the offing for that property. Re-alignment of the road at the top of the draw may also help alleviate ice build-ups and allow for a grade reduction from the present 11%. The entrance for the gravel pit is presently at a point where three (3) roads-all intersect, and all of them are on downhill grades at this intersection. Since this intersection is on a north-sloping hillside it too creates hazards for drivers in the winter so we would ask consideration be given to recommendations for how we might mitigate or even alieviate this problem. Development on the west bench will only make this intersection more of a problem in the years this project is suppose to cover.

The 1997 CIP recommends construction of a road between the Ski Hill Road on the South and highway 78 on the North. Presumably, this road would follow along the west side of the airport and fairgrounds properties and then along the edge of the west 'bench before intersecting with the Ski Hill Road at Tipi Village. I note the 10/24/04 Memo envisions widening at least one curve (a good plan), but I am wondering if MDT could also provide professional assistance in how and where we might create a better flow of traffic by acting on the link recommended in 1997. This would not be a so-called "by-pass", but rather a connecting artery between two roads now under-going reengineering studies to help disperse increasing traffic due to development occurring west and north of Red Lodge.

Another connection MDT might consider in dispersing traffic would be a connection between highways 212 and 78 north of the country club estates. Recently, the local hospital decided to re-locate its operations to a site at the north end of the golf course in country club estates on Highway 212, so bringing a road in from the west to this point would provide alternative emergency access.

Please accept this letter as comments by the City on the October 24, 2004 Memo. They are neither exhaustive, nor final comments, but rather part of an on-going attempt by the City to provide positive informational input for the process. We appreciate any consideration you might give this letter.

City of Red Lodge

₽v Rodney B. Protitt Office of the Mayor

cc. file



June 13, 2005

Received

JUN 15 2005

Kadrmas Lee & Jackson

Sheri G. Lares Kadrmas, Lee & Jackson, Engineers P.O. Box 80303 Billings, MT 59108-0303

Re: <u>MDOT Project – Red Lodge North Design Study</u> Request for Cooperating Agency Status

Dear Ms. Lares

Please accept this letter in confirmation of our phone conversation of late last month. The City has received your letter of May 19, 2005 and would appreciate being designated as a cooperating agency for the project STPP 28-2 (25) 70. Please send all background information available on this project to date so that we might be better acquainted with the project, its status, and the scoping done on this project to date.

We are advised this project will entail reconstruction of the right-of-way in two or more phases. The work will be phased from the City of Red Lodge to the unincorporated area known as Roberts. The first phase would extend from the City to mile post 71.54, and will include everything north of the intersection of 8th Street and Broadway Avenue within the City of Red Lodge. A letter outlining some of the City's concerns and willingness to participate in planning for this project was sent to Bruce Barrett on or about December 8, 2004 with subsequent follow-up communication. Some of the comments made previously follow.

The public meeting conducted in Red Lodge made clear the project's southern starting point was an important factor for the City. Every intersection within the project area has issues that need to be dealt with in this project. At least those areas south from the intersection of highways 78 and 212 are expected to re-engineered as urban corridor. The area north from there to the Two Mile Bridge is no longer rural, but an emerging growth area for the City.

 The City has an historic lighting project in progress. This lighting project will include the area south from Eighth Street. The City would ask that the project include compatible lighting, sidewalks, and curb & gutter to the highway's intersection with State Highway 78 and all other lighting contemplated also be as compatible as possible to the character of the City.

- The City just held a public forum on streets and sidewalks, and several people
 mentioned the drainage problems. Including the area north from Eighth Street in
 this project would alleviate some of the drainage problems the public has brought
 to our attention, and could be the basis we need to work from in developing a
 drainage plan for the City.
- The local hospital has a purchase agreement for land to build a new hospital, and associated care facilities just north of the REA building. The hospital wants city services to this site. The site under contract lies between the golf course and the northerly City limits on the west side of U.S. Highway 212; all within the City's jurisdiction.
- The hospital anticipates not just heavy traffic in and out of its facility, but also
 emergency traffic from both the north and south. The City is reviewing its options
 in extending at least one street south to alleviate some traffic congestion, but we
 are not all sure we will be able to acquire all the right-of-way to accomplish this.
 The City recommends MDT design the highway to be compatible with needs of
 this hospital project and the development this project is likely to spawn.
- The right-of-way will necessarily require consideration of three lanes, turn lanes, acceleration lanes, drainage, utility corridors, and limited access points to assure good and extended vision. The hospital projects it will need city services by late 2006 so the City would appreciate highway improvements by that date also. At the very minimum if the right-of-way issues could be resolved to allow utility easements, it would helpful. It would also be helpful to us to know what MDT's timeline for reconstructing phase 1 is so we can keep the hospital informed.
- The City understands certain property owners along the highway have taken possession of the former railroad right-of-way, which we believe to be public land suitable for a separated recreational trail. Red Lodge is very much interested MDOT assert whatever ownership rights may exist, and incorporate a trails component to the design of this roadway.
- Emergency vehicles are entering the intersection of highways 78 and 212 under all conditions. At the same time, we have tourists stopping for the chamber and people trying to access businesses and the downtown. We have also been contacted by a developer inquiring about putting in multi-use, high density development where the trailer court is (just north and west of the intersection) and we would want to plan for that eventuality as well. A traffic study is needed to determine just what is needed.
- The City has initiated discussions with the Carbon County Arts and with the owner of what is commonly referred to as the "Pony Express" building about those properties lying between Lions Club park (west) and the highway between 6th and 8th streets. The City is interested in providing public restrooms for the park, expanding parking for the businesses located in this area, providing better sidewalk access in this area, and supporting our arts and crafts community.

 The City is very concerned about the traffic patterns at the interchange of highways 78 and 212. This intersection needs a complete overhaul. Please consider putting in a "round-about", recommending closure of certain city streets at that same location, and other possible alternatives to the present situation. We have been talking to the owner of the trailer park on the north, and the City owns land south of the intersection, so adjustments could be made either north or south to create a safer, more efficient, intersection.

Thank you in advance for your kind consideration of the City's concerns. We appreciate Gary Neville's efforts to arrange a meeting to discuss these points with more specificity. Please feel free to contact me with any questions or comments on this letter at your convenience.

City of Red Lodge

By: Rodney B. Proffitt Office of the Mayor City of Red Lodge e-mail: RLadmin@vcn.net

cc. Richard C. Gessling, Mayor

file

CHAYON ROLL MILLO Montana 590 SEP 19 2005 ()nt Desig: MASTER PARAP 対わた Litregy Chief Bruce Barrett, District 5 Administrator Consultant plans Eng Ocsign Superviso: Montana Department of Transportation CTEP Engineer P.O. Box 20437 Billings, MT 59104-0437 TIM KELLY (K17) Re: Montana State Highway 78 Response to Notice of 9/14/05 JOHN SHOFF Dear Mr. Barrett: (HK M) Notice of a public meeting to discuss the above-captioned project appeared in the Billings Gazette, and this letter is written in response to that notice. The City of Red Lodge has already requested cooperating agency status with regard to re-engine the

plans for U.S. Highway 212 north of Red Lodge, and has sought to participate in planning for reconstruction of the Beartooth All-American Road and the Ski Run Road.

With so many different projects being implemented within such a short time span, the City very much wants to make sure planning going into these projects is coordinated, and adapts these access corridors to the needs of the City for the 21st century rather than simply re-working transportation corridors for needs long abandoned from the 19th Century. Needs have changed a great deal for this community since the days of the railroad and coal mining.

The City recently witnessed the dire economic consequences resulting from the loss of access to this community from a natural disaster. Now, planning is underway for closure of key transportation access points from these projects. The City is concerned about negative impacts these projects will have on the local economy; and therefore, request that substantive efforts be undertaken to minimize the impediments to traffic, and expedite construction during the key summer tourist season. Any detours, closures, etc. necessary to construction need to be coordinated with local authorities to assure that economic impacts are mitigated to the fullest extent possible.

The State Highway 78 corridor is seeing a great deal of development. Although the corridor is becoming increasing urbanized, the County continues to conduct land use planning as it did fifty years ago. One of the few influences on how development will occur in outlying areas to the City is transportation, and we are hopeful, this project will consider the urbanization of this transportation corridor. The City would request consideration of a north-south roadway at just west of the Red Lodge Country Club extending past the fairgrounds to the Ski Run Road; and in the alternative for a re-routing of 78 to downtown Red Lodge (business route). The City would also ask for consideration of a re-routing 78 along the north side of the Red Lodge Country Club to U.S. Highway 212.

Capes To Council And Cescol Republic Control Cescol Republic Control Cescol WHALLE NOTH

Just west of the aforementioned intersection, Highway 78 begins a steep incline to reach the top of the bench. The grade through this climb makes it treacherous driving, but other hazards also exist along this stretch of highway. The City alerted MDT months ago that the house just south of the highway on the grade was for sale along with 4 acres. No action was taken and the land was sold. It is unfortunate because the location of this house creates ice flows onto the highway, and there is a pond on the property that poses significant issues for the highway as well.

The City has three (3) intersections along the aforementioned incline. One intersecting street is not on the City plat and the City would like for MDT to advise the City on whether this intersection should remain open. This is Word Avenue. It is a blind intersection, very narrow, and not necessary to the City's transportation planning.

The second existing intersection is at Airport Road and 78. Airport Road enters on the down-hill from the south, which makes it hard to keep dry in winter. This intersection may soon have an off-set intersecting access from development planned for what is commonly known as "the gravel pit". This would mean that three roads are all intersecting at the same location and all are downhill, narrow, roadways.

The final intersection is at Lazy M Street and 78 almost at the top of the aforementioned grade up to the west bench. This last intersection is a major collector street for the Red Lodge Country Club subdivision and gets a lot of traffic, which will continue to increase in coming years. The intersection is also at the end of a long downhill curve for traffic coming from the west. The City believes this is potentially a dangerous intersection. Visability for traffic entering onto 78 from Lazy M to go east is particularly difficult in the mornings. The City would request recommendations from MDT how we might partner with MDT to relocate or at least mitigate these issues as to all three of these existing intersections, and how best to locate the roadway yet in the planning stage.

The Beartooth Hospital has recently purchased land at the north edge of the City. In its present location, emergency vehicles could bypass the downtown by using the Airport Road to reach the hospital, but once relocated, the emergency vehicles will have a more convoluted route. The City is most concerned about the most likely new route to the hospital, because ambulances will soon be passing the Fire Station at the corner of highways 78 and 212, which could exacerbate a high-risk intersection. The City has already asked that a roundabout be considered at that intersection in the 212 planning, and we hope that plans for this project will cooperate in evaluating opportunities to better control and disburse traffic at that intersection.

Finally, the City has budgeted to do a comprehensive trails plan for the City this year. In a previous memorandum, mention was made of the City's desire to have a separated trail along 78 so I understand MDT is aware of the City's intentions. Any input into our process of developing a trails system MDT has to coordinate our efforts to mutual benefit would be appreciate. Once the trails plan is completed the City would have the expectation that MDT would follow it as much as possible during this project and accommodate the plan into the design of the re-engineered highway.

÷.

Please accept this letter as preliminary comments providing direction to scooping for the projects enumerated earlier in this letter. They are neither exhaustive, nor final, but rather part of an on-going attempt by the City to provide positive informational input for the process. If cooperating agency status is available that would allow the City to become a full participant in this project planning, we would appreciate being considered for that status.

The City appreciates any consideration the Montana Department of Transportation gives this letter. We look forward to working with you on planning, design and construction of this project and the others in and around Red Lodge over the next few years.

City of Red Lodge

By: < Rodnoy B. Proffitt Office of the Mayor

cc. Richard C. Gessling, Mayor
 Jim Lynch, Director MDT
 Tom Kaiserski, Office of the Governor
 file

RESOLUTION NO. 3223

A RESOLUTION OF THE CITY COUNCIL IDENTIFYING AND SELECTING CERTAIN PREFERRED ALTERNATIVES PROPOSED BY THE MONTANA DEPARTMENT OF TRANSPORTATION FOR MDT PROJECT # STPP 28-2 (2570), KNOWN AS THE CORRIDOR STUDY RED LODGE NORTH.

WHEREAS, the Montana Department of Transportation (MDT) has proposed rebuilding highway 212 that runs through the City of Red Lodge and has requested the City Council's input and selection of certain preferred alternatives for specific areas and intersections within and/or near the City; and

WHEREAS, The City Council appreciates the opportunity to cooperate with MDT to identify preferred alternatives for the 212 Corridor Project prior to when the designs are prepared and finalized.

WHEREAS, based on the alternatives provided by MDT, the City Council hereby selects and supports the following preferred alternatives:

- The one-way design alternative, as generally described by the attached concept, for Oakes Avenue between 8th and 7th Streets, with that one way being southbound, to improve the safety and enjoyment of pedestrian traffic as well as to increase opportunities for parking and a flexible street scene for civic and cultural events, with parking being angled.
- Construction of a roundabout at the intersection of Highways 78 and 212, to realize the safety advantages that roundabouts provide, capture the cost advantages of roundabouts over time, and provide a distinctive entrance into the City of Red Lodge.
- 3. Further exploration of alternative options to a continuous two-way left turn lane from the intersection of Highways 78 and 212 to the Two-Mile Bridge Road and continuing North, while considering the current transportation design and the future impact the proposed design may have on current planned and unplanned land use patterns for the North Corridor of Red Lodge.
- 4. Creation of a controlled access corridor, designed appropriately through consultation and future legislation by and between the City Council and Carbon County to identify, adopt and implement appropriate and sufficient land use regulations to support a controlled access roadway along the 212 Corridor, with the acknowledgement that MDT has no authority to create or enforce land use regulations in either the City or Carbon County. The concept for a controlled access corridor includes identifying appropriate spacing for vehicular access.

NOW THEREFORE, BE IT RESOLVED, that the City Council hereby supports the preferred alternatives listed herein and further requests that MDT consider the City's selection of preferred alternatives when designing the overall project.

PASSED and APPROVED by the Red Lodge City Council this 10^{+-} day of September 2006.

FOR THE CITY OF RED LODGE, MT

By: Brian C. Roat, Mayor Attest:

Debbie Tomicich, City Clerk

RESOLUTION NO. 3228

A RESOLUTION OF THE CITY COUNCIL IDENTIFYING AND SELECTING CERTAIN PREFERRED ALTERNATIVES PROPOSED BY THE MONTANA DEPARTMENT OF TRANSPORTATION FOR MDT PROJECT # STPP 28-2 (2570), KNOWN AS THE CORRIDOR STUDY RED LODGE NORTH.

WHEREAS, the Montana Department of Transportation (MDT) has proposed rebuilding highway 212 that runs through the City of Red Lodge and has requested the City Council's input and selection of certain preferred alternatives for specific areas and intersections within and/or near the City; and

WHEREAS, The City Council appreciates the opportunity to cooperate with MDT to identify preferred alternatives for the 212 Corridor Project prior to when the designs are prepared and finalized.

WHEREAS, based on the alternatives provided by MDT, the City Council hereby selects and supports the following preferred alternatives:

- 1. Construction of an urban cross-section consistent with a City approved streetscape plan and MDT Urban Design Standards from 8th Street to the intersection of Highways 78 and 212.
- Construction of roundabouts at all full access intersections, including that of Highways 78 and 212, to realize the safety advantages that roundabouts provide, capture the cost advantages of roundabouts over time, and provide a distinctive entrance into the City of Red Lodge.
- 3. Construction of a continuous two-way left turn lane from the intersection of Highways 78 and 212 extending approximately ½ Mile North, in consideration of current access issues along this portion of the corridor.
- 4. Construction of a controlled access corridor as proposed along the 212 North Corridor, including a raised median with appropriate lighting. Implementation of a controlled access corridor may require the city and or county to implement land use planning and regulation legislation supporting the access control corridor. MDT shall have no authority to adopt or implement any land use regulations.

NOW THEREFORE, BE IT RESOLVED, that the City Council hereby supports the preferred alternatives listed herein and further requests that MDT consider the City's selection of preferred alternatives when designing the overall project.

PASSED and APPROVED by the Red Lodge City Council this 27^{++} day of March 2007.

FOR THE CITY OF RED LODGE, MT By: Brian C. Roat, Mayor Attest: Debbie Tomicich, City Clerk

Appendix B

NRCS Coordination and CPA-106 Form

United States Department of Agriculture

ONRCS

Natural Resources Conservation Service 606 West Front Street PO Box 510 Joliet, MT 59041-0229

BATOC C1

File 2201107

Received

DEC -2 2002

November 29,2002

Kadrmas Lee & Jackson

Dear M. Wanbolt, Enclosed is a map with the prime farmland highlighted in yellow for the highway from Boyd to Red Lodge, the any portion with prime fainland is on the end near Boyd, assuming that all your work would take place within the existing right of way, prime fainland would not be a factor. If additional land beyond that would be included then the prime fainland issue would be relevant. Please advise me if you need any additional data.

Aincerely, A. Ray M. Phil District Conserva tionist



U.S. DEPARTMENT OF AGRICULTURE Natural Resources Conservation Service 08-01-07 07:38 (Rev 10)

FARMLAND CONVERSION IMPACT RATING FOR CORRIDOR TYPE PROJECTS

PART I (To be completed by Federal Agency)		3. Date	of Land Evaluation	Request	7/11/07	4. Sheet 1 c	of <u>1</u>		
1. Name of Project US Highway 212 STPP 28-2 (25) 70		5. Federal Agency Involved							
2. Type of Project Reconstruction 6. 0			County and State Carbon County, Montana						
PART II (To be completed by NRCS) 1		1. Date F	Date Request Received by NRCS 2. Person Completing Form Toot Rolle						
3. Does the corridor contain prime, unique statewide or local in (If no, the FPPA does not apply - Do not complete additional	mportant farmland? al parts of this form).	، ب	YES INNO		4. Acres	Irrigated Average	Farm Size		
5. Major Crop(s)	6. Farmable Land	in Govern	ament Jurisdiction		7. Amoun	t of Farmland As D	efined in FPPA		
HAN Badey, wheat was	Acres: 82.4	387	+ % 7	7	Acres	: 220.94	13 %20		
8. Name OPLand Evaluation System Used	9. Name of Local	Site Asse	ssment System		10. Date Land Evaluation Returned by NRCS				
LESA	NA					8/13/07			
DADT III (To fee a survey late of the Feedbard (Anomaly)			Alternative Corridor For Segment						
PART III (To be completed by rederal Agency)			Corridor A	Corri	dor B	Corridor C	Corridor D		
A. Total Acres To Be Converted Directly	······································		266						
B. Total Acres To Be Converted Indirectly, Or To Receive	Services								
C. Total Acres In Corridor			266	0		0	0		
PART IV (To be completed by NRCS) Land Evaluat	ion Information								
A. Total Acres Prime And Unique Farmland		·····	86.9						
B. Total Acres Statewide And Local Important Farmland			48.7	1					
C. Percentage Of Farmland in County Or Local Govt. Un	it To Be Converted		DDDA				-		
D. Percentage Of Farmland in Govt, Jurisdiction With Sam	e Or Higher Relativ	e Value	51	f					
PART V (To be completed by NRCS) Land Evaluation Inf	ormation Criterion I	Relative		1					
value of Farmland to Be Serviced or Converted (Scale	of 0 - 100 Points)		61						
PART VI (To be completed by Federal Agency) Corrid	or M	aximum		1					
Assessment Criteria (These criteria are explained in 7	CFR 658.5(c))	Points							
1. Area in Nonurban Use		15	15	-		·····			
2. Perimeter in Nonurban Use		10	9				1		
3. Percent Of Corridor Being Farmed		20	18	1					
4. Protection Provided By State And Local Governmer	nt	20	0						
5. Size of Present Farm Unit Compared To Average		10	10						
6. Creation Of Nonfarmable Farmland		25	0			· · · · · · · · · · · · · · · · · · ·			
7. Availablility Of Farm Support Services		5	5		· · · · ·		1		
8. On-Farm Investments		20	15						
9. Effects Of Conversion On Farm Support Services			0						
10. Compatibility With Existing Agricultural Use		10	0						
TOTAL CORRIDOR ASSESSMENT POINTS		160	72	0		0	0		
PART VII (To be completed by Federal Agency)									
Relative Value Of Farmland (From Part V)		100	61						
Total Corridor Assessment (From Part VI above or a loc	al site	4.9.5	4 -				1		
assessment)		160	-12	0		0	0		
TOTAL POINTS (Total of above 2 lines)		260	133	0		0	0		
1. Corridor Selected: 2. Total Acres of Far Converted by Pro	mlands to be 3. ject:	Date Of a	Selection:	4. Was	A Local Sit	te Assessment Us	ed?		
					YES [] NO []			

5. Reason For Selection:

DATE

Appendix C

March 2002 Agency Scoping

List of Commenting Agencies Corridor Study – Red Lodge North Environmental Assessment Carbon County, Montana

Federal Agencies

Letter #

State Agencies

Montana Department of Environmental Quality	6
Montana Department of Natural Resources and Conservation	7 and 8
Montana Fish, Wildlife and Parks	9

Local Agencies

BNSF Railway Company	10
Montana Land Reliance	11
Red Lodge Fire Department	12
Red Lodge Parks Board	13
Roberts Public Schools	14
Roberts School District #5	15
Roberts School District #5 Response from MDT	16



DEPARTMENT OF THE ARMY CORPS OF ENGINEERS, OMAHA DISTRICT 106 SOUTH 15TH STREET OMAHA, NEBRASKA 68102-1618

REPLY TO ATTENTION OF:

April 16, 2002

Received APR 2 2 2002 Kadrmas Lee & Jackson

Planning, Programs and Project Management Division

Mr. Mike Wamboldt Kadrmas, Lee & Jackson 2611 Gabel Road PO Box 80303 Billings, Montana 59108

Dear Mr. Wamboldt:

We have reviewed your letter dated March 13, 2002 regarding the proposed construction on US Highway 212 from Red Lodge to Roberts, Montana, reference Project STPP 28-2 (25) 70, Control No. 4375, and we offer the following comments.

It should be ensured that the proposed project is in compliance with flood plain management criteria of Carbon County and the State of Montana. As a minimum, the design should ensure that the 100-year flood water surface elevation of any stream affected, that has a designated floodway, is not increased relative to pre-project conditions. If a designated floodway has not been identified then the design should ensure that the 100-year flood water surface elevation is not increased by more than one-foot relative to pre-project conditions. It is desirable, however, that water surface elevations either remain the same or decrease as a result of this project.

Your plans should be coordinated with the U.S. Environmental Protection Agency, which is currently involved in a program to protect groundwater resources.

If you have not already done so, we recommend that you consult with the U.S. Fish and Wildlife Service and the Montana Department of Fish, Wildlife and Parks regarding fish and wildlife resources. In addition, the Montana Historic Preservation Office should be contacted for information and recommendations on potential cultural resources in the project area.

If construction activities involve any work in waters of the United States, a Section 404 permit may be required. For a detailed review of permit requirements, final project plans should be sent to:

Mr. Allen Steinle U.S. Army Corps of Engineers Helena Regulatory Office 301 South Park Drawer 10014 Helena, Montana 59626-0014 If you have any questions, please contact Lauren Deane of our staff at (402) 221-4598. Thank you for the opportunity to review this proposal.

j

}

Sincerely,

Candace Gorton

Candace Gorton Chief, Environmental, Economics and Cultural Resources Section Planning Branch 「「大学のない」と言いていたいで、

÷

ł





U.S. ARMY CORPS OF ENGINEERS HELENA REGULATORY OFFICE 10 WEST 15TH STREET, SUITE 2200 HELENA, MONTANA 59626

REPLY TO ATTENTION OF:

November 27, 2002

Helena Regulatory Office (406) 441-1375 Phone (406) 441-1380 Fax Received DEC - 2 2002

Kadrmas Lee & Jackson

Subject:

Corps File Number 2002-90-186 Red Lodge – Roberts STPP 28-2(25)70, MDT Control Number 4375

Mr. Mike Wamboldt, P.E. Project Engineer Kadrmas Lee & Jackson PO Box 80303 Billings, Montana 59108

Dear Mr. Wamboldt:

Reference is made to your request for preliminary comments on the subject project. The proposed highway reconstruction project is located on US Highway 212 between the communities of Red Lodge and Boyd in Carbon County, Montana.

Under the authority of Section 404 of the Clean Water Act, Department of the Army permits are required for the discharge of fill material below the ordinary high water mark of our nation's rivers, streams, lakes or wetlands. A review of the limited project information submitted to this office has shown there will likely be work in Waters of the United States; as a result, this project is subject to Department of Army regulatory authorities, and permits will be required from the US Army Corps of Engineers.

This office will provide specific comments upon receipt of more specific project information, including additional environmental documentation and project design plans and details.

If you have any questions, please call Todd Tillinger of this office at (406) 441-1375, and reference Corps File Number 2002-90-186.

Sincerely,

Allan Steinle

Allan Steinle Montana Program Manager

Copy Furnished:

Jean Riley, Montana Department of Transportation - Helena

United States Department of the Interior



IN REPLY REFER TO:

BUREAU OF INDIAN AFFAIRS Rocky Mountain Regional Office 316 North 26th St. Billings, Montana 59101

Received

MAR 2 2 2002 MAR 2 1 2002

Kadrmas Lee & Jackson

reducinas Lee & Jacks

Mike Wamboldt, Project Engineer Kadrmas, Lee, and Jackson P.O. Box 80303 Billings, Montana 59108

Dear Mr. Wamboldt:

These comments relate to the US Highway 212 project. The existing right of way (ROW) and area adjacent to that ROW are far from the boundary of the nearest Indian reservation under the jurisdiction of this office. As such, we have very few comments that we can make about the project. However, the project is located within an area historically occupied by the Crow Tribe. The historic occupation could have left archaeologic or culturally significant sites. If the reconstructed road remains near the present ROW, there is very little likelihood of finding/disturbing artifacts.

If dramatic changes to the alignment are proposed, the possibility of archaeologic discoveries will increase. For most environmental assessments developed for Montana Department of Transportation projects, an archaeologic or cultural inventory and report is prepared. We assume this will be the case for this project as well. If you were not planning to do such, we would encourage it.

Questions may be directed to me at 406/247-7911.

Sincerely;

Alfani

Chief, Environmental Services



United States Department of the Interior

FISH AND WILDLIFE SERVICE MONTANA FIELD OFFICE 100 N. PARK, SUITE 320 HELENA, MT 59601 PHONE (406) 449-5325, FAX (406) 449-5339

MAR 2 8 2002

Received

Kadrmas Lee & Jackson

File: M.29 (I)

March 25, 2002

Mike Wamboldt, Kadrmas Lee and Jackson P.O. Box 80303 Billings, Montana 59108

Dear Mr. Wamboldt:

This is in response to your March 14, 2002 letter regarding a proposal for reconstructing U.S. Highway 212 from Red Lodge to Roberts in Carbon County, Montana (STPP 28-2(25)70; Control No.4375). The project is being proposed by the Montana Department of Transportation. Your letter requested comments from the U.S. Fish and Wildlife Service (Service) on the proposal.

The Service has reviewed the proposed action and determined that threatened bald eagles (*Haliaeetus leucocephalus*) and nonessential experimental gray wolves (*Canis lupus*) may occur in the project area. To meet section 7 requirements a biological assessment should prepared regarding these species. Please keep the Service apprised as the details of this proposal develop.

Your letter also requested any comments we may have regarding any property the Service may own or have interest in adjacent to the project corridor. Based on the information we have in this office and the map you included with your request, there are no such lands administered by the Service in the immediate vicinity of the proposed project location.

The Service has responsibility, under a number of authorities, for conservation and management of fish and wildlife resources. Chief among the federal statutes with which our office deals are the Fish and Wildlife Coordination Act, Endangered Species Act of 1973, and the National Environmental Policy Act. The Coordination Act requires that fish and wildlife resources be given equal consideration in the planning, implementation, and operation of Federal and federally funded, permitted, or licensed water resource related projects.

In Montana, habitats frequently used by important fish and wildlife resources are wetlands, streams, and riparian (streamside) woodlands. Special attention is given to proposed developments that include modification of wetlands, stream alterations, or contamination of important habitats. The Service recommends ways to avoid, minimize, rectify, reduce, or compensate for damaging impacts

to important fish and wildlife resources and their habitats that may be attributed to land and water resource development proposals.

We have reviewed the plans for the proposed activities and offer the following comments concerning wetlands and stream channels:

- a. In connection with the stream crossings, we recommend that you work closely with the Corps of Engineers (COE) Regulatory Office (406) 441-1375 regarding any Section 404 permits that may be needed. With regard to such permits, depending on permit type and other factors the Service may be required to review permit applications and recommend fish and wildlife protection or mitigation measures to the COE as appear reasonable and prudent at that time.
- b. The Service recommends clear spanning the stream channel, if possible, to avoid placement of structures in the stream channel. Bridge abutments and piers, and their attendant riprap, that are located in or encroach upon the stream channel can constrict flows, increase erosion and affect bedload movement up and down stream of the structure, resulting in significant effect to the physical, chemical and biological dynamic of the stream and its associated aquatic resources. The Service recommends that if instream structures are proposed, the direct, indirect, and cumulative impacts of those structures be analyzed, along with future activities related to scour protection and bank stabilization that are required to maintain such structures.

Should you have any further questions, please contact Mr. Scott Jackson within our office at (406) 449-5225, extension 201. Thank you for the opportunity to offer comments.

Sincerely,

R. Mark u

R. Mark Wilson Field Supervisor

N SK United States Department of the Interior VED^{FISH} AND WILDLIFE SERVICE ECOLOGICAL SERVICES MONTANA FIELD OFFICE MAY 1 8 2005 100 N. PARK, SUITE 320 MASTER FILE INVIRONNELITAPHONE (406) 449-52230FAX4306944980335 COPY Date S/a3/as Routies Buscap Crief CC. Kaul Holvik - Consultant L M.44 MDT (I) Conserved Print Plans Barg May 12, 2005 Deuge Sagetteens Bonnie Stag- Env. C. YELK Frequences Paul Sturm Bruce Barrett - Billings Holsik Montana Department of Transportation Paul Ferry - Highways En Environmental Services 2701 Prospect Avenue Mark Goodman Hydraulics KET JUC. P.O. Box 201001 SENT Shallos Art Jacobsan- Enu. Helena, Montana 59620-1001 Paul Sturm - Env. Dear Mr. Sturm: File- Env. ٦ί١

This is in response to your letter dated April 14, 2005, in which you requested the U.S. Fish and Wildlife Service's (Service) concurrence with your determination that the Montana Department of Transportation's (Department) proposed improvements to U.S. Highway 212, primarily between the towns of Red Lodge and Boyd in Carbon County, Montana (STPP 28-2(25)70; Control No. 4375), would not be likely to adversely affect federally-listed threatened or endangered species. This proposed project would entail reconstruction of 21.3 miles of U.S. Highway 212 and would include both urban and rural sections. Your letter stated that the Department intends to phase this proposed project under two or more construction projects, depending on the availability of funding and that your determinations of effect and concurrence request pertain to the entire road corridor. However, as each construction phase is designed, the Department would submit plans for that section to the Service to see if any new information would necessitate reinitiation.

Your letter transmitted a biological assessment (BA) for this project that concluded that this project would not be likely to adversely affect threatened bald eagles (Haliaeetus leucocephalus) and would not be likely to jeopardize the continued existence of the Yellowstone non-essential experimental population of gray wolves (Canis hupus). Based on information provided in your letter and the BA for this project, the Service concurs with the Department's determination that this project, as proposed, would not adversely affect listed species. This concurrence is based upon the projects in this corridor being designed and constructed as described and upon implementation of the coordination measures stated in the BA that are intended to protect listed species. These measures should be implemented for each separate project that is constructed within this corridor. In addition, we acknowledge your determination that this project would not affect endangered black-footed ferrets (Mustela nigripes).

This concludes informal consultation pursuant to regulations 50 CFR § 402.13 implementing the Endangered Species Act (Act). This project should be re-analyzed if new information reveals
effects of the action that may affect threatened or endangered species or if the project is modified in a manner that causes an effect not considered in this consultation.

If you have questions about this letter or your responsibilities under the Act, please contact Scott Jackson at (406) 449-5225, extension 201.

Sincerely, R. Marke

R. Mark Wilson Field Supervisor

Copy to: FWS-ES, Billings Suboffice

". .



Judy H. Martz, Governor

P.O. Box 200901 • Helena, MT 59620-0901 • (406) 444-2544 • Website: www.deq.state.mt.us

March 22, 2002

Mike Wamboldt Kadrmas, Lee & Jackson P.O. Box 80303 Billings, MT 59108

Subject: Project STPP 28-2 (25) 70 - Red Lodge to Roberts

Dear Mr. Wamboldt:

I am responding to your request for information concerning contaminated sites on or near the above referenced project. Sources used to supply information for your data request include four Department of Environmental Quality (DEQ) - Remediation Division databases which contain information on the nature and location of Underground Storage Tank sites (USTs), Leaking Underground Storage Tank sites (LUSTs), Abandoned Mine sites, and State Superfund sites (CECRA). The State Superfund database also includes information on sites affected under the Water Quality Act (WQA) and pesticide-contaminated sites addressed under the Montana Agricultural Chemical Groundwater Protection Act. Site information from this database search revealed several UST and LUST sites in the reference area (SEE ATTACHED LISTS). The databases were searched using the following criteria: sites with identified addresses in Carbon county near cities of Red Lodge, Roberts, or Boyd.

More information can be found on the DEQ's website. The list of CECRA sites (State Superfund) can be found on the Web at http://www.deq.state.mt.us/Rem/hwc/Srs/cecralistformats.asp. The list of Federal National Priority List (Federal Superfund) sites can be found at http://www.deq.state.mt.us/rem/mwc/feds.asp. The list of active Underground Storage Tanks (USTs) and Leaking Underground Storage Tanks (LUSTs) can be found at http://www.deq.state.mt.us/rem/tsb/iss/ustdownloads.asp. For information concerning inactive mines, contact our office at 444-0475 or visit the Mine Waste Bureau's website at http://www.deg.state.mt.us/rem/mwc/index.asp. Information on hazardous waste generators or hazardous waste permitted facilities can be obtained from the DEQ's Air and Waste Management Bureau at 444-3490. For lists of permitted landfills and landfarms, contact DEQ's Community Services Bureau, at 444-4400. Finally, contact the DEQ's Complaint Management Section at 444-0379 for a list of the sites/spills about which DEQ has received complaints. The State does not have a comprehensive inventory program; therefore, a site could be contaminated and not be listed on any of the lists maintained by DEQ. For further information on your site area of interest, you are welcome to inspect and make copies of our files during normal business hours.

Please also be aware that new contaminated sites are discovered every day. You may encounter contamination at a location that we are not yet aware of. If you encounter soil or groundwater contamination at the subject property, please call our office at (406) 444-1420 to report contamination from USTs or the Complaint Management Section at 444-2964 to report all other contamination.

Feel free to call me at (406) 444-0474, if you have any questions or comments about this data request.

Sincerely Jack Yates

Information Services/Section Technical Services Bureau Montana Dept. of Environmental Quality 444-0474 jyates(a)state.mt.us

qryLustSitesByCity-jy

)

03/22/2002

a de la companya de l

.

Contraction of the second second

......

2297 ANDERSON CONOCO910 W VILLARDRed Lodge07/22/19211 BEARTOOTH ELECTRIC CO-OP INCBOX 1119Red Lodge07/20/19108 BOYD STOREBOX 236Boyd06/11/19979 CARBON COUNTY ABSTRACT TITLE C105 N BROADWAYRed Lodge10/24/191324 CARBON COUNTY CATTLE COBOX 156Boyd10/07/19751 CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314 County ShopWhite Ave S R.L.Red Lodge06/12/19313 COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	88 Carbon 93 Carbon 93 Carbon 89 Carbon 93 Carbon 91 Carbon 98 Carbon 91 Carbon
211 BEARTOOTH ELECTRIC CO-OP INCBOX 1119Red Lodge07/20/19108 BOYD STOREBOX 236Boyd06/11/19979 CARBON COUNTY ABSTRACT TITLE C105 N BROADWAYRed Lodge10/24/191324 CARBON COUNTY CATTLE COBOX 156Boyd10/07/19751 CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314 County ShopWhite Ave S R.L.Red Lodge06/17/19313 COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	93 Carbon 93 Carbon 89 Carbon 93 Carbon 91 Carbon 98 Carbon 91 Carbon
108BOYD STOREBOX 236Boyd06/11/19979CARBON COUNTY ABSTRACT TITLE C105 N BROADWAYRed Lodge10/24/191324CARBON COUNTY CATTLE COBOX 156Boyd10/07/19751CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314County ShopWhite Ave S R.L.Red Lodge06/17/19313COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007Former AST Service Station#1 Railroad AveRoberts03/31/192567LAUREL COOP ASSOCBOX 11Roberts10/30/19260LONE PINE RANCHCLEAR CREEKRoberts06/22/19	93 Carbon 89 Carbon 93 Carbon 91 Carbon 98 Carbon 91 Carbon
979CARBON COUNTY ABSTRACT TITLE C105 N BROADWAYRed Lodge10/24/191324CARBON COUNTY CATTLE COBOX 156Boyd10/07/19751CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314County ShopWhite Ave S R.L.Red Lodge06/17/19313COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007Former AST Service Station#1 Railroad AveRoberts03/31/192567LAUREL COOP ASSOCBOX 11Roberts10/30/19260LONE PINE RANCHCLEAR CREEKRoberts06/22/19	89 Carbon 93 Carbon 91 Carbon 98 Carbon 91 Carbon
1324 CARBON COUNTY CATTLE COBOX 156Boyd10/07/15751 CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314 County ShopWhite Ave S R.L.Red Lodge06/17/19313 COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	93 Carbon 91 Carbon 98 Carbon 91 Carbon
751CARTER'S BULK PLANTHWY 212Red Lodge08/01/193314County ShopWhite Ave S R.L.Red Lodge06/17/19313COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007Former AST Service Station#1 Railroad AveRoberts03/31/192567LAUREL COOP ASSOCBOX 11Roberts10/30/19260LONE PINE RANCHCLEAR CREEKRoberts06/22/19	91 Carbon 98 Carbon 91 Carbon
3314 County ShopWhite Ave S R.L.Red Lodge06/17/19313 COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	98Carbon 91Carbon
313 COWGER, NICKRTE 1 BOX 4165Red Lodge06/12/193007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	91 Carbon
3007 Former AST Service Station#1 Railroad AveRoberts03/31/192567 LAUREL COOP ASSOCBOX 11Roberts10/30/19260 LONE PINE RANCHCLEAR CREEKRoberts06/22/19	
2567 LAUREL COOP ASSOC BOX 11 Roberts 10/30/19 260 LONE PINE RANCH CLEAR CREEK Roberts 06/22/19	97 Carbon
260 LONE PINE RANCH CLEAR CREEK Roberts 06/22/19	91Carbon
	93Carbon
188 MT DEPT HWY-RED LODGE SITE BOX 445 Red Lodge 09/10/19	96Carbon
1601 RANSDELL UNION 76 BULK PLANT Address Unknown Red Lodge 08/20/19	93Carbon
3622 Ray Judd Ford 116 N Broadway Red Lodge 07/07/19	99Carbon
428 RED LODGE EXXON 524 S. BROADWAY Red Lodge 10/12/19	94 Carbon
289 RED LODGE MTN SKI RESORT W OF RED LODGE Red Lodge 03/31/19	89Carbon
3346 Red Lodge Travel Center 403 S Broadway Red Lodge 07/30/19	98 Carbon
734 ROCK CREEK C STORE 1022 S ADAMS Red Lodge 06/25/19	96 Carbon
1744 ROCK CREEK C STORE 1022 S ADAMS Red Lodge 03/09/19	89Carbon
312 T & D PUMP HWY 212 N Red Lodge 12/09/19	93 Carbon
2999 Uncle Milty's Drive-In 704 S Broadway Red Lodge 03/07/19	97 Carbon
517 US HWY 212 BROADWAY & 17TH Red Lodge 10/30/19	95 Carbon
1155 Y-STOP BOX 85 Roberts 05/27/19	02 Carbon

03/22/2002

.....

ł

;

ł

qselActiveFacilities

AltFacilityID	Facility Name	Street Address	City	County
05-04228	T & D Pump	Hwy 212 N	Red Lodge	Carbon
05-05655	Ray Judd Ford	116 N Broadway	Red Lodge	Carbon
05-06599	Y-Stop C Store	Hwy 212 & Cooney Rd S	Roberts	Carbon
05-06961	Pony Express	401 N Broadway Ave	Red Lodge	Carbon
05-06970	Anderson's Conoco	224 S Broadway	Red Lodge	Carbon
05-09748	Rock Creek Convenience Store	1022 S Adams	Red Lodge	Carbon
05-09926	Beartooth Hospital & Health Center	600 W 21st Street	Red Lodge	Carbon
05-13924	Rock Creek North	902 N Broadway	Red Lodge	Carbon

· .

)

DEPARTMENT OF NATURAL RESOURCES AND CONSERVATION

Orig - Sheri varia
cc: Kurtiss.
File
7
1



JUDY MARTZ GOVERNOR

DIRECTOR'S OFFICE (406) 444-2074 TELEFAX NUMBER (406) 444-2684

WATER RESOURCES DIVISION (406) 444-6601 TELEFAX NUMBERS (406) 444-0533 / (406) 444-5918 http://www.dnrc.state.mt.us/wrd/home.htm

48 NORTH LAST CHANCE GULCH PO BOX 201601 HELENA, MONTANA 59620-1601

December 12, 2002

Mr. Mike Wamboldt Kadrmas, Lee & Jackson 2611 Gabel Road P.O. Box 80303 Billings, MT 59108

Received

DEC 16 2002

Kadrmas Lee & Jackson

RE: Comments on Preliminary Field Review Report – Project STPP 28-2 (25) 70, U.S. Hwy 212 from Red Lodge to Roberts

Dear Mr. Wamboldt:

Based on my review of the information provided for this project, I offer the following comments:

There are no state-owned water projects or Water Resources Division property within the project area, so I would not anticipate any impacts; however, there are many private irrigation canals and ditches that cross under the existing highway. Culverts of adequate size would have to be placed for these canals and ditches. Also, it is important to place these culverts on grade (i.e. not raise them). A significant amount of wastewater collects in the existing borrow areas along the road. Adequate drainage would have to be planned for to deal with this water. There are also several springs and ponds about 1 mile north of Red Lodge that cause seepage under the highway.

I discussed this proposal with Mr. Keith Kerbel, the Water Resources Division Regional Manager in our Billings Office. He provided me with this information and is very familiar with the water related issues along this stretch of highway. Keith and I would like to be placed on the distribution list for the Environmental Assessment that will be completed for this project. Our mailing addresses are as follows:

Keith Kerbel, Regional Manager Montana DNRC, Water Resources Division Billings Regional Office Airport Business Park 1371 Rimtop Drive Billings, MT 59105-1978 (406) 247-4415 e-mail kkerbel@state.mt.us

STATE WATER PROJECTS BUREAU (406) 444-6646

WATER MANAGEMENT BUREAU (406) 444-6637 WATER OPERATIONS BUREAU (406) 444-0860 WATER RIGHTS BUREAU (406) 444-6610 James P. Domino Montana DNRC, State Water Projects Bureau 48 N. Last Chance Gulch P.O. Box 201601 Helena, MT 59620-1601 (406) 444-6622 e-mail jdomino@state.mt.us

Thank you for the opportunity to review and comment on the Preliminary Field Review Report. Please contact me if you have any additional questions.

Sincerely,

Janos P Domino

James P. Domino Environmental Specialist State Water Projects Bureau

c: Keith Kerbel, Billings Regional Office Kevin Smith

April 9, 2002

Mike Wamboldt 2611 Gabel Road P.O. Box 80303 Billings, MT 59108

Received APR 1 0 2002 Kadrmas Lee & Jackson

Dear Mr. Wamboldt,

This letter is in response to the request for information about the fishery and aquatic habitat in the vicinity of US Highway 212 from Red Lodge to Roberts. The fisheries habitat within this section of stream supports a thriving population of wild brown, rainbow and brook trout. Habitat types include riffles, glides and pools that provide critical spawning and rearing habitat for fish. There is an active floodplain throughout much of this reach of stream and because of the intensity of spring floods and past stream manipulations the river channel is continually migrating. However, US Highway 212 is far enough from the stream throughout nearly the entire project (except at the bridge crossing near Roberts) that direct interaction between the reconstruction and the stream is unlikely. Indirect effects of road reconstruction on fisheries could include the transportation of disturbed soils during rain events from construction areas to the stream. Fine sediment such as sand and clay entering the stream can smother fish eggs incubating in stream gravels and cause other detrimental effects to aquatic organisms. Erosion control measures should be employed to regulate surface erosion of soils until natural vegetation can be reestablished.

Two stream crossings are associated with this reconstruction project: the crossing over Stanley Creek, a small tributary to Rock Creek between Boyd and Roberts and the bridge over Rock Creek near Roberts. I am currently unaware of the presence of a fishery in Stanley Creek, but if there is a viable fish population present, proper passage would need to be addressed so that fish from Rock Creek can access Stanley Creek and visa versa. If fish passage is an issue, a hydrologist would need to be consulted for proper culvert/bridge design. The current bridge over Rock Creek is a span bridge that does not restrict fish passage. However, angler access at public road crossings is an important concern and it is my desire that public access be improved at the bridge crossing Rock Creek. I am not aware if there is currently public access to Rock Creek from the rest area on the northeast side of the bridge, but if public access were available from the rest area, then access at the bridge would be less of a concern.

The fish population in Rock Creek includes predominantly rainbow, brook and brown trout, mountain whitefish, and mottled sculpin with a few Yellowstone cutthroat trout longnose dace and several species of suckers. Average length of trout within this section of river is roughly 10 in with fish ranging in size from 3 in to 18 in. Spawning and egg incubation times (i.e., times when fish are most susceptible to the effects of siltation) for rainbow and cutthroat trout are approximately from April 1 to July 15 and for brown and brook trout from October 1 to April 1. If you have any question or comments, please feel free to contact me (406-322-1162).

Second Second

Sincerely,

,

Jim Olsen

Regional Fisheries Biologist, Columbus Montana Fish Wildlife and Parks P.O. Box 85 Columbus, MT 59019 March 13, 2002

Kadrmas Lee & Jackson

Received

MAR 2 6 2002

John M. Cowles, Public Projects Manager BNSF Railway Company Suite 1A 2454 Occidental Avenue South Seattle, WA 98134-1451

Subject: Project STPP 28-2 (25) 70 Control No. 4375 US Highway 212 from Red Lodge to Roberts

Dear Mr. Cowles:

Kadrmas, Lee, & Jackson has been selected by the Montana Department of Transportation to complete the project design through the right-of-way phase on the above referenced project. This includes the access management study and the environmental document development. We are requesting information from your office for the environmental documentation on this proposed highway project. Attached is a copy of the Preliminary Field Review Report describing the proposed project and a Project Location Map.

To ensure that all social, economic and environmental effects are considered in the development of this project, we are soliciting your views and comments on the proposed project pursuant to Section 102(2) (D) (IV) of the National Environmental Policy Act of 1969, as amended. We are particularly interested in any property that your department may own or have an interest in and which would be adjacent to the proposed highway improvement. We would also appreciate being made aware of any proposed developments your department may be contemplating in the areas under consideration for the proposed highway facility. It is our opinion that the project will not have a significant effect on the quality of the human environment. Any information or comments relating to environmental matters that you might furnish will be appreciated.





Page 2 Re: Project STPP 28-2 (25) 70 Control No. 4375 US Highway 212 from Red Lodge to Roberts

We would appreciate receiving a reply by April 10, 2002 so that this information can be considered early in the environmental analysis process. If you have any questions, please contact me at (406) 245-5499. Thank you in advance for your cooperation.

Sincerely. Kadrmas, Lee & Jackson

TIN-

Mike Wamboldt, PE Project Engineer-

Attachments

Cc: Bruce H. Barrett, Administrator - MDT Billings District No. 5 Joseph P. Kolman, P.E. - MDT Bridge Engineer Carl S. Peil, P.E. - MDT Preconstruction Engineer Tom S. Martin, P.E. - MDT Consultant Design Engineer John H. Horton, Chief - MDT Right-Of-Way Bureau Timothy W. Reardon, Chief Counsel - MDT Legal Services Stan Sternberg, Acting Environmental Manager -MDT Environmental Services Bob Harrington, Area Manager - DNR&C Southern Land Office Walt Scott, Supervisor - MDT Utilities Section

The burgers it is my The purjuit have in the privation of the private have in the private of th

J:\trans\mtdot\2201102\sov\SOV general BNSF.doc

ana LAND RELIANCE

Received

NOV 27 2001

26 November 2001

Mr. Clay Schwartz Kadrmas, Lee & Jackson P.O. Box 80303 Billings, MT 59108

Kadrmas Lee & Jackson

Re: Montana Department of Transportation Project No. STTP 28-2(25)70, C.N.4375 Red Lodge North

Dear Mr. Schwartz,

I am writing on behalf of the Montana Land Reliance (MLR) regarding a proposed road expansion of Highway 212 from Red Lodge to Boyd, Montana. The MLR is a private, non-profit land trust dedicated to providing permanent protection for private lands that are significant for agricultural production, for fish and wildlife habitat, and for their scenic open space values.

The MLR holds conservation easements on three properties which border Highway 212 between Red Lodge and Boyd. They are the Conlon Property, the Hoiness Property, and the Lay Property. The former is near Red Lodge, Montana; the latter two are near Boyd, Montana. The conservation easements contain terms which limit potential expansion possibilities. At this time, I would like to know (1) the source of funding for this expansion project and (2) the approximate anticipated date this project is to begin.

In future, please be sure MLR receives any correspondence or ongoing notification of the progress of this project with respect to the three properties identified above. Feel free to call me at our Helena office if you have any questions.

Sincerely. norrather k

Noorjahan Parwana Land Steward

GLACIER FLATHEAD OFFICE

470 Electric Ave.

PO Box 460
Bigfork, Montana 59911-0460
406/837-2178

Fax 406/837-4980
email mlrnw@digisys.net

MAIN OFFICE

324 Fuller Ave. • PO Box 355 Helena, Montana 59624-0355 406/443-7027 • Fax 406/443-7061 email mtland@mt.net EASTERN OFFICE

2320 Third Ave. N. • PO Box 171 Billings, Montana 59103-0171 406/259-1328 • Fax 406/259-1437 email mlr@mcn.net



Red Lodge Fire Department P.O. Box 9 • Red Lodge, Montana 59068 (406) 446-2320 Fax (406) 446-3936



Kadrmas Lee & Jackson

03-22-02

To: Mike Wambolt PE Project Engineer Kadrmas, Lee, and Jackson 2611 Gabel Road Billings MT. 59102

From: Jerry Ballard Fire Chief Red Lodge Fire Dept.

Subject: Project STPP 28-2 (25) 70 Control Number 4375 US Highway 212 from Red Lodge to Roberts.

Dear Mike Wambolt

I am not the Police Chief for Red Lodge I am the Fire Chief. If information about your project needs to go to the Red Lodge Police Chief that information can be addressed to Richard Pringle P.O. Box 9 Red Lodge MT. 59068

The concern I have for this project is that it includes a turn lane onto Highway 78 from 212 at the junction at 3rd street in Red Lodge. Thank you

Jerry Ballord

Jerry Ballard

RED LODGE PARKS BOARD CITY OF RED LODGE, MONTANA PO BOX 9, RED LODGE MT 59068

March 27, 2002

Montana Department of Transportation Planning Division Helena, Montana 59620

Dear Planning Division:

The Red Lodge Parks Board has begun the effort to secure funds to construct a city-wide pedestrian and bicycle trail system. Two components have been partially constructed (not paved) and another is under consideration by the MT Fish, Wildlife and Parks Recreational Trails Program.

We ask that you consider including additional trail components in your planning for reconstruction of US Highway 212. A trail along the west side of this highway from the northern end of our city limits to Fifth Street is requested. In addition, a trail from the Golf Course and the Apartment Complex descending "Brewery Hill" on Highway 78 is critically needed. Residents of these areas walk and bike along the edge of this narrow, dangerous road. A safe path leading into town and to schools is requested.

Attached are maps showing the proposed city-wide trail system and the Highway 212 and Highway 78 sections. Additional information is available on request.

Thank you for your consideration of this project.

Sincerely,

Members of the Red Lodge Parks Board PO Box 598 Red Lodge, MT 59068 406-446-3847 etafoya@vcn.com





Superintendent Randy C. Durr MAS 9 HLE

District Clerk

Roberts Public Schools

Athletic Director George Nelson P.O. Box 78 * 106 Maple * Roberts, MT 59070

RECEIVED

SEP 2 3 2002

September 20, 2002

MONTANA DEPT. OF THANSPORTATION BILLINGS DISTRICT BILLINGS, MONTANA

Mr. Bruce Barette 424 Morey, Box 20437 Billings, MT 59104-0437

Re: Guardrail

Dear Mr. Barette:

It is the feeling of several parents of children at Roberts School that the playground area is in a dangerous location to the highway and that children may be unsafe. In consideration of that feeling, it is the wish of the Roberts School Trustees that some barrier be erected outside the fence enclosing the playground.

In talking to the highway department they said we should contact you and enlist your help in erecting a guardrail on the inside of the curve opposite the Y-Stop Convenience Store and the Roberts School Playground. Since it is a safety issue for school children, they felt you would be willing to do this.

Please let us know your feelings and what actions you could take regarding this as soon as possible.

Thank you,

Randy C. Durr / Superintendent

/dmt

(406) 445-2421 FAX: (406) 445-2506

Sent By: Kadrmas, Lee and Jackson; 4062945502; Sep-22-05 1;33PM; Page 2/2 Roberts School District #5 P O Box 78 Roberts, MT 59070 (406) 445-2421 FAX (406) 445-2506 September 15, 2005 Receiv Mike Wamboldt, Chief Engineer

Kadrmas, Lee & Jackson, Inc. 2611 Gabel Rd. PO Box 80303 Billings, MT 59108-0303 SEP Z 1 2005 Kadrmas Lee & Jackson

RE: STPP 28-2(25)70 Corridor Study Red Lodge - N School Bus Turnarounds

Dear Mr. Wamboldt:

Following our September 13th board meeting, the Roberts Board of Trustees has indicated that the turnarounds proposed by Board Member Joe Niemi to be the ones that we would like to request. Those turnarounds are at the following locations:

1. One mile north of Fox at the present State gravel pile

2. At the District boundary approximately three miles south of Boyd

The Board of Trustee feels that these two turnarounds would adequately provide the school district with the means of turning around safely. The district would like to reiterate the importance of these turnarounds in providing a safe environment for our children on our school buses.

Sincerely,

Jeffrey Bermes, Superintendent Roberts School

Cc: File Randy Wiemer, Chairman, Board of Trustees

Jeff Bermes Superintendent Jalayne Obert District Clerk Randy Weimer Chairman of the Board

ł

Montana Department of Transportation

David A. Gail, Diractor Judy Martz, Governor

\$C ~ }

Billings District Office 424 Morey Street PO Box 20437 Billings, MT 59104-0437

October 16, 2002

Roberts Public Schools Attn: Randy C. Durr, Principal PO Box 78 Roberts, MT 59070

Subject: Guardrail Request Roberts School Grounds US 212/P-28

Both the District Maintenance Chief and Traffic Engineer examined the playground area you requested for a possible guardrail installation. We also examined accident data from the Safety Section for the last 10 years and found no accident history at this location.

The roadway near your playground area has many safe features: It has speed limit control with flashing lights; there is a prominent school crossing with new school waining signs, and your playground area is on the inside of the curve; set outside our standard clear zone. The playground also has a good chain link fence separating it from the roadway, and has an irrigation channel skirting the south end of the playground; both of which are capable of restraining an errant vehicle in many cases.

National experience has determined that guardrail is a significant hazard in itself, and should only be used as a last resort. A vehicle that would normally come to a controlled stop off road, loses all control when contacting a guardrail or other barrier, and presents a significantly greater danger to people near the roadside as well as to the vehicle occupants.

While there is always the possibility of freak accident, your playground area has very reasonable protection features in place; and the existing potential hazards do not outweigh that foreseen for a guardrail installation. Our recommendation is not to install guardrail, or other similar barrier.

Respectfully,

UUS

Bruce H. Barrett District Administrator - Billings

10p28robertsPS:smj

copies: Traffic File

Appendix D

May 2007 Agency Scoping

List of Commenting Agencies Corridor Study – Red Lodge North Environmental Assessment Carbon County, Montana

Federal Agencies

<u>Letter #</u>

US Department of Agriculture – Natural Resources Conservation Service	1
US Department of Defense – Army Corps of Engineers, Omaha District	2
US Department of the Interior - Fish and Wildlife Service, Montana Field Office	3

State Agencies

Montana Department of Environmental Quality	.4
Montana Fish, Wildlife and Parks	5
Montana Natural Heritage Program	6

Local Agencies

City of Red Lodge Parks Board	7
Montana Land Reliance	8 and 9

United States Department of Agriculture

MASTER FILE

COPY



Natural Resources Conservation Service Federal Building, Room 443 10 East Babcock Bozeman, MT 59715

May 14, 2007

Ms. Heidy Bruner Environmental Services Montana Department of Transportation P.O. Box 201001 Helena, Montana 59620-1001

Dear Ms. Bruner:

We are responding to your letter of May 2, 2007, requesting information for Project Number STPP 28-2(25)70, Reconstruction of State Highway 212, between Red Lodge and Boyd in Carbon County, Montana.

Your letter indicates that the Natural Resources Conservation Service (NRCS) had not responded to previous requests for information regarding this project. This statement is in error as Mr. Tony Rolfes, a Resource Soil Scientist for NRCS, responded on September 9, 2005, to the firm of Kadramas, Lee and Jackson, a consultant for the Department of Transportation on this project. The correspondence provided the status of important farmlands in the highway corridor. NRCS administers the Farmland Protection Policy Act which requires evaluation of impacts to important farmland as a result of federal actions to include federal funding.

Should you need assistance in locating this correspondence, please contact Mr. Gordon Hill, P.O. Box 510, Joliet, Montana 59041-0229. Mr. Hill's phone number is (406) 962-3641, Ext. 101.

CARRIE MOSLEY Acting State Conservationist

cc:

Ronald Nadwornick, State Resource Conservationist, NRCS, Bozeman, Montana Dennis Loreth, Assistant State Conservationist for Field Operations, NRCS, Bozeman, Montana Gordon Hill, District Conservationist, NRCS, Joliet, Montana Tony Rolfes, Resource Soil Scientist, NRCS, Bozeman, Montana

> HELPING PEOPLE HELP THE LAND An Equal Opportunity Provider and Employer

RECEIVED

MAY 1 7 2007

ENVIRONMENTAL

Office: (406) 587-6811 Fax: (406) 587-6761



DEPARTMENT OF THE ARMY CORPS OF ENGINEERS, OMAHA DISTRICT 106 SOUTH 15TH STREET OMAHA NE 68102-1618

REPLY TO ATTENTION OF

May 17, 2007

Planning, Programs, and Project Management Division

Ms. Heidy Bruner Montana Department of Transportation 2701 Prospect Avenue Helena, Montana 59620-1001

Dear Ms. Bruner:

The U.S. Army Corps of Engineers, Omaha District (Corps) has reviewed your letter dated May 2nd, 2007 regarding the proposed US Highway 212 improvements. The Corps offers the following comments:

It should be ensured that the addition to the proposed project is in compliance with flood plain management criteria of Carbon County and the State of Montana. As a minimum, the design should ensure that the 100-year flood water surface elevation of any stream affected that has a designated floodway, is not increased relative to pre-project conditions. If a designated floodway has not been identified then the design should ensure that the 100-year flood water surface elevation is not increased by more than 1-foot relative to pre-project conditions. It is desirable, however, that water surface elevations either remain the same or decrease as a result of this project.

Your plans should be coordinated with the U.S. Environmental Protection Agency, which is currently involved in a program to protect ground water resources. If you have not already done so, it is recommended you consult with the U.S. Fish and Wildlife Service and the Montana Department of Fish, Wildlife and Parks regarding fish and wildlife resources. In addition, the Montana State Historic Preservation Office should be contacted for information and recommendations on potential cultural resources in the project area.

If construction activities involve any work in waters of the United States, a Section 404 permit may be required. For a detailed review of permit requirements, preliminary and final project plans should be sent to:

U.S. Army Corps of Engineers Helena Regulatory Office Attention: CENWO-OD-R-MT/Steinle 10 West 15th Street, Suite 2200 Helena, Montana 59626



RECEIVED

MAY 2 2 2007

ENVIRONMENTAL



If you have any questions, please contact Mr. Dave Crane at (402) 221-4882.

Sincerely, Vev Larry D. Janis, Chief

Environmental, Economics, and Cultural Resources Section Planning Branch

United States Department of the Interior

FISH AND WILDLIFE SERVICE ECOLOGICAL SERVICES MONTANA FIELD OFFICE 585 SHEPARD WAY HELENA, MONTANA 59601 PHONE (406) 449-5225, FAX (406) 449-5339 RECEIVED

JUN 1 2 2007

ENVIRONMENTAL



M.44 MDT (I)

June 8, 2007

Heidy Bruner Montana Department of Transportation Environmental Services 2701 Prospect Avenue PO Box 201001 Helena, Montana 59620-1001

Dear Ms. Bruner:

Thank you for your May 2, 2007 letter in which you invited comments from the US Fish and Wildlife Service (Service) relative to the Montana Department of Transportation's (Department) Red Lodge North Corridor Study (STPP 28-2(25)70; Control No. 4375). Proposed corridor improvements would occur along approximately 21.2 miles of US Highway 212 from Red Lodge to Boyd in Carbon County, Montana. Your letter referenced a letter from the Service to the Department dated March 25, 2002 pertaining to this project, and asked if that information was still current and if the Service had additional project-related comments.

A review of our file for this project indicates that in addition to our March 25, 2002 letter to your office, we also issued a letter to the Department on May 12, 2005 (enclosed) in which we concurred with the Department's determination that the proposed actions in that corridor would not be likely to adversely affect federally-listed threatened or endangered species. To the best of our knowledge, this information is still current and no further consultation is required pursuant to Section 7 of the Endangered Species Act. If the Department has new information related to the effects of this proposed action on federally-listed species, consultation with the Service should be reinitiated.

We appreciate the opportunity to comment on this project. If you have questions related to this letter, please contact Scott Jackson at (406)449-5225, extension 201.

Sincerely,

R. Mark Wilson Field Supervisor

Enclosure



United States Department of the Interior

FISH AND WILDLIFE SERVICE ECOLOGICAL SERVICES MONTANA FIELD OFFICE 100 N. PARK, SUITE 320 HELENA, MONTANA 59601 PHONE (406) 449-5225, FAX (406) 449-5339

M.44 MDT (I)

May 12, 2005

Paul Sturm Montana Department of Transportation Environmental Services 2701 Prospect Avenue P.O. Box 201001 Helena, Montana 59620-1001

Dear Mr. Sturm:

This is in response to your letter dated April 14, 2005, in which you requested the U.S. Fish and Wildlife Service's (Service) concurrence with your determination that the Montana Department of Transportation's (Department) proposed improvements to U.S. Highway 212, primarily between the towns of Red Lodge and Boyd in Carbon County, Montana (STPP 28-2(25)70; Control No. 4375), would not be likely to adversely affect federally-listed threatened or endangered species. This proposed project would entail reconstruction of 21.3 miles of U.S. Highway 212 and would include both urban and rural sections. Your letter stated that the Department intends to phase this proposed project under two or more construction projects, depending on the availability of funding and that your determinations of effect and concurrence request pertain to the entire road corridor. However, as each construction phase is designed, the Department would submit plans for that section to the Service to see if any new information would necessitate reinitiation of consultation.

Your letter transmitted a biological assessment (BA) for this project that concluded that this project would not be likely to adversely affect threatened bald eagles (Haliaeetus leucocephalus) and would not be likely to jeopardize the continued existence of the Yellowstone non-essential experimental population of gray wolves (Canis lupus). Based on information provided in your letter and the BA for this project, the Service concurs with the Department's determination that this project, as proposed, would not adversely affect listed species. This concurrence is based upon the projects in this corridor being designed and constructed as described and upon implementation of the coordination measures stated in the BA that are intended to protect listed species. These measures should be implemented for each separate project that is constructed within this corridor. In addition, we acknowledge your determination that this project would not affect endangered black-footed ferrets (Mustela nigripes).

This concludes informal consultation pursuant to regulations 50 CFR § 402.13 implementing the Endangered Species Act (Act). This project should be re-analyzed if new information reveals

effects of the action that may affect threatened or endangered species or if the project is modified in a manner that causes an effect not considered in this consultation.

If you have questions about this letter or your responsibilities under the Act, please contact Scott Jackson at (406) 449-5225, extension 201.

Sincerely, R. Ma

R. Mark Wilson Field Supervisor

Copy to: FWS-ES, Billings Suboffice

Red Lodge North Record of Telephone Conversation

Date: 06/11/07 2:25 p.m.	Project Number: 2201102
Recorded By: Becky Rude	Phase Number:
•	•
Talked With: Kent Harris	
Representing: MDEQ	
Address:	
City: State: Zip:	
Phone Number: 406-841-5048	
Email:	
•••••••••••••••••••••••••••••••••••••••	· · · · · · · · · · · · · · · · · · ·

Subject of Conversation: The Ski Station underground storage tank

Items Discussed:

Mr. Harris returned my phone call regarding the history of the Ski Station (Site 6015056). There is no report for KLJ to obtain. Kirth Erickson was present when the kerosene tanks were removed from the ground. Nine soil samples were taken using the EPH method and the highest levels recorded were 1300 ppm and 640 ppm. They dug up everything that they could and Mr. Harris believes it would be very unlikely that we would come into contact with contaminated soil during the construction of our project.

Distribution: CKL&J

(Names) (Names) 开 -/



From: Charlotte Brett [charlotte.brett@kljeng.com]
Sent: Friday, May 11, 2007 4:04 PM
To: mike.wamboldt@kljeng.com; 'Becky Rude'
Subject: FW: 4375; STPP 28-2(25)70; Red Lodge North --Phone Response from MT FWP

----Original Message----From: Bruner, Heidy [mailto:hbruner@mt.gov]
Sent: Friday, May 11, 2007 4:50 PM
To: Priebe, Gabe; charlotte.brett@kljeng.com
Cc: Bruner, Heidy; Semmens, Bill
Subject: 4375; STPP 28-2(25)70; Red Lodge North --Phone Response from MT FWP

Hi Folks:

We received a phone message from Jim Olsen of MT FWP (328.4636) confirm that he is aware of no additional impacts or concerns related to this project.

Cheers, Heidy

Heidy Bruner Great Falls District Project Development Engineer MDT Environmental Services 406.444.7203



P.O. Box 201800 • 1515 East Sixth Avenue • Helena, MT 59620-1800 • fax 406.444.0581 • tel 406.444.5354 • http://mtnhp.org

井(0

August 17, 2007

Becky Rude Kadrmas, Lee & Jackson, Inc. PO Box 937 Valley City, ND 58072-0937

Dear Becky,

I am writing in response to your request for information on plant and animal species of special concern in the vicinity of U.S. Highway 212 from Red Lodge to Boyd. We checked our databases for information in this general area and have enclosed 6 species of concern reports, 8 animal inferred extent reports, 4 ecological site reports, one map and explanatory material.

Please keep in mind the following when using and interpreting the enclosed information and maps:

- (1) These materials are the result of a search of our database for species of concern that occur in an area defined by requested road segment with an additional one-mile buffer surrounding the requested area. This is done to provide a more inclusive set of records and to capture records that may be immediately adjacent to the requested area. Reports are provided for the species of concern that are located in your requested area with approximately a one-mile buffer. Species of concern outside of this area may be depicted on the map but are not reported.
- (2) On the map, polygons represent one or more source features as well as the locational uncertainty associated with the source features. A source feature is a point, line, or polygon that is the basic mapping unit of an EO Representation. The recorded location of the occurrence may vary from its true location due to many factors, including the level of expertise of the data collector, differences in survey techniques and equipment used, and the amount and type of information obtained. Therefore, this inaccuracy is characterized as locational uncertainty, and is now incorporated in the representation of an EO. If you have a question concerning a specific EO, please do not hesitate to contact us.
- (3) This report may include sensitive data, and is not intended for general distribution, publication or for use outside of your agency. In particular, public release of specific location information may jeopardize the welfare of threatened, endangered, or sensitive species or communities.

- (4) The accompanying map(s) display management status, which may differ from ownership. Also, this report may include data from privately owned lands, and approval by the landowner is advisable if specific location information is considered for distribution. Features shown on this map do not imply public access to any lands.
- (5) Additional information on species habitat, ecology and management is available on our web site in the Plant and Animal Field Guides, which we encourage you to consult for valuable information. You can access these guides at <u>http://mtnhp.org</u>. General information on any species can be found by accessing the link to NatureServe Explorer.

The results of a data search by the Montana Natural Heritage Program reflect the current status of our data collection efforts. These results are not intended as a final statement on sensitive species within a given area, or as a substitute for on-site surveys, which may be required for environmental assessments. The information is intended for project screening only with respect to species of concern, and not as a determination of environmental impacts, which should be gained in consultation with appropriate agencies and authorities.

I hope the enclosed information is helpful to you. If in the future you would prefer to receive a digital PDF file instead of paper, just let know. Please feel free to contact me at (406) 444-3290 or via my e-mail address, below, should you have any questions or require additional information.

Sincerely,

Main R. Malle,

Martin P. Miller Montana Natural Heritage Program martinm@mt.gov

To: Bruce Barrett MDT Billings District Administrator 424 Morey St. PO Box 20437 Billings, MT 59104-0437 bbarrett@mt.gov

September 24, 2007

Dear Mr. Barrett,

The City of Red Lodge Parks Board appreciates the chance it had to meet on August 22 with MDT personnel Leroy Wosoba and Gary Neville at Red Lodge to discuss CTEP project opportunities in the Red Lodge area. We also appreciate the May 2, 2007 letter from Heidi Bruner requesting comments on the Highway 212 Corridor Study. The City of Red Lodge Parks Board looks forward to working with MDT to accommodate current and future transportation needs in Carbon County.

The City of Red Lodge Parks Board requests that MDT consider trails planned in the 2006 City of Red Lodge Comprehensive Trails Plan (RLTP) and that trails associated with highway projects be constructed according to the RLTP. This plan included extensive public involvement and participation. The RLTP has been officially adopted by the City of Red Lodge. MDT officials were provided a copy of the RLTP at our 8/22/07 meeting. Additionally, the RLTP is available online at: http://www.beartoothtrails.org/RLTrailPlan.pdf.

In addition, we ask that MDT consider the following comments, issues and concerns regarding MDT's Highway 78 (including the Highway 78 Corridor Study and the Red Lodge Northwest Project Number STPP 78-1-(8)-0), and scheduled studies and improvements on US Highway 212 (including the Corridor Study for the Red Lodge North Project Number STPP 28-2(25)70) and Highway 308:

1. Alternative transportation – MDT's reconstruction of Highways 212 and 78 should accommodate alternative transportation needs. The City of Red Lodge Parks Board requests that a non-motorized trail be developed and constructed adjacent to or near the Highway 78 and 212 corridors. Trails should meet needs for pedestrians, bicyclists, wheelchairs, and other non-motorized human traffic. Trails along Highways 212 and 78 should connect to trails and trail segments leading into the City of Red Lodge, outlying subdivisions, and other communities and subdivisions in Carbon and Stillwater Counties. The City of Red Lodge Parks board encourages MDT to eventually develop trails along Highway 212 from the Wyoming border to Laurel. Given that highway reconstruction projects do not often occur, it is imperative that such projects be planned and constructed

to accommodate future growth and demand for alternative transportation, recreation needs, and potential visual concerns.

- 2. Parking Regularly spaced parking facilities should be developed within highway right-of-ways along the Highway 212 and 78 corridors. This could be accomplished by retaining abandoned sections of highway, ensuring regularly spaced turnouts, and by constructing wide parking areas at junctions with secondary roads. Parking areas will provide opportunities for car pool users to park cars, safer school bus stops, future rest areas, and trail user parking along trail systems and equestrian routes/trails. Such parking areas should include signs indicating parking opportunities and also informational signs, such as roadside history, environmental interpretation, or providing information about the incredible scenery along both these highways. The City of Red Lodge Parks board would be willing to assist MDT in developing language for such signs.
- 3. Brewery Hill trail As detailed on pages 27 to 28 of the RLTP, "MDT officials have publicly stated their desire to construct a pedestrian/bike path that would promote travel along Brewery Hill but not directly adjacent to Highway 78. MDT has cited environmental and engineering constraints as a deterrent to adding a pedestrian/bike path adjacent to the road corridor. This plan suggests two possible alternative routes to avoid a pedestrian/bike path directly adjacent to Highway 78/Brewery Hill." Potential RLTP routes include two options connecting the West Bench area to the downtown Red Lodge area. The City of Red Lodge Parks board would be willing to assist MDT in working with private property owners to secure any easements needed for these routes.
- 4. Trail accessibility The City of Red Lodge Parks Board recommends that trails adjacent to Highways 212 & 78 provide maximum accessibility for the widest variety of user groups. As stated in the RLTP, "Accessible trails benefit not only people with mobility impairments, but older adults and families with young children. Designing a trail to accommodate bicyclists inherently creates a facility which is accessible to mobility impaired individuals. Such accessible trails are generally paved with asphalt or concrete, but may also use prepared surfaces such as aggregate or soil stabilizing agents mixed with native soils or aggregates. Critical features to consider for accessible trails include grade, cross slope, width, surface, and vertical clearance. Wherever possible, trails should meet grade, cross slope, tread width, tread obstacle, and surface material standards, such as those detailed in the US Department of Transportation Federal Highway Administration publication 'Designing Sidewalks and Trails for Access.'"
- 5. Road shoulders The road corridor should include road shoulders on both sides of the highway that are wide enough to safely accommodate bicycle use. This would allow road bicyclists to travel at high speeds along the highway in a corridor and keep this higher-speed non-motorized traffic separate from that of any trail system traffic and separate from automobile traffic.

- 6. Equestrian use The City of Red Lodge Parks Board encourages MDT to retain or purchase roadside easements adjacent to highways that will allow development of an equestrian trail system. A trail or corridor adequate for equestrian use should be large enough to accommodate two-way horse traffic while keeping equestrian traffic separated from other users, such as pedestrians, people walking dogs, and bicyclists.
- 7. Environmental impacts The City of Red Lodge Parks Board recommends that, in designing and constructing this trail, MDOT mitigate or eliminate impacts to the environment. This includes consideration of impacts to water quality, wetlands, sensitive plant species, noxious weed proliferation, wildlife, cultural resources, air quality, etc. Specific considerations should include:
 - a) Hydrologic impacts ensuring that bridge replacements, in-stream work, and construction activities do not adversely impact water quality or fisheries in East Rosebud Creek, Willow Creek (which is listed as impaired under section 303 d of the Clean Water Act), Red Lodge Creek, Rock Creek or adjacent wetlands and riparian areas. New culverts or bridge replacements should accommodate the passage of fish, amphibians, and other riparian-dependent species that are known to be present in each specific stream or riparian areas. Wherever possible, impacts to wetlands (both isolated wetlands and wetlands connected to flowing streams) and riparian areas should be avoided. Where this is not possible, wetland mitigation or improvement elsewhere should occur. In the event that wetland or stream mitigation sites are needed, the City of Red Lodge Parks Board would be interested in working with MDOT to locate potential wetland or riparian mitigation sites in and adjacent to City of Red Lodge property. It is likely that several such sites could be identified along Rock Creek in Red Lodge.
 - b) Surveying for noxious weed presence any detection of noxious weeds should include completion of chemical and/or mechanical weed treatment within and immediately adjacent to road corridor areas that will be disturbed.
 Proliferation of noxious weeds along the road and/or trail corridor could potentially lead to the spread of weeds into City of Red Lodge Parks and elsewhere in Carbon County and Montana.
 - c) Revegetation of disturbed areas to prevent introduction of noxious weeds and reduce impacts to native grassland species, prompt revegetation of disturbed areas should occur. This should include seeding or hydroseeding of disturbed areas and roadside ditches with mixes of native plant species appropriate to this area. All seed used should be certified as noxious weed free. Visually attractive native species, such as blue flax, blanket flower, cone flower, lupine, etc. should be included in seed mixes adjacent to any trail corridor. Where they will not interfere with highway visibility standards, willow slips and/or native deciduous tree seedlings should be planted adjacent to any disturbed riparian areas or irrigation ditches. In areas devoid of organic material, we recommend that MDOT consider use of topsoil placement, mulching, and/or hydroseeding. Revegetation efforts should be monitored until revegetation occurs. Criteria defining revegetation should be defined specific to each project and a revegetation monitoring plan should be

developed. Such a plan should include means for additional seeding and revegetation efforts in the event that initial methods are not fully successful.

- d) Wildlife impacts to reduce impacts to wildlife, we recommend that seed used for revegetation consist primarily of species deemed less palatable or desirable to wildlife. We also suggest that MDOT consider the use of wildlife fences and/or wildlife crossing structures to reduce potential for auto/wildlife collisions.
- e) Air quality please ensure that construction and related activities in and near Red Lodge and other population centers include measures to address impacts to air quality, such as dust abatement.
- f) Recycling we encourage MDOT to utilize recycling technology, such as crushed glass, recycled pavement, or shredded tires, to reconstruct highways and construct any associated trails.
- g) Salvage of trees the Parks Board requests that MDT allow the salvage of trees and shrubs that will be removed as part of highway reconstruction or ongoing highway right-of-way clearing in the area. Such trees and shrubs could be transplanted to City of Red Lodge property, including parks and street sides. This would not only save the trees and save City money for the purchase of new trees, but it would also provide a myriad of environmental and social benefits associated with placement of urban vegetation in neighborhoods and parks. Since fall or early spring would provide the best opportunities for transplant survivability, we would appreciate advance notice of any such salvage availability. This would allow scheduling of a professional arborist or tree service to salvage and transplant the trees. It may also be advantageous to allow tree salvage a year or two in advance of construction to ensure that salvage operations don't interfere with construction activities. We would also appreciate information on any applicable safety measures, such as signing or the use of flaggers, that would be required for this type salvage operation.
- 8. Use of abandoned road infrastructure for new trails In the event of road relocation or road segment relocation, the City of Red Lodge Parks Board encourages MDOT to locate trails or trail segments in areas currently occupied by highway infrastructure. Examples would include use of existing pavement, bridges, and culverts that will no longer be used by motorized traffic as part of the Highway 78 or 212 highways. Sections of abandoned pavement could also be utilized as parking areas, roadside parks, viewing platforms, or rest areas. Such use of portions of the highway to be abandoned would decrease costs of trail construction and costs of reclaiming these areas. In the event that such abandoned portions of the highway cannot be utilized for trail construction, we recommend that infrastructure (i.e. pavement, culverts, bridges, etc.) be removed and the areas are reclaimed and revegetated so as to prevent future erosion and/or noxious weed spread.

Please feel free to contact myself or any Parks Board members if you have any questions or need more information relating to City of Red Lodge trails or parks. Thank you for this opportunity to comment on the Highway 212 and 78 projects.

Please continue to keep the City of Red Lodge Parks Board on the mailing and meeting notification lists for all ongoing and future MDT or FHWA projects occurring in Carbon County on Highways 212, 78, and 308.

Sincerely,

/S/ DAN SEIFERT

Dan Seifert, Red Lodge Parks Board Chairman e-mail: <u>danerinn@copper.net;</u> phone: 406-446-1902 *for* The City of Red Lodge Parks Board c/o City of Red Lodge PO Box 9 Red Lodge, MT 59068

ana LAND RELIANCE

MASTER FILE COPY RECEIVED MAY 1 1 2007

ENVIRONMENTAL

May 10, 2007

Heidy Bruner Montana Dept. of Transportation P.O. Box 201001 Helena, MT 59620

Re: Request for Information - Project #STTP 28-2(25)70

Dear Ms. Bruner:

I just wanted to follow-up with you from your letter of 5/2/07. I appreciate the heads-up regarding the highway expansion on US 212. The Montana Land Reliance currently holds conservation easements on three properties that may be impacted by your planned activities. The issue of greatest concern would be the acquisition of additional right-of-way. Generally, it is MLR's policy to cooperate to the greatest extent possible on these projects. The problem we face is that in order for you to proceed absent a condemnation proceeding, we will have to go through the appropriate steps necessary to extinguish the easement on the property taken-up by the ROW. This can be a lengthy process for us and will require that you submit the appropriate documentation proving the public necessity for the action. As you plan for this project, you should anticipate a couple of months lead time in order for us to gain the necessary approvals from our own staff and Board of Directors prior to signing a quitclaim for the ROW.

Please let me know if you have any additional questions or concerns regarding this matter. The attached map should give you a good idea of the conservation easements we hold in this area. You can reach me at 443-7027 or via email at: <u>michael@mtlandreliance.org</u>

Sincerely,

Michael Downey Land Steward / GIS Analyst

GLACIER FLATHEAD OFFICE

470 Electric Ave. • PO Box 460 Bigfork, Montana 59911-0460 406/837-2178 • Fax 406/837 4980 email mirnw@mtlandreliance.org

MAIN OFFICE

324 Fuller Ave. * PO Box 355 Helena, Montana 59624-0355 406/443-7027 * Fax 406/443-7061 email info@mtlandreliance.org

EASTERN OFFICE

3318 3rd Ave. N., Suite 20/ * PO Box 3/1 Billings, Montana 59103-0171 406/259-1328 * Fax 406/259-1437 email mlrb@mtlandreliance.org
MLR Projects along U.S. Hwy 212



Disclaimer: information provided on this map is for illustrative purposes only and should not be relied upon to identify legal boundaries.

Red Lodge North Record of Telephone Conversation

9

جمعہ: محمد

Date: 6/12/2007 12:05 p.m.	Project Number: 2201102
Recorded By: Becky Rude	Phase Number:
•	•••••••••••••••••••••••••••••••••••••••
Talked With: Michael Downey	
Representing: Montana Land Reliance	
Address:	
City: State: Zip:	
Phone Number: 406-443-7027	
Email: michael@mtlandreliance.org	
\$	•••••••••••••••••••••••••••••••••••••••
Subject of Conversation: Easements	
Items Discussed: Mr. Downey informed me that none of the easements identifi Section 4(f) properties. He also informed me that the norther the correct lines, even though it does go over the roadway, b easement for right-of-way. He said typically in the older ease greater amount of right-of-way than what that right-of-way ea	ed along Highway 212 are nmost parcel probably has ut MDT likely has an ments MDT purchases a sement provides.
•	• • • • • • • • • • • • • • • • • • • •

Distribution: 🗌 KL&J

(Names) (Names)

Appendix E

Cultural Resources



Company Elevator (24CB1830), the Dairy Delite Drive-In (24CB1833), the O'Shea House (24CB1705), the Silakka House (24CB1712), the Boyd Mercantile (24CB1831), and the Maryott Ranch/Carbon County Dairy (24CB1336). There would be No Adverse Effect to the St. Thomas Church (24CB1717), the Monahan House /(24CB1720), the Kent Dairy Round Barn (24CB1320), and the Maryott Residence (24CB1339. There would be No Effect to the Red Lodge Commercial Historic District (24CB145) and the Hi Bug Historic District (24CB1030). We request your concurrence.

If you have any questions, please contact me at 444-6258.

on An Jon Axline, Historian

Environmental Services

Enclosure

cc: Bruce Barrett, Billings District Administrator Carl Peil, P.E., Preconstruction Bureau Gordon Stockstad, Resources Section

CONCUR MONTANA SHI DATE 22 APT 03 SIGNED

Received

JUN - 6 2003

Kadrmas Lee & Jackson

file; MDT/2003

Environmental Services Unit Phone: (406) 444-7228 Pax: (406) 444-7245

An Equal Opportunity Employer

Weo Fage: www.mot.state.mt.us Roed Roport: (400) 226-76:2 ITT: (800) 335-7592

										-#	2
	Sent	By: Kadrm	Ias, Lee	ano Jackson;		4062945502;	۲	90-17-V4 12:12	(PM)	Page 2/3	for the second s
15)		2	-		200	03091103	ř	۵.			
m	in at i	SP	Trén-	. :" M	ontana	Department of Tr	ensportatio	on	David A. C	alt, Director	
		ana aquit from beer	RE	ACTEDE	ic	2701 Prospect Avenu PO Box 201001	не		• Judy Mai	tz, Governor	
-			100	CODY		Helena MT 59620-10	X01			·	
			L			Josef	•			4.5	
F 39		S	eptembe	er 10, 2003	1	MOT	1.				
		X	4			EFFEL 70 2	4CB12B3	his 12 years			
7.00J		IV S	hark Bai Itate Hisl	imier, Ph.D. Ioric Preservation	o Office	APACIACE	LAIL WAT	BY:	€ ¹ C		
		l- P	410 8 th / O Box 3	Avenue 201202			6 1173	• • <u>-</u> > 动竹菜 网络竹子 分上	*****		
raj		· H	Helena, N	AT 59620-1202			attabarrat de C				
and the second se		S	ubject:	ŠTPP 28-2	(25)70	1		CONCL			Λ
T-88			-	Corridor St Control No	udy – R . 4375	ed Lodge North	MC	MIANA	SHPU	1	h
		C	`n An <u></u> 1	4 2002 me auto	sittad o) Determination o	DATE CA 250	USSIGNED De	- K- KAN	<u>arn</u> an	L.
r.q		ir	mpacts to	o twelve historic	sites loc	cated in the Area	of Potent	al Effect (APE)	for the	,	
Ì		a) 2	bove pro 003. Ho	oject in Carbon C owever, we inadv	ounty. crtently	You concurred v omitted assessir	vith our de	termination on the Rocky Fo	April 22, rk Branch o	f	
ring.		1	he North	em Pacific Raily	vay (24(CB1283).	.	· · ·			
l		r and	The Rock	y Fork Branch v	as com	pleted in 1889 ar	nd function	ned primarily as	a coal and		
-		a, ra	gricultu ailroad a	ral carrier betwee bandoned the br	n Red I e with s	odge and the rai	l yards/ju ade return	nction at Laurel. ed to agricultura	. In 1983, th I use by	e	
		a	djacent l	landowners, con-	rerted in	ito a roadbed or	allowed to	deteriorate. Th	e Rocky Fo	rk	
7 000		C C	Criterion	A in 1992.	iginic it	a nic ivadonal K	chizici oi	matoric macca	unoci .		
		. γ	What rem	nains of the aban	doned g	rade parallels m	ich of the	above 20± mile	project area		
1		V	While mo	ost of it lies outsi	de the p	roposed R/W an	d construc	tion limits boun	dary, ithin the		
101		p	roject ar	ea and may be in	npacted	by the proposed	project.	The grade would	d be impacte	d	
		b	y realign	nment and widen	ing of t l	ie existing roadv	vay.				
b		V	We have	determined, how	ever, th	at the proposed p Branch Line (2)	project wo	uld have No Ad	lverse Effec	ť	1
2 N		l	983 with	h the segment no	th of R	ockvale to Laure	still acti	ve and maintain	ed by the	•	
		B h	Burlingto lowever,	n Northern-Sant has been signifie	a Fe Rai cantly in	Iroad. The 32-n npacted since 19	nile segme 83. The b	nt south of Roc oridges, rails, tie	kvale, s. ballast and	ł	
		9: fr	ssociated	d features (i.e. sp	ikes, tie	plates, signal m	asts, etc.)	have been long	removed		
		R	tohn me . toberts.	Segments of the	line hav	e been converte	d into loci	al access roads o	or were turne	d	
1		oʻ th	ver to ad	ijacent landowne Other segments l	rs who have bee	either allowed it in abandoned an	to deterio d are deter	rate or otherwis	e obliterated ick of		
No.		n	naintenai	nce. There has a	lso been	considerable re	sidential c	levelopment wit	hin the		
		C	unnaor, '	which has purine	i impaci	ico me integrity	VI INC FAIL	ivao ime.			

Environmental Services Unit Phone: (405) 444–7228 F8x; (406) 444–7245

An Equal Opportunity Employer

Web Page: www.indtataio.nit.ua Road Report. (800) 226–7693 TTY: (800) 935–7552

wire wy nourmas, Lee and Jackson;

4002340002,

reu-11-04 12:12PM;

Page 3/3

Based on the existing condition of the line, the proposed project would not significantly alter its appearance, integrity, or ability to convey its historic significance to the development of Rock Creek valley. The most intact segments south of Roberts are located outside the proposed R/W boundary and are not included within the construction limits of the project. Those segments are currently being used as local access roads and would be perpetuated. There are no railroad-related structures that would be impacted by the project. The setting would not be adversely effected in that residential and commercial development all along the line in the project area over the last 20 years has already had a significant impact to the setting of the area. Because of the Rocky Fork branch line's significance to the development of Red Lodge and the Rock Creek valley, the MDT would install an interpretive marker adjacent to U.S. 212 near where a relatively intact segment of the line remains. We request your concurrence that the proposed project would have **No Adverse** Effect to the Rocky Fork Branch of the Northern Pacific Railroad (24CB1283).

If you have any questions, please contact me at 444-6258.

Ì

Son Axi in

Jon Axline, Historian Environmental Services

cc: Bruce Barrett, Billings District Administrator Carl Peil, P.E., Preconstruction Bureau Bonnic Steg, Resources Section

Way-02-05	10:09am Fro	m-MOT CONSULTANT DESIGN	SECTION	4054445253	T-131	P.002/003	F-484
Survey 4	B-J-J- Douwithsrido	Monfanc	<mark>i Departme</mark> i 2701 Praspe PO Box 2 Helena MT	nt of Transportation ct Avenue 201001 59620-1001	පිරි	Jim Lyncl an Schweitzer,	<u>1, Director</u> Governor
Л	April 25, 2005			Consultant Design	MAS	TER F	ILE
N S I F F	Mark Baumler, State Historic F 410 8 th Avenu ⁹ O Box 20120 Helena, MT 59	Ph.D. Preservation Office le 20 9620-1202		Rouling Buregiu Chine Canstablert Pachas Eng Doskań Silporzhof CTOP Englient Halici Ic	· · · · · · · · · · · · · · · · · · ·		
Ç L	Subject: STPP Red I Contr	28-2(25)70 .odge – North Corrid ol No. 4375	or Study	KLJ ENGINEERS			
1	Dear Mark:			auto martine for			

With the demise of the Irrigation Ditch Programmatic Agreement, FHWA has directed that the MDT complete the full Section 106 process on historic irrigation ditches for projects that were previously treated under the PA, but have not yet had their environmental documentation completed. Consequently, we have determined that the eleven ditches on the above project are ineligible for the National Register of Historic Places. The ditches are: the Brandt Ditch 24CB1722), Rule-Thompson Ditch (24CB1723), Duncan-Aikon Ditch (24CB1724), Bernhardt Ditch (24CB1725), Hunts Ditch (24CB1726), Highline Ditch (24CB1727), Rooney Ditch (24CB) 728), Drakes Ditch (24CB1729), Ward Ditch (24CB1730), Carbonado Ditch (24CB1731), and the Hoyle Ditch (24CB1761). We have based our determination on the following grounds. Although the ditches are shown on the maps in the Carbon County Water Resources Survey, they are privately owned and operated and are not part of any irrigation company or government irrigation project. Consequently, there is no information about the ditches in the water resources survey or in the local histories. The exact length of the facilities are not known, as are the construction dates, and photographs of them shown in the 2002 cultural resource survey for this project indicate that they are very basic facilities with few or no diversion structures, turn-outs, check structures, etc. They also appear to serve one or two users. A review of the names associated with the ditches do not include any individuals who were particularly important to the history of Carbon County or the areas where the ditches are located. The lack of historical information also makes it difficult to place the ditches in any kind of historic context. To review the site forms, we request that you refer to Laurie Travis, "Red Lodge North, Highway Improvements: A Cultural Resource Inventory from Roberts to Boyd, Carbon County, Montana," (Metcalf Archaeological Consultants, October, 2002) in your files.

The proposed MDT project, moreover, consists primarily of replacing c.1985 corrugated steel or reinforced concrete culverts with reinforced concrete culverts and minor rechanneling within and adjacent to the R/W to improve the hydraulies of the structures.

Environmental Schvides Bureau Phone: (406) 444-7228 Fox: 1406/ 444-7245

An Squal Opportunity Employer

Engineering Division TTY: 19001 335-7992 WED Page: www.mdf.mt.gov If you have any questions, please contact me at 444-6258.

pm A - T

Jon Axline, Historian Environmental Services

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section

Jun-20-05	08:18am	From-MDT C	CONSULTANT	OESIGN	SECTION	4064448253	1-211	P.002/005	F-610
			IVEZ	alana	Departm	ent of Transportation		Jim Lynch	. Director
eereding you	and the provide of the	IUN !	4 2005		2701 Pros PO Ba Hetena N	pect Avenue 5x 201001 11 59620-1001	Brian	i Schweitzer,	Governor
	May 19, 2	ALAN	INMEN	LP.	2005	052001	~		·
	,,,							т. 19 19 10 10 10 10 10 10 10 10 10 10 10 10 10	
	Mark Bau	ımler, Pl	n.D.			14 s	1		
	State Hist	loric Pre	servation	Offic	3	S .	See 3 29	* .	
	-1410.8^{th} /	Avenue				, UY:S	HPO	1	a
	P O Box 3	201202				in the second se	and the second	e 105	et
	Helena, N	AT 5962	20-1202				ت ۱ ۱۰ ب	Rod	r Lodae -
	Subject:	ST Re	PP 28-2(d Lodge	25)70 Nori	h Corrid	or Study	ONCUR	N co	rridar Study
		Co	ori iotino	. 4375		MONT	rana sh	LPO.	А
	Dear Mai	k:				DATE JOJUN DS SI	GNED Jan	Julai	huch
	The MD?	f has pro	gramme	d a pro	viect to re	construct 19.81 miles	s of U.S. Highw:	av 212	

The MDT has programmed a project to reconstruct 19.81 miles of U.S. Highway 212 from the north side of Red Lodge to the community of Boyd. On September 10, 2003, your office concurred with our Determination of Effect regarding the historic properties located along the road corridor. There are eleven privately-owned irrigation ditches located along that segment of the highway. They are: the Brandt Ditch 24CB1722), Rule-Thompson Ditch (24CB1723), Duncan-Aiken Ditch (24CB1724), Bernhardt Ditch (24CB1725), Hunts Ditch (24CB1726), Highline Ditch (24CB1727), Rooney Ditch (24CB1728), Drakes Ditch (24CB1729), Ward Ditch (24CB1730), Carbonado Ditch (24CB1731), and the Hoyle Ditch (24CB1761). Site forms for these ditches are included in Laurie Travis, "Red Lodge North, Highway Improvements: A Cultural Resource Inventory from Roberts to Boyd, Carbon County, Montana," (Metcalf Archaeological Consultants, October, 2002). Based on their contribution to the agricultural development of the lower Rock Creek Valley, the ditches are eligible for the National Register of Historic Places under Criterion A.

U.S. Highway 212 was originally constructed in 1936 and 1939 to provide a connection between U.S. 10, Red Lodge, and the recently completed Beartooth Highway. The roadway was last reconstructed in 1985 and all the culverts that would be impacted as part of this job were installed at that time. The Reinforced Concrete Pipe culverts proposed for this project are of the same dimensions as those installed in 1985. The culverts are, moreover, part of the roadway and not part of the ditch system. The following is a list of the ditches, the proposed impacts to them, and the effects of the proposed project on them.

Brandt Ditch (24CB1722). It is the intent of the project to replace the existing corrugated steel pipe (CSP) with a reinforced concrete pipe (RCP). The ditch is approximately two miles in length and would not be rechanneled where it crosses under U.S. Highway 212. The existing and historic function of the ditch would be perpetuated as would its alignment where it crosses under the roadway. None of the criteria of Adverse Effect could be applied to this ditch as a result of the project. The proposed project would, therefore, have No Effect to the Brandt Ditch.

2005 file: MOT

Environmental Services Unit Phone: (406) 444–7228 Fax: (406) 444–7245

An Equal Opportunity Employer

Web Page: www.nidt.state.mt.us Road Report: (800) 226–7623 TTY: (800) 335–7592 Jun-20-05

The <u>Rule-Thompson Ditch (24CB1723)</u> is about 2½ miles in length and consists of a crude field irrigation ditch. The MDT intends to replace the existing RCP crossing under the roadway with a new RCP as part of the project. In addition, about 180-feet of ditch would be rechanneled to accommodate a new crossing under the roadway. This constitutes 1.3% of the entire length of the ditch. Evidence suggests that the ditch was originally rechanneled in 1936 to accommodate the roadway. There would not be a significant change in the setting of the ditch as a result of the project and its existing and historic function would be perpetuated. The proposed width of the rechanneled portion of the ditch would match the existing width of the facility. There would be No Adverse Effect to the Rule-Thompson Ditch as a result of the project.

The <u>Duncan-Aiken Ditch (24CB1724)</u> is approximately 3 miles in length and also consists of a small field ditch. The MDT intends to replace the existing CSP with a new RCP at the point where the ditch crosses under U.S. Highway 212. The ditch would not be rechanneled to accommodate the pipe. The setting of the site would be retained and the historic function of the ditch perpetuated. There would be No Effect to the Duncan-Aiken Ditch as a result of the proposed project.

The Bernhardt Ditch (24CB1725) is about 1½ miles in length and is similar in design, appearance, and usage. According to the preliminary plan sheets, the existing culvert would not be replaced as part of the proposed project. There would be no change in the alignment of the ditch and no alteration of its current dimensions or use as a field ditch. There would, therefore, be No Effect to the Bernhardt Ditch as a result of the project.

The Hunts Ditch (24CB1726) crosses under U.S. Highway 212 at two points in Section 29, T5S, R21E. The ditch has a total length of about two miles and is similar in appearance to the Bernhardt Ditch. It is the intent of the MDT to replace the existing CSP's with RCP's. About 351-feet of ditch would also be rechanneled to accommodate the crossings. This constitutes about 3.3% of the total length of the ditch. There would not, however, be any change in the function of the facility and it would continue to provide water to adjacent farmlands near the roadway. The ditch, moreover, was rechanneled in 1936 when the roadway was originally constructed. The setting of the site would remain intact as would the majority of the alignment and configuration of the ditch. The MDT project would have No Adverse Effect to the Hunts Ditch.

The <u>Highline Ditch (24CB1727)</u> roughly parallels U.S. Highway 212 for $10\pm$ miles in Section 21, T5S, R21E. It is the intent of the MDT to replace the existing RCP with a new RCP. The project would also include the rechanneling of 148-feet of the 52,800± foot ditch. This would constitute about .24% of the entire ditch system. There would be no change in the historic function of the ditch and it would continue to irrigate farm fields in central Carbon County. The dimensions of the rechanneled portion of the ditch would closely match the existing width and depth of the facility. There would be no significant change in the setting of the facility. The proposed MDT project would, therefore, have No Adverse Effect to the Highline Ditch.

The <u>Rooney Ditch (24CB1728)</u> is 3± miles in length and carries water to agricultural property in the general vicinity of U.S. Highway 212. The MDT intends to replace the

4064446253

existing RCP with a new RCP. Along with the RCP, it is the intent of the project to rechannel about 754-feet of the ditch to better accommodate the new pipe. There would be no change in the historic function of the ditch, which was originally rechanneled in 1936 because of highway construction. The dimensions of the rechanneled portion of the ditch would closely match the segments located outside the project area. The setting of the property would also remain largely intact. The proposed project would have No Adverse Effect to the Rooney Ditch.

Drakes Ditch (24CB1729) is a small field ditch that is about two miles in length and carries water to one or two users. It is the intent of the MDT to replace the existing RCP with a new RCP and rechannel about 207-feet of the ditch to accommodate the new pipe and improve its hydraulics. This would constitute 1.9% of the entire length of the system. The ditch would continue to function in its historic capacity as a ditch and there would be no change to most of the facility. The dimensions of the rechanneled portions of the ditch would closely match the unchanged segment of the facility. There would be no change in the setting of the ditch.

The <u>Ward Ditch (24CB1730)</u> is approximately five miles in length and irrigates farmland for one or two users in Section 2, T5S, R21E and Section 35, T4S, R21E. The MDT intends to replace the existing RCP with a new RCP that would better accommodate the hydraulics of the site. About 541-feet or 2% of the ditch would be rechanneled in conjunction with the installation of the new pipe. There would be no change in the historic function or capacity of the ditch as a result of the project. The setting of the historic property would also remain largely intact with most of the ditch located outside the APE of the project. The setting would also remain intact. The proposed MDT project would have No Adverse Effect to the Ward Ditch.

The <u>Carbonado Ditch (24CB1731)</u> is also about five miles in length and carries water to one or two users in the lower Rock Creek valley. It is the intent of the project to replace the existing RCP with a new RCP and rechannel 105-feet of the ditch to accommodate the replacement. That would constitute about .39% of the entire length of the ditch. The historic function and capacity of the ditch would be perpetuated with more than 99% of the ditch located outside the APE of the project. The proposed project would have No Effect to the Carbonado Ditch.

About three miles of the <u>Hoyle Ditch (24CB1761)</u> is located near the Red Lodge – North project area with about 2,500-feet located within the existing R/W of U.S 212. It is the intent of the project to relocate 2,401-feet of the ditch outside the R/W boundary and, thus, remove, a hazard adjacent to the road. The existing ditch parallels the roadway. The proposed new alignment would also parallel the existing alignment – except it would be located outside the R/W line. There would be no change in the historic function of the ditch and its dimensions would be matched for its new alignment. None of the criteria for Adverse Effect would apply to the rechanneling and the proposed project would have No Adverse Effect to the Hoyle Ditch.

In summary, the proposed project would have No Effect to 24CB1722, 24CB1724, 24CB1725, and 24CB1731. There would be No Adverse Effect to 24CB1723,

24CB1726, 24CB1727, 24CB1728, 24CB1729, 24CB1730, and 24CB1761. We request your concurrence.

If you have any questions, please contact me at 444-6258.

Q On Jon Axline, Historian

Environmental Services

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section

Jul-14-05	02:40pm	From-MDT CONSULTANT	DESIGN SECTION	4064446253	T-240	P.002/003	F-660
Serving you	D	RECEIVE	2701 Prospect 2701 Prospect PO Box 20 Helena MT 59	of Transportation Avenue 1001 620-1001	Briar	Jim Lynor n Schweitzer,	1, Director Gavernor
		ENVIRONMENTE	, P				
Ju	ine 28, 20	05		វាធិន ជួរផ្ទុក			
M Si 1 ² P H	fark Baum ate Histo (10 8 th Ax O Box 20 elena, M ²	iler, Ph.D. ric Preservation Off venue 01202 f 59620-1202	ice B	<u>ү. ты го</u>		, José M D CORP REN L Effe	F T 1008 STUDY - 2056 AORTH: T PEASUES(Mbn
S	ubject: S' C C ear Mark	TPP 28-2(25)70 orridor Study – Red ontrol No. 4375 :	Lodge North	CONCI PROJECT WILL NO EFFECT ON NEH MOMTANA DATE(2-1-105 SIGNED	UR Have IP qualities SKPO	Nuclu	h

On April 22, 2003 your office concurred with our determination that the above project would have No Adverse Effect to the St. Thomas Church (24CB1717), the Monahan House (24CB1720), the Kent Dairy Round Barn (24CB1320), and the Maryott Residence (24CB1339) as a result of the above project. Recently, however, the MDT has made some changes to the preliminary design of the project to minimize impacts to four NRHP-eligible and listed sites above.

At the St. Thomas Church, the centerline would be shifted 9-feet away from the church. The existing centerline is 82-feet from the property and the existing pavement edge is 52½ feet from the church. The proposed centerline would be located 91-feet from the property and the pavement edge 61 feet from the church. The pavement edge, therefore, would be 8½ feet further from the property than it is now. In 2003, the proposed pavement edge would have been located 2-feet closer to the church. There would be no R/W acquisition at the property and the setting would be perpetuated. The proposed project would have No Effect to the St. Thomas Church (24CB1717).

The existing centerline would be perpetuated at 134-feet from the Monahan House. The existing pavement edge is located at 117½ feet from the residence. Widening of the roadway would place the proposed pavement edge 4-feet closer to the property at 113½ feet. All construction activities would be confined to the existing R/W and there would be no construction permits needed at the site. The widening, moreover, would consist of the addition of paved shoulders to the roadway. The driving lanes would remain the same distance from the property and there would be no significant change in the setting of the site. None of the criteria of adverse effect can be applied to the property by this proposed project. There would, therefore, be No Effect to the NRHP-eligible Monahan House (24CB1720) as a result of the proposed project.

At the NRHP-listed Kent Round Dairy Barn the existing centerline would be perpetuated at the site. The existing/proposed centerline would be 126-feet from the site and the existing pavement

le: M.DT/2005

Environmental Services Bureau Phone: (406) 444-7228 Fax: (406) 444-7245

An Equal Opportunity Employer

Engineering Division TTY: (800) 335-7592 Wob Page: www.mdt.mt.gov

4064446253

edge is 115-feet from the site. Widening would place the proposed pavement edge about 4½ feet closer to the property at 110½ feet. The would be no R/W acquired from the site, but a construction permit would be needed to reconstruct a fill slope about 13-feet beyond the R/W boundary. A fill slope currently exists at the site. The reconstruction would allow for the wider roadway and addition of the paved shoulders. The reconstructed fill slope would extend to the parking lot that is currently located between the brick barn and the roadway. There would be no physical encroachment on the site and the building. There would be no significant change in the setting of the property as a fill slope would be perpetuated. The site would retain its historic appearance and none of the criteria of adverse effect applies to the project and this site in this instance. There would, therefore, be **No Effect** to the NRHP-eligible Kent Road Dairy Barn (24CB1320) because of the MDT's proposed project.

The existing centerline would be perpetuated at the Maryott Residence. The existing centerline is 131-feet from the residence while the pavement edge is 117½ feet from the residence. Widening would put the proposed pavement edge about 4-feet closer to the site at 113½ feet from the site. The widening would consist of a paved shoulder. In order to reconstruct the existing fill slope to accommodate the wider roadway, the construction limits would extend 15-feet beyond the existing R/W to within 81-feet of the project. There would be no physical encroachment on the residence or any of its outbuildings. Indeed, the setting of the property would remain largely unchanged as the construction beyond the existing/proposed R/W boundary would consist of the modification of an existing fill slope. A two-lane facility would be perpetuated and the widening in the vicinity of the site would consist of paved shoulders to improve highway safety. Other than the grassy fill slope, no other vegetation would be impacted at the site and it would appear as it does currently. The proposed project would have No Effect to the Maryott Residence (24CB1339).

We request your concurrence that the proposed project would have No Effect to the St. Thomas Church (24CB1717), the Monahan House (24CB1720), the Kent Dairy Round Barn (24CB1320), and the Maryott Residence (24CB1339). In all cases the MDT has made modifications to the plans to avoid impacting these sites as much as possible and still accomplish the goals of the proposed project.

If you have any questions, please contact me at 444-6258.

Jon Axline, Historian

Environmental Services

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section

Monte	ana Department of Transportation	Jim Lynch, Director
ing you with pride	2701 Prospect Avenue PO Box 201001 Helena ME 585 E POVE E	Brian Schweitzer, Governor
September 4, 2007	SEP 2 1 2007	And Ander - North
Mark Baumler, Ph.D. State Historic Preservation Office 1410 8 th Avenue P O Box 201202 Helena, MT 59620-1202	ENVIRONMENTAL	5-47-0
	CONCUR	
Subject: STPP 28-2(25)70 Red Lodge – North Control No. 4375	MONTANA SHPO DATE <u>1754007 SIGNED WW</u>	lich
Dear Mark:		

The Carnegie Library (24CB145) was inadvertently left out of the Determination of Effect we submitted to your office on June 28, 2005. The library is located at the intersection of US 212 (Broadway), Oakes Avenue, and 8th Street on the north side of Red Lodge. Based on the latest plans for that intersection, there would be some modifications made to it to facilitate traffic movement in that area. The existing alignments of all three streets would be perpetuated, but the sidewalks and curb and gutter would be changed. US 212/Broadway would remain a two-lane facility, but bulb-outs would be constructed at the intersection of 8th Street to assist pedestrians crossing the street there. The bulb-out adjacent to the library would require the acquisition of additional Right-of-Way (R/W). Also a sliver of R/W would be required at the northwest corner of the lot and east side of the property to accommodate a new sidewalk. All other construction activities would be confined to the existing R/W. The attached preliminary plan for the library indicates there would be no physical encroachment on the library and the lot upon which it is located would remain mostly intact. Based on the plan, we have determined that the proposed project would have No Adverse Effect to the Carnegie Library. There would be no physical change to the building and no significant change in the triangle between the three streets upon which it sits. The building would continue to function as the city library, although the triangle would be changed somewhat to facilitate both motor vehicle and pedestrian traffic there. We request your concurrence.

If you have any questions, please contact me at 444-6258.

on Axline

æ 4

Jon Axline, Historian Environmental Services

Attachment

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section Heidy Bruner, P.E., Engineering Services Appendix F

Section 4(f) Properties

F2 VV H



Montana Fish, Wildlife & Parks

MAR 2 8 2003

1420 East Sixth Avenue P O Box 200701 Helena, MT. 59620-0701 Phone: (406)444-3939 FAX: (406)444-3023 March 25, 2003

Sheri G. Lares Senior Environmental Planner Kadrmas Lee & Jackson 3237 E.Broadway P O Box 1157 Bismarck ND 58502

۲,

Dear Ms. Lares:

1

We received your letter and great aerials yesterday for proposed project STPP28-2(25)70. Unfortunately most of your questions will need to be answered by regional personnel in the FWP Billings office. I do not think it would be appropriate for FWP to evaluate or answer several of the items particularly item 15 when we do not have specifics on right of way required other than the areas highlighted on the aerials. FWP does own the sites identified as Horse Thief Station, Beaver Lodge, Bull Springs, and Water Birch. It does not own Rock Creek Fishing Trail, but perhaps the regional personnel will be able to shed some light on that one for you.

I have forwarded your letter, aerials and forms to that office and you should hear directly from them on this.

Sincerely,

Duby Dels

Debby Dils Land Section Supervisor

Cc: Harvey Nyberg, Walt Timmerman



Montana Fish, Wildlife & ParKs

#2

20 Mar Brown

- / /

Sheri Lates K, L #J Bismarck, ND 58502

(• '	
APR	2	003	Ì
man.			

Dear Ms. Lares, Here are the forms you requested. Please excuse the hand written note built I am on the road. Please contact me if you have any questions. With Regards Doug Habermann Regional Parks Manager 2300 Lake Elmo Dr Billings MT 59105 406-247-2954

		Hote: this so owned and open				
	PUBLIC RECREATI	MDOT.				
Facili	ty Name: Kark CREEK FISHING	6 TIZATL				
Facili	ty Location:	۶				
1.	Nearest crossroads (landmarks): <u>B</u> <i>Rock, Creek</i>	vidge crossing on				
	Segment/Concept plan sheet number:					
	Approximate distance from existing rig	ght-of-way (metric/english):				
2.	Who owns the facility? Any applicate easements, restrictions, conditions)?_	ole clause affecting the ownership (lease, MDOT Crestarea)				
3.	What is the total size of the facility (he devoted to recreational facilities (hecta	ectare/acre)? What is the approximate area acre/acre)?				
4.	Type of Section 4(f) property (park, re highway rest area	creation, historic) and park classification:				
5.	What types of activities are available a	at the facility?				
	E Playground Equipment	🗖 Biking Trails				
	E Tennis Courts	C Volleyball Courts				
	E Picnic Area	☐ Shelters				
	E Basketball Courts	C Swimming Pool				
	T Lake	C Skating Rink				
	☐ Soccer Fields	Golf Course (# of holes)				
	T Walking Trails	T Other:				
	Highway restared					
6.	Description and location of existing face Rest area with bay	cilities (ball diamonds, tennis courts, etc.):				
7.	Description and location of planned fu	ture facilities:				
8.	Any identified facility deficiencies?	?				
		·				

1. 10

, 8	Are recreational facilities open to the general public? What restrictions to access are there? <u>Yes - during summer season</u>
- E	Estimated number of visitors per year:
ļ	Are there other similar facilities in the vicinity?
⊥ t	Are there unusual characteristics of the property that either reduce or enhance the value of the property? (flooding problems, terrain conditions, other features)
	Statement of significance from the official who has jurisdiction over the entire Section 4(f) property:
١	What impact would the proposed improvements have on this facility?
e a	nd Title of Person Interviewed: Doug Habermann-Regional Parks
	: 2300 Lake Elmo Dr Billingo MT 59105
ess	

Ĩ.

Ē

224

10 march 10

S

A survey by

بر م


1420 East Sixth Avenue P O Box 200701 Helena, MT. 59620-0701 Phone: (406) 444-3939 June 17, 2005

Sheri G. Lares Senior Environmental Engineer Kadrmas, Lee & Jackson 3237 E. Broadway Bismarck, ND 58501

Dear Ms. Lars:

Montana Fish, Wildlife & Parks has not acquired any new property in the Highway 212 project area since your last formal inquiry in 2003. The following sites are still owned by this agency and will qualify for special treatment due to federal funding (LWCF & Dingell Johnson) if impacted by reconstruction:

Beaver Lodge Fishing Access Site (DJ) Bull Springs Fishing Access Site (LWCF)

Horse Thief Fishing Access Site (LWCF) Water Birch Fishing Access Site (LWCF)

The FWP land records do not show any information on the Rock Creek Fish Trail although its sounds as if it should be a 4(f) property. We may have discussed this several years ago, but I did not keep any notes resolving FWP involvement in that property.

Sincerely,

Doby Dels

Debby Dils Land Section Supervisor

Enclosure C: R5; Walt Timmerman



Received

PUBLIC RECREATIONAL FACILITIES (SCHOOLS, HIGHER EDUCATION)

AUG 2 2 2005

Kadrmas Lee & Jackson

School Name: Roberts Public Schools

School Location: 106 East Maple Street Roberts, MT 59070

 Nearest crossroads (landmarks): <u>North of the intersection of East Maple Street</u> and US Highway 212

Segment/Concept plan sheet number: Between RP 214 and 217 (approximate)

Approximate distance from existing right-of-way (metric/english): <u>Adjacent</u>

- 2. Who owns the school? Any applicable clause affecting the ownership (lease, easements, restrictions, conditions)? Tax payers of District #5
- 3. What is the total size of the school (hectare/acre)? What is the approximate area devoted to recreational facilities (hectare/acre)? Total 10 Rec Oug. 5
- Type of Section 4(f) property (park, recreation, historic) and park classification: School playground (recreation)
- 5. What types of activities are available at the park?

Playground Equipment	C Biking Trails
r Tennis Courts	
F Picnic Area	☐ Shelters
𝒢 Basketball Courts	C Swimming Pool
☐ Lake	Skating Rink
T Handball/Multi-Use	☐ Golf Course (# of holes)
Jズ Baseball Fields	Softball Fields
C Soccer Fields	🖵 Walking Trails
ROther Feelball Fields / Terry	<u>1</u> <u>K</u>

- 6. Description and location of existing facilities (ball diamonds, tennis courts, etc.): See WRANCE Attricture Description-
- 7. Description and location of planned future facilities:
- 8. Any identified recreation deficiencies? Lack of Restrocms

9.	Describe access to the school (pedestrian, vehicular): The awar is a work of working on the west side of school building along Warmer 1212
10.	Are recreational facilities open to the general public? What restrictions to access are there? The Recreational Facilities are open to the general public all year.
11,	Estimated number of visitors per year: 300 0
12.	Are there other similar facilities in the vicinity?
13.	Are there unusual characteristics of the property that either reduce or enhance the value of the property? (flooding problems, terrain conditions, other features) <u>Locatton rest to a major highway</u> , Ditch right way blo school and play ground area
14.	Statement of significance from the official who has jurisdiction over the entire Section 4(f) property: Record of Trustees for School District #5
15.	What impact would the proposed improvements have on these facilities? Server issues with closness of Highway, Loss of Land Area for the scient law ground area
Name	and Title of Person Interviewed: J. Freezy Brenes Superinterview
Addre	ss: PU Bur 78 Roberts, MT 59070
Telepl	hone: 445-, 421
Intervi	Date of Interview: B/15/05
	, ,

•

J_

Description of the Area

The area in question is north of the school. The approximate acreage is 5 acres. Within those five acres is a football field surrounded by a five line track. On the west side of the football field/ track is the playground. The playground is fenced off from the highway with a five foot chain link fence. The Roberts Community Foundation has created an area within the playground area for the community to enjoy by adding benches, new playground equipment and lighting. The area also has a location to accommodate softball and little league baseball to the east of the playground equipment. This summer we added a basketball court west of the grandstands by pouring a new cement pad. This court lies in-between the track and the fence. We have also put in underground sprinklers to irrigate the area with the pump house located next to the crow's nest.

Future Plans

The designing of the area was set up to take place in three stages. We have currently completed stage 2 with the addition of the basketball court. Stage 3 is still in the planning stages. Early proposals include finishing the fence along the eastside of the track to secure the area from the irrigation ditch, lighting more of the area such as the new basketball court, and the addition of public restrooms. There is also talk about lighting the area inside of the track.

Concerns:

The greatest concern is the safety of our children with the highway being moved closer to the playground area. Also, there is the loss of area for our children to play. These are two areas that we feel are very important to understand and rectify before we proceed. OCT 1 2 2006





Red Lodge Area Chamber of Commerce

September 11, 2006

Mayor Roat and City Council PO Box 9 Red Lodge, MT 59068

RE: Highway 212 design-roundabout alternative

Dear Mayor Roat and City Council:

The Red Lodge Area Chamber of Commerce is giving its strong support for the proposed roundabout design presented by the engineering firm Kadrmas Lee & Jackson and Montana Department of Transportation for the reconstruction of the intersection of State Highway 78 and U.S. 212.

Reasons for this support are as follows:

- Goal 8 of the recently completed <u>Downtown Red Lodge Assessment & Action</u> <u>Plan</u> states – "Develop attractive entryways that create a positive first impression of Red Lodge, and clearly welcome visitors and guide them into downtown." The chamber believes the roundabout plan for this intersection would be a major improvement to the north entrance to the town. The center section of the roundabout can be a wonderful area for art and welcoming flower gardens. The roundabout would say – this is Red Lodge and welcome to our unique and beautiful city.
- 2. A modern designed roundabout will accommodate all sizes of vehicles legally allowed on our state highway system.
- Traffic engineers and traffic studies support the use of the design at similar major intersections for their safety history and ease of use. These studies and reports testify to the safe and efficient use of similar installations.
- 4. We support the cost effectiveness of the roundabouts over the possibility of the MDOT paying for and maintaining a full signalized intersection. We understand the proposed roundabout design is less expensive to construct in the beginning and much easier and less costly to maintain on an annual basis.
- 5. Red Lodge resident's takes pride in not having a "street stop light" in our town. Since this intersection most likely would be the first to meet state warrants for a signalized placement this roundabout will forestall the need for such a "street stop light" installation for many years.

We have heard many of the negative issues regarding roundabouts, but find them wanting for facts. This type of intersection design may be new to Montanans, but it certainly is not new to persons who have traveled in other states and other parts of the world. We, who have used them, even in Europe, have found them to be safe and effective in moving auto, bike and pedestrian travel though highly traveled intersections such as S.H. 78 and U.S. 212.

We trust you will take our comments and support MDOT in its effort to make our town a safer and more attractive place to visit, live and travel through by supporting the roundabout design alternative.

Sincerely yours, Don L. Kinne President,

CC Bruce Barrett, MDOT

.

#6

NDD	Montana Department of Transportation	Jim Lynch, Director
August 22, 2006	2701 Prospect Avenue PO Box 201001 Helena MT 59620-1001	Brian Schweitzer, Governor
City of Red Lodge Mayor 1 South Platt P.O. Box 9 Red Lodge, MT 59068	and City Council RFTEIVED OCI 1 2 2006 ENVIRONMENTAL	MASTER FILE COPY
Re: Section 4(f) Regu MDT Project: R	alations: Significance of Grassy Area Adjacent to Red Lodge North	Visitors Center

Dear Mayor and City Council:

Control Number: 4375

Project Number: STPP 28-2(25)70

The Montana Department of Transportation (MDT) is evaluating potential environmental impacts that may be associated with the above project. With this letter we are requesting your assistance in providing public land information on the grassy area north of and adjacent to the Red Lodge Visitors Center and Chamber of Commerce (from this point forward, referred to as "the site"). We need your information to determine if a certain federal regulation might be applicable to this site.

The federal regulation we are specifically interested in is found at 49 USC 303 Section 4(f) and 23 CFR 771.135 and is referred to as the Section 4(f) Regulation. Potentially applicable portions of the Section 4(f) Regulation state that the Federal Highway Administration (FHWA) can approve projects requiring the use of publicly owned land of a public park or recreation area only if there is no feasible and prudent alternative to such use and only if the project includes all possible planning to minimize harm.

The project under consideration involves reconstructing US Highway 212 from 8th Street in Red Lodge to north of Boyd. As you know, in the immediate vicinity of the Red Lodge Visitors Center/Chamber of Commerce, the project includes two improvement options for the MT Highway 78/US Highway 212 intersection: a roundabout and a signalized intersection. In general, a conscious effort was made to keep construction limits of this project within the existing right of way as much as possible. However, both build options under consideration at the MT 78/US 212 intersection would require additional right of way from this site.

Before the project can proceed, it must be determined if the 4(f) Regulation is applicable to that site. MDT cannot determine applicability of this regulation. "Officials having jurisdiction" must determine applicability of the 4(f) Regulation by evaluating the major purposes and functions of the site and the significance of the site. For purposes of applying this regulation, the City of Red Lodge should consider four criteria in your evaluation of the site. Those criteria are outlined below.

First, the site must be publicly owned. Our records indicate this site is publicly owned and therefore, the first criterion is met. Please inform us if our understanding is incorrect.

Second, the site must be open to the public. Our understanding is that the area is open to the public at all times. As a result, the second criterion is met. Please inform us if our understanding is incorrect.

Third, one of the major purposes and functions of the site must be a park or recreation area. Please note that incidental, secondary, occasional or dispersed recreational activities do not constitute a major purpose.¹ Our understanding is that the site is used for incidental, secondary, occasional or dispersed recreational activities. Please inform us if our understanding is incorrect.

¹ US Department of Transportation Federal Highway Administration, Office of Planning, Environment and Realty Project-Development and Environmental Review, FHWA Section 4(f) Policy Paper, page 11, March 1, 2005.

Red Lodge City Planner	Red Lodge North Project
August 22, 2006	STPP 28-2(25)70
Page 2 of 2	Control Number 4375

If the third criterion is met, then the fourth criterion must be considered. For the fourth criterion to be met, the site must be a "significant property." Significance means that in comparing the availability and function of this site with the park and recreation objectives of the community or authority, this site plays an important role in meeting those objectives. Management plans or other official forms of documentation regarding the land, if available and up-to-date, are important in this determination.² Our understanding is that the potentially impacted portions of this site do not play an important role in meeting community overall recreation objectives. Please inform us if we are incorrect and the City of Red Lodge determines that this site is a significant property.

Based on our preliminary review of the impacts to the site, it appears that this site may not meet the criteria for Section 4(f) applicability. However, the City of Red Lodge, as the officials with jurisdiction over the site, must make that applicability determination.

If you determine that one of the primary purposes and functions of the site is not recreation and/or the site is not significant, the Section 4(f) regulations would not apply. Please sign below if you concur. If you do not concur, please respond with a letter.

If you have any questions or concerns, please phone me at 406.444.7203 or Bruce Barrett at 406.657.0210. We will be pleased to assist you. Thank you for your assistance in this matter.

Sincerely,

1 Suner

Heidy Bruner Environmental Services Bureau Project Development Engineer

The City of Red Lodge concurs that the grassy area adjacent to the Red Lodge Visitors Center/Chamber of Commerce is not "significant" to the City's overall recreation system.

cc: Jean A. Riley, PE Bruce Barrett Gabe Priebe, PE Tom Martin, PE Jim Mullins Heidy Bruner Alan Woodmansey, PE File MDT Environmental Service Bureau Chief MDT Billings District Administrator MDT Consultant Design MDT Consultant Design MDT Right of Way MDT Environmental Services FHWA

S:\PROJECTS\BILLINGS\4000-4999\4375\4375EN4fCSP001.DOC

2 Ibidem, page 12.

Red Lodge North Record of Telephone Conversation

Date: 07/13/07 10:15 a.m.	Project Number: 2201102
Recorded By: Becky Rude	Phase Number:
•	•••••••••••••••••••••••••••••••••••••••
Talked With: Hugh Huntley	
Representing: Magic City Fly Fishers	
Address:	
City: State: Zip:	
Phone Number: 406-652-2261	
Email: hughmaryhunt@yahoo.com	
\$	•••••••••••••••••••••••
Subject of Conversation: Rock Creek Fishing Trail	
Items Discussed: Mr. Huntley informed me that the Rock Creek Fishing Trail is far as he knows it is publicly owned. They have been mainta for the past 10 to 15 years. The only involvement Montana F was that they put up the sign near the trail. The trail runs app the rest area parking lot to the Creek.	open to the public and as ining it about twice a year ish, Wildlife & Parks had proximately 100 yards from

Distribution: CKL&J

(Names) (Names)

Appendix G

Section 4(f) De Minimis Evaluations



U.S. Department of Transportation

Federal Highway Administration Montana Division

585 Shepard Way Helena, MT 59601

January 8, 2008

Mark Baumler State Historic Preservation Office PO Box 201202 Helena, MT 59620-1202

In Reply Refer To: HDA-MT

Subject: **De minimis Finding** Project Name: Corridor Study – Red Lodge North Project Number: STPP 28-2(25)70 Control Number: 4375

Dear Mr. Baumler:

By way of this letter and attachments, the Federal Highway Administration (FHWA) is requesting written concurrence from the Montana State Historic Preservation Office (SHPO) that the determinations of effect as listed below are still applicable:

Boyd Mercantile (24CB1831) No Effect.

The Preferred Alternative would avoid impacts to the Boyd Mercantile and minimize right-ofway impacts to this property. At this location, highway-related storm water drainage would be diverted to the west side of the roadway, eliminating the need for a full ditch on the east side of the roadway. In addition, a reverse curb would be used to delineate the Boyd Country Store parking lot from the roadway which would improve the safety and functionality of the parking lot. A temporary construction easement would be required at this location to facilitate construction. Also, MDT has existing easements for the existing roadway that would be incorporated into permanent right-of-way.

Carnegie Library (24CB145) No Adverse Effect.

The Preferred Alternative would avoid impacts to the Carnegie Library structure and minimize/mitigate impacts to the Carnegie Library property, as described below:

- The Carnegie Library building would be avoided with the use of a bulb-out at the 8th Street / US Highway 212 intersection.
- A sidewalk would be added on the east and west sides of US Highway 212 from 8th Street to the north; this would improve pedestrian access to the Library.
- Additional public parking spaces would be added on the west side of the Library along Oakes Avenue, which would improve vehicular access to the Library.
- The existing Mountain Ash tree on the southeast corner of the Library property would be avoided; however, the sign may need to be relocated.



• MDT will work with Carnegie Library representatives during project design to determine an appropriate treatment for the triangular area on the north end of the property (such as seeding with grass seed).

Kent Dairy Round Barn (24CB1320) No Adverse Effect.

The Preferred Alternative would avoid impacts to the Kent Dairy Round Barn structure and minimize impacts to the property. Right-of-way impacts would be minimized with the use of a buried culvert rather than an open ditch at this location.

Rocky Fork Branch of the Northern Pacific Railroad (24CB1283) No Adverse Effect

The Preferred Alternative would impact the Rocky Fork Branch of the Northern Pacific Railroad in approximately eight locations. Impacts at four of these locations would be due to relocation of irrigation ditches outside of the proposed right-of-way per standard MDT procedures. At two locations, the impact would be a result of the construction of the highway ditch. One location would be impacted by the addition of a northbound passing lane. Lastly, one location would be impacted by installation of a new culvert under the railroad bed to improve storm water drainage for the community of Roberts. The site consists of approximately 22 acres, of which approximately 3.2 acres would be disturbed by the Preferred Alternative.

This site has already been significantly impacted. The bridges, rails, ties, ballast, and associated features have been long removed from the line. Also, segments of the line have been converted into local access roads, residential developments have encroached on the line, and lack of maintenance has led to deterioration of the line's integrity. The minor impacts from the Preferred Alternative would not substantially alter the railroad line's historical integrity, as it has already been significantly impacted.

In addition to Section 106 of the National Historic Preservation Act (NHPA), FHWA must comply with the provisions of Section 4(f) of the 1966 Department of Transportation Act. Historically, Section 4(f) has required that prior to approval of any federally-funded highway project resulting in the "use" of listed or eligible historic properties under the NHPA; the FHWA must perform an avoidance analysis to determine whether there is a "feasible and prudent" alternative that would avoid the Section 4(f) resource.

In August of 2005, Section 138 of title 23, USC was amended under the Safe, Accountable, Flexible, and Efficient Transportation Act: A Legacy for Users (SAFETEA-LU). Section 6009 of SAFETEA-LU provided new legislative authority to address programs and projects with minor or 'de minimis' impacts on a Section 4(f) resource.

More specifically, Section 6009(b) (2) of SAFETEA-LU states:

(2) HISTORIC SITES.--With respect to historic sites, the Secretary may make a finding of *de minimis impact* only if--

(A) the Secretary has determined, in accordance with the consultation process required under section 106 of the National Historic Preservation Act (16 U.S.C. 470f), that--

(i) the transportation program or project will have no adverse effect on the historic site; or

(ii) there will be no historic properties affected by the transportation program or project;

(B) the finding of the Secretary has received written concurrence from the applicable State historic preservation officer or tribal historic preservation officer (and from the Advisory Council on Historic Preservation if the Council is participating in the consultation process); and

(C) the finding of the Secretary has been developed in consultation with parties consulting as part of the process referred to in subparagraph (A).

This new provision of Section 4(f) is the basis of this letter, and of the FHWA's determination of *de minimis* impacts.

De Minimis Determination

The findings of "no effect" and "no adverse effect" reflect a conclusion that the uses identified in the attached exhibits will not "alter, directly or indirectly, any of the characteristics of [the] historic property that qualify the property for inclusion in the National Register in a manner that would diminish the integrity of the property's location, design, setting, materials, workmanship, feeling, or association."

If you concur in the "no effect" and "no adverse effect" determinations, FHWA intends to make a finding that impacts to historic resources that would result from implementation of the subject project would be *de minimis* for purposes of Section 4(f), as recently amended by Congress.

Request for Concurrence

The FHWA requests the written concurrence of the Montana SHPO in the above-described findings of "no effect" and "no adverse effect" on historic resources from the subject project. This written concurrence will be evidence that the concurrence and consultation requirements of Section 6009 of SAFETEA-LU, as they will be codified at 23 U.S.C. § 138(b) (2) (B) & (C), and 49 U.S.C. § 303 (d) (2) (B) and (C) are satisfied. Concurrence can be provided either by signing and dating this letter or by separate letter from the Montana SHPO to the Federal Highway Administration, 585 Shepard Way, Helena, MT 59601.

Sincerely,

Juny Ke

Kevin L. McLaury, P.E. Division Administrator

Attachments

cc: Gabe Priebe - MDT, Engineering Project Manager Carl James - FHWA, Transportation Specialist

File: STPP 28-2(25)70 aw/lw



PAGE 02/02 KADRMAS LEE JACKSON 7018450252 03/08/2094 08:37 Sheri Love 2003040801 Montana Department of Transportation David A. Galt, Director RECENTED 01 Prospect Avenue Judy Martz, Governor Helene MT 59520-1001 APR 2 4 2003 April 4, 2003 APR 0 8 2003 Carridon Stud Ximmedon Stud ENVIRONIZENTAL Mark Baumler, Ph.D. State Historic Preservation Office 1410 8th Avenue P O Box 201202 Helena, MT 59620-1202 STPP 28-2(25)70 Subject: Corridor Study - Red Lodge North Control No. 4375 Enclosed is the Determination of Effect for the above project in Carbon County. We have determined that the proposed project would have No Effect to Finley House (24CB1822), the Richardson Bungalow (24CB1287), the Montana Dakota Grain Company Elevator (24CB1830), the Dairy Delite Drive-In (24CB1833), the O'Shea House (24CB1705), the Silakka House (24CB1712), the Boyd Mercantile (24CB1831), and the Maryott Ranch/Carbon County Dairy (24CB1336). There would be No Adverse Effect to the St. Thomas Church (24CB1717), the Monahan House /(24CB1720), the Kent Dairy Round Barn (24CB1320), and the Maryott Residence (24CB1339. There would be No Effect to the Red Lodge Commercial Historic District (24CB145) and the Hi Bug Historic District (24CB1030). We request your concurrence. If you have any questions, please contact me at 444-6258. CONCUR Jon Axline, Historian MONTANA SHPO **Environmental Services** DATE 22 Apr 03 SIGNED Enclosure Bruce Barrett, Billings District Administrator cc: Carl Peil, P.E., Preconstruction Bureau Gordon Stockstad, Resources Section Received. JUN - 6 2003 Kadrmas Lee & Jackson

file: MDT/2003

Ervironmental Services Unit Phono: (606) 444–7228 Fox: (406) 444–7245

An Equal Opportunity Employer

Web Page: www.mot.state.mi.u4 Robo Raport: (ROD) 326-7623 TTY: (800) 335-7592

Monte	ana Department of Transportation	Jim Lynch, Director
arviog you with pride	2701 Prospect Avenue PO Box 201001 Heleng MESCEPVEL	Brian Schweitzer, Governar
September 4, 2007	SEP 2 1 2007	
Mark Baumler, Ph.D. State Historic Preservation Office 1410 8 th Avenue P O Box 201202 Helena, MT 59620-1202	ENVIRONMENTAL	541 F 2
	Concur	
Subject: STPP 28-2(25)70 Red Lodge – North Control No. 4375	MONTANA SHPO	-lih
Dear Mark:		
The Camegie Library (24CB145)	was inadvertently left out of the Deter	mination of Effect we

submitted to your office on June 28, 2005. The library is located at the intersection of US 212 (Broadway), Oakes Avenue, and 8th Street on the north side of Red Lodge. Based on the latest plans for that intersection, there would be some modifications made to it to facilitate traffic movement in that area. The existing alignments of all three streets would be perpetuated, but the sidewalks and curb and gutter would be changed. US 212/Broadway would remain a two-lane facility, but bulb-outs would be constructed at the intersection of 8th Street to assist pedestrians crossing the street there. The bulb-out adjacent to the library would require the acquisition of additional Right-of-Way (R/W). Also a sliver of R/W would be required at the northwest corner of the lot and east side of the property to accommodate a new sidewalk. All other construction activities would be confined to the existing R/W. The attached preliminary plan for the library indicates there would be no physical encroachment on the library and the lot upon which it is located would remain mostly intact. Based on the plan, we have determined that the proposed project would have No Adverse Effect to the Carnegie Library. There would be no physical change to the building and no significant change in the triangle between the three streets upon which it sits. The building would continue to function as the city library, although the triangle would be changed somewhat to facilitate both motor vehicle and pedestrian traffic there. We request your concurrence.

If you have any questions, please contact me at 444-6258.

Jon Axline, Historian Environmental Services

Attachment

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section Heidy Bruner, P.E., Engineering Services
rep-1/-04 12112MM **ア208 273** 40029400021 Sent By: Kaurmas, Lee and Jackson; 1. 2003091103 David A. Gall, Director Montana Department of Transportation Judy Martz, Govarnal 2701 Prospect Avenue P0 801 201001 MASTER FILE! Helena MT 59620-1001 COPY Jose September 10, 2003 REPLODE NOTH: EFFEL TO ZYCBIZB3 A PACIFIC RAILINAY Mark Baumler, Ph.D. State Historic Preservation Office 1410 8th Avenue P O Box 201202 Helena, MT 59620-1202 CONCUR STPP 28-2(25)70 Subject: MONTANA SHPG Corridor Study - Red Lodge North Control No. 4375 DATE 25 Seg 03 SIGNED On April 4, 2003, we submitted a Determination of Effect to your office describing impacts to twelve historic sites located in the Area of Potential Effect (APE) for the above project in Carbon County. You concurred with our determination on April 22, 2003. However, we inadvertently omitted assessing effects to the Rocky Fork Branch of the Northern Pacific Railway (24CB1283). The Rocky Fork Branch was completed in 1889 and functioned primarily as a coal and agricultural carrier between Red Lodge and the rail yards/junction at Laurel. In 1983, the railroad abandoned the line with sections of the grade returned to agricultural use by adjacent landowners, converted into a roadbed or allowed to deteriorate. The Rocky Fork Branch was determined eligible for the National Register of Historic Places under Criterion A in 1992. What remains of the abandoned grade parallels much of the above 20± mile project area. While most of it lies outside the proposed R/W and construction limits boundary, approximately 2.8-miles of non-continuous abandoned railroad grade lies within the project ares and may be impacted by the proposed project. The grade would be impacted by realignment and widening of the existing roadway. We have determined, however, that the proposed project would have No Adverse Effect to the NRHP-cligible Rocky Fork Branch Line (24CB1283). The line was abandoned in 1983 with the segment north of Rockvale to Laurel still active and maintained by the Burlington Northern-Santa Fe Railroad. The 32-mile segment south of Rockvale, however, has been significantly impacted since 1983. The bridges, rails, ties, ballast and associated features (i.e. spikes, tie plates, signal masts, etc.) have been long removed from the line. However, depots and grain elevators still exist in Red Lodge, Fox, and Roberts. Segments of the line have been converted into local access roads or were turned over to adjacent landowners who either allowed it to deteriorate or otherwise obliterated the line. Other segments have been abandoned and are deteriorating from lack of maintenance. There has also been considerable residential development within the corridor, which has further impacted the integrity of the railroad line. Web Page: mmuodi slala.mi.ue Roed Report: (800) 226-7623 TTY: (800) 335-7562 Environmental Services Unit An Equal Opportunity Employe Phone: (405) 444-7228 14061 444-7245

оне ота проглава, сее вно звокариј

47058433065

reuriz-us idrigrig

Fage 3/3

Based on the existing condition of the line, the proposed project would not significantly alter its appearance, integrity, or ability to convey its historic significance to the development of Rock Creek valley. The most intact segments south of Roberts are located outside the proposed R/W boundary and are not included within the construction limits of the project. Those segments are currently being used as local access roads and would be perpetuated. There are no railroad-related structures that would be impacted by the project. The setting would not be adversely effected in that residential and commercial development all along the line in the project area over the last 20 years has already had a significant impact to the setting of the area. Because of the Rocky Fork branch line's significance to the development of Red Lodge and the Rock Creek valley, the MDT would install an interpretive marker adjacent to U.S. 212 near where a relatively infact segment of the line remains. We request your concurrence that the proposed project would have No Adverse Effect to the Rocky Fork Branch of the Northern Pacific Railroad (24CB1283).

If you have any questions, please contact me at 444-6258.

Jon Axline, Historian Environmental Services

cc: Bruce Barrett, Billings District Administrator Carl Peil, P.E., Preconstruction Bureau Bonnic Steg, Resources Section



Carnegie Library (24CB145) Section 4(f) De Minimis Impacts











Subject: De minimis Finding

Project Name: Corridor Study - Red Lodge North Project Number: STPP 28-2(25)70 Control Number: 4375

Dear Mr. Baumler:

MOVING THE AMERICAN ECONOMY

By way of this letter, the Federal Highway Administration (FHWA) is requesting written concurrence from the Montana State Historic Preservation Office (SHPO) that the concurrence dated June 10, 2005, is still applicable. The determinations were for the following Historic Irrigation Ditches:

C. BRET

FILE PROJECT

losef

REPLODGE

DE MINIMIS EINDI/16

4 HIMA

NORTH

Brandt Ditch 24CB1722 No Effect Rule-Thompson Ditch 24CB1723 No Adverse Effect Duncan-Aiken Ditch 24CB1724 No Effect Hunts Ditch 24CB1726 No Adverse Effect Highline Ditch 24CB1727 No Adverse Effect Rooney Ditch 24CB1728 No Adverse Effect Drakes Ditch 24CB1729 No Adverse Effect Ward Ditch 24CB1730 No Adverse Effect Carbonado Ditch 24CB1731 No Effect Hoyle Ditch 24CB1761 No Adverse Effect

The preferred alternative would impact 10 historic irrigation ditches along the project corridor. These irrigation ditches fall within proposed right-of-way and would require relocation outside of the right-of-way, per MDT's standard practice. Within the community of Roberts, the preferred alternative would require approximately 55 feet of right-of-way (including existing right-of-way and easements) along Brandt Ditch; therefore, requiring relocation of the ditch. Also within Roberts, the realignment of Cooney Dam Road with US Highway 212 would require the relocation of the Rule-Thompson Ditch outside of the road's proposed right-of-way. Within the rural segments of the project corridor, in this case between Roberts and Boyd, a standard 80foot right-of-way is generally proposed. The remaining eight historic irrigation ditches fall within this 80-foot right-of-way and would require the relocation of the ditches outside of the proposed right-of-way.

In addition to Section 106 of the National Historic Preservation Act (NHPA), FHWA must comply with the provisions of Section 4(f) of the 1966 Department of Transportation Act. Historically, Section 4(f) has required that prior to approval of any federally-funded highway project resulting in the "use" of listed or eligible historic properties under the NHPA; the FHWA must perform an avoidance analysis to determine whether there is a "feasible and prudent" alternative that would avoid the Section 4(f) resource.

In August of 2005, Section 138 of title 23, USC was amended under the Safe, Accountable, Flexible, and Efficient Transportation Act: A Legacy for Users (SAFETEA-LU). Section 6009 of SAFETEA-LU provided new legislative authority to address programs and projects with minor or '*de minimis*' impacts on a Section 4(f) resource.

More specifically, Section 6009(b) (2) of SAFETEA-LU states:

(2) HISTORIC SITES.--With respect to historic sites, the Secretary may make a finding of *de minimis impact* only if--

(A) the Secretary has determined, in accordance with the consultation process required under section 106 of the National Historic Preservation Act (16 U.S.C. 470f), that--

(i) the transportation program or project will have no adverse effect on the historic site; or

(ii) there will be no historic properties affected by the transportation program or project;

(B) the finding of the Secretary has received written concurrence from the applicable State historic preservation officer or tribal historic preservation officer (and from the Advisory Council on Historic Preservation if the Council is participating in the consultation process); and

(C) the finding of the Secretary has been developed in consultation with parties consulting as part of the process referred to in subparagraph (A).

This new provision of Section 4(f) is the basis of this letter, and of the FHWA's determination of *de minimis* impacts.

De Minimis Determination

The findings of "no effect" or "no adverse effect" reflect a conclusion that the uses identified in the attached exhibits will not "alter, directly or indirectly, any of the characteristics of [the] historic property that qualify the property for inclusion in the National Register in a manner that would diminish the integrity of the property's location, design, setting, materials, workmanship, feeling, or association."

If you concur in the "no effect" or "no adverse effect" determinations, FHWA intends to make a finding that impacts to historic resources that would result from implementation of the subject project would be *de minimis* for purposes of Section 4(f), as recently amended by Congress.

Request for Concurrence

The FHWA requests the written concurrence of the Montana SHPO in the above-described findings of "no effect" or "no adverse effect" on historic resources from the subject project. This written concurrence will be evidence that the concurrence and consultation requirements of Section 6009 of SAFETEA-LU, as they will be codified at 23 U.S.C. § 138(b) (2) (B) & (C), and 49 U.S.C. § 303 (d) (2) (B) and (C) are satisfied. Concurrence can be provided either by signing and dating this letter or by separate letter from the Montana SHPO to the Federal Highway Administration, 585 Shepard Way, Helena, MT 59601.

Sincerely,

Kevin L. McLaury, P.E. Division Administrator

Attachments

cc: Gabe Priebe, P.E., MDT Consultant Design Project Engineer Alan Woodmansey, P.E., FHWA, Operations Engineer Carl James, P.E., FHWA, Transportation Specialist

File: STPP 28-2(25)70 aw/lw

CONCUR MONTANA SHPO

Apr 2008 SIGNED DATE

From-HOT CONSULTANT DESIGN SECTION 4064446253 T-211 P.002/005 Jun-20-05 88:18aa F-610 RECEIVED Department of Transportation Jim Lynch, Director Brion Schweitzer, Governor JUN 1 4 2005 PO Box 201001 Helena MT 59620-1001 May 19, 2000 TRONMENTAL 2005052001 Mark Baumler, Ph.D. State Historic Preservation Office BY: SHPO 1410 8th Avenue ي زوندر Juspl P O Box 201202 Helena, MT 59620-1202 Red Lodge. STPP 28-2(25)70 Subject: Corridor CONCUR Red Lodge - North Corridor Study Control No. 4375 MONTANA SHPO Dear Mark: DATE JOJUNOS SIGNED The MDT has programmed a project to reconstruct 19.81 miles of U.S. Highway 212 from the north side of Red Lodge to the community of Boyd. On September 10, 2003, your office concurred with our Determination of Effect regarding the historic properties located along the road confidor. There are eleven privately-owned irrigation ditches located along that segment of the highway. They are: the Brandt Ditch 24CB1722), Rule-Thompson Ditch (24CB1723), Duncan-Aiken Ditch (24CB1724), Bernhardt Ditch (24CB1725), Hunts Ditch (24CB1726), Highline Ditch (24CB1727), Rooney Ditch (24CB1728), Drakes Ditch (24CB1729), Ward Ditch (24CB1730), Carbonado Ditch (24CB1731), and the Hoyle Ditch (24CB1761). Site forms for these ditches are included in Laurie Travis, "Red Lodge North, Highway Improvements: A Cultural Resource Inventory from Roberts to Boyd, Carbon County, Montana," (Metealf Archaeological Consultants, October, 2002). Based on their contribution to the agricultural development of the lower Rock Creek Valley, the ditches are eligible for the National Register of Historic Places under Criterion A. U.S. Highway 212 was originally constructed in 1936 and 1939 to provide a connection between U.S. 10, Red Lodge, and the recently completed Beartooth Highway. The roadway was last reconstructed in 1985 and all the culverts that would be impacted as part of this job were installed at that time. The Reinforced Concrete Pipe culverts proposed for this project are of the same dimensions as those installed in 1985. The culverts are, moreover, part of the roadway and not part of the ditch system. The following is a list of the ditches, the proposed impacts to them, and the effects of the proposed project on them. Brandt Ditch (24CB1722). It is the intent of the project to replace the existing corrugated steel pipe (CSP) with a reinforced concrete pipe (RCP). The ditch is approximately two miles in length and would not be rechanneled where it crosses under U.S. Highway 212. The existing and historic function of the ditch would be perpetuated as would its alignment where it crosses under the roadway. None of the criteria of Adverse Effect could be applied to this ditch as a result of the project. The proposed project would, therefore, have No Effect to the Brandt Ditch. てつうち file: MO Environmental Services Unit Web Page: www.mdLstate.nc.us Road Report (800) 225-7623 TTY: (800) 335-7592 An Equal Opportunity Employer Phone: (406) 444-7228 Fax: (406) 444-7245

08:18am From-MDT CONSULTANT DESIGN SECTION

4054446258

The <u>Rule-Thompson Ditch (24CB1723)</u> is about 2½ miles in length and consists of a crude field irrigation ditch. The MDT intends to replace the existing RCP crossing under the roadway with a new RCP as part of the project. In addition, about 180-feet of ditch would be rechanneled to accommodate a new crossing under the roadway. This constitutes 1.3% of the entire length of the ditch. Evidence suggests that the ditch was originally rechanneled in 1936 to accommodate the roadway. There would not be a significant change in the setting of the ditch as a result of the project and its existing and historic function would be perpetuated. The proposed width of the rechanneled portion of the ditch would match the existing width of the facility. There would be No Adverse Effect to the Rule-Thompson Ditch as a result of the project.

The <u>Duncan-Aiken Ditch (24CB1724)</u> is approximately 3 miles in length and also consists of a small field ditch. The MDT intends to replace the existing CSP with a new RCP at the point where the ditch crosses under U.S. Highway 212. The ditch would not be rechanneled to accommodate the pipe. The setting of the site would be retained and the historic function of the ditch perpenated. There would be No Effect to the Duncan-Aiken Ditch as a result of the proposed project.

The <u>Bernhardt Ditch (24CB1725)</u> is about 1½ miles in length and is similar in design, appearance, and usage. According to the preliminary plan sheets, the existing culvert would not be replaced as part of the proposed project. There would be no change in the alignment of the ditch and no alteration of its current dimensions or use as a field ditch. There would, therefore, be No Effect to the Bernhardt Ditch as a result of the project.

The <u>Hunts Ditch (24CB1726)</u> crosses under U.S. Highway 212 at two points in Section 29, T5S, R21E. The ditch has a total length of about two miles and is similar in appearance to the Bernhardt Ditch. It is the intent of the MDT to replace the existing CSP's with RCP's. About 351-feet of ditch would also be rechanneled to accommodate the crossings. This constitutes about 3.3% of the total length of the ditch. There would not, however, be any change in the function of the facility and it would continue to provide water to adjacent farmlands near the roadway. The ditch, moreover, was rechanneled in 1936 when the roadway was originally constructed. The setting of the site would remain intact as would the majority of the alignment and configuration of the ditch. The MDT project would have No Adverse Effect to the Hunts Ditch.

The <u>Highline Ditch (24CB1727)</u> roughly parallels U.S. Highway 212 for 10± miles in Section 21, T5S, R21E. It is the intent of the MDT to replace the existing RCP with a new RCP. The project would also include the rechanneling of 148-feet of the 52,800± foot ditch. This would constitute about 24% of the entire ditch system. There would be no change in the historic function of the ditch and it would continue to irrigate farm fields in central Carbon County. The dimensions of the rechanneled portion of the ditch would closely match the existing width and depth of the facility. There would be no significant change in the setting of the facility. The proposed MDT project would, therefore, have No Adverse Effect to the Highline Ditch.

The <u>Roonev Ditch (24CB1728)</u> is 3± miles in length and carries water to agricultural property in the general vicinity of U.S. Highway 212. The MDT intends to replace the

Jun-20-85

existing RCP with a new RCP. Along with the RCP, it is the intent of the project to rechannel about 754-feet of the ditch to better accommodate the new pipe. There would be no change in the historic function of the ditch, which was originally rechanneled in 1936 because of highway construction. The dimensions of the rechanneled portion of the ditch would closely match the segments located outside the project area. The setting of the property would also remain largely intact. The proposed project would have No Adverse Effect to the Rooney Ditch.

Drakes Ditch (Z4CB1729) is a small field ditch that is about two miles in length and carries water to one or two users. It is the intent of the MDT to replace the existing RCP with a new RCP and rechannel about 207-feet of the ditch to accommodate the new pipe and improve its hydraulics. This would constitute 1.9% of the entire length of the system. The ditch would continue to function in its historic capacity as a ditch and there would be no change to most of the facility. The dimensions of the rechanneled portions of the ditch would closely match the unchanged segment of the facility. There would be no change in the setting of the ditch. There would be No Adverse Effect to the Drakes Ditch.

The Ward Ditch (24CB1730) is approximately five miles in length and irrigates familand for one or two users in Section 2, TSS, R21B and Section 35, T4S, R21E. The MDT intends to replace the existing RCP with a new RCP that would better accommodate the hydraulics of the site. About 541-feet or 2% of the ditch would be rechanneled in conjunction with the installation of the new pipe. There would be no change in the historic function or capacity of the ditch as a result of the project. The setting of the historic property would also remain largely intact with most of the ditch located outside the APE of the project. The setting would also remain intact. The proposed MDT project would have No Adverse Effect to the Ward Ditch.

The <u>Carbonado Ditch (24CB1731)</u> is also about five miles in length and carries water to one or two users in the lower Rock Creek valley. It is the intent of the project to replace the existing RCP with a new RCP and rechannel 105-feet of the ditch to accommodate the replacement. That would constitute about .39% of the entire length of the ditch. The historic function and capacity of the ditch would be perpetuated with more than 99% of the ditch located outside the APE of the project. The proposed project would have No Effect to the Carbonado Ditch.

About three miles of the <u>Hoyle Ditch (24CB1761)</u> is located near the Red Lodge – North project area with about 2,500-feet located within the existing R/W of U.S 212. It is the intent of the project to relocate 2,401-feet of the ditch outside the R/W boundary and, thus, remove, a hazard adjacent to the road. The existing ditch parallels the roadway. The proposed new alignment would also parallel the existing alignment – except it would be located outside the R/W line. There would be no change in the historic function of the ditch and its dimensions would be matched for its new alignment. None of the criteria for Adverse Effect would apply to the rechanneling and the proposed project would have No Adverse Effect to the Hoyle Ditch.

In summary, the proposed project would have No Effect to 24CB1722, 24CB1724, 24CB1725, and 24CB1731. There would be No Adverse Effect to 24CB1723,

Jun-20-05 D8:18am From-NOT CONSULTANT DESIGN SECTION

4064446253

T-211 P.005/005 F-610

24CB1726, 24CB1727, 24CB1728, 24CB1729, 24CB1730, and 24CB1761. We request your concurrence.

If you have any questions, please contact me at 444-6258.

Q Non Axtine, Historian

Environmental Services

cc: Bruce Barrett, Billings District Administrator Tom Martin, P.E., Consultant Design Bonnie Steg, Resources Section

